

Affidavit of Michèle Hay #3 Sworn: May 3, 2023

> No. S233209 Vancouver Registry

IN THE SUPREME COURT OF BRITISH COLUMBIA

IN THE MATTER OF THE COMPANIES' CREDITORS ARRANGEMENT ACT, R.S.C. 1985, c. C - 36 AS AMENDED

AND

IN THE MATTER OF SLP HOLDINGS LTD, STRUCTURLAM MASS TIMBER CORPORATION, STRUCTURLAM MASS TIMBER U.S., INC., NATURAL OUTCOMES, LLC

APPLICATION OF SLP HOLDINGS LTD. UNDER SECTION 46 OF THE COMPANIES' CREDITORS ARRANGEMENT ACT, R.S.C. 1985, c. C-36, AS AMENDED

PETITIONER

AFFIDAVIT

- I, **MICHÈLE HAY**, Legal Administrative Assistant, of Suite 2300 550 Burrard Street, Vancouver, British Columbia, V6C 2B5, SWEAR THAT:
- 1. I am employed by Gowling WLG (Canada) LLP, counsel for SLP Holdings Ltd., the Foreign Representative under Part IV of the *Companies' Creditors Arrangement Act* for recognition of US Chapter 11 Proceedings, and as such have personal knowledge of the facts and matters hereinafter deposed to, save and except where the same are stated to be based upon information and belief, and where so stated I verily believe the same to be true.
- 2. Attached hereto and marked as Exhibit "A" to this my Affidavit is a copy of the Final Order dated May 16, 2023, (I) authorizing the Debtors to (A) continue to operate their cash management system, (B) honor certain prepetition obligations related thereto, (C) maintain existing business forms, and (D) continue to perform intercompany transactions, (II) granting administrative expense status to post-petition intercompany balances, and (III) granting related relief.
- 3. Attached hereto and marked as Exhibit "B" to this my Affidavit is a copy of the Final Order dated May 16, 2023, (I) authorizing the Debtors to (A) pay their obligations under insurance

policies entered into prepetition and (B) renew, supplement, modify, or purchase insurance coverage, and (II) granting related relief.

- 4. Attached hereto and marked as Exhibit "C" to this my Affidavit is a copy of the Final Order dated May 16, 2023, (I) determining adequate assurance of payment for future utility services, (II) prohibiting utility providers from altering, refusing, or discontinuing utility services, (III) establishing procedures for determining adequate assurance of payment, and (IV) granting related relief.
- 5. Attached hereto and marked as Exhibit "D" to this my Affidavit is a copy of the Final Order dated May 16, 2023, (I) authorizing the Debtors to (A) pay prepetition wages, salaries, other compensation, and reimbursable expenses and (B) continue certain employee benefits programs and (II) granting related relief.
- 6. Attached hereto and marked as Exhibit "E" to this my Affidavit is a copy of the Final Order dated May 16, 2023, (I) authorizing the payment of certain prepetition taxes and fees and (II) granting related relief.
- 7. Attached hereto and marked as Exhibit "F" to this my Affidavit is a copy of the Order dated May 16, 2023, (A) authorizing the Debtors to reject the amended and restated Walmart Realty Supplier Agreement Goods not for resale, effective as of the petition date, and (B) granting related relief.
- 8. Attached hereto and marked as Exhibit "G" to this my Affidavit is a copy of the Order dated May 16, 2023, (I) authorizing and approving procedures to reject executory contracts and unexpired leases, and (II) granting related relief.
- 9. Attached hereto and marked as Exhibit "H" to this my Affidavit is a copy of the Final Order dated May 17, 2023, authorizing, but not directing, the Debtors to pay certain pre-petition claims of critical vendors, shippers, and 503(b)(9) claimants and granting related relief.
- 10. Attached hereto and marked as Exhibit "I" to this my Affidavit is a copy of the Final Order dated May 19, 2023, (I) authorizing debtors and debtors in possession to (A) obtain post-petition financing, (B) use cash collateral, (C) grant liens and super-priority claims, and (D) grant adequate protection; (II) modifying the automatic stay; and (III) granting related relief.

- 11. Attached hereto and marked as Exhibit "J" to this my Affidavit is a copy of the Order dated May 25, 2023, authorizing the Debtors to employ and retain Kurtzman Carson Consultants LLC as administrative advisor effective as of the Petition Date.
- 12. Attached hereto and marked as Exhibit "K" to this my Affidavit is a copy of the Order dated May 25, 2023, establishing bar dates and related procedures for filing proofs of claim (including for administrative expense claims arising under section 503(b)(9) of the Bankruptcy Code) and approving the form and manner of notice.
- 13. Attached hereto and marked as Exhibit "L" to this my Affidavit is a copy of the Order (I) authorizing (A) sale of assets free and clear of all liens, claims, encumbrances and interests and (B) the debtors' assumption and assignment of certain executory contracts and unexpired leases; and (II) granting related relief.
- 14. Attached hereto and marked as Exhibit "M" to this my Affidavit is a copy of the Order dated May 30, 2023, (I) authorizing the Debtors to implement Key Employee incentive and retention plans and (II) granting related relief.

SWORN BEFORE WE at Vancouver, BC, this day of May, 2023.

A Commissioner for taking Affidavits within British Columbia

MICHELE H

JONATHAN B. ROSS

GOWLING WLG (CANADA) LLP
BARRISTER & SOLICITOR
550 BURRARD STREET - SUITE 2300
BENTALL 5 - VANCOUVER, B.C. V6C 2B5
TELEPHONE: (604) 891-2778

1

This is Exhibit "<u>A</u>" referred to in the Affidavit of **MICHÈLE HAY**, sworn before me at Vancouver, BC, this 30th day of May, 2023.

A Commissioner for Making Affidavits in British Columbia

IN THE UNITED STATES BANKRUPTCY COURT FOR THE DISTRICT OF DELAWARE

STRUCTURLAM MASS TIMBER U.S., INC., et al., 1)	-1\
Debtors.) (Joint Administration Requeste) Related Docket Nos. 8 and 38	·

FINAL ORDER (I) AUTHORIZING
THE DEBTORS TO (A) CONTINUE TO OPERATE
THEIR CASH MANAGEMENT SYSTEM, (B) HONOR
CERTAIN PREPETITION OBLIGATIONS RELATED
THERETO, (C) MAINTAIN EXISTING BUSINESS FORMS, AND
(D) CONTINUE TO PERFORM INTERCOMPANY TRANSACTIONS,
(II) GRANTING ADMINISTRATIVE EXPENSE STATUS TO POSTPETITION
INTERCOMPANY BALANCES, AND (III) GRANTING RELATED RELIEF

Upon the motion (the "Motion")² of the above-captioned debtors and debtors in possession (collectively, the "<u>Debtors</u>") for entry of a final order (this "<u>Final Order</u>"): (a) authorizing the Debtors to (i) continue to operate their Cash Management System, (ii) honor certain prepetition obligations related thereto, (iii) maintain existing Business Forms in the ordinary course of business, and (iv) continue to perform Intercompany Transactions consistent with historical practice; (b) granting administrative expense status to postpetition Intercompany Balances; and (c) granting related relief, all as more fully set forth in the Motion; and upon the First Day Declaration; and this Court having jurisdiction over this matter pursuant to 28 U.S.C. §§157 and 1334 and the *Amended Standing Order of Reference* from the United States District Court for

The Debtors in these chapter 11 cases, along with the last four digits of each Debtor's federal tax identification number or Canadian business number, as applicable, include: Structurlam Mass Timber U.S., Inc. (6287); Natural Outcomes, LLC (n/a); Structurlam Mass Timber Corporation (5050); and SLP Holdings Ltd. (3114); The location of the Debtors' headquarters is: 2176 Government Street, Penticton, British Columbia, Canada V2A 8B5. The address of the registered agent for Structurlam Mass Timber U.S., Inc. is: 8 The Green, Suite A, Dover, Delaware 19901.

² Capitalized terms used but not otherwise defined herein have the meanings given to such terms in the Motion.

the District of Delaware, dated February 29, 2012; and this Court having found that this is a core proceeding pursuant to 28 U.S.C. § 157(b)(2), and that this Court may enter a final order consistent with Article III of the United States Constitution; and this Court having found that venue of this proceeding and the Motion in this district is proper pursuant to 28 U.S.C. §§ 1408 and 1409; and this Court having found that the relief requested in the Motion is in the best interests of the Debtors' estates, their creditors, and other parties in interest; and this Court having found that the Debtors' notice of the Motion and opportunity for a hearing on the Motion were appropriate and no other notice need be provided; and this Court having reviewed the Motion and having heard any statements in support of the relief requested therein at a hearing before this Court (the "Hearing"); and this Court having determined that the legal and factual bases set forth in the Motion and at the Hearing establish just cause for the relief granted herein; and upon all of the proceedings had before this Court; and after due deliberation and sufficient cause appearing therefor, it is HEREBY ORDERED THAT:

- 1. The Motion is granted on a final basis as set forth herein.
- 2. The Debtors are authorized, but not directed, on a final basis, to: (a) continue operating the Cash Management System, as described in the Motion, with all pre-petition agreements related to the Bank Accounts remaining in full force and effect; (b) honor their prepetition obligations related thereto; (c) use, in their present form, all preprinted correspondence and Business Forms (including letterhead and checks) without reference to the Debtors' status as debtors in possession, subject to paragraph 4 below; and (d) continue to perform Intercompany Transactions consistent with historical practice.
- 3. The Debtors are further authorized, but not directed, on a final basis, to:
 (a) continue to use, with the same account numbers, the Bank Accounts in existence as of the

Petition Date, including those accounts identified on **Exhibit 1** attached to the Interim Order; (b) treat the Bank Accounts for all purposes as accounts of the Debtors as debtors in possession; (c) deposit funds in and withdraw funds from the Bank Accounts by all usual means, including checks, wire transfers, and other debits; (d) pay the Bank Fees (including any prepetition amounts); and (e) pay, pursuant to those certain existing deposit, cash management, and credit card agreements between the Debtors and the Cash Management Banks, any ordinary course Bank Fees, including attorneys' fees, incurred in connection with the Bank Accounts, irrespective of whether such fees arose prior to the Petition Date, and to otherwise perform their obligations under the documents governing the Bank Accounts. Any postpetition fees, costs, charges and expenses, including Bank Fees, or charge-backs payable to the Cash Management Banks that are not so paid shall be entitled to priority as administrative expenses pursuant to section 503(b)(1) of the Bankruptcy Code.

- 4. In the event the Debtors need to purchase additional check stock during the pendency of these Chapter 11 Cases, such new check stock shall include a legend referring to the Debtors as a "Debtors in Possession" or "DIP." Further, any electronically produced checks shall reflect the Debtors' status as "Debtors in Possession."
- 5. The Cash Management Banks are authorized to continue to maintain, service, and administer the Bank Accounts as accounts of the Debtors as debtors in possession, without interruption and in the ordinary course, and to receive, process, honor, and pay, to the extent of available funds, any and all checks, drafts, wires, credit card payments, and ACH transfers issued and drawn on the Bank Accounts after the Petition Date by the holders or makers thereof, as the case may be; *provided*, however, the Debtors will take all reasonable steps to stop payment on any checks, drafts, wires, or ACH transfers drawn or issued by the Debtors before, but which did not

clear the Bank Accounts prior to, the Petition Date; provided further that (a) those certain existing deposit, cash management, and credit card agreements between the Debtors and the Cash Management Banks shall continue to govern the postpetition cash management relationship between the Debtors and the respective Cash Management Bank, and all of the provisions of such agreements, including the termination and fee provisions and any provisions relating to offset or charge back rights with respect to returned items, shall remain in full force and effect, and (b) the Debtors and the Cash Management Banks may, without further order of this Court, agree to and implement changes to the Cash Management System and procedures related thereto in the ordinary course of business pursuant to terms of those certain existing deposit, cash management, and credit card agreements, including the closing of Bank Accounts or the opening of new bank accounts; provided, however, that, except as provided in this Final Order, none of the Cash Management Banks shall sweep, without an order from this Court, any cash postpetition in the Bank Accounts to set off or otherwise pay down or diminish any prepetition debt owed to such Cash Management Bank.

- 6. The relief granted in this Final Order is extended to any new bank account opened by the Debtors after the date hereof, which account shall be deemed a Bank Account, and to the bank at which such account is opened, which bank shall be deemed a Cash Management Bank.
- 7. All banks, including the Cash Management Banks, provided with notice of this Final Order maintaining any of the Bank Accounts shall not honor or pay any bank payments drawn on the listed Bank Accounts, or otherwise issued before the Petition Date, absent further direction from the Debtors.
- 8. The Debtors will maintain records in the ordinary course reflecting transfers of cash, if any, so as to permit all such transactions to be ascertainable.

- 9. In the course of providing cash management services to the Debtors, each of the Cash Management Banks is authorized, in the ordinary course of business without further order of this Court, to deduct, pursuant to those certain existing deposit, cash management, and credit card agreements between the Debtors and the Cash Management Banks, the applicable fees and expenses (including any costs incurred by the Cash Management Banks to comply with any requirements of the U.S. Trustee, including posting collateral and all letter of credit fees and expenses) associated with the nature of the deposit, cash management, and credit card services rendered to the Debtors, whether arising prepetition or postpetition, from the appropriate accounts of the Debtors, and further, to charge back to the appropriate accounts of the Debtors any amounts resulting from returned checks or other returned items, including returned items that result from wire transfers, ACH transactions, or other electronic transfers of any kind, regardless of whether such items were deposited or transferred prepetition or postpetition and regardless of whether the returned items relate to prepetition or postpetition items or transfers.
- 10. Each of the Cash Management Banks is authorized to debit the Debtors' accounts in the ordinary course and without further order of this Court on account of: (a) all checks drawn on the Debtors' accounts that have been cashed at such banks' counters or exchanged for cashier's or official checks by the payees thereof prior to the Petition Date; (b) all checks or other items deposited in one of the Debtors' accounts with such bank prior to the Petition Date which have been dishonored or returned unpaid for any reason, together with any fees and costs in connection therewith, to the same extent the Debtors were responsible for such items prior to the Petition Date; and (c) all undisputed prepetition amounts outstanding as of the date hereof, if any, owed to any bank as to service charges for the maintenance of the Cash Management System.

- Any bank, including a Cash Management Bank, may rely upon the representations of the Debtors with respect to whether any check, draft, wire, or other transfer drawn or issued by the Debtors prior to the Petition Date should be honored pursuant to any order of this Court, and no bank that honors a prepetition check or other item drawn on any account that is the subject of this Final Order (a) at the direction of the Debtors, (b) in a good faith belief that this Court has authorized such prepetition check or item to be honored, or (c) as a result of an innocent mistake made despite implementation of customary item handling procedures, shall be deemed to be nor shall be liable to the Debtors, their estates, or any other party on account of such prepetition check or other item being honored postpetition, or otherwise deemed to be in violation of this Final Order.
- 12. Any and all banks, including the Cash Management Banks, are further authorized to (a) honor the Debtors' directions with respect to the opening and closing of any Bank Account and (b) accept and hold the Debtors' funds in accordance with the Debtors' instructions; *provided* that the Cash Management Banks shall not have any liability to any party for relying on such representations to the extent such reliance otherwise complies with applicable law.
- Bank Accounts as they may deem necessary and appropriate; provided that such actions are not prohibited or restricted by the terms of the DIP Order or any of the DIP Documents; provided, further, that the Debtors shall give notice within five business days to the U.S. Trustee and any statutory committee appointed in these Chapter 11 Cases of the opening or closing of any Bank Accounts and such opening or closing shall be timely indicated on the Debtors' monthly operating reports; provided, further, that the Debtors shall open any such new Bank Account at banks that have executed a Uniform Depository Agreement with the U.S. Trustee for the District of Delaware or at such banks that are willing to immediately execute such an agreement.

- 14. The requirements of section 345(b) of the Bankruptcy Code are waived as to the Cash Management Banks at which the Debtors hold Bank Accounts that are not party to a Uniform Depositary Agreement with the U.S. Trustee for the District of Delaware.
- 15. The requirement to establish separate accounts for cash collateral and/or tax payments is hereby waived.
- Notwithstanding anything to the contrary set forth herein, the Debtors are 16. authorized, but not directed, to continue Intercompany Transactions arising from or related to the operation of their business in the ordinary course. The Debtors shall not be authorized by this Final Order to undertake any Intercompany Transactions that are (a) not on the same terms as, or materially consistent with, the Debtors' operation of their business in the ordinary course during the prepetition period or (b) prohibited or restricted by the terms of the DIP Order or any of the All postpetition payments from a Debtor to another Debtor under any postpetition Intercompany Transactions authorized hereunder are hereby accorded administrative expense status under section 503(b) of the Bankruptcy Code, subject and junior to the claims, including adequate protection and superpriority claims, granted in connection with any DIP Orders. In connection with the Intercompany Transactions, the Debtors shall continue to maintain current records with respect to all transfers of cash so that all Intercompany Transactions may be readily ascertained, traced, and properly recorded on intercompany accounts consistent with historical practice as described in the Motion; provided that such records shall distinguish between prepetition and postpetition transactions. The Debtors shall not make any intercompany loans to non-debtor entities absent further order of the court.
- 17. Nothing contained in the Motion or this Final Order shall be construed to (a) create or perfect, in favor of any person or entity, any interest in cash of a Debtor that did not exist as of

the Petition Date or (b) alter or impair any security interest or perfection thereof, in favor of any person or entity, that existed as of the Petition Date.

- 18. Notwithstanding the Debtors' use of a consolidated cash management system, the Debtors shall calculate quarterly fees under 28 U.S.C. § 1930(a)(6) based on the disbursements of each Debtor, regardless of which entity pays those disbursements.
- 19. The Debtors are authorized to issue postpetition checks, or to effect postpetition fund transfer requests, in replacement of any checks or fund transfer requests that are dishonored as a consequence of these Chapter 11 Cases with respect to prepetition amounts owed in connection with any Bank Fees.
- 20. Notwithstanding the relief granted in this Final Order and any actions taken pursuant to such relief, nothing in this Final Order shall be deemed: (a) an admission as to the validity, priority, or amount of any particular claim against a Debtor entity; (b) a waiver of the Debtors' or any other party in interest's right to dispute any particular claim on any grounds; (c) a promise or requirement to pay any particular claim; (d) an implication or admission that any particular claim is of a type specified or defined in this Final Order or the Motion; (e) a request or authorization to assume any agreement, contract, or lease pursuant to section 365 of the Bankruptcy Code; (f) a waiver or limitation of the Debtors' or any other party in interest's rights under the Bankruptcy Code or any other applicable law; or (g) a concession by the Debtors or any other party in interest that any liens (contractual, common law, statutory, or otherwise) satisfied pursuant to this Final Order are valid and the Debtors and all other parties in interest expressly reserve their rights to contest the extent, validity, or perfection or to seek avoidance of all such liens. Any payment made pursuant to this Final Order should not be construed as an admission as

10

Case 23-10497-CTG Doc 117 Filed 05/16/23 Page 9 of 9

to the validity, priority, or amount of any particular claim or a waiver of the Debtors' or any other

party in interest's rights to subsequently dispute such claim.

21. The banks and financial institutions on which checks were drawn or electronic

payment requests made in payment of the prepetition obligations approved herein are authorized

to receive, process, honor, and pay all such checks and electronic payment requests when presented

for payment, and all such banks and financial institutions are authorized to rely on the Debtors'

designation of any particular check or electronic payment request as approved by this Final Order

without any duty of further inquiry.

22. Notwithstanding Bankruptcy Rule 6004(h), the terms and conditions of this Final

Order are immediately effective and enforceable upon its entry.

23. The Debtors are authorized to take all actions necessary to effectuate the relief

granted in this Final Order in accordance with the Motion.

24. This Court retains jurisdiction with respect to all matters arising from or related to

the implementation, interpretation, and enforcement of this Final Order.

Dated: May 16th, 2023 Wilmington, Delaware CRAIG T. GOLDBLATT UNITED STATES BANKRUPTCY JUDGE

Cy Dullan

9

This is Exhibit "B" referred to in the Affidavit of **MICHÈLE HAY**, sworn before me at Vancouver, BC, this 30^{th} day of May, 2023

A Commissioner for taking Affidavits in British Columbia

IN THE UNITED STATES BANKRUPTCY COURT FOR THE DISTRICT OF DELAWARE

In re:) Chapter 11
STRUCTURLAM MASS TIMBER U.S., INC., et al., 1	Case No. 23-10497 (CTG)
Debtors.) (Jointly Administered)) Related Docket Nos. 10 and 42

FINAL ORDER (I) AUTHORIZING THE DEBTORS
TO (A) PAY THEIR OBLIGATIONS UNDER INSURANCE POLICIES
ENTERED INTO PREPETITION AND (B) RENEW, SUPPLEMENT, MODIFY,
OR PURCHASE INSURANCE COVERAGE AND (II) GRANTING RELATED RELIEF

Upon the motion (the "Motion")² of the above-captioned debtors and debtors in possession (collectively, the "Debtors") for entry of a final order (this "Final Order"): (a) authorizing, but not directing, the Debtors to (i) pay their obligations under the insurance policies entered into prepetition and (ii) renew, supplement, modify, or purchase insurance coverage in the ordinary course; and (b) granting related relief, all as more fully set forth in the Motion; and upon the First Day Declaration; and this Court having jurisdiction over this matter pursuant to 28 U.S.C. §§ 157 and 1334; and this Court having found that this is a core proceeding pursuant to 28 U.S.C. § 157(b)(2), and that this Court may enter a final order consistent with Article III of the United States Constitution; and this Court having found that venue of this proceeding and the Motion in this district is proper pursuant to 28 U.S.C. §§ 1408 and 1409; and this Court having found that the relief requested in the Motion is in the best interests of the Debtors' estates, their

The Debtors in these chapter 11 cases, along with the last four digits of each Debtor's federal tax identification number or Canadian business number, as applicable, include: Structurlam Mass Timber U.S., Inc. (6287); Natural Outcomes, LLC (n/a); Structurlam Mass Timber Corporation (5050); and SLP Holdings Ltd. (3114). The location of the Debtors' headquarters is: 2176 Government Street, Penticton, British Columbia, Canada V2A 8B5. The address of the registered agent for Structurlam Mass Timber U.S., Inc. is: 8 The Green, Suite A, Dover, Delaware 19901.

Capitalized terms used but not otherwise defined herein have the meanings given to such terms in the Motion.

creditors, and other parties in interest; and this Court having found that the Debtors' notice of the Motion and opportunity for a hearing on the Motion were appropriate and no other notice need be provided; and this Court having reviewed the Motion and having heard any statements in support of the relief requested therein at a hearing before this Court (the "Hearing"); and this Court having determined that the legal and factual bases set forth in the Motion and at the Hearing establish just cause for the relief granted herein; and upon all of the proceedings had before this Court; and after due deliberation and sufficient cause appearing therefor, it is HEREBY ORDERED THAT:

- 1. The Motion is granted on a final basis as set forth herein.
- 2. The Debtors are authorized, but not directed, on a final basis to:
 - a) continue the Insurance Policies and to pay any prepetition or postpetition obligations under the Insurance Policies, the Insurance Broker Fees, and any other amounts related to the Insurance Policies, including any amounts owed to the Insurance Broker;
 - b) renew, amend, supplement, extend, reduce, or purchase insurance policies to the extent that the Debtors determine that such action is in the best interest of their estates;
 - c) honor the terms of the Premium Financing Agreements and pay any installment payments; and
 - d) enter into, amend, supplement, or extend the Premium Financing Agreements as necessary, in the ordinary course of business consistent with past practices to the extent the Debtors determine that such action is in the best interests of their estates.
- 3. Notwithstanding the relief granted in this Final Order and any actions taken pursuant to such relief, nothing in this Final Order shall be deemed: (a) an admission as to the validity, priority, or amount of any particular claim against a Debtor entity; (b) a waiver of the Debtors' or any other party in interest's right to dispute any particular claim on any grounds; (c) a promise or requirement to pay any particular claim; (d) an implication or admission that any particular claim is of a type specified or defined in this Final Order or the Motion; (e) a request or

authorization to assume any agreement, contract, or lease pursuant to section 365 of the Bankruptcy Code; (f) a waiver or limitation of the Debtors' or any other party in interest's rights under the Bankruptcy Code or any other applicable law; or (g) a concession by the Debtors or any other party in interest that any liens (contractual, common law, statutory, or otherwise) satisfied pursuant to this Final Order are valid and the Debtors and all other parties in interest expressly reserve their rights to contest the extent, validity, or perfection or to seek avoidance of all such liens. Any payment made pursuant to this Final Order should not be construed as an admission as to the validity, priority, or amount of any particular claim or a waiver of the Debtors' or any other party in interest's rights to subsequently dispute such claim. The relief granted by this Final Order is not intended to and does not grant the Debtors any authority to use cash collateral of Bank of Montreal or to use proceeds of any postpetition financing provided by Bank of Montreal in any way inconsistent with the provisions of any interim or final order of this Court approving the Debtors' use of cash collateral or authorizing the Debtors to obtain debtor-in-possession financing (a "DIP Order"). For the avoidance of doubt, in case of a conflict between the provisions of this Final Order and a DIP Order, the provisions of such DIP Order shall control.

- 4. The banks and financial institutions on which checks were drawn or electronic payment requests made in payment of the prepetition obligations approved herein are authorized to receive, process, honor, and pay all such checks and electronic payment requests when presented for payment, and all such banks and financial institutions are authorized to rely on the Debtors' designation of any particular check or electronic payment request as approved by this Final Order without any duty of further inquiry and without liability for following the Debtors' instructions.
- 5. The Debtors are authorized to issue postpetition checks, or to effect postpetition fund transfer requests, in replacement of any checks or fund transfer requests that are dishonored

as a consequence of these Chapter 11 Cases with respect to prepetition amounts owed in connection with any Insurance Policies and the Premium Financing Agreements.

- 6. Notwithstanding Bankruptcy Rule 6004(h), the terms and conditions of this Final Order are immediately effective and enforceable upon its entry.
- 7. The Debtors are authorized to take all actions necessary to effectuate the relief granted in this Final Order in accordance with the Motion.
- 8. This Court retains jurisdiction with respect to all matters arising from or related to the implementation, interpretation, and enforcement of this Final Order.

Cin Dellas

Dated: May 16th, 2023 Wilmington, Delaware CRAIG T. GOLDBLATT UNITED STATES BANKRUPTCY JUDGE This is Exhibit "<u>C</u>" referred to in the Affidavit of **MICHÈLE HAY**, sworn before me at Vancouver, BC, this 30^{th} day of May, 2023.

A Commissioner for taking Affidavits in British Columbia

IN THE UNITED STATES BANKRUPTCY COURT FOR THE DISTRICT OF DELAWARE

Debtors.) Re: Docket Nos. 11 and 39
)) (Jointly Administered)
STRUCTURLAM MASS TIMBER U.S., INC., et al., 1) Case No. 23-10497 (CTG)
In re:) Chapter 11

FINAL ORDER (I) DETERMINING ADEQUATE
ASSURANCE OF PAYMENT FOR FUTURE UTILITY
SERVICES, (II) PROHIBITING UTILITY PROVIDERS
FROM ALTERING, REFUSING, OR DISCONTINUING UTILITY
SERVICES, (III) ESTABLISHING PROCEDURES FOR DETERMINING
ADEQUATE ASSURANCE OF PAYMENT, AND (IV) GRANTING RELATED RELIEF

Upon the motion (the "Motion")² of the above-captioned debtors and debtors in possession (collectively, the "Debtors") for entry of a final order (this "Final Order"): (a) approving the Debtors' Proposed Adequate Assurance; (b) prohibiting Utility Providers from altering, refusing, or discontinuing services; (c) approving the Debtors' proposed procedures for resolving any Additional Assurance Requests; and (d) granting related relief; all as more fully set forth in the Motion and upon the First Day Declaration; and this Court having jurisdiction over this matter pursuant to 28 U.S.C. §§ 157 and 1334 and the *Amended Standing Order of Reference* from the United States District Court for the District of Delaware, dated February 29, 2012; and this Court having found that this is a core proceeding pursuant to 28 U.S.C. § 157(b)(2), and that this Court may enter a final order consistent with Article III of the United States Constitution; and this Court

The Debtors in these chapter 11 cases, along with the last four digits of each Debtor's federal tax identification number or Canadian business number, as applicable, include: Structurlam Mass Timber U.S., Inc. (6287); Natural Outcomes, LLC (n/a); Structurlam Mass Timber Corporation (5050); and SLP Holdings Ltd. (3114). The location of the Debtors' headquarters is: 2176 Government Street, Penticton, British Columbia, Canada V2A 8B5. The address of the registered agent for Structurlam Mass Timber U.S., Inc. is: 8 The Green, Suite A, Dover, Delaware 19901.

² Capitalized terms used but not otherwise defined herein have the meanings given to such terms in the Motion.

having found that venue of this proceeding and the Motion in this district is proper pursuant to 28 U.S.C. §§ 1408 and 1409; and this Court having found that the relief requested in the Motion is in the best interests of the Debtors' estates, their creditors, and other parties in interest; and this Court having found that the Debtors' notice of the Motion and opportunity for a hearing on the Motion were appropriate and no other notice need be provided; and this Court having reviewed the Motion and having heard the statements in support of the relief requested therein at a hearing before this Court (the "Hearing"); and this Court having determined that the legal and factual bases set forth in the Motion and at the Hearing establish just cause for the relief granted herein; and upon all of the proceedings had before this Court; and after due deliberation and sufficient cause appearing therefor, it is HEREBY ORDERED THAT:

- 1. The Motion is granted on a final basis as set forth herein.
- 2. All Utility Providers are prohibited from altering, refusing, or discontinuing services on account of any unpaid prepetition charges, the commencement of these Chapter 11 Cases, or any perceived inadequacy of the Proposed Adequate Assurance.
 - 3. The following adequate Assurance Procedures are hereby approved on a final basis:
 - a. The Debtors will serve a copy of the Motion and this Final Order to each Utility Provider within two business days after entry of this Interim Order.
 - b. Subject to paragraphs (f)—(j) herein, the Debtors will deposit the Adequate Assurance Deposit in the Adequate Assurance Account within ten business days after entry of the Interim Order.
 - c. The funds in the Adequate Assurance Account shall be held for the benefit of each Utility Provider in the amount set forth for such Utility Provider in the column labeled "Proposed Adequate Assurance" on the Utility Services List attached hereto as Exhibit C.
 - d. If an amount relating to Utility Services provided postpetition by a Utility Provider is unpaid, and remains unpaid beyond any applicable grace period, such Utility Provider may request a disbursement from the Adequate Assurance Account by giving notice to: (a) the Debtors, 2176 Government Street, Penticton, British Columbia, Canada V2A 8B5, Attn: Matthew

Karmel (mkarmel@structurlam.com); (b) proposed counsel to the Debtors, Chipman Brown Cicero & Cole, LLP, Hercules Plaza, 1313 North Market Street, Suite 5400, Wilmington, Delaware 19801, Attn: William E. Chipman, Jr. (chipman@chipmanbrown.com); Robert A. Weber Desgrosseilliers (weber@chipmanbrown.com); Mark L. Olivere and Mark (desgross@chipmanbrown.com); (olivere@chipmanbrown.com); (c) the Office of the United States Trustee, 844 King Street, Suite 2207, Lockbox 35, Wilmington, Delaware 19801, Attn: Benjamin A. Hackman (Benjamin.a.hackman@usdoj.gov); (d) counsel to the Prepetition Lender and DIP Lender, (i) Blake, Cassels & Graydon LLP, 855 - 2nd Street S.W., Suite 3500, Bankers Hall East Tower, Bourassa Kelly 4J8, Attn: Alberta, T2P Calgary, (kelly.bourassa@blakes.com), Christopher Keliher Fleming Erik (christopher.keliher@blakes.com), (erik.fleming@blakes.com), and Austin Beck (Austin.Beck@blakes.com), (ii) Chapman and Cutler LLP, 320 South Canal Street, Chicago, Illinois 60606 (Attn: Stephen R. Tetro II and James P. Sullivan), email: stetro@chapman.com and jsulliva@chapman.com, and (iii) Womble Bond Dickinson (US) LLP, 1313 North Market Street, Suite 1200, Wilmington, Delaware 19801 (Attn: Matthew P. Ward), email: matthew-ward@wbdus.com; and (e) counsel to any statutory committee appointed in the Chapter 11 Cases.. The Debtors shall honor such request within five business days after the date the request is received by the Debtors, subject to the ability of the Debtors and any such requesting Utility Provider to resolve any dispute regarding such request without further order of the Court. To the extent a Utility Provider receives a disbursement from the Adequate Assurance Account, the Debtors shall replenish the Adequate Assurance Account in the amount disbursed.

- e. The portion of the Adequate Assurance Deposit attributable to each Utility Provider will be returned to the Debtors on the earlier of (i) reconciliation and payment by the Debtors of the Utility Provider's final invoice in accordance with applicable nonbankruptcy law following the Debtors' termination of Utility Services from such Utility Provider, (ii) the effective date of any Court order directing the dismissal of these Chapter 11 Cases, (iii) the effective date of any Court order directing the conversion of these Chapter 11 Cases to cases under chapter 7 of the Bankruptcy Code, or (iv) the effective date of any chapter 11 plan confirmed in these Chapter 11 Cases.
- f. Any Utility Provider that (i) objects to the Debtors' Proposed Adequate Assurance or (ii) desires additional assurances of payment in the form of deposits, prepayments, or otherwise must serve a request for additional assurance (an "Additional Assurance Request") on the Notice Parties.
- g. Any Additional Assurance Request must: (i) be in writing; (ii) identify the location for which the Utility Services are provided; (iii) summarize the

Debtors' payment history relevant to the affected account(s); (iv) certify the amount that is equal to one-half of the monthly Utility Services the Utility Provider supplied to the Debtors in the Prepetition Utility Period; (v) certify that the Utility Provider does not already hold a deposit equal to or greater than one-half of the monthly Utility Services; (vii) provide evidence that the Debtors have a direct obligation to the Utility Provider; and (viii) explain why the Utility Provider believes the Proposed Adequate Assurance is not sufficient adequate assurance of future payment.

- h. An Additional Assurance Request may be made at any time. If a Utility Provider fails to file and serve an Additional Assurance Request, the Utility Provider shall be (i) deemed to have received "satisfactory" adequate assurance of payment in compliance with section 366 of the Bankruptcy Code and (ii) forbidden from discontinuing, altering, or refusing Utility Services to, or discriminating against, the Debtors on account of any unpaid prepetition charges, or requiring additional assurance of payment other than the Proposed Adequate Assurance.
- i. Upon a Utility Provider's service of an Additional Assurance Request in accordance with these Adequate Assurance Procedures, the Debtors shall have 21 days from the receipt of such Additional Assurance Request (the "Resolution Period") to negotiate with the Utility Provider to resolve the Utility Provider's Additional Assurance Request.
- j. Without further order of the Court, the Debtors may enter into agreements granting additional adequate assurance to a Utility Provider serving an Additional Assurance Request if the Debtors determine that the Additional Assurance Request is reasonable.
- k. If the Debtors determine that the Additional Assurance Request is not reasonable and the Debtors are unable to reach an alternative resolution with the Utility Provider, the Debtors, during or immediately after the Resolution Period, will request a hearing (a "<u>Determination Hearing</u>") before the Court to determine the adequacy of assurance of payment with respect to that Utility Provider pursuant to section 366(c)(3) of the Bankruptcy Code.
- 1. Pending resolution of such dispute at a Determination Hearing, the relevant Utility Provider shall be prohibited from altering, refusing, or discontinuing service to the Debtors on account of: (i) unpaid charges for prepetition services; (ii) a pending Additional Assurance Request; or (iii) any objections filed in response to the Proposed Adequate Assurance.
- m. The Adequate Assurance Deposit deposited into the Adequate Assurance Account on behalf of any Utility Provider (including any additional amount deposited upon request of any applicable Utility Provider), or any portion thereof, shall be returned to the Debtors, <u>less</u> any amounts owed on account of unpaid, postpetition Utility Services, by no later than five business days

following the earlier of the date upon which (i) a chapter 11 plan becomes effective after being confirmed in these Chapter 11 Cases, (ii) a Court order dismissing these Chapter 11 Cases becomes effective, (iii) a Court order converting these Chapter 11 Cases to cases under chapter 7 of the Bankruptcy Code becomes effective, or (iv) the Debtors provide notice to a Utility Provider that services provided to the Debtors by such Utility Provider will no longer be needed or will be reduced.

- 4. The Utility Providers are prohibited from requiring additional adequate assurance of payment other than pursuant to the Adequate Assurance Procedures.
- 5. If the Debtors identify any Utility Provider following the entry of this Final Order, the Debtors shall (i) cause a copy of this Final Order, including the Adequate Assurance Procedures, to be served on any subsequently identified Utility Provider and any such Utility Provider shall be bound by the Adequate Assurance Procedures, (ii) file an updated list of Utility Providers, and (iii) increase the amount of Adequate Assurance, as needed.
- 6. Any deposit provided to a Utility Provider by the Debtors prior to the Petition Date must be returned to the Debtors within 21 days of receiving a notice that the services provided to the Debtors by such Utility Provider will no longer be needed.
- 7. Notwithstanding the relief granted in this Final Order and any actions taken pursuant to such relief, nothing in this Final Order shall be deemed: (a) an admission as to the validity of any particular claim against a Debtor entity; (b) a waiver of the Debtors' or any other party in interest's right to dispute any particular claim on any grounds; (c) a promise or requirement to pay any particular claim; (d) an implication or admission that any particular claim is of a type specified or defined in this Final Order or the Motion; (e) a request or authorization to assume any agreement, contract, or lease pursuant to section 365 of the Bankruptcy Code; (f) a waiver or limitation of the Debtors' or any other party in interest's rights under the Bankruptcy Code or any other applicable law; or (g) a concession by the Debtors or any other party in interest that any liens (contractual, common law, statutory, or otherwise) satisfied pursuant to this Final Order are valid

and the Debtors and all other parties in interest expressly reserve their rights to contest the extent, validity, or perfection or to seek avoidance of all such liens. Any payment made pursuant to this Final Order should not be construed as an admission as to the validity of any particular claim or a waiver of the Debtors' or any other party in interest's rights to subsequently dispute such claim. The relief granted by this Final Order is not intended to and does not grant the Debtors any authority to use cash collateral of Bank of Montreal or to use proceeds of any postpetition financing provided by Bank of Montreal in any way inconsistent with the provisions of any interim or final order of this Court approving the Debtors' use of cash collateral or authorizing the Debtors to obtain debtor-in-possession financing (a "DIP Order"). For the avoidance of doubt, in case of a conflict between the provisions of this Final Order and a DIP Order, the provisions of such DIP Order shall control.

- 8. The banks and financial institutions on which checks were drawn or electronic payment requests made in payment of the prepetition obligations approved herein are authorized to receive, process, honor, and pay all such checks and electronic payment requests when presented for payment, and all such banks and financial institutions are authorized to rely on the Debtors' designation of any particular check or electronic payment request as approved by this Final Order without any duty of further inquiry.
- 9. The Debtors are authorized to issue postpetition checks, or to effect postpetition fund transfer requests, in replacement of any checks or fund transfer requests that are dishonored as a consequence of these chapter 11 cases with respect to prepetition amounts owed in connection with any Utility Services.

- 10. Notice of the Motion as provided therein shall be deemed good and sufficient notice of such Motion and the requirements of Bankruptcy Rule 6004(a) and the Local Rules are satisfied by such notice.
- 11. Notwithstanding Bankruptcy Rule 6004(h), the terms and conditions of this Final Order are immediately effective and enforceable upon its entry.
- 12. The Debtors are authorized to take all actions necessary to effectuate the relief granted in this Final Order in accordance with the Motion.
- 13. This Court retains jurisdiction with respect to all matters arising from or related to the implementation, interpretation, and enforcement of this Final Order.

Dated: May 16th, 2023 Wilmington, Delaware

CRAIG T. GOLDBLATT UNITED STATES BANKRUPTCY JUDGE

my Dollar

This is Exhibit "____" referred to in the Affidavit of **MICHÈLE HAY**, sworn before me at Vancouver, BC, this 30th day of May, 2023

A Commissioner for taking Affidavits in British Columbia

IN THE UNITED STATES BANKRUPTCY COURT FOR THE DISTRICT OF DELAWARE

)	Re: Docket Nos. 7 and 37
Debtors.) (Jointly Administered)
STRUCTURLAM MASS TIMBER U.S., INC., et al., 1	Case No. 23-10497 (CTG
In re:) Chapter 11

FINAL ORDER (I) AUTHORIZING THE DEBTORS TO (A) PAY PREPETITION WAGES, SALARIES, OTHER COMPENSATION, AND REIMBURSABLE EXPENSES AND (B) CONTINUE CERTAIN EMPLOYEE BENEFITS PROGRAMS AND (II) GRANTING RELATED RELIEF

Upon the motion (the "Motion")² of the above-captioned debtors and debtors in possession (collectively, the "Debtors") for entry of a final order (this "Final Order"): (a) authorizing, but not directing, the Debtors to (i) pay certain prepetition employee wages, salaries, other compensation, reimbursable employee expenses, and (ii) continue employee benefits programs in the ordinary course, including payment of certain prepetition obligations related thereto; and (b) granting related relief; all as more fully set forth in the Motion and upon the First Day Declaration; and this Court having jurisdiction over this matter pursuant to 28 U.S.C. §§ 157 and 1334 and the Amended Standing Order of Reference from the United States District Court for the District of Delaware, dated February 29, 2012; and this Court having found that this is a core proceeding pursuant to 28 U.S.C. § 157(b)(2), and that this Court may enter a final order consistent with Article III of the United States Constitution; and this Court having found that venue of this proceeding and the

The Debtors in these chapter 11 cases, along with the last four digits of each Debtor's federal tax identification number or Canadian business number, as applicable, include: Structurlam Mass Timber U.S., Inc. (6287); Natural Outcomes, LLC (n/a); Structurlam Mass Timber Corporation (5050); and SLP Holdings Ltd. (3114). The location of the Debtors' headquarters is: 2176 Government Street, Penticton, British Columbia, Canada V2A 8B5. The address of the registered agent for Structurlam Mass Timber U.S., Inc. is: 8 The Green, Suite A, Dover, Delaware 19901.

² Capitalized terms used but not otherwise defined herein have the meanings given to such terms in the Motion.

Motion in this district is proper pursuant to 28 U.S.C. §§ 1408 and 1409; and this Court having found that the relief requested in the Motion is in the best interests of the Debtors' estates, their creditors, and other parties in interest; and this Court having found that the Debtors' notice of the Motion and opportunity for a hearing on the Motion were appropriate and no other notice need be provided; and this Court having reviewed the Motion and having heard the statements in support of the relief requested therein at a hearing before this Court (the "Hearing"); and this Court having determined that the legal and factual bases set forth in the Motion and at the Hearing establish just cause for the relief granted herein; and upon all of the proceedings had before this Court; and after due deliberation and sufficient cause appearing therefor, it is HEREBY ORDERED THAT:

- 1. The Motion is granted on a final basis as set forth herein.
- 2. The Debtors are authorized, but not directed, in their sole discretion, to: (a) continue, modify, change, and discontinue the Employee Compensation and Benefits in the ordinary course of business during these Chapter 11 Cases and without the need for further Court approval, subject to applicable law, and (b) pay and honor prepetition Employee Obligations in the ordinary course of business.
- 3. Pursuant to section 362(d) of the Bankruptcy Code, Employees are authorized to proceed with their claims under the Workers' Compensation Program in the appropriate judicial or administrative forum and the Debtors are authorized, but not directed, to continue the Workers' Compensation Program in the ordinary course. This modification of the automatic stay pertains solely to claims under the Workers' Compensation Program.
- 4. The Debtors are authorized, but not directed, to forward any unpaid amounts on account of Withholding Obligations to the appropriate third-party recipients or taxing authorities in accordance with the Debtors' prepetition policies and practices.

- 5. Notwithstanding anything to the contrary contained herein, with respect to the Severance Programs, no payments, absent further Court order, shall be made to or on behalf of any "insider" as that term is defined in section 101(31) of the Bankruptcy Code other than with respect to payments on account of Severance Programs that are required by applicable law.
- 6. Nothing herein shall be deemed to authorize the payment of any amounts that violate section 503(c) of the Bankruptcy Code.
- 7. Nothing contained herein is intended or should be construed to create an administrative priority claim on account of the Employee Compensation and Benefits obligations.
- 8. Notwithstanding the relief granted in this Final Order and any actions taken pursuant to such relief, nothing in this Final Order shall be deemed: (a) an admission as to the validity, priority, or amount of any particular claim against a Debtor entity; (b) a waiver of the Debtors' or any other party in interest's right to dispute any particular claim on any grounds; (c) a promise or requirement to pay any particular claim; (d) an implication or admission that any particular claim is of a type specified or defined in this Final Order or the Motion; (e) a request or authorization to assume any agreement, contract, or lease pursuant to section 365 of the Bankruptcy Code; (f) a waiver or limitation of the Debtors' or any other party in interest's rights under the Bankruptcy Code or any other applicable law; or (g) a concession by the Debtors or any other party in interest that any liens (contractual, common law, statutory, or otherwise) satisfied pursuant to this Final Order are valid and the Debtors and all other parties in interest expressly reserve their rights to contest the extent, validity, or perfection or to seek avoidance of all such liens. Any payment made pursuant to this Final Order should not be construed as an admission as to the validity, priority, or amount of any particular claim or a waiver of the Debtors' or any other party in interest's rights to subsequently dispute such claim. The relief granted by this Final Order

is not intended to and does not grant the Debtors any authority to use cash collateral of Bank of Montreal or to use proceeds of any postpetition financing provided by Bank of Montreal in any way inconsistent with the provisions of any interim or final order of this Court approving the Debtors' use of cash collateral or authorizing the Debtors to obtain debtor-in-possession financing (a "<u>DIP Order</u>"). For the avoidance of doubt, in case of a conflict between the provisions of this Final Order and a DIP Order, the provisions of such DIP Order shall control.

- 9. The banks and financial institutions on which checks were drawn or electronic payment requests made in payment of the prepetition obligations approved herein are authorized to receive, process, honor, and pay all such checks and electronic payment requests when presented for payment, and all such banks and financial institutions are authorized to rely on the Debtors' designation of any particular check or electronic payment request as approved by this Final Order without any duty of further inquiry and without liability for following the Debtors' instructions.
- 10. The Debtors are authorized to issue postpetition checks, or to effect postpetition fund transfer requests, in replacement of any checks or fund transfer requests that are dishonored as a consequence of these Chapter 11 Cases with respect to prepetition amounts owed in connection with the Employee Compensation and Benefits Programs.
- 11. Notwithstanding Bankruptcy Rule 6004(h), the terms and conditions of this Final Order are immediately effective and enforceable upon its entry.
- 12. The Debtors are authorized to take all actions necessary to effectuate the relief granted in this Final Order in accordance with the Motion.

13. This Court retains jurisdiction with respect to all matters arising from or related to the implementation, interpretation, and enforcement of this Final Order.

Dated: May 16th, 2023 Wilmington, Delaware CRAIG T. GOLDBLATT UNITED STATES BANKRUPTCY JUDGE

Cry Dellas

This is Exhibit "___" referred to in the Affidavit of MICHÈLE HAY, sworn before me at Vancouver, BC, this 30th day of May, 2023.

A Commissioner for taking Affidavits in British Columbia

IN THE UNITED STATES BANKRUPTCY COURT FOR THE DISTRICT OF DELAWARE

Debtors.	Re: Docket Nos. 14 and 40
Deletano) (Jointly Administered)
STRUCTURLAM MASS TIMBER U.S., INC., et al., 1) Case No. 23-10497 (CTG)
In re:) Chapter 11

FINAL ORDER (I) AUTHORIZING THE PAYMENT OF CERTAIN PREPETITION TAXES AND FEES AND (II) GRANTING RELATED RELIEF

Upon the motion (the "Motion")² of the above-captioned debtors and debtors in possession (collectively, the "Debtors") for entry of a final order (this "Final Order"): (a) authorizing the payment of certain prepetition taxes and fees; and (b) granting related relief; all as more fully set forth in the Motion and upon the First Day Declaration; and this Court having jurisdiction over this matter pursuant to 28 U.S.C. §§ 157 and 1334 and the Amended Standing Order of Reference from the United States District Court for the District of Delaware, dated February 29, 2012; and this Court having found that this is a core proceeding pursuant to 28 U.S.C. § 157(b)(2), and that this Court may enter a final order consistent with Article III of the United States Constitution; and this Court having found that venue of this proceeding and the Motion in this district is proper pursuant to 28 U.S.C. §§ 1408 and 1409; and this Court having found that the relief requested in the Motion is in the best interests of the Debtors' estates, their creditors, and other parties in interest; and this Court having found that the Debtors' notice of the Motion and opportunity for a

The Debtors in these chapter 11 cases, along with the last four digits of each Debtor's federal tax identification number or Canadian business number, as applicable, include: Structurlam Mass Timber U.S., Inc. (6287); Natural Outcomes, LLC (n/a); Structurlam Mass Timber Corporation (5050); and SLP Holdings Ltd. (3114). The location of the Debtors' headquarters is: 2176 Government Street, Penticton, British Columbia, Canada V2A 8B5. The address of the registered agent for Structurlam Mass Timber U.S., Inc. is: 8 The Green, Suite A, Dover, Delaware 19901.

² Capitalized terms used but not otherwise defined herein have the meanings given to such terms in the Motion.

hearing on the Motion were appropriate and no other notice need be provided; and this Court having reviewed the Motion and having heard the statements in support of the relief requested therein at a hearing before this Court (the "Hearing"); and this Court having determined that the legal and factual bases set forth in the Motion and at the Hearing establish just cause for the relief granted herein; and upon all of the proceedings had before this Court; and after due deliberation and sufficient cause appearing therefor, it is HEREBY ORDERED THAT:

- 1. The Motion is granted on a final basis as set forth herein.
- 2. The Debtors are authorized, but not directed, to: (a) negotiate, pay, and remit the obligations that arose or accrued in the ordinary course of business on account of the Taxes and Fees prior to the Petition Date and that will become due and owing in the ordinary course of business during the pendency of these Chapter 11 Cases at such time when the Taxes and Fees are payable in an aggregate amount not to exceed \$10,153 (CAD) and \$145,000 (USD) absent further order of this Court; and (b) negotiate and pay Taxes and Fees that arise or accrue in the ordinary course of business on a postpetition basis.
- 3. The Debtors are authorized, but not directed, to (a) participate in any amnesty program or (b) otherwise settle tax disputes pursuant to Bankruptcy Rule 9019.
- 4. Notwithstanding the relief granted in this Final Order and any actions taken pursuant to such relief, nothing in this Final Order shall be deemed: (a) an admission as to the validity, priority, or amount of any particular claim against a Debtor entity; (b) a waiver of the Debtors' or any other party in interest's right to dispute any particular claim on any grounds; (c) a promise or requirement to pay any particular claim; (d) an implication or admission that any particular claim is of a type specified or defined in this Final Order or the Motion; (e) a request or authorization to assume any agreement, contract, or lease pursuant to section 365 of the

Bankruptcy Code; (f) a waiver or limitation of the Debtors' or any other party in interest's rights under the Bankruptcy Code or any other applicable law; or (g) a concession by the Debtors or any other party in interest that any liens (contractual, common law, statutory, or otherwise) satisfied pursuant to this Final Order are valid and the Debtors and all other parties in interest expressly reserve their rights to contest the extent, validity, or perfection or to seek avoidance of all such liens. Any payment made pursuant to this Final Order should not be construed as an admission as to the validity, priority, or amount of any particular claim or a waiver of the Debtors' or any other party in interest's rights to subsequently dispute such claim. The relief granted by this Final Order is not intended to and does not grant the Debtors any authority to use cash collateral of Bank of Montreal or to use proceeds of any postpetition financing provided by Bank of Montreal in any way inconsistent with the provisions of any interim or final order of this Court approving the Debtors' use of cash collateral or authorizing the Debtors to obtain debtor-in-possession financing (a "DIP Order"). For the avoidance of doubt, in case of a conflict between the provisions of this Final Order and a DIP Order, the provisions of such DIP Order shall control.

- 5. The banks and financial institutions on which checks were drawn or electronic payment requests made in payment of the prepetition obligations approved herein are authorized to receive, process, honor, and pay all such checks and electronic payment requests when presented for payment, and all such banks and financial institutions are authorized to rely on the Debtors' designation of any particular check or electronic payment request as approved by this Final Order without any duty of further inquiry and without liability for following the Debtors' instructions.
- 6. The Debtors are authorized to issue postpetition checks, or to effect postpetition fund transfer requests, in replacement of any checks or fund transfer requests that are dishonored

as a consequence of these Chapter 11 Cases with respect to prepetition amounts owed in connection with the relief granted in this Final Order.

- 7. The Debtors are not authorized to pay any amounts on account of past-due taxes.
- 8. Notwithstanding Bankruptcy Rule 6004(h), the terms and conditions of this Final Order are immediately effective and enforceable upon its entry.
- 9. The Debtors are authorized to take all actions necessary to effectuate the relief granted in this Final Order in accordance with the Motion.
- 10. This Court retains jurisdiction with respect to all matters arising from or related to the implementation, interpretation, and enforcement of this Final Order.

Dated: May 16th, 2023 Wilmington, Delaware CRAIG T⊮GOLDBLATT UNITED STATES BANKRUPTCY JUDGE This is Exhibit "<u>F</u>" referred to in the Affidavit of **MICHÈLE HAY**, sworn before me at Vancouver, BC, this 30^{th} day of May, 2023.

A Commissioner for taking Affidavits in British Columbia

IN THE UNITED STATES BANKRUPTCY COURT FOR THE DISTRICT OF DELAWARE

Debtors.	Re: Docket No. 64
Deleteur) (Jointly Administered)
STRUCTURLAM MASS TIMBER U.S., INC., et al., 1) Case No. 23-10497 (CTG
In re:) Chapter 11
	}

ORDER (A) AUTHORIZING THE DEBTORS TO REJECT THE AMENDED AND RESTATED WALMART REALTY SUPPLIER AGREEMENT – GOODS NOT FOR RESALE, EFFECTIVE AS OF THE PETITION DATE,

AND (B) GRANTING RELATED RELIEF

This matter coming before the Court upon the Motion of the Debtors for Entry of an Order (A) Authorizing the Debtors to Reject the Amended and Restated Walmart Realty Supplier Agreement – Goods Not for Resale, Effective as of the Petition Date, and (B) Granting Related Relief (the "Motion"),² all as further described in the Motion; the Court having found that (i) the Court has jurisdiction over this matter pursuant to 28 U.S.C. §§ 157 and 1334 and the Amended Standing Order of Reference from the United States District Court for the District of Delaware, dated February 29, 2012, (ii) venue is proper in this district pursuant to 28 U.S.C. §§ 1408 and 1409, (iii) this is a core proceeding pursuant to 28 U.S.C. § 157(b), (iv) notice of the Motion is adequate and appropriate under the particular circumstances; the Court having reviewed the Motion and the First Day Declaration and having considered the statements of counsel and the

The Debtors in these chapter 11 cases, along with the last four digits of each Debtors' federal tax identification number or Canadian business number, as applicable, include: Structurlam Mass Timber U.S., Inc. (6287); Natural Outcomes, LLC (n/a); Structurlam Mass Timber Corporation (5050); and SLP Holdings Ltd. (3114). The location of the Debtors' headquarters is: 2176 Government Street, Penticton, British Columbia, Canada V2A 8B5. The address of the registered agent for Structurlam Mass Timber U.S., Inc. is: 8 The Green, Suite A, Dover, Delaware 19901.

² Capitalized terms used but not otherwise defined in this Order shall have the meanings ascribed to them in the Motion.

evidence adduced with respect to the Motion; after due deliberation, and good and sufficient cause having been shown and in order to avoid immediate and irreparable harm to the Debtors and their estates,

IT IS HEREBY ORDERED THAT:

- 1. The Motion is GRANTED as set forth herein.
- 2. The Supplier Agreement is rejected, but only to the extent it is an executory contract, pursuant to sections 105(a) and 365(a) of the Bankruptcy Code and Bankruptcy Rule 6006, effective as of the Petition Date.
- 3. The entry of this Order is not a determination on the merits that the terminated Supplier Agreement is an executory contract.
- 4. Any claim arising from the rejection of the Supplier Agreement shall be filed with the Debtors' claims and noticing agent on or before the later of (i) the deadline for filing proofs of claim established in these Chapter 11 Cases, if any, and (ii) the date that is thirty (30) days from the date of entry of this Order. The proof of claim form and instructions for submission may be obtained on the website of the Debtors' claim and noticing agent.
- 5. Notwithstanding the relief granted in this Order and any actions taken pursuant to such relief, nothing in this Order shall be deemed: (a) an admission as to the validity, priority, or amount of any particular claim against a Debtor entity; (b) a waiver of the Debtors' or any other party-in-interest's rights to dispute any particular claim on any grounds; (c) a promise or requirement to pay a particular claim; (d) an implication or admission that any particular claim is of a type specified or defined in this motion; or (e) a waiver or limitation of the Debtors' or any other party-in-interest's rights under the Bankruptcy Code or any other applicable law.

6. By entry of this Order, the Debtors do not waive any claims that they may have against Walmart, whether or not such claims arise under, are related to the rejection of, or are independent of the Supplier Agreement.

7. Notice of the Motion is deemed good and sufficient notice of such Motion, and the requirements of Bankruptcy Rule 6004(a) and the Local Rules of Bankruptcy Practice and Procedure of the United States Bankruptcy Court for the District of Delaware are waived by such notice.

8. Notwithstanding Bankruptcy Rule 6004(h), the terms and conditions of this Order are effective immediately and enforceable upon its entry.

9. The Debtors are authorized to take all actions necessary to effectuate the relief granted in this Order in accordance with the Motion.

10. This Court retains jurisdiction with respect to all matters arising from or related to the implementation, interpretation, or enforcement of this Order.

Dated: May 16th, 2023 Wilmington, Delaware CRAIG [†] GOLDBLATT UNITED STATES BANKRUPTCY JUDGE This is Exhibit "<u>____</u>" referred to in the Affidavit of **MICHÈLE HAY**, sworn before me at Vancouver, BC, this 30th day of May, 2023.

A Commissioner for taking Affidavits in British Columbia

IN THE UNITED STATES BANKRUPTCY COURT FOR THE DISTRICT OF DELAWARE

In re:) Chapter 11
STRUCTURLAM M.	ASS TIMBER U.S., INC., et al.,1) Case No. 23-10497 (CTG)
Debtors	Debtors.) (Jointly Administered)
DC	Jeolois.	Related Docket No. 63

ORDER (I) AUTHORIZING AND APPROVING PROCEDURES TO REJECT EXECUTORY CONTRACTS AND UNEXPIRED LEASES AND (II) GRANTING RELATED RELIEF

Upon the motion (the "Motion")² of the above-captioned debtors and debtors-in-possession (the "Debtors") for entry of an order (this "Order") (a) authorizing and approving procedures to reject executory contracts and unexpired leases and (b) granting related relief, all as more fully set forth in the Motion; and this Court having jurisdiction over this matter pursuant to 28 U.S.C. §§ 157 and 1334 and the Amended Standing Order of Reference, dated February 29, 2012; and consideration of the Motion and the relief requested therein being a core proceeding pursuant to 28 U.S.C. §§ 157(b); and this Court having found that venue of this proceeding and the Motion in this district is proper pursuant to 28 U.S.C. §§ 1408 and 1409; and this Court having found that the Debtors' notice of the Motion and opportunity for a hearing on the Motion were appropriate under the circumstances and no other notice need be provided; and this Court having determined that the legal and factual bases set forth in the Motion establish just cause for the relief

The Debtors in these chapter 11 cases, along with the last four digits of each Debtors' federal tax identification number or Canadian business number, as applicable, include: Structurlam Mass Timber U.S., Inc. (6287); Natural Outcomes, LLC (n/a); Structurlam Mass Timber Corporation (5050); and SLP Holdings Ltd. (3114). The location of the Debtors' headquarters is: 2176 Government Street, Penticton, British Columbia, Canada V2A 8B5. The address of the registered agent for Structurlam Mass Timber U.S., Inc. is: 8 The Green, Suite A, Dover, Delaware 19901.

² Capitalized terms used and not defined herein have the meanings given to them in the Motion.

granted herein; and upon all of the proceedings had before this Court; and after due deliberation and sufficient cause appearing therefor,

IT IS HEREBY ORDERED THAT:

- 1. The Motion is granted as set forth herein.
- 2. The following Rejection Procedures are approved in connection with rejecting Contracts and Leases:
 - (a) Rejection Notice. The Debtors shall file a notice substantially in the form attached to the Order as Exhibit 1 (the "Rejection Notice") indicating the Debtors' intent to reject a Contract(s) or Lease(s) pursuant to section 365 of the Bankruptcy Code, which Rejection Notice shall set forth, among other things: (i) the Contract(s) or Lease(s) to be rejected; (ii) the Debtor or Debtors party to such Contract(s) or Lease(s); (iii) the names and addresses of the counterparties to such Contract(s) or Lease(s) (each, a "Rejection Counterparty"); (iv) the proposed effective date of rejection for such Contract(s) or Lease(s) (the "Rejection Date"); and (v) the deadlines and procedures for filing objections to the Rejection Notice (as set forth below). The Rejection Notice may list multiple Contracts or Leases; provided that the number of counterparties to Contracts and Leases listed on each Rejection Notice shall be limited to no more than one hundred (100). To the extent that a Rejection Notice includes abandonment of property, subsection (h) shall also apply.
 - (b) Service of the Rejection Notice. The Debtors will cause the Rejection Notice to be served: (i) by email, if known, and by overnight delivery service upon the Rejection Counterparties affected by the Rejection Notice at the notice address provided in the applicable Contract or Lease (and upon such Rejection Counterparty's counsel, if known); and (ii) by first class mail, email, or fax, upon (a) the U.S. Trustee, (b) counsel to the Prepetition Lender and the DIP Lender, (c) counsel to any Committee appointed in these Chapter 11 Cases, (d) any party that has requested notice pursuant to Bankruptcy Rule 2002, and (e) third parties known to have an interest in any Abandoned Property that is subject to such Rejection Notice (collectively, the "Master Notice Parties").
 - (c) Objection Procedures. Parties objecting to a proposed rejection or abandonment described in a Rejection Notice must file and serve a written objection³ so that such objection is filed with this Court on the docket of these Chapter 11 Cases no later than fourteen (14) days after the date the Debtors' file and serve the relevant Rejection Notice (the "Rejection Objection Deadline") and promptly serve such objection on the Master Notice Parties and the following parties (collectively, the "Objection Service Parties): (i) proposed counsel to the Debtors, Chipman Brown

An objection to the rejection of any particular Contract or Lease listed on a Rejection Notice shall not constitute an objection to the rejection of any other Contract or Lease listed on such Rejection Notice.

Cicero & Cole, LLP, Hercules Plaza, 1313 North Market Street, Suite 5400, William Chipman, Ε. Delaware 19801, Attn: (chipman@chipmanbrown.com); Robert A. Weber (weber@chipmanbrown.com); Mark L. Desgrosseilliers (desgross@chipmanbrown.com); and Mark D. Olivere (olivere@chipmanbrown.com); (ii) counsel to any Committee appointed in these Chapter 11 Cases; (iii) counsel to the Prepetition Lender and the DIP Lender, (a) Blake, Cassels & Graydon LLP, 855 - 2nd Street S.W., Suite 3500, Bankers Hall Kelly Bourassa Alberta, T₂P 4J8, Attn: Calgary, Tower, Keliher Christopher (kelly.bourassa@blakes.com), (christopher.keliher@blakes.com), Erik Fleming (erik.fleming@blakes.com), and Austin Beck (Austin Beck@blakes.com), (b) Chapman and Cutler LLP, 320 South Canal Street, Chicago, Illinois 60606, Attn: Stephen R. Tetro II and James P. Sullivan, (stetro@chapman.com) and (jsulliva@chapman.com), and (c) Womble Bond Dickinson (US) LLP, 1313 North Market Street, Suite 1200, Wilmington, Delaware 19801, Attn: Matthew P. Ward, (matthew.ward@wbd-us.com); and (iv) the Office of the United States Trustee for the District of Delaware, 855 King Street, Suite 2207, Lockbox 35, Wilmington, Delaware 19801, Attn: Benjamin A. Hackman (Benjamin.a.hackman@usdoj.gov).

- No Objection Timely Filed. If no objection to the rejection of any Contract or (d) Lease or abandonment of any Abandoned Property described in a Rejection Notice is timely filed, each Contract or Lease listed therein shall be rejected as of the applicable Rejection Date, or such other date as the Debtors and the applicable Rejection Counterparty agrees; provided, however, that the Rejection Date for a rejection of a lease of non-residential real property shall not occur until the later of (i) the Rejection Date set forth in the Rejection Notice and (ii) the date the Debtors relinquish control of the property by notifying the affected party in writing of the Debtors' surrender of the property and (A) turning over keys, key codes, and security codes, if any, to the affected party or (B) notifying the affected party in writing that the keys, key codes, and security codes, if any, are not available but the party may reclaim the leased property; provided, further, that the Rejection Date for a lease of non-residential real property rejected pursuant to these Rejection Procedures shall not occur earlier than the date the Debtors filed and served the applicable Rejection Notice.
- (e) Unresolved Timely Objection. If an objection to a Rejection Notice is timely filed and properly served as specified above and not withdrawn or resolved, the Debtors shall schedule a hearing on such objection and shall provide at least fourteen (14) days' notice of such hearing to the applicable Rejection Counterparty and the other Objection Service Parties. If such objection is overruled or withdrawn, such Contract or Lease shall be rejected as of the applicable Rejection Date set forth in the Rejection Notice or such other date to which the Debtors and the applicable Rejection Counterparty agree, or as ordered by the Court.
- (f) Removal from Schedule. The Debtors reserve the right to remove any Contract or Lease from the schedule to a Rejection Notice at any time prior to the Rejection Date, and shall notify the counterparty in writing of such removal.

- (g) No Application of Deposits. If the Debtors have deposited monies with a Rejection Counterparty as a security deposit or other arrangement, such Rejection Counterparty may not set off or recoup or otherwise use such deposit without the prior approval of the Court, unless the Debtors and the applicable Rejection Counterparty otherwise agree.
- Abandoned Property. The Debtors are authorized, but not directed, at any time (h) through and including the applicable Rejection Date, to remove or abandon any of the Debtors' personal property that may be located on the Debtors' leased premises that are subject to a rejected Contract (the "Abandoned Property"). The Debtors shall provide a general description of any known Abandoned Property in the applicable Rejection Notice, shall use reasonable good faith efforts to (i) identify therein third parties with a known interest in such Abandoned Property, if any, and (ii) describe such Abandoned Property in sufficient detail to permit identification of the same, and shall serve such Rejection Notice on any third parties known to have an interest in such Abandoned Property. Absent a timely objection, and with the consent of the Debtors' DIP Lenders and Prepetition Lenders or as otherwise permitted by the Credit Documents, any and all property located on the Debtors' leased premises on the Rejection Date of the applicable Lease of non-residential real property shall be deemed abandoned pursuant to section 554 of the Bankruptcy Code, as is, effective as of the Rejection Date. ⁴ After the Rejection Date, contract counterparties to leases may, in their sole discretion and without further notice or order of this Court, utilize and/or dispose of such property without liability to the Debtors or (to the extent consistent with due process) third parties and, to the extent applicable, the automatic stay is modified to allow such disposition. Notwithstanding the foregoing, the personal property that the Debtors are authorized to abandon hereunder shall not include documents (including electronic documents) that are known to contain personally identifying information or business records relating to potential estate claims, causes of action or other litigation.
- (i) **Proofs of Claim.** Claims arising out of the rejection of Contract(s) or Lease(s), if any, must be filed on or before the later of (i) the deadline for filing proofs of claim established in these Chapter 11 Cases, if any, and (ii) 30 days after the later of (A) if no objection is timely filed, the Rejection Objection Deadline, and (B) if an objection is timely filed, the date that all such filed objections have either been overruled or withdrawn, or (C) the date the Court enters an Order, if any, approving the applicable rejection. If no proof of claim is timely filed, such claimant shall be forever barred from asserting a claim for damages arising from the rejection and from participating in any distributions on such a claim that may be made in connection with these Chapter 11 Cases.

If the Rejection Date specified in the Rejection Notice with respect to any Contract that is a lease of nonresidential real property predates the removal of any property not otherwise generally described in the Rejection Notice, the Rejection Date will not become effective until such date the property is removed from the leased premises.

- 3. Approval of the Rejection Procedures and this Order will not prevent the Debtors from seeking to reject a Contract or Lease by separate motion.
- 4. Notwithstanding the relief granted in this Order and any actions taken pursuant to such relief, nothing in this Order shall be deemed: (a) an admission as to the validity, priority, or amount of any particular claim against a Debtor entity; (b) a waiver of the Debtors' or any other party-in-interest's right to dispute any particular claim on any grounds; (c) a promise or requirement to pay any particular claim; (d) an implication or admission that any particular claim is of a type specified or defined in this Order or the Motion; (e) a waiver or limitation of the Debtors' or any other party-in-interest's rights under the Bankruptcy Code or any other applicable law; or (f) a concession by the Debtors or any other party-in-interest that any liens (contractual, common law, statutory, or otherwise) satisfied pursuant to this Order are valid and the Debtors and all other parties-in-interest expressly reserve their rights to contest the extent, validity, or perfection or to seek avoidance of all such liens. Any payment made pursuant to this Order should not be construed as an admission as to the validity, priority, or amount of any particular claim or a waiver of the Debtors' or any other party-in-interest's rights to subsequently dispute such claim.
- 5. All rights and defenses of the Debtors are preserved, including all rights and defenses of the Debtors with respect to a claim for damages arising as a result of a Contract or Lease rejection, including any right to assert an offset, recoupment, counterclaim, or deduction. In addition, nothing in this Order or the Motion shall limit the Debtors' ability to subsequently assert that any particular Contract or Lease is terminated and is no longer an executory contract or unexpired lease.

45

Case 23-10497-CTG Doc 115 Filed 05/16/23 Page 6 of 6

6. The Debtors are authorized to take all actions necessary to effectuate the relief

granted pursuant to this Order in accordance with the Motion, the Rejection Notices, and the

Rejection Procedures.

7. Notice of the Motion as provided therein shall be deemed good and sufficient notice

of such Motion, and the requirements of Bankruptcy Rule 6004(a) and the Local Rules are satisfied

by such notice.

8. Notwithstanding Bankruptcy Rule 6004(h), the terms and conditions of this Order

are immediately effective and enforceable upon its entry.

9. The Debtors are authorized to take all actions necessary to effectuate the relief

granted in this Order in accordance with the Motion.

10. This Court retains exclusive jurisdiction with respect to all matters arising from or

related to the implementation, interpretation, and enforcement of this Order.

Dated: May 16th, 2023 Wilmington, Delaware CRAIG T. GOLDBLATT

UNITED STATES BANKRUPTCY JUDGE

1 Musico

EXHIBIT 1

IN THE UNITED STATES BANKRUPTCY COURT FOR THE DISTRICT OF DELAWARE

)	Related Docket No.
Debtors.	(Jointly Administered)
STRUCTURLAM MASS TIMBER U.S., INC., et al., 1	Case No. 23-10497 (CTG)
In re:	Chapter 11

NOTICE OF REJECTION OF CERTAIN EXECUTORY CONTRACTS AND/OR UNEXPIRED LEASES

PARTIES RECEIVING THIS NOTICE SHOULD LOCATE THEIR NAMES AND THEIR CONTRACTS OR LEASES ON SCHEDULE 2 ATTACHED HERETO AND READ THE CONTENTS OF THIS NOTICE CAREFULLY

PLEASE TAKE NOTICE that on ______, 2023 the United States Bankruptcy Court for the District of Delaware (the "Court") entered an order on the motion (the "Motion")² of debtors and debtors-in-possession (the "Debtors"), approving procedures for the rejection of executory contracts and unexpired leases and granting related relief [Docket No. _____] (the "Procedures Order") attached hereto as Schedule 1.³

PLEASE TAKE FURTHER NOTICE that, pursuant to the Procedures Order and by this written notice (this "Rejection Notice"), the Debtors hereby notify you that they have determined,

The Debtors in these chapter 11 cases, along with the last four digits of each Debtors' federal tax identification number or Canadian business number, as applicable, include: Structurlam Mass Timber U.S., Inc. (6287); Natural Outcomes, LLC (n/a); Structurlam Mass Timber Corporation (5050); and SLP Holdings Ltd. (3114). The location of the Debtors' headquarters is: 2176 Government Street, Penticton, British Columbia, Canada V2A 8B5. The address of the registered agent for Structurlam Mass Timber U.S., Inc. is: 8 The Green, Suite A, Dover, Delaware 19901.

² Capitalized terms used and not otherwise defined herein have the meanings given to them in the Motion.

The Procedures Order is also available on the website of the Debtors' claims and noticing agent at https://kccllc.net/structurlam.

in the exercise of their business judgment, that each Contract or Lease set forth on <u>Schedule 2</u> attached hereto is hereby rejected effective as of the date (the "Rejection Date") set forth in <u>Schedule 2</u>, or such other date as the Debtors and the counterparty or counterparties to any such Contract or Lease agree.⁴

PLEASE TAKE FURTHER NOTICE that parties seeking to object to the proposed rejection of any of the Contracts or Leases must file and serve a written objection so that such objection is filed with the Court on the docket of the Debtors' Chapter 11 Case no later than fourteen (14) days after the date that the Debtors served this Notice and promptly serve such objection on the following parties: (i) proposed counsel to the Debtors, Chipman Brown Cicero & Cole, LLP, Hercules Plaza, 1313 North Market Street, Suite 5400, Wilmington, Delaware 19801, Attn: Weber Robert (chipman@chipmanbrown.com); William E. Chipman, Jr. (weber@chipmanbrown.com); Mark L. Desgrosseilliers (desgross@chipmanbrown.com); and Mark D. Olivere (olivere@chipmanbrown.com); (ii) counsel to any Committee appointed in these Chapter 11 Cases; (iii) counsel to the Prepetition Lender and the DIP Lender, (a) Blake, Cassels & Graydon LLP, 855 - 2nd Street S.W., Suite 3500, Bankers Hall East Tower, Calgary, Alberta, Kelly Bourassa (kelly.bourassa@blakes.com), Christopher Keliher T2P 4J8, Attn: (christopher.keliher@blakes.com), Erik Fleming (erik.fleming@blakes.com), and Austin Beck (Austin Beck@blakes.com), (b) Chapman and Cutler LLP, 320 South Canal Street, Chicago, Illinois 60606, Attn: Stephen R. Tetro II and James P. Sullivan, (stetro@chapman.com) and (jsulliva@chapman.com), and (c) Womble Bond Dickinson (US) LLP, 1313 North Market Street,

An objection to the rejection of any particular Contract or Lease listed in this Rejection Notice shall not constitute an objection to the rejection of any other Contract or Lease listed in this Rejection Notice. Any objection to the rejection of any particular Contract or Lease listed in this Rejection Notice must state with specificity the Contract or Lease to which it is directed. For each particular Contract or Lease whose rejection is not timely or properly objected to, such rejection will be effective in accordance with this Rejection Notice and the Procedures Order.

Suite 1200, Wilmington, Delaware 19801, Attn: Matthew P. Ward, (matthew.ward@wbd-us.com); and (iv) the Office of the United States Trustee for the District of Delaware, 855 King Street, Suite 2207, Lockbox 35, Wilmington, Delaware 19801, Attn: Benjamin A. Hackman (Benjamin.a.hackman@usdoj.gov).

PLEASE TAKE FURTHER NOTICE that, absent an objection being timely filed, the rejection of each Contract or Lease shall become effective on the applicable Rejection Date set forth in <u>Schedule 2</u>, or such other date as the Debtors and the counterparty or counterparties to such Contract or Lease agree.

PLEASE TAKE FURTHER NOTICE that, if an objection to the rejection of any Contract or Lease is timely filed and not withdrawn or resolved, the Debtors shall file a notice for a hearing to consider the objection for the Contract(s) or Lease(s) to which such objection relates. If such objection is overruled or withdrawn, such Contract or Lease shall be rejected as of the applicable Rejection Date set forth in <u>Schedule 2</u> attached hereto or such other date as the Debtors and the counterparty or counterparties to any such Contract or Lease agree.

PLEASE TAKE FURTHER NOTICE that, pursuant to the terms of the Procedures Order, if the Debtors have deposited monies with a Contract or Lease counterparty as a security deposit or other arrangement, the Contract or Lease counterparty may not set off or recoup or otherwise use such monies without further order of the Court, unless the Debtors and the counterparty or counterparties to such Contracts and Leases otherwise agree.

PLEASE TAKE FURTHER NOTICE that, absent timely objection, any personal property of the Debtors that remains on any of the premises subject to a rejected Contract or Lease shall be deemed abandoned as of the Rejection Date.

PLEASE TAKE FURTHER NOTICE that to the extent you wish to assert a claim with respect to rejection of your Contract(s) or Lease(s), you must do so by the later of (i) the deadline for filing proofs of claim established in these Chapter 11 Cases, if any, and (ii) thirty (30) calendar days after the later of (A) if no objection is timely filed, the Rejection Objection Deadline, and (B) if an objection is timely filed, the date that all such filed objections have either been overruled or withdrawn, or (C) the date the Court enters an order, if any, approving the applicable rejection. IF YOU FAIL TO TIMELY SUBMIT A PROOF OF CLAIM IN THE APPROPRIATE FORM BY THE DEADLINE SET FORTH HEREIN, YOU WILL BE, FOREVER BARRED, ESTOPPED, AND ENJOINED FROM (1) ASSERTING SUCH CLAIM AGAINST THE DEBTORS AND THEIR CHAPTER 11 ESTATES, (2) VOTING ON ANY CHAPTER 11 PLAN OF REORGANIZATION FILED IN THESE CASES ON ACCOUNT OF SUCH CLAIM, AND (3) PARTICIPATING IN ANY DISTRIBUTION IN THE DEBTORS' CHAPTER 11 CASES ON ACCOUNT OF SUCH CLAIM.

Dated: ______, 2023 Wilmington, Delaware CHIPMAN BROWN CICERO & COLE, LLP

/s/ DRAFT

William E. Chipman, Jr. (No. 3818)

Robert A. Weber (No. 4013)

Mark L. Desgrosseilliers (No. 4083)

Mark D. Olivere (No. 4291)

Hercules Plaza

1313 North Market Street, Suite 5400

Wilmington, Delaware 19801

Telephone:

(302) 295-0191

Email:

chipman@chipmanbrown.com

weber@chipmanbrown.com desgross@chipmanbrown.com olivere@chipmanbrown.com

Proposed Counsel to Debtors and Debtors in Possession

This is Exhibit "H" referred to in the Affidavit of MICHÈLE HAY, sworn before me at Vancouver, BC, this 30th day of May, 2023.

A Commissioner for taking Affidavits in British Columbia

IN THE UNITED STATES BANKRUPTCY COURT FOR THE DISTRICT OF DELAWARE

In re:) Chapter 11
STRUCTURLAM MASS TIMBER U.S., INC., et al., 1) Case No. 23-10497 (CTG)
Debtors.)) (Jointly Administered)
	Re: Docket Nos. 13 and 43

FINAL ORDER AUTHORIZING, BUT NOT DIRECTING, THE DEBTORS TO PAY CERTAIN PREPETITION CLAIMS OF CRITICAL VENDORS, SHIPPERS, AND 503(B)(9) CLAIMANTS AND GRANTING RELATED RELIEF

Upon the motion (the "Motion")² of the above-captioned debtors and debtors in possession (collectively, the "Debtors") for entry of a final order (this "Final Order"): (a) authorizing, but not directing, the Debtors to pay certain prepetition amounts owing to Critical Vendors, Shippers, and 503(b)(9) Claimants (collectively, the "Critical Vendor Claims") in an amount not to exceed the Final Order Cap, absent further order of the Court; and (b) granting related relief; all as more fully set forth in the Motion and upon the First Day Declaration; and this Court having jurisdiction over this matter pursuant to 28 U.S.C. §§ 157 and 1334 and the Amended Standing Order of Reference from the United States District Court for the District of Delaware, dated February 29, 2012; and this Court having found that this is a core proceeding pursuant to 28 U.S.C. § 157(b)(2), and that this Court may enter a final order consistent with Article III of the United States Constitution; and this Court having found that venue of this proceeding and the Motion in this district is proper

The Debtors in these chapter 11 cases, along with the last four digits of each Debtor's federal tax identification number or Canadian business number, as applicable, include: Structurlam Mass Timber U.S., Inc. (6287); Natural Outcomes, LLC (n/a); Structurlam Mass Timber Corporation (5050); and SLP Holdings Ltd. (3114). The location of the Debtors' headquarters is: 2176 Government Street, Penticton, British Columbia, Canada V2A 8B5. The address of the registered agent for Structurlam Mass Timber U.S., Inc. is: 8 The Green, Suite A, Dover, Delaware 19901.

Capitalized terms used but not otherwise defined herein have the meanings given to such terms in the Motion.

pursuant to 28 U.S.C. §§ 1408 and 1409; and this Court having found that the relief requested in the Motion is in the best interests of the Debtors' estates, their creditors, and other parties in interest; and this Court having found that the Debtors' notice of the Motion and opportunity for a hearing on the Motion were appropriate and no other notice need be provided; and this Court having reviewed the Motion and having heard the statements in support of the relief requested therein at a hearing before this Court (the "Hearing"); and this Court having determined that the legal and factual bases set forth in the Motion and at the Hearing establish just cause for the relief granted herein; and upon all of the proceedings had before this Court; and after due deliberation and sufficient cause appearing therefor, it is HEREBY ORDERED THAT:

- 1. The Motion is granted on a final basis as set forth herein.
- 2. The Debtors are authorized, but not directed, in their sole discretion, to pay all or part of, and discharge, on a case-by-case basis, the Critical Vendor Claims in an aggregate amount not to exceed \$750,000 (CAD) on a final basis, absent further order of the Court.
- 3. Nothing herein shall impair or prejudice the Debtors' ability to contest, in their sole discretion, the extent, perfection, priority, validity, or amounts of any claims held by any Shipper, Critical Vendor, or 503(b)(9) Claimant. The Debtors do not concede that any claims satisfied pursuant to this Final Order are valid, and the Debtors expressly reserve all rights to contest the extent, validity, or perfection or to seek the avoidance of all such liens or the priority of such claims.
- 4. Notwithstanding the relief granted in this Final Order and any actions taken pursuant to such relief, nothing in this Final Order shall be deemed: (a) an admission as to the validity, priority, or amount of any particular claim against a Debtor entity; (b) a waiver of the Debtors' or any other party in interest's right to dispute any particular claim on any grounds; (c) a

promise or requirement to pay any particular claim; (d) an implication or admission that any particular claim is of a type specified or defined in this Final Order or the Motion; (e) a request or authorization to assume any agreement, contract, or lease pursuant to section 365 of the Bankruptcy Code; (f) a waiver or limitation of the Debtors' or any other party in interest's rights under the Bankruptcy Code or any other applicable law; or (g) a concession by the Debtors or any other party in interest that any liens (contractual, common law, statutory, or otherwise) satisfied pursuant to this Final Order are valid and the Debtors and all other parties in interest expressly reserve their rights to contest the extent, validity, or perfection or to seek avoidance of all such liens. Any payment made pursuant to this Final Order should not be construed as an admission as to the validity, priority, or amount of any particular claim or a waiver of the Debtors' or any other party in interest's rights to subsequently dispute such claim. The relief granted by this Final Order is not intended to and does not grant the Debtors any authority to use cash collateral of Bank of Montreal or to use proceeds of any postpetition financing provided by Bank of Montreal in any way inconsistent with the provisions of any interim or final order of this Court approving the Debtors' use of cash collateral or authorizing the Debtors to obtain debtor-in-possession financing (a "DIP Order"). For the avoidance of doubt, in case of a conflict between the provisions of this Final Order and a DIP Order, the provisions of such DIP Order shall control.

5. The banks and financial institutions on which checks were drawn or electronic payment requests made in payment of the prepetition obligations approved herein are authorized to receive, process, honor, and pay all such checks and electronic payment requests when presented for payment, and all such banks and financial institutions are authorized to rely on the Debtors' designation of any particular check or electronic payment request as approved by this Final Order without any duty of further inquiry and without liability for following the Debtors' instructions.

The Debtors are authorized to issue postpetition checks, or to effect postpetition 6.

fund transfer requests, in replacement of any checks or fund transfer requests that are dishonored

as a consequence of these Chapter 11 Cases with respect to prepetition amounts owed in

connection with any Shipper Charge, Critical Vendor Claim, or 503(b)(9) Claim.

Notwithstanding Bankruptcy Rule 6004(h), the terms and conditions of this Final 7.

Order are immediately effective and enforceable upon its entry.

8. The Debtors are authorized to take all actions necessary to effectuate the relief

granted in this Final Order in accordance with the Motion.

9. This Court retains jurisdiction with respect to all matters arising from or related to

the implementation, interpretation, and enforcement of this Final Order.

Dated: May 17th, 2023

Wilmington, Delaware

Cry Daller

CRAIG T. GOLDBLATT

UNITED STATES BANKRUPTCY JUDGE

This is Exhibit "<u>I</u>" referred to in the Affidavit of **MICHÈLE HAY**, sworn before me at Vancouver, BC, this 30th day of May, 2023.

A Commissioner for taking Affidavits in British Columbia

IN THE UNITED STATES BANKRUPTCY COURT FOR THE DISTRICT OF DELAWARE

Debuis.	Related Docket Nos. 5 and 36
Debtors.) (Jointly Administered)
STRUCTURLAM MASS TIMBER U.S., INC., et al., 1) Case No. 23-10497 (CTG)
In re:) Chapter 11

FINAL ORDER (I) AUTHORIZING DEBTORS AND DEBTORS
IN POSSESSION TO (A) OBTAIN POSTPETITION FINANCING, (B) USE
CASH COLLATERAL, (C) GRANT LIENS AND SUPER-PRIORITY
CLAIMS, AND (D) GRANT ADEQUATE PROTECTION; (II) MODIFYING
THE AUTOMATIC STAY; AND (III) GRANTING RELATED RELIEF

Upon the motion (the "Motion")² of the above-captioned debtors and debtors in possession (collectively, the "Debtors") pursuant to sections 105, 361, 362, 363, 364(c)(1), 364(c)(2) and 364(c)(3), 364(d), 503, 506, 507 and 552 of title 11 of the United States Code (the "Bankruptcy Code"), Rules 2002, 4001, 9013 and 9014 of the Federal Rules of Bankruptcy Procedure (as amended, the "Bankruptcy Rules"), and Rules 2002-1, 4001-1, 4001-2 and 9013-1 of the Local Rules of Bankruptcy Practice and Procedure for the United States Bankruptcy Court for the District of Delaware (the "Local Rules"), inter alia seeking, among other things:

(1) authorization (a) for Structurlam Mass Timber U.S., Inc., a Delaware corporation, as debtor and debtor in possession and Structurlam Mass Timber Corporation, a corporation

Capitalized terms used but not defined in this Final Order shall have the same meanings ascribed to such terms in the DIP Credit Agreement.



The Debtors in these chapter 11 cases, along with the last four digits of each Debtor's federal tax identification number or Canadian business number, as applicable, include: Structurlam Mass Timber U.S., Inc. (6287); Natural Outcomes, LLC (n/a); Structurlam Mass Timber Corporation (5050); and SLP Holdings Ltd. (3114). The location of the Debtors' headquarters is: 2176 Government Street, Penticton, British Columbia, Canada V2A 8B5. The address of the registered agent for Structurlam Mass Timber U.S., Inc. is: 8 The Green, Suite A, Dover, Delaware 19901.

governed by the laws of the Province of British Columbia, as debtor and debtor in possession (each individually, a "Borrower" and collectively, "Borrowers"), to obtain, and (b) for each Borrower, SLP Holdings Ltd., a corporation governed by the laws of the Province of British Columbia ("SLP") and Natural Outcomes, LLC, a Delaware limited liability company ("Natural" and together with SLP, each a "Guarantor" and collectively the "Guarantors" and together with the Borrowers, each a "Loan Party" and collectively the "Loan Parties") to guarantee, unconditionally, on a joint and several basis, post-petition financing in the form of a revolving credit facility in accordance with the terms and conditions set forth in the Senior Secured Super-Priority Debtorin-Possession Financing Credit Agreement, substantially in the form attached hereto as Exhibit 1 (as amended, supplemented or otherwise modified from time to time in accordance with the terms and conditions set forth herein, the "DIP Credit Agreement"), by and among the Borrowers, the Guarantors, and Bank of Montreal (the "DIP Lender"), and the other Documents (as defined in the DIP Credit Agreement and referred to herein as the "DIP Loan Documents"), and in accordance with this order (this "Final Order"), secured by perfected senior priority security interests in and liens on the Collateral (as defined herein) pursuant to sections 364(c)(2) and 364(c)(3), and 364(d) of the Bankruptcy Code (subject only to the Carve-Out and the Permitted Liens (each as defined herein));

authorization for the Loan Parties to apply all collections, asset proceeds and payments in accordance with the Agreed Budget and DIP Credit Agreement (i) first to the Pre-Petition Lender for application or deemed application to up to Cdn. \$3,900,000 of Pre-Petition Obligations (as defined herein) until such obligations are fully repaid in accordance with the Pre-Petition Credit Agreement (as defined herein) and other Pre-Petition Loan Documents (as defined herein) and (ii) to the DIP Lender for application and repayment of all Post-Petition Obligations

(as defined herein) in accordance with the DIP Credit Agreement and the other DIP Loan Documents;

- (3) authorization for the Debtors to grant superpriority administrative claim status, pursuant to section 364(c)(1) of the Bankruptcy Code, to the DIP Lender in respect of all Post-Petition Obligations (subject to the Carve-Out (as defined herein));
- (4) as set forth below, approval of certain stipulations by the Debtors as set forth in this Final Order in connection with the Pre-Petition Credit Agreement and Pre-Petition Loan Documents;
- (5) as set forth below, authorization to provide adequate protection to Pre-Petition Lender;
- (6) authorization to use the Cash Collateral (as defined in section 363 of the Bankruptcy Code) subject to the terms of this Final Order;
- (7) the waiver of the Debtors' right to assert claims to surcharge against the Collateral (as defined herein) or the Pre-Petition Collateral (as defined herein) pursuant to section 506(c) of the Bankruptcy Code;
- (8) the modification of the automatic stay imposed by section 362 of the Bankruptcy Code to the extent necessary to implement and effectuate the terms and provisions of this Final Order to the extent hereinafter set forth; and
 - (9) related relief.

The initial hearing on the Motion having been held by this Court on April 26, 2023 (the "Interim Hearing"), and upon the record made by the Debtors at the Interim Hearing, including the Motion, the Declaration of Matthew Karmel, Interim Chief Executive Officer of the Debtors, in Support of Debtors' Chapter 11 Petitions and First Day Motions [ECF No. 9] (the "First Day

Declaration") and the Declaration of Kevin Haggard in Support of Debtors' Motion for Entry of Interim and Final Orders (I) Authorizing Debtors and Debtors In Possession to (A) Obtain Postpetition Financing and (B) Use Cash Collateral, (C) Granting Liens and Superpriority Administrative Expense Claims, (D) Grant Adequate Protection; (II) Modifying Automatic Stay, (III) Scheduling a Final Hearing; and (IV) Granting Related Relief [ECF No. 6] (the "DIP Declaration", and together with the First Day Declaration, collectively, the "Declarations"); and the Court having entered the Interim Order (I) Authorizing Debtors and Debtors In Possession to (A) Obtain Postpetition Financing and (B) Use Cash Collateral, (C) Granting Liens and Superpriority Administrative Expense Claims, (D) Grant Adequate Protection; (II) Modifying Automatic Stay, (III) Scheduling a Final Hearing; and (IV) Granting Related Relief [ECF No. 30] (the "Interim Order"); and the final hearing on the Motion having been held by this Court on May 19, 2023 (the "Final Hearing"); and upon the record made by the Debtors at the Final Hearing, including the Motion and the filings and pleadings in the above-captioned chapter 11 cases (the "Chapter 11 Cases"), the Court having found that the relief requested in the Motion is in the best interests of the Debtors, their estates, their creditors and other parties in interest; and appropriate notice of the Motion, the relief requested therein, and the Final Hearing (the "Notice") having been provided; and the Notice having been served by the Debtors in accordance with Bankruptcy Rule 4001 on (i) counsel to the DIP Lender and Pre-Petition Lender; (ii) the office of the United States Trustee (the "U.S. Trustee"), (iii) the holders of the 30 largest unsecured claims, on a consolidated basis, against the Debtors' estates (the "30 Largest Unsecured Creditors"), (iv) the Internal Revenue Service and applicable state taxing authorities, (v) the United States Attorney's Office for the District of Delaware; (vi) any party that has asserted or may assert a lien in the Debtors' assets; and (vii) all parties entitled to notice pursuant to Local Rules 2002-1(b) and 90131(m) (collectively, the "<u>Noticed Parties</u>"), and the opportunity for a hearing on the Motion was appropriate and no other notice need be provided; and after due deliberation sufficient cause appearing therefor;

BASED UPON THE RECORD ESTABLISHED AT THE FINAL HEARING BY THE DEBTORS, INCLUDING THE SUBMISSIONS OF THE DECLARATIONS, THE FACTS IN THE MOTION AND THE REPRESENTATIONS OF COUNSEL, THE COURT HEREBY MAKES THE FOLLOWING FINDINGS OF FACT AND CONCLUSIONS OF LAW:³

- A. <u>Petition.</u> On April 21, 2023 (the "<u>Petition Date</u>"), each Debtor filed a voluntary petition (each, a "<u>Petition</u>") under chapter 11 of the Bankruptcy Code. The Debtors continue to operate their businesses and manage their properties as debtors-in-possession pursuant to sections 1107(a) and 1108 of the Bankruptcy Code.
- B. <u>Disposition</u>. The Motion is hereby granted in accordance with and to the extent set forth in this Final Order. Any objections to the Motion with respect to the entry of this Final Order that have not been withdrawn, waived, resolved, or settled are hereby denied and overruled.
- C. <u>Jurisdiction and Venue</u>. The Court has jurisdiction of this proceeding and the parties and property affected hereby pursuant to 28 U.S.C. §§ 157(b) and 1334, as well as the Amended Standing Order of Reference from the United States District Court for the District of Delaware, dated February 29, 2012. The Motion is a "core" proceeding as defined in 28 U.S.C. §§ 157(b)(2)(A), (D) and (M). This Court may enter a final order consistent with Article III of the United States Constitution. Venue of the Chapter 11 Cases and the Motion in this Court is proper

The findings and conclusions set forth herein constitute the Court's findings of fact and conclusions of law pursuant to Bankruptcy Rule 7052. To the extent any findings of fact constitute conclusions of law, they are adopted as such. To the extent any conclusions of law constitute findings of fact, they are adopted as such.

pursuant to 28 U.S.C. §§ 1408 and 1409. The statutory and legal predicates for the relief sought herein are sections 105, 361, 362, 363, 364, 503, 506, 507 and 552 of the Bankruptcy Code and Bankruptcy Rules 2002, 4001, 9013 and 9014 and the applicable provisions of the Local Rules.

- D. <u>Committee Formation</u>. On May 3, 2023, the U.S. Trustee appointed an official committee of unsecured creditors in these Chapter 11 Cases pursuant to section 1102 of the Bankruptcy Code (a "Committee").⁴
- E. <u>Notice</u>. Notice of the Motion has been provided in accordance with the Bankruptcy Code, the Bankruptcy Rules, and the Local Rules, and no other or further notice of the Motion with respect to the relief requested at the Final Hearing or the entry of this Final Order shall be required.
- F. <u>Acknowledgments and Agreements of Debtors</u>. After consultation with their attorneys and financial advisors, and without prejudice to the rights of the Committee or other parties-in-interest as and to the extent set forth in Section 4.1 of this Final Order, the Debtors, on their behalf and on behalf of their estates, admit, stipulate, acknowledge and agree that (collectively, the "Stipulations"):
- (i) <u>Pre-Petition Loan Documents</u>. Prior to the commencement of the Chapter 11 Cases, the Pre-Petition Lender (as defined herein) made loans, advances and provided other financial accommodations to the Loan Parties pursuant to the terms and conditions set forth in (1) that certain Amended and Restated Letter Agreement dated December 3, 2019 among the Borrowers as borrowers, SLP and Natural as guarantors, and Bank of Montreal, as lender (the "<u>Pre-Petition Lender</u>") as amended by a waiver and first amendment agreement dated

⁴ See Notice of Appointment of Creditors' Committee [ECF No. 65] and Amended Notice of Appointment of Creditors' Committee [ECF No. 73].

as of February 2, 2021, a waiver and second amendment agreement dated as of September 29, 2021, a waiver and third amendment agreement dated as of January 24, 2022, a fourth amendment agreement dated as of June 17, 2022, a fifth amendment agreement dated as of June 27, 2022, a consent and sixth amending agreement dated as of February 7, 2023, a seventh amending agreement dated as of March 3, 2023, and an eighth amending agreement dated as of April 21, 2023 (collectively, the "Pre-Petition Credit Agreement"); and (2) all other agreements, documents and instruments executed and/or delivered with, to, or in favor of the Pre-Petition Lender in connection with the Pre-Petition Credit Agreement, including, without limitation, all security agreements, notes, guarantees, mortgages, Uniform Commercial Code financing statements, PPSA financing statements, and all other related agreements, documents and instruments executed and/or delivered in connection therewith or related thereto (all of the foregoing, together with the Pre-Petition Credit Agreement, as all of the same have heretofore been amended, supplemented, modified, extended, renewed, restated and/or replaced at any time prior to the Petition Date, collectively, the "Pre-Petition Loan Documents"). Copies of the operative Pre-Petition Loan Documents are in the possession of counsel to the Debtors and available upon reasonable request.

(ii) <u>Pre-Petition Loan Obligations</u>. As of the Petition Date, the Loan Parties were indebted to the Pre-Petition Lender under the Pre-Petition Loan Documents in respect of (a) a committed revolving credit facility in the maximum principal amount of Cdn. \$3,000,000, (b) a non-revolving term credit facility in the original maximum principal amount of Cdn. \$21,333,317, (c) a revolving letter of credit facility in the maximum principal amount of Cdn. \$413,000, (d) a committed, reducing, non-revolving term credit facility in the original maximum principal amount of US \$35,529,007, (e) a MasterCard Facility providing corporate credit card services in the maximum principal amount of US \$50,000 and Cdn. \$100,000 and (f) an

uncommitted non-revolving demand facility in the maximum principal amount of Cdn. \$900,000 (each as described and defined in the Pre-Petition Credit Agreement) in an aggregate outstanding principal amount of not less than US \$50,065,515.23,5 plus interest accrued and accruing thereon, together with all costs, fees, expenses (including attorneys' fees and legal expenses) and other charges accrued, accruing or chargeable with respect thereto, each in accordance with the terms of the Pre-Petition Credit Agreement (collectively, the "Pre-Petition Obligations", as such term is further defined in the DIP Credit Agreement). The Pre-Petition Obligations constitute allowed, legal, valid, binding, enforceable and non-avoidable obligations of the Loan Parties, and are not subject to any offset, defense, counterclaim, avoidance, recharacterization or subordination pursuant to the Bankruptcy Code or any other applicable law; the Loan Parties do not possess, shall not assert, hereby forever release, and are forever barred from bringing any claim, counterclaim, setoff or defense of any kind, nature or description, in any such case, arising out of, connected with, or relating to any and all acts, omissions or events occurring prior to the entry of this Final Order, which would in any way affect the validity, enforceability and non-avoidability of any of the Pre-Petition Obligations or liens and security interest securing the same described in clause (E)(iii) below.

Pre-Petition Collateral. As of the Petition Date, the Pre-Petition Obligations were fully secured pursuant to the Pre-Petition Loan Documents by valid, perfected, enforceable and non-avoidable first-priority security interests and liens granted by the Loan Parties to Pre-Petition Lender under the Pre-Petition Loan Documents, upon all of the Pre-Petition Collateral (as defined in the DIP Credit Agreement and hereinafter referred to as the "Pre-Petition Collateral").

Reflects conversion of Canadian dollar amounts to US dollar amounts calculated as of April 20, 2023.

Petition Collateral"), subject only to the liens specifically permitted under Section 9.2(b) of the Pre-Petition Credit Agreement to the extent that such security interests, liens or encumbrances are (1) valid, perfected, and unavoidable senior priority liens and security interests existing as of the Petition Date securing valid, binding and unavoidable debt permitted under the Pre-Petition Loan Documents, and (2) senior to and have not been and are not subject to being subordinated to Pre-Petition Lender's liens on and security interests in the Pre-Petition Collateral under the Pre-Petition Loan Documents or otherwise avoided, and, in each instance, only for so long as and to the extent that such encumbrances are and remain senior and outstanding (hereinafter referred to as the "Permitted Liens"). None of the Loan Parties possess or will assert any claim, counterclaim, setoff or defense of any kind, nature or description which would in any way affect the validity, enforceability and non-avoidability of any of Pre-Petition Lender's liens, claims or security interests in the Pre-Petition Collateral.

- (iv) Subject to section 363(k) of the Bankruptcy Code, Section 4.1 of this Final Order, and the rights of the DIP Lender, the Pre-Petition Lender has the right to "credit bid" the amount of its claims that are Pre-Petition Obligations arising under the terms of the Pre-Petition Loan Documents, during any sale of all or substantially all of the Debtors' assets, including without limitation, sales occurring pursuant to section 363 of the Bankruptcy Code or included as part of any restructuring plan subject to confirmation under section 1129(b)(2)(A)(ii)-(iii) of the Bankruptcy Code.
- (v) <u>Proof of Claim.</u> The acknowledgment by the Loan Parties of the Pre-Petition Obligations and the liens, rights, priorities and protections granted to or in favor of Pre-Petition Lender in respect of the Pre-Petition Collateral as set forth herein and in the Pre-

Petition Loan Documents shall be deemed a timely filed proof of claim on behalf of Pre-Petition Lender in each of these Chapter 11 Cases.

G. Findings Regarding the Post-Petition Financing.

(i) <u>Post-Petition Financing</u>. The Loan Parties have requested from the DIP Lender, and the DIP Lender is willing to extend, certain loans, advances and other financial accommodations on the terms and conditions set forth in this Final Order, the DIP Credit Agreement and the other DIP Loan Documents, in a principal amount of up to Cdn. \$4,000,000 during the Interim Financing Period and up to Cdn. \$7,500,000 following entry of this Final Order.

Need for Post-Petition Financing. The Debtors do not have (ii) sufficient available sources of working capital, including Cash Collateral, to operate their businesses in the ordinary course of business and fund these Chapter 11 Cases without access to the financing provided under the DIP Credit Agreement as requested in the Motion and the authority to use Cash Collateral pursuant to the terms of this Final Order. The Debtors' ability to maintain business relationships with their vendors, suppliers and customers, to pay their employees, and to otherwise fund their operations and these Chapter 11 Cases is essential to the Debtors' continued viability as the Debtors seek to maximize the value of the assets of the Estates (as defined herein) for the benefit of all creditors of the Debtors. The ability of the Debtors to obtain sufficient working capital and liquidity through the proposed post-petition financing arrangements with the DIP Lender as set forth in this Final Order, the DIP Credit Agreement, and other DIP Loan Documents, as applicable, is vital to the preservation and maintenance of the going concern value of each Debtor. Accordingly, the Debtors have an immediate need to obtain the post-petition financing in order to, among other things, permit the orderly continuation of the operation of their businesses, minimize the disruption of their business operations, and preserve and maximize the value of the assets of the Debtors' bankruptcy estates (as defined under section 541 of the Bankruptcy Code, the "Estates").

No Credit Available on More Favorable Terms. The Debtors are unable to procure financing in the form of unsecured credit allowable as an administrative expense under sections 364(a), 364(b), or 503(b)(1) of the Bankruptcy Code or in exchange for the grant of a superpriority administrative expense, junior liens on encumbered property of the Estates, or liens on property of the Estates not subject to a lien pursuant to sections 364(c)(1), 364(c)(2) or 364(c)(3) of the Bankruptcy Code, without granting to the DIP Lender, the DIP Liens and the DIP Loan Superpriority Claim (each as defined herein) and incurring the Adequate Protection Obligations (as defined herein), in each case as provided for herein subject to the Carve-Out to the extent set forth herein, under the terms and conditions set forth in this Final Order and in the DIP Loan Documents. The Debtors have been unable to procure the necessary financing on terms more favorable, taken as a whole, than the financing offered by the DIP Lender pursuant to the DIP Loan Documents.

to the DIP Lender an initial 13-week budget (the "<u>DIP Budget</u>"), a summary of which is attached to this Final Order as **Exhibit 2**. Such DIP Budget has been thoroughly reviewed by the Loan Parties and their management and sets forth, among other things, the projected cash receipts and disbursements of the Loan Parties for the periods covered thereby. The DIP Lender is relying upon the Loan Parties' compliance with the DIP Budget (subject to the variances permitted under the DIP Credit Agreement including Sections 8.1(p), 8.1(s), 8.2(r) of the DIP Credit Agreement and the definition of "Agreed Budget" therein (the "<u>Permitted Variances</u>")) and related covenants in accordance with the DIP Credit Agreement including Sections 8.1 and 8.2 of the DIP Credit

Agreement, the other DIP Loan Documents, and this Final Order in determining to enter into the post-petition financing arrangements provided for herein.

- Section 364(e). The extensions of credit under the DIP Credit Agreement and the other DIP Loan Documents and this Final Order are fair, just and reasonable under the circumstances, are ordinary and appropriate for secured financing to debtors-in-possession, reflect the Debtors' exercise of their prudent business judgment consistent with their fiduciary duties, and are supported by reasonably equivalent value and fair consideration. The terms and conditions of the DIP Credit Agreement and the other DIP Loan Documents and this Final Order have been negotiated in good faith and at arms' length by and among the Loan Parties and the DIP Lender with all parties being represented by counsel. Any credit extended under the terms of the Interim Order and this Final Order shall be deemed to have been extended in good faith by the DIP Lender, as that term is used in section 364(e) of the Bankruptcy Code.
- (vi) <u>Good Cause</u>. The relief requested in the Motion is necessary, essential and appropriate, and is in the best interest of and will benefit the Debtors, their creditors and their Estates, as its implementation will, among other things, provide the Debtors with the necessary liquidity to (1) minimize disruption to the Debtors' businesses and operations, (2) preserve and maximize the value of the Debtors' Estates for the benefit of all the Debtors' creditors, and (3) avoid immediate and irreparable harm to the Debtors, their creditors, their businesses, their employees, and their assets.
- (vii) <u>Immediate Entry</u>. Sufficient cause exists for immediate entry of this Final Order pursuant to Bankruptcy Rule 4001(c)(2). No party appearing in the Chapter 11 Cases has filed or made an objection to the relief sought in the Motion or the entry of

this Final Order, or any objections that were made (to the extent such objections have not been withdrawn, waived, resolved, or settled) are hereby overruled.

Based upon the foregoing, and after due consideration and good cause appearing therefor; IT IS HEREBY ORDERED THAT:

Section 1. Authorization and Conditions to Financing.

- 1.1 <u>Motion Granted</u>. The Motion is granted in accordance with Bankruptcy Rule 4001(c)(2) to the extent provided in this Final Order. Except as otherwise expressly provided in this Final Order, any objection to the entry of this Final Order that has not been withdrawn, waived, resolved or settled, is hereby denied and overruled on the merits.
- hereby authorized and empowered to immediately borrow and obtain Loans and to incur indebtedness and other obligations (collectively, the "Post-Petition Obligations"), and each Loan Party is hereby authorized to guaranty such Post-Petition Obligations in an amount of (i) up to Cdn. \$7,500,000 in principal amount, plus (ii) all interest, costs, and fees, accrued or accruing under the DIP Credit Agreement and other DIP Loan Documents, all pursuant to the terms and conditions of this Final Order, the DIP Credit Agreement, the other DIP Loan Documents, during the period commencing on the date of the Interim Order through and including the Maturity Date (the "Financing Period"). Subject to the terms and conditions contained in this Final Order and the DIP Loan Documents, Borrowers shall use the proceeds of the Loans and other credit and financial accommodations provided by the DIP Lender under the DIP Credit Agreement and the other DIP Loan Documents solely for payment of expenses set forth in the DIP Budget and amounts owing to the DIP Lender in accordance with the terms and conditions of the DIP Loan Documents and this Final Order.

1.3 Financing Documents

- Authorization. The Loan Parties are hereby authorized to enter into, (a) execute, deliver, perform, and comply with all of the terms, conditions and covenants of the DIP Credit Agreement and the other DIP Loan Documents. Upon execution and delivery of the DIP Credit Agreement and the other DIP Loan Documents, such agreements and documents shall constitute valid and binding obligations of the Loan Parties, enforceable against each Loan Party in accordance with the terms of such agreements, documents and this Final Order. No obligation, payment, transfer or grant of security under the DIP Credit Agreement, the other DIP Loan Documents or this Final Order shall be stayed, restrained, voidable, or recoverable under the Bankruptcy Code or under any applicable state, federal or foreign law (including, without limitation, under chapter 5 of the Bankruptcy Code or under any applicable state Uniform Fraudulent Transfer Act, Uniform Fraudulent Conveyance Act or similar statute or foreign law) or be subject to any defense, reduction, setoff, counterclaim, recoupment, offset, recharacterization, subordination (whether equitable, contractual or otherwise), cross-claims or any other challenge under the Bankruptcy Code or any applicable law, rule or regulation by any person or entity, other than a Challenge (as defined below) to the Roll-Up (as defined below) pursuant to Section 4.1 of this Final Order.
- (b) Approval; Evidence of Borrowing Arrangements. All terms, conditions and covenants set forth in the DIP Loan Documents (including, without limitation, the DIP Credit Agreement) are approved on a final basis. All such terms, conditions and covenants shall be sufficient and conclusive evidence of (a) the borrowing arrangements by and among the Loan Parties and the DIP Lender, and (b) each Loan Party's agreement to all of the terms, conditions, and covenants of the DIP Credit Agreement and the other DIP Loan Documents for all purposes, including, without limitation, to the extent applicable, the payment of all Post-Petition

Obligations arising thereunder, including, without limitation, all principal, interest, fees and other expenses, including, without limitation, all of the DIP Lender's consultant fees, professional fees, attorney fees and legal expenses, whether incurred pre-petition or post-petition, as more fully set forth in the DIP Loan Documents.

- Amendment. Subject to the terms and conditions of the DIP Credit (c) Agreement and the other DIP Loan Documents, the Loan Parties and the DIP Lender may amend, modify, supplement or waive any provision of the DIP Loan Documents (a "DIP Loan Amendment") without further approval or order of the Court, so long as (a) such DIP Loan Amendment is not material (for purposes hereof, a "material" DIP Loan Amendment shall mean any DIP Loan Amendment that operates to increase the interest rate other than as currently provided in the DIP Loan Documents, increase the maximum credit available pursuant to the DIP Credit Agreement, increase or add an additional early termination fee, prepayment or repayment premium, add specific new events of default or enlarge the nature and extent of default remedies available to the DIP Lender following an event of default) and is undertaken in good faith by the DIP Lender and the Loan Parties; (b) the Debtors provide prior written notice of the DIP Loan Amendment (the "DIP Loan Amendment Notice") to (i) the U.S. Trustee, and (ii) counsel to the Committee; (c) the Debtors file the DIP Loan Amendment Notice with the Court; and (d) no objection to the DIP Loan Amendment is filed with the Court within five (5) business days from the date the DIP Loan Amendment Notice is filed with the Court in accordance with this Paragraph. Any material DIP Loan Amendment to the DIP Loan Documents must be approved by the Court to be effective.
- 1.4 <u>Payment of Pre-Petition Debt</u>. The Debtors are authorized to pay all Pre-Petition Obligations in accordance with the Pre-Petition Credit Agreement (subject to Paragraph

4.1(b) of this Final Order), the DIP Credit Agreement, the other DIP Loan Documents to the extent authorized by this Final Order, including, without limitation, Paragraph 1.5 of this Final Order, provided that the Roll-Up (as defined below) shall be subject to the Challenge Period (as defined below).

1.5 Application of Payments and Collateral Proceeds; Roll-Up.

Cash Collateral and proceeds of Collateral in excess of the aggregate (a) receipts forecast in the initial DIP Budget attached hereto as Exhibit 2 shall be applied, first, to repayment of the Pre-Petition Obligations up to the amount of Cdn. \$3,900,000, (the "Roll-Up") (subject to the Challenge Period (as defined below) described in Section 4.1 of this Final Order), and, second, to the DIP Indebtedness until paid in full, in accordance with the Payment Protocol as defined in the DIP Credit Agreement. Except to the extent permitted otherwise pursuant to the preceding sentence, all proceeds of Pre-Petition Collateral shall be applied to the Pre-Petition Indebtedness (in accordance with the Pre-Petition Loan Documents) until paid in full; provided, however, that the Debtors shall not make any non-ordinary course payments to the Pre-Petition Lender that are not expressly authorized herein absent further order of the Court. The Debtors shall not have the right to direct the manner of application of any payments to the Pre-Petition Lender or the DIP Lender or any other receipts by the Pre-Petition Lender or the DIP Lender of proceeds of any of the Pre-Petition Collateral or Collateral other than in the manner set forth in this ordering paragraph, the Pre-Petition Credit Agreement and the DIP Credit Agreement. The Debtors are authorized to make all payments and transfers of Estate property to the DIP Lender as provided for, permitted and/or required under the DIP Credit Agreement and the other DIP Loan Documents, and any such payments and transfers (other than the Roll-Up (as defined below)) shall not be avoidable or recoverable from the DIP Lender under sections 547, 548, 550, 553 or any other section of the Bankruptcy Code, or by reason of any other claim, charge, assessment, or other liability, whether by application of the Bankruptcy Code, other law or otherwise.

- (b) Without limiting the generality of Paragraph 1.5(a), the Debtors are authorized without further order of this Court to pay or reimburse the DIP Lender for all costs and expenses, including, without limitation, all professional fees, consultant fees and legal fees and expenses paid or incurred by the DIP Lender in connection with the financing transactions as provided in this Final Order and the DIP Loan Documents, whether incurred pre-petition or postpetition, all of which shall be and are included as part of the principal amount of the Post-Petition Obligations and secured by the Collateral (as defined herein); provided that any such legal fees shall be paid only in accordance with the procedures set forth in Paragraph 5.12 of this Final Order.
- 1.6 Continuation of Pre-Petition Procedures. Except to the extent expressly set forth in the DIP Loan Documents or any cash management order entered in these Chapter 11 Cases, all pre-petition practices and procedures for the payment and collection of proceeds of the Collateral, the turnover of cash, including any deposit account control agreement (a "Control Agreement") and any other similar lockbox or blocked depository bank account arrangements, are hereby approved, ratified and shall continue without interruption after the commencement of the Chapter 11 Cases and are further approved and deemed applicable for the payment and collection of proceeds of the Collateral, the turnover of cash, and the delivery of property to the DIP Lender in accordance with the terms of the DIP Loan Documents.

Section 2. <u>Post-Petition Lien; Superpriority Administrative Claim Status.</u>

2.1 <u>Post-Petition Liens</u>. To secure the prompt payment and performance of any and all Post-Petition Obligations of the Loan Parties to the DIP Lender of whatever kind, nature or description, absolute or contingent, now existing or hereafter arising, the following valid,

binding, continuing, enforceable and non-avoidable security interests and liens (all security interests and liens granted to the DIP Lender, pursuant to this Final Order and the DIP Loan Documents, the "<u>DIP Liens</u>") are hereby granted to the DIP Lender (all property identified in clauses (a) through (f) below, including any property recovered as a result of transfers or obligations avoided or actions maintained or taken pursuant to sections 542, 544, 545, 547, 548, 549, 550, 551, 553(b) or 724(a) of the Bankruptcy Code ("<u>Avoidance Actions</u>" and the proceeds thereof, the "Avoidance <u>Proceeds</u>") being collectively referred to as the "<u>Collateral</u>"):

- (a) <u>Priming Liens on Encumbered Property.</u> Pursuant to section 364(d)(1) of the Bankruptcy Code, valid and perfected first priority security interests and liens, superior to all other liens, claims or security interests that any creditor of any of the Debtors' Estates may have (the "<u>Pre-Petition Liens</u>") (but subject to the Carve-Out (as defined herein) and the Permitted Liens), in and upon all tangible and intangible pre- and post-petition property of the Debtors that is subject to a security interest or lien as of the Petition Date, including the Pre-Petition Collateral;
- (b) <u>Liens on Unencumbered Property</u>. Pursuant to section 364(c)(2) of the Bankruptcy Code, a valid, binding, continuing, enforceable, fully-perfected first priority senior security interest in and lien upon all tangible and intangible pre- and post-petition property of the Debtors, whether existing on the Petition Date or thereafter acquired, that, on, after or as of the Petition Date, is not subject to a valid, perfected and non-avoidable lien (collectively, the "<u>Unencumbered Property</u>"), including any and all unencumbered cash of the Debtors and any investment of such cash, inventory, accounts receivable, other rights to payment whether arising before or after the Petition Date, claims, causes of action, commercial tort claims, contracts, properties, plants, fixtures, machinery, equipment, general intangibles, documents, instruments,

securities, chattel paper, interests in leaseholds, real properties, deposit accounts, patents, intellectual property, copyrights, trademarks, trade names, rights under license agreements and other intellectual property, capital stock of subsidiaries, and including the Debtors' rights in any deposit or escrowed funds delivered by any proposed buyer of assets of any of the Debtors, whether delivered prior to or after the Petition Date, and including any Funded Reserve Account (as defined below) or any similar account and the contents thereof, in each case wherever located and whenever arising, and the proceeds, products, rents and profits of the foregoing, whether arising under section 552(b) of the Bankruptcy Code or otherwise, of all the foregoing, in each case other than Avoidance Actions but including the Avoidance Proceeds; provided, however, that the DIP Lender will be paid from all Collateral other than Avoidance Proceeds before receiving payment from the Avoidance Proceeds, and will only seek and receive payment from the Avoidance Proceeds to the extent that payment from all other Collateral does not result in the Payment in Full of all Post-Petition Obligations;

- (c) <u>Lien Priority in Collateral</u>. The liens and security interests of the DIP Lender granted under the DIP Loan Documents and this Final Order in the Collateral securing all Post-Petition Obligations shall be first and senior in priority to all other interests and liens of every kind, nature and description, whether created consensually, by an order of the Court or otherwise, including, without limitation, liens or interests granted in favor of third parties in conjunction with sections 363, 364 or any other section of the Bankruptcy Code or other applicable law; provided, however, that the DIP Lender's liens on and security interests in the Collateral shall be subject only to (a) the Permitted Liens and (b) the Carve-Out (as defined herein).
- (d) <u>Right of Repayment</u>. The right to repayment of the DIP Lender granted under the DIP Loan Documents and this Final Order from the sale or other disposition of

the Collateral or any proceeds thereof shall be first and senior in priority to all other rights of repayment of every kind, nature and description, subject in each case to the Permitted Liens.

Post-Petition Lien Perfection. This Final Order shall be sufficient (e) and conclusive evidence of the priority, perfection and validity of the post-petition liens and security interests granted herein, effective as of the Petition Date (and, in the case of the Roll-Up, subject to the Challenge Period (as defined below) described in Section 4.1 of this Final Order), without any further act and without regard to any other federal, state or local requirements or law requiring notice, filing, registration, recording or possession of the Collateral, or other act to validate or perfect such security interest or lien, including without limitation, control agreements with any financial institution(s) party to a Control Agreement with respect to any deposit account or securities account held by such financial institution(s) or other depository account consisting of Collateral (a "Perfection Act"). Notwithstanding the foregoing, if the DIP Lender, shall, in its sole discretion, elect for any reason to file, record or otherwise effectuate any Perfection Act, then the DIP Lender is authorized to perform such act, and the Debtors are authorized to perform such act to the extent necessary or required by the DIP Lender, which act or acts shall be deemed to have been accomplished as of the date and time of entry of this Final Order notwithstanding the date and time actually accomplished, and in such event, the subject filing or recording office is authorized to accept, file or record any document in regard to such act in accordance with applicable law. The DIP Lender may choose to file, record or present a certified copy of this Final Order in the same manner as a Perfection Act, which shall be tantamount to a Perfection Act, and, in such event, the subject filing or recording office is authorized to accept, file or record such certified copy of this Final Order in accordance with applicable law. Should the DIP Lender so choose and attempt to file, record or perform a Perfection Act, no defect or failure in connection with such attempt shall in any way limit, waive or alter the validity, enforceability, attachment, or perfection of the post-petition liens and security interests granted herein by virtue of the entry of this Final Order.

Nullifying Pre-Petition Restrictions to Post-Petition Financing. (f) Notwithstanding anything to the contrary contained in any pre-petition agreement, contract, lease, document, note or instrument to which any Debtor is a party or under which any Debtor is obligated, except as otherwise permitted under the DIP Loan Documents, any provision that restricts, limits or impairs in any way any Debtor from granting the DIP Lender security interests in or liens upon any of the Debtors' assets or properties (including, among other things, any antilien granting or anti-assignment clauses in any leases or other contractual arrangements to which any Debtor is a party) under the DIP Loan Documents or this Final Order, as applicable, or otherwise entering into and complying with all of the terms, conditions and provisions hereof or of the DIP Loan Documents, shall not (a) be effective and/or enforceable against any such Debtor(s), or the DIP Lender, as applicable, or (b) adversely affect the validity, priority or enforceability of the liens, security interests, claims, rights, priorities and/or protections granted to the DIP Lender pursuant to this Final Order or the DIP Loan Documents, in each case, to the maximum extent permitted under the Bankruptcy Code and other applicable law that would not render any unexpired lease or executory contract non-assumable or non-assignable pursuant to section 365 of the Bankruptcy Code.

2.2 Superpriority Administrative Expenses.

(a) <u>DIP Loans</u>. For all Post-Petition Obligations now existing or hereafter arising pursuant to this Final Order, the DIP Loan Documents or otherwise, the DIP Lender is granted an allowed superpriority administrative claim pursuant to section 364(c)(1) of

the Bankruptcy Code, having priority in right of payment over any and all other obligations, liabilities and indebtedness of Debtors (other than the Carve-Out), whether now in existence or hereafter incurred by Debtors, and over any and all administrative expenses or priority claims of the kind specified in, or ordered pursuant to, inter alia, sections 105, 326, 328, 330, 331, 364(c)(1), 503(b), 506(c), 507(a), 507(b), 546(c), 726, 1113 or 1114 of the Bankruptcy Code, whether or not such expenses or claims may become secured by a judgment lien or other non-consensual lien, levy or attachment, which allowed superpriority administrative claim shall be payable from all prepetition and post-petition property of the Debtors and all proceeds thereof (the "DIP Loan Superpriority Claim"). Notwithstanding the foregoing, any superpriority administrative claim granted to the DIP Lender on account of the Roll-Up shall remain subject to the Challenge Period (as defined below).

2.3 Carve-Out.

(a) As used in this Final Order, the "Carve-Out" means the sum of (i) all fees required to be paid to the Clerk of the Court and to the Office of the United States Trustee under 28 U.S.C. § 1930(a) plus interest at the statutory rate through the Carve-Out Termination Date (defined below) (collectively, the "Statutory Fees"); (ii) all reasonable fees and expenses up to \$25,000 incurred by a trustee under section 726(b) of the Bankruptcy Code; (iii) to the extent allowed at any time, whether by interim order, procedural order, or other order of this Court, all unpaid fees and expenses (the "Allowed Professional Fees") incurred by persons or firms retained by the Debtors pursuant to sections 327, 328, or 363 of the Bankruptcy Code (the "Debtor Professionals"), including any Debtor Professionals appointed to act on behalf of the Debtors before the Supreme Court of British Columbia and retained by the Debtors in such capacity under the foregoing provisions of the Bankruptcy Code, and the Committee pursuant to sections 328 or

1103 of the Bankruptcy Code (the "Committee Professionals" and, together with the Debtor Professionals, the "Professional Persons") at any time before or on the last day of the week in which an Enforcement Notice is served, less the amount of any remaining and unapplied fee retainers held by any such Professional Person pursuant to sections 327, 328 or 1103 of the Bankruptcy Code, provided, that the Allowed Professional Fees of the Committee Professionals shall not at any time exceed the aggregate amount of the fees and expenses identified in the DIP Budget for each such Committee Professional covering the period of time commencing on the Petition Date through the earlier of (A) the date of determination, and (B) the last day of the week in which an Enforcement Notice is served; provided, further, that solely in the event that a sale of all or substantially all of the Debtors' assets does not close by June 30, 2023 (unless such delay or failure to close is caused by or supported by the Committee), the Carve-Out for the Allowed Professional Fees of the Committee Professionals incurred prior to the last day of the week in which an Enforcement Notice is served shall not exceed the greater of (a) \$500,000 and (b) the lesser of (1) the amount in the Funded Reserve Account after payment of all Allowed Professional Fees of the Debtor Professionals and (2) \$650,000; and (iv) Allowed Professional Fees of Professional Persons in an aggregate amount not to exceed \$500,000 incurred after the first business day following delivery by the DIP Lender of the Enforcement Notice, to the extent allowed at any time, whether by interim order, procedural order, or other order of this Court (the amounts set forth in this clause (iv) being the "Post-Notice Carve-Out Cap").

For the avoidance of doubt, (i) so long as an Enforcement Notice has not been issued, the Debtors shall be permitted to pay Allowed Professional Fees as the same may be due and payable to the extent set forth in the DIP Budget and in accordance with the DIP Credit Agreement and

this Final Order; and (ii) nothing herein shall be construed to impair the ability of any party to object to the fees, expenses, reimbursement or compensation described herein.

- (b) <u>Funded Reserve Account</u>. The Debtors shall establish and fund a segregated account (the "<u>Funded Reserve Account</u>") for purpose of funding the Carve-Out and for holding funds for the Professional Persons for payment of such Professional Persons' Allowed Professional Fees. The Funded Reserve Account will remain DIP Collateral and will be funded first, from identifiable proceeds of any DIP Collateral, and, solely to the extent that the amount of the identifiable proceeds of any DIP Collateral is insufficient to fund the Funded Reserve Account, the proceeds of the DIP Loans.
- (i) Commencing on the later of May 25, 2023 or such later day as the Debtors have established the Funded Reserve Account (in either case, or the first business day thereafter), the Debtors shall deposit into the Funded Reserve Account an amount equal to 115% of the aggregate amount of budgeted professional fees of the Professional Persons as provided on the DIP Budget for the period from the Petition Date through May 31, 2023.
- Debtors have established the Funded Reserve Account (in either case, or the first business day thereafter), and continuing on the first business day of each month thereafter, the Debtors shall deposit in the Funded Reserve Account an amount equal to 115% of the aggregate amount of budgeted professional fees and expenses as provided on the DIP Budget for the prior month (the "Monthly Funded Reserve Amount"). By no later than the 15th of each month, starting in May 2023 (or such other date that may be established in any order authorizing professional compensation procedures), each Professional Person shall deliver to the Debtors a notice of the cumulative total amount of actual fees and expenses incurred in the preceding month ("Actual").

Accrued Prior Month's Fees and Expenses") (each such statement, a "Fee Statement") for which such Professional Person has or will seek payment as Allowed Professional Fees. For any Professional Person, to the extent budgeted fees and expenses, are greater than Allowed Professional Fees, or the Monthly Funded Reserve Amount exceeds the Actual Accrued Prior Month's Fees and Expenses, such Professional Person's Monthly Funded Reserve Amount for the following month shall be reduced accordingly.

- (iii) The Funded Reserve Account shall be maintained, and the funds therein (the "Funded Reserve Amount") shall constitute Carve-Out funds and be held solely to pay Professional Persons until all Allowed Professional Fees are paid in full. Until all Allowed Professional Fees are paid in full, any and all amounts in the Funded Reserve Account shall not be subject to any cash sweep and/or foreclosure provisions in the documents governing the Prepetition Credit Facilities or the DIP Loan Documents, and the DIP Lender shall not be entitled to sweep or foreclose on such amounts notwithstanding any provision to the contrary in the documents governing the Prepetition Credit Facilities or the DIP Loan Documents. Notwithstanding the foregoing, any and all payments to Professional Persons shall be paid first from the Funded Reserve Account. For the avoidance of doubt, all Professionals seeking payment of Allowed Professional Fees from the Funded Reserve Account must first submit and obtain approval from the Court of a fee application for such Allowed Professional Fees.
- (c) <u>Carve-Out Reserves</u>. On the day on which an Enforcement Notice is given by the DIP Lender to the Debtors with a copy to counsel to the Committee (the "<u>Termination Declaration Date</u>"), the Enforcement Notice shall constitute a demand to the Debtors to utilize all cash on hand as of such date, including cash in the Funded Reserve Account, and any available cash thereafter held by any Debtor to fund a reserve in an amount equal to the then-

unpaid amounts of the Allowed Professional Fees. The Debtors shall deposit and hold such amounts in a segregated account at the DIP Lender in trust to pay such then unpaid Allowed Professional Fees (the "Pre-Notice Carve-Out Reserve") prior to any and all other claims. On the Termination Declaration Date, the Enforcement Notice shall also constitute a demand to the Debtors to utilize all cash on hand as of such date and any available cash thereafter held by any Debtor, after funding the Pre-Notice Carve-Out Reserve, to fund a reserve in an amount equal to the Post-Notice Carve-Out Cap. The Debtors shall deposit and hold such amounts in a segregated account at the DIP Lender in trust to pay such Allowed Professional Fees benefiting from the Post-Notice Carve-Out Cap (the "Post-Notice Carve-Out Reserve" and, together with the Pre-Notice Carve-Out Reserve, the "Carve-Out Reserves") prior to any and all other claims. All funds in the Pre-Notice Carve-Out Reserve shall be used first to pay the obligations set forth in clauses (i) through (iii) of the definition of Carve-Out set forth above (the "Pre-Notice Carve Out Amounts"), but not, for the avoidance of doubt, the Post-Notice Carve-Out Cap, until paid in full, and then, to the extent the Pre-Notice Carve-Out Reserve has not been reduced to zero, to the DIP Lender for application to the Post-Petition Obligations, unless the Post-Petition Obligations have been indefeasibly paid in full, in cash, and all commitments under the DIP Loan Documents have been terminated, in which case any such excess shall be paid to the Pre-Petition Lender in accordance with its rights and priorities hereunder. All funds in the Post-Notice Carve-Out Reserve shall be used first to pay the obligations set forth in clause (iv) of the definition of Carve-Out set forth above (the "Post-Notice Carve-Out Amounts"), and then, to the extent the Post-Notice Carve-Out Reserve has not been reduced to zero, to the DIP Lender for application to the Post-Petition Obligations, unless the Post-Petition Obligations have been indefeasibly paid in full, in cash, and all commitments under the DIP Loan Documents have been terminated, in which case any such excess shall be paid to the Pre-Petition Lender in accordance with its rights and priorities hereunder. Notwithstanding anything to the contrary in the DIP Loan Documents, or this Final Order, if either of the Carve-Out Reserves is not funded in full in the amounts set forth in this paragraph 2.3, then, any excess funds in one of the Carve-Out Reserves following the payment of the Pre-Notice Carve-Out Amounts and Post-Notice Carve-Out Amounts, respectively, shall be used to fund the other Carve-Out Reserve, up to the applicable amount set forth in this paragraph 2.3, prior to making any payments from the Carve-Out Reserve to the DIP Lender or the Pre-Petition Lender, as applicable. Notwithstanding anything to the contrary in the DIP Loan Documents or this Final Order, following delivery of an Enforcement Notice, the DIP Lender and the Pre-Petition Lender shall not sweep or foreclose on cash (including cash received as a result of the sale or other disposition of any assets) of the Debtors until the Carve-Out Reserves have been fully funded, but shall have a security interest in any residual interest in the Carve-Out Reserves, with any excess paid to the DIP Lender for application in accordance with the DIP Loan Documents and the Pre-Petition Lender in accordance with the Pre-Petition Loan Documents pursuant to their relative rights and priorities as set forth herein. Further, notwithstanding anything to the contrary in this Final Order, (i) disbursements by the Debtors from the Carve-Out Reserves shall not constitute Loans (as defined in the DIP Credit Agreement) or increase or reduce the Post-Petition Obligations, (ii) the failure of the Carve-Out Reserves to satisfy in full the Allowed Professional Fees shall not affect the priority of the Carve-Out, (iii) in no way shall the DIP Budget, Carve-Out, Post-Notice Carve-Out Cap, Carve-Out Reserves, or any of the foregoing be construed as a cap or limitation on the amount of the Allowed Professional Fees or Statutory Fees due and payable by the Debtors and (iv) the DIP Lender shall have no obligation to advance any loans or other amounts after the Termination Declaration Date. For the avoidance of doubt and notwithstanding anything to the contrary in this Final Order, the DIP Loan Documents, or in any Prepetition Loan Documents, the Carve-Out shall be senior to all liens and claims securing the Post-Petition Obligations and the Superpriority Claim, and any and all other forms of adequate protection, liens, or claims securing the Post-Petition Obligations or the Pre-Petition Obligations.

- (d) <u>Excluded Professional Fees</u>. Notwithstanding anything to the contrary in this Final Order, neither the Carve-Out nor the proceeds of Collateral or any Loans or any other credit or financial accommodations provided under or in connection with the DIP Loan Documents shall be used to pay any Allowed Professional Fees or any other fees or expenses incurred by any Professional Person in connection with any of the following:
- (i) an assertion or joinder in any claim, counter-claim, action, proceeding, application, motion, objection, defense or other contested matter seeking any order, judgment, determination or similar relief: (A) challenging the legality, validity, priority, perfection, or enforceability of (1) the Pre-Petition Obligations or Pre-Petition Lender's liens on and security interests in the Pre-Petition Collateral or (2) the Post-Petition Obligations or the DIP Lender's liens on and security interests in the Collateral; (B) invalidating, setting aside, avoiding or subordinating, in whole or in part, (1) the Pre-Petition Obligations or Pre-Petition Lender's liens on and security interests in the Pre-Petition Collateral, or (2) the Post-Petition Obligations or the DIP Lender's liens on and security interests in the Collateral; or (C) preventing, hindering or delaying the DIP Lender's assertion or enforcement of any lien, claim, right or security interest or realization upon any Collateral in accordance with the terms and conditions of the DIP Credit Agreement, the DIP Loan Documents, and this Final Order;
- (ii) a request, without the prior written consent of the DIP Lender, to use Cash Collateral (as such term is defined in section 363 of the Bankruptcy Code) in

any manner that is not in accordance with the terms and conditions of this Final Order and the DIP Loan Documents;

(iii) a request, without the prior written consent of the DIP Lender, for authorization to obtain debtor in possession financing or other financial accommodations pursuant to section 364(c) or (d) of the Bankruptcy Code that does not provide for the Payment in Full of all the Post-Petition Obligations on terms and conditions on the date any such financing or financial accommodation is approved by this Court; ⁶

(iv) the commencement or prosecution of any action or proceeding of any claims, causes of action or defenses against the DIP Lender or any of its respective officers, directors, employees, agents, attorneys, affiliates, predecessors, successors or

All references to "payment in full" or "Payment in Full" or "paid in full" or "Paid in Full" with respect to the Pre-Petition Obligations or Post-Petition Obligations, as the case may be, means the indefeasible repayment in full in cash of all such obligations (including principal, interest, fees, expenses, and indemnities), except that for contingent obligations it shall include arrangements and amounts reasonably satisfactory to the applicable lender, whether in the form of cash collateral or a backstop letter of credit, and including in any event in addition to such arrangements as to letters of credit and similar instruments, treasury and cash management obligations, hedging obligations and other bank product obligations, the pledge of cash collateral in such amounts as the applicable lender or issuing bank may reasonably require to secure any reimbursement, indemnification or similar continuing obligations of the Debtors in favor of the applicable lender, in each case, in accordance with the terms of the applicable Pre-Petition Loan Document or DIP Loan Document (it being understood that the reference to cash collateral herein under the Pre-Petition Credit Agreement or the DIP Credit Agreement means the pledge and deposit with and delivery to the lender, as the case may be, as collateral for the obligations specified above in such amounts as Pre-Petition Lender or the DIP Lender, as the case may be, determines pursuant to documentation in form and substance reasonably satisfactory to such lender and that the Pre-Petition Lender, or the DIP Lender, as the case may be, shall receive a release from each Debtor and the Committee of and from all demands, actions, causes of action, suits, covenants, contracts, controversies, agreements, promises, sums of money, accounts, bills, reckonings, damages and any and all other claims, counterclaims, defenses, rights of set-off, demands and liabilities (including, without limitation any obligations or liabilities of any kind related to the Carve-Out upon payment in full of the Pre-Petition Obligations or Post-Petition Obligations) in form and substance acceptable to Pre-Petition Lender or the DIP Lender, as applicable or in the case of the Committee, instead of such release, (A) if the Challenge Period (as defined in this Final Order) has not elapsed, a written notice or other confirmation that no Challenge or any other claims of any kind (including with respect to the Carve-Out upon payment in full of the Pre-Petition Obligations or Post-Petition Obligations) will be asserted or (B) if the Challenge Period has elapsed, then no Challenge or any other claim has been asserted or any Challenge or other claim asserted has been dismissed pursuant to a final, non-appealable order of a court of competent jurisdiction and a written notice or other confirmation that no claim of any kind with respect to the Carve-Out will be asserted against the Pre-Petition Lender or the DIP Lender.

assigns, including, without limitation, any attempt to recover or avoid any claim or interest from the DIP Lender under chapter 5 of the Bankruptcy Code;

- (v) the cost of investigation into any claims against Pre-Petition

 Lender arising under or in connection with the Pre-Petition Loan Documents in an aggregate

 amount that exceeds \$30,000;
- (vi) seeking relief under the Bankruptcy Code, including, without limitation, in each case under section 105 of the Bankruptcy Code, to the extent such relief would restrict or impair the rights and remedies of the DIP Lender as set forth herein, the other DIP Loan Documents or any order of the Bankruptcy Court; or
- (vii) any act which has or could directly, materially and adversely modify or compromise the rights and remedies of the DIP Lender under this Final Order, or which directly results in the occurrence of an Event of Default under any DIP Loan Documents or this Final Order.
- <u>Date</u>. Any payment or reimbursement made prior to the occurrence of the Termination Declaration Date in respect of any Statutory Fees or Allowed Professional Fees of the Debtor Professionals shall not reduce the Carve Out. Any payment or reimbursement made prior to the occurrence of the Termination Declaration Date in respect of any Statutory Fees or Allowed Professional Fees shall not reduce the Post-Notice Carve-Out Cap.

2.5 Payment of Carve-Out On or After the Termination Declaration Date.

(a) Any payment or reimbursement made on or after the occurrence of the Termination Declaration Date in respect of any Allowed Professional Fees shall permanently reduce the Carve-Out on a dollar-for-dollar basis. Any funding of the Carve-Out, if any, shall be added to, and made a part of, the Post-Petition Obligations secured by the DIP Collateral and shall be otherwise entitled to the protections granted under this Final Order, the DIP Loan Documents, the Bankruptcy Code, and applicable law.

Lender or Pre-Petition Lender shall not and shall not be deemed to reduce the Pre-Petition Obligations or Post-Petition Obligations, and shall not and shall not be deemed to subordinate any liens and security interests of the DIP Lender or the Pre-Petition Lender in the Pre-Petition Collateral or Collateral, the Superpriority Claim (as defined herein), or the DIP Loan Superpriority Claim (as defined herein) to any junior pre- or post-petition lien, interest or claim in favor of any other party. The DIP Lender and the Pre-Petition Lender shall not, under any circumstance, be responsible for the direct payment or reimbursement of any fees or disbursements of any Professionals incurred in connection with the Chapter 11 Cases under any chapter of the Bankruptcy Code, and nothing in this Final Order or otherwise shall be construed to obligate the DIP Lenders or the Pre-Petition Lender in any way, to directly pay compensation to or to reimburse expenses of any Professional, or to ensure that the Debtors have sufficient funds to pay such compensation or reimbursement.

2.6 <u>Limited Use of Cash Collateral; Adequate Protection.</u>

(a) <u>Authorization to Use Cash Collateral</u>. Subject to the terms and conditions of this Final Order, the DIP Credit Agreement, the DIP Loan Documents, and in accordance with the DIP Budget (subject to the Permitted Variances), Borrowers shall be and are hereby authorized to use the Cash Collateral (as defined in section 363 of the Bankruptcy Code) for the period commencing on the date of this Final Order and terminating upon the earlier of (i) the date that is the final day of the Financing Period; and (ii) the date on which the DIP Lender

delivers an Enforcement Notice (as defined herein) to counsel for the Loan Parties, counsel for the Committee, and the U.S. Trustee, subject to the liens and security interests granted to the DIP Lender. Nothing in this Final Order shall authorize the disposition of any assets of the Debtors or their Estates outside the ordinary course of business, or any Borrower's use of Cash Collateral or other proceeds resulting therefrom, except as expressly permitted in this Final Order, the DIP Loan Documents and in accordance with the DIP Budget.

(b) Replacement Liens.

Replacement Lien. As adequate protection for any (i) diminution in value of their interests in the Pre-Petition Collateral (including Cash Collateral) on account of the Debtors' use of such Collateral (including Cash Collateral), the imposition of the automatic stay and the subordination to the DIP Liens and the Carve-Out on a dollar-for-dollar basis (collectively, the "Diminution in Value"), the Pre-Petition Lender is hereby granted pursuant to sections 361 and 363 of the Bankruptcy Code, and solely to the extent of the Diminution in Value, valid, binding, enforceable and perfected replacement liens upon and security interests in all Collateral (the "Replacement Lien"), which shall be subject to the Challenge Period (as defined below). The Replacement Lien shall be junior and subordinate only to (A) the Carve-Out (B) the Permitted Liens, and (C) DIP Liens, and shall otherwise be senior to all other security interests in, liens on, or claims against any of the Collateral; provided, however, that the Pre-Petition Lender will be paid from all Collateral other than the Avoidance Proceeds prior to payment from the Avoidance Proceeds, and will only seek and receive payment from the Avoidance Proceeds to the extent that payment from all other Collateral does not result in the Payment in Full of all Pre-Petition Obligations and Adequate Protection Obligations.

(c) Additional Adequate Protection.

- (i) <u>Superpriority Claim.</u> As adequate protection for the Diminution in Value, the Pre-Petition Lender is hereby granted as and to the extent provided by section 507(b) of the Bankruptcy Code, and to the extent of the Diminution in Value, an allowed superpriority administrative expense claim in each of the Chapter 11 Cases and any successor bankruptcy cases (the "<u>Superpriority Claim</u>"), which shall be subject to the Challenge Period (as defined below). The Superpriority Claim shall be junior only to (A) the Carve-Out and (B) the DIP Loan Superpriority Claim, and shall otherwise have priority over all administrative expense claims, including administrative expenses of the kinds specified in or ordered pursuant to sections 503(b) and 507(b) of the Bankruptcy Code, and unsecured claims against Debtors and their Estates now existing or hereafter arising, of any kind or nature whatsoever.
- (ii) <u>Payment of Professional Fees</u>. As further adequate protection, the Debtors are authorized to pay, in accordance with the terms of Paragraph 5.12 of this Final Order, all reasonable and documented fees and expenses of the Pre-Petition Lender, whether incurred pre-petition or post-petition (the "<u>Pre-Petition Lender Professional Fees</u>"), which shall be subject to the Challenge Period (as defined below).
- will pay the Pre-Petition Lender cash interest at the non-default rate on the Pre-Petition Obligations as and when due under the Pre-Petition Credit Agreement (the "Adequate Protection Interest Payments," together with the Replacement Lien, Superpriority Claim and Pre-Petition Lender Professional Fees, the "Adequate Protection Obligations"), which shall be subject to the Challenge Period (as defined below). For the avoidance of doubt, (A) in the event it is subsequently determined that the Pre-Petition Lender is undersecured and the Pre-Petition Lender is not entitled to the accrual of any post-petition interest, any post-petition interest paid to the Pre-Petition Lender

shall be re-allocated to payment of principal amount of the Pre-Petition Obligations and (B) interest on the Pre-Petition Obligations shall accrue at the default rate at all times following the Petition Date (less any amounts actually paid hereunder as Adequate Protection Interest Payments).

Section 3. <u>Default; Rights and Remedies; Relief from Stay.</u>

- 3.1 <u>Events of Default</u>. The occurrence of any of the following events shall constitute an "<u>Event of Default</u>" under this Final Order: (a) any Debtor's failure to perform, in any respect, any of its obligations under this Final Order; or (b) an "Event of Default" under the DIP Credit Agreement or any of the other DIP Loan Documents, including but not limited to any failure to comply with the Milestones.
- Alghts and Remedies upon Event of Default. Upon the occurrence of and during the continuance of an Event of Default, (a) the Debtors shall be bound by all restrictions, prohibitions and other terms as provided in this Final Order, the DIP Credit Agreement and the other DIP Loan Documents, and (b) the DIP Lender shall be entitled to take any act or exercise any right or remedy (subject to Paragraph 3.4 below) as provided in this Final Order or any DIP Loan Document, as applicable, including, without limitation, declaring all Post-Petition Obligations immediately due and payable, accelerating the Post-Petition Obligations, ceasing to extend Loans, setting off any Post-Petition Obligations with Collateral or proceeds in the DIP Lender's possession, and enforcing any and all rights with respect to the Collateral; provided that none of the foregoing shall diminish the obligations to fund the Carve-Out as set forth in this Final Order. Except as expressly set forth in Paragraph 2.4, the DIP Lender shall have no obligation to lend or advance any additional funds to or on behalf of Debtors, or provide any other financial accommodations to Debtors, immediately upon or after the occurrence of an Event of Default or

upon the occurrence of any act, event, or condition that, with the giving of notice or the passage of time, or both, would constitute an Event of Default.

- Borrowers' authority to borrow and obtain other credit accommodations from the DIP Lender pursuant to the terms of this Final Order and the DIP Loan Documents (except if such authority shall be extended with the prior written consent of the DIP Lender, which consent shall not be implied or construed from any action, inaction or acquiescence by the DIP Lender), and (b) the delivery by the DIP Lender of written notice pursuant to Section 13.4 of the DIP Credit Agreement upon the occurrence of an Event of Default, above, all of the Post-Petition Obligations shall immediately become due and payable and the DIP Lender shall have no obligation whatsoever to make or extend any loans, advances, provide any financial or credit accommodations to Debtors or permit the use of Cash Collateral, except, with respect to use of Cash Collateral, to the extent necessary to fund the Carve-Out.
- 3.4 <u>Modification of Automatic Stay.</u> The automatic stay provisions of section 362 of the Bankruptcy Code and any other restriction imposed by an order of the Court or applicable law are hereby modified without further notice, application or order of the Court to the extent necessary to permit the DIP Lender and the Borrowers to perform, immediately upon entry of this Final Order and at any time thereafter, any act authorized or permitted under or by virtue of this Final Order or the DIP Loan Documents, as applicable, including, without limitation, (a) to implement the post-petition financing arrangements authorized by this Final Order and pursuant to the terms of the DIP Loan Documents, (b) to take any act to create, validate, evidence, attach or perfect any lien, security interest, right or claim in the Collateral, and (c) to assess, charge, collect, advance, deduct and receive payments with respect to the Pre-Petition Obligations or the Post-

Petition Obligations, as applicable, including, without limitation, all interests, fees, costs and expenses permitted under the DIP Loan Documents (subject to Paragraph 5.12 of this Final Order) and apply such payments to the Pre-Petition Obligations or Post-Petition Obligations pursuant to the DIP Loan Documents and this Final Order, as applicable. Without limiting the foregoing, upon the occurrence of an Event of Default and the DIP Lender providing four (4) business days (the "Default Notice Period") prior written notice, which period may be extended by written agreement (including e-mail) of the Borrowers and the DIP Lender, (the "Enforcement Notice") to (i) counsel for the Loan Parties, (ii) counsel for the Committee, and (iii) the U.S. Trustee, the DIP Lender shall be entitled without further notice, application or order of the Court to take any action and exercise all other rights and remedies provided to it by this Final Order, the DIP Loan Documents or applicable law that the DIP Lender may deem appropriate in its sole discretion to proceed against and realize upon the Collateral or any other assets or properties of Debtors' Estates upon which the DIP Lender has been or may hereafter be granted liens or security interests to obtain the indefeasible repayment in full of all Post-Petition Obligations. Notwithstanding anything to the contrary, any action that the DIP Lender is otherwise permitted to take pursuant to this Final Order to (i) terminate the commitments under the DIP Loan Documents, (ii) accelerate the Loans, (iii) send blocking notices or activation notices pursuant to the terms of any Blocked Account Agreement, (iv) repay any amounts owing in respect of the Post-Petition Obligations (including, without limitation, fees, indemnities and expense reimbursements) and (v) cash collateralize letters of credit and bank products issued pursuant to the DIP Loan Documents, in each case, shall not require any advance notice to the Debtors. During the Default Notice Period, the Debtors and the Committee shall be entitled to seek an emergency hearing, and the DIP Lender shall consent to such emergency hearing within the Default Notice Period.

Section 4. Representations; Covenants; and Waivers.

- 4.1 Effect of Stipulations on Third Parties.
- (a) Subject to Paragraph 4.1(b) hereof, each stipulation, admission, and agreement contained in this Final Order including, without limitation, the Stipulations, shall be binding upon the Debtors, their Estates and any successor thereto (including, without limitation, any chapter 7 or chapter 11 trustee appointed or elected for any of the Debtors) under all circumstances and for all purposes, and the Debtors are deemed to have irrevocably waived and relinquished all Challenges (as defined herein) as of the date of the Petition Date.
- Nothing in this Final Order shall prejudice the rights of the Committee or any other party in interest, if granted standing by the Court, to seek, solely in accordance with the provisions of this Paragraph 4.1, to assert claims against the Pre-Petition Lender on behalf of the Debtors or the Debtors' creditors or to otherwise challenge the Stipulations, findings, waivers, releases, affirmations, and other stipulations, including, but not limited to those in relation to (i) the validity, extent, priority, or perfection of the security interests, and liens of the Pre-Petition Lender, (ii) the validity, allowance, priority, or amount of the Pre-Petition Obligations, or (iii) any liability of the Pre-Petition Lender with respect to anything arising from the Pre-Petition Loan Documents. Notwithstanding the immediately preceding sentence, the Committee or any other party in interest must, after obtaining standing approved by the Court, commence a contested matter or adversary proceeding raising such claim, objection, or challenge, including, without limitation, any claim or cause of action against the Pre-Petition Lender, (each, a "Challenge") no later than (i) with respect to the Committee, the date that is sixty (60) days after the Committee's formation, or (ii) with respect to any chapter 11 trustee appointed in the Chapter 11 Cases, or any chapter 7 trustee appointed in case of conversion of the Chapter 11 Cases (a "Successor Case"), prior to the expiration of the periods set forth in subsections (i) and (ii) above, no later than the

date that is the later of (A) fourteen (14) days after the appointment of such trustee or (B) the expiration of the time periods set forth in the foregoing subsections (i) and (ii) above (collectively, the "Challenge Period"); provided, that if a motion for standing attaching a draft complaint is timely filed by the Committee, the Challenge Period shall be extended solely for the Committee and solely as to the claims asserted in such complaint until three (3) business days after the Court disposes of such motion. The Challenge Period may only be extended with the written consent of the Pre-Petition Lender prior to the expiration of the Challenge Period, and for the avoidance of doubt, any such extension shall only apply to the Pre-Petition Lender providing such written consent, as applicable. Unless otherwise ordered by the Court, only those parties in interest who commence a Challenge within the Challenge Period may prosecute such Challenge. As to (x) any parties in interest, including the Committee, who fail to file a Challenge within the Challenge Period, or if any such Challenge is filed and overruled or otherwise finally resolved or adjudicated in favor of the Pre-Petition Lender or (y) any and all matters that are not expressly the subject of a timely Challenge: (1) any and all such Challenges by any such party (including, without limitation, the Committee, any chapter 11 trustee, any examiner or any other estate representative appointed in the Debtors' Chapter 11 Cases, or any chapter 7 trustee, appointed in any Successor Case), shall be deemed to be forever waived and barred, (2) all of the findings, Stipulations, waivers, releases, affirmations, and other stipulations shall be of full force and effect and forever binding upon the Debtors' bankruptcy estates and all such creditors, interest holders, and other parties in interest in the Chapter 11 Cases and any Successor Cases, and (3) any and all claims or causes of action against the Pre-Petition Lender related in any way to the Debtors, these Chapter 11 Cases, the Pre-Petition Obligations or the Pre-Petition Loan Documents shall be released by the Debtors' estates, all creditors, interest holders, and other parties in interest in the Chapter 11 Cases and any Successor Cases.

- (c) Nothing in this Final Order vests or confers on any person (as defined in the Bankruptcy Code), including the Committee, standing or authority to pursue any cause of action belonging to the Debtors or their Estates, including, without limitation, any Challenge with respect to the Post-Petition Obligations or the Pre-Petition Obligations.
- (d) Notwithstanding anything to the contrary set forth herein, in any pleading, or filing, any DIP Loan Document or otherwise, prior to the expiration of the Challenge Period no act or omission by or on behalf of Pre-Petition Lender or the DIP Lender shall be construed as a waiver of any rights or remedies with respect any Loan Party or any of their respective, successors and assigns, present and former shareholders, affiliates, subsidiaries, divisions, predecessors, directors, officers, attorneys, employees and other representatives.
- d.2 <u>Debtors' Waivers.</u> At all times during the Chapter 11 Cases, and whether or not an Event of Default has occurred, the Debtors irrevocably waive any right that they may have to seek further authority (a) to use Cash Collateral of the DIP Lender or the Pre-Petition Lender under section 363 of the Bankruptcy Code, other than as provided in this Final Order or as may be otherwise expressly permitted pursuant to the DIP Loan Documents or other agreement with or consent of the DIP Lender or the Pre-Petition Lender, as applicable, (b) to obtain post-petition loans or other financial accommodations pursuant to section 364(c) or 364(d) of the Bankruptcy Code that does not provide for the indefeasible repayment in full of all Pre-Petition Obligations and Post-Petition Obligations in cash at the time any such post-petition loans or financial accommodations are provided, extended or otherwise made available to Debtors, (c) to challenge, contest or otherwise seek to impair or object to the validity, extent, enforceability or

priority of the DIP Lender's post-petition liens and claims, (d) to challenge the application of any payments authorized by this Final Order as pursuant to section 506(b) of the Bankruptcy Code, or to assert that the value of the Pre-Petition Collateral is less than the Pre-Petition Obligations, (e) to propose, support or have a plan of reorganization or liquidation that does not provide for the indefeasible payment in cash in full and satisfaction of all Post-Petition Obligations (other than unmatured indemnity obligations for which claims (i) have not been asserted and (ii) are not reasonably expected to be asserted at any time in the future) on the effective date of such plan in accordance with the terms and conditions set forth in the DIP Credit Agreement, (f) to surcharge the Collateral or the Pre-Petition Collateral pursuant to 506(c) of the Bankruptcy Code or (g) to seek relief under the Bankruptcy Code, including without limitation, under section 105 of the Bankruptcy Code, to the extent any such relief would in any way restrict or impair the rights and remedies of the DIP Lender as provided in this Final Order and the DIP Loan Documents or the DIP Lender's exercise of such rights or remedies (other than to object to the exercise of the rights and remedies within the Default Notice Period on the grounds set forth in Section 3.4 of this Final Order); provided, however, that the DIP Lender may otherwise consent in writing, but no such consent shall be implied from any other action, inaction, or acquiescence by the DIP Lender; provided further, however, that nothing herein shall prohibit the Debtors from obtaining postpetition loans or other financial accommodations pursuant to section 364(c) or 364(d) of the Bankruptcy Code if the proceeds of such financing are used to indefeasibly pay (i) all Pre-Petition Obligations and the Pre-Petition Obligations are satisfied in full in accordance with the terms of the Pre-Petition Credit Agreement and the other Pre-Petition Loan Documents, and (ii) all Post-Petition Obligations are satisfied in full in accordance with the terms of the DIP Credit Agreement and the other DIP Loan Documents.

- 4.3 <u>Section 506(c) Claims</u>. No costs or expenses of administration which have or may be incurred in the Chapter 11 Cases shall be charged against the DIP Lender, the Pre-Petition Lender, or their respective claims or the Collateral or the Pre-Petition Collateral pursuant to section 506(c) of the Bankruptcy Code without the prior written consent of such agent or lender, and no such consent shall be implied from any other action, inaction or acquiescence by the DIP Lender, or the Pre-Petition Lender.
- 4.4 <u>Collateral Rights</u>. Until all Post-Petition Obligations shall have been indefeasibly paid and satisfied in full in accordance with the terms of the DIP Credit Agreement and the other DIP Loan Documents:
- (a) no party other than the DIP Lender shall foreclose or otherwise seek to enforce any junior lien or claim in Collateral; and
- (b) upon and after the declaration of the occurrence of an Event of Default, and subject to the expiration of the Default Notice Period, in connection with a liquidation of any of the Collateral or the DIP Lender's exercise of remedies in respect of the Collateral, the DIP Lender (or any of its employees, agents, consultants, contractors or other professionals) shall have the right, at the sole cost and expense of the Loan Parties, to: (i) enter upon, occupy and use any real or personal property, fixtures, equipment, leasehold interests or warehouse arrangements owned or leased by Debtors and (ii) use any and all trademarks, trade names, copyrights, licenses, patents or any other similar assets of Debtors, which are owned by or subject to a lien of any third party and which are used by Debtors in their businesses. The DIP Lender will be responsible for the payment of any applicable fees, rentals, royalties or other amounts due such lessor, licensor or owner of such property for the period of time that the DIP Lender actually uses the equipment or the intellectual property (but in no event for any accrued and unpaid fees, rentals or other amounts

due for any period prior to the date that the DIP Lender actually occupies or uses such assets or properties or for any fees, rentals or other amounts that may become due following the end of the DIP Lender's occupation or use).

Releases. Subject to Section 4.1 above, each Debtor, on behalf of itself and 4.5 its successors and assigns (collectively, the "Releasors"), hereby forever releases, discharges and acquits the Pre-Petition Lender, and its successors and assigns, present and former shareholders, affiliates, subsidiaries, divisions, predecessors, directors, officers, attorneys, employees and other representatives in their respective capacities as such (collectively, the "Pre-Petition Releasees") of and from any and all claims, demands, liabilities, responsibilities, disputes, remedies, causes of action, indebtedness and obligations, of every kind, nature and description, including, without limitation, any so-called "lender liability" claims or defenses, that Releasors had, have or hereafter can or may have against Pre-Petition Releasees, or any of them, as of the date hereof, in respect of events that occurred on or prior to the date hereof with respect to the Debtors, the Pre-Petition Obligations, the Pre-Petition Loan Documents and any Loans, Credit Facilities, the L/C Facility, the Mastercard Facility and other Obligations (as such terms are defined in the Pre-Petition Credit Agreement) or other financial accommodations made by any Pre-Petition Releasee to any Debtor. In addition, notwithstanding anything to the contrary set forth herein, upon the repayment of all Post-Petition Obligations owed to the DIP Lender by Debtors and termination of the rights and obligations arising under the DIP Loan Documents (which payment and termination shall be on terms and conditions acceptable to the DIP Lender), the DIP Lender shall be released from any and all obligations, liabilities, actions, duties, responsibilities and causes of action arising or occurring, on or prior to the date of such repayment and termination, in connection with or related to the Debtors, DIP Loan Documents, the Interim Order or this Final Order (including without limitation any obligation or responsibility whether direct or indirect, absolute or contingent, due or not due, primary or secondary, liquidated or unliquidated to pay or otherwise fund the Carve-Out and/or the Professional Fee Carve-Out).

Section 5. Other Rights and Post-Petition Obligations.

- 5.1 No Modification or Stay of This Final Order. Notwithstanding (a) any stay, modification, amendment, supplement, vacating, revocation or reversal of this Final Order, the DIP Loan Documents or any term hereunder or thereunder, or (b) the dismissal or conversion of one or more of the Chapter 11 Cases (each, a "Subject Event"), (x) the acts taken by the DIP Lender in accordance with this Final Order, and (y) the Post-Petition Obligations incurred or arising prior to the DIP Lender's actual receipt of written notice from Debtors expressly describing the occurrence of such Subject Event shall, in each instance, be governed in all respects by the original provisions of this Final Order, and the acts taken by the DIP Lender in accordance with this Final Order, and the liens granted to the DIP Lender in the Collateral, and all other rights, remedies, privileges, and benefits in favor of the DIP Lender pursuant to this Final Order and the DIP Loan Documents shall remain valid and in full force and effect to the extent provided in to section 364(e) of the Bankruptcy Code. For purposes of this Final Order, the term "appeal", as used in section 364(e) of the Bankruptcy Code, shall be construed to mean any proceeding for reconsideration, amending, rehearing, or re-evaluating this Final Order by this Court or any other tribunal.
- 5.2 <u>Power to Waive Rights; Duties to Third Parties</u>. The DIP Lender shall have the right to waive any of the terms, rights and remedies provided or acknowledged in this Final Order that are in favor of the DIP Lender (the "<u>DIP Lender Rights</u>") and shall have no obligation or duty to any other party with respect to the exercise or enforcement, or failure to exercise or enforce, any DIP Lender Right(s). Any waiver by the DIP Lender of any DIP Lender Rights shall

not be or constitute a continuing waiver unless expressly specified in writing by the DIP Lender. Any delay in or failure to exercise or enforce any DIP Lender Right shall neither constitute a waiver of such DIP Lender Right, subject the DIP Lender to any liability to any other party, nor cause or enable any party other than the Debtors to rely upon or in any way seek to assert as a defense to any obligation owed by the Debtors to the DIP Lender.

- 5.3 <u>Disposition of Collateral</u>. The Debtors shall not sell, transfer, lease, encumber or otherwise dispose of any portion of the Collateral, other than pursuant to the terms of the DIP Loan Documents, without the prior written consent of the DIP Lender (and no such consent shall be implied, from any other action, inaction or acquiescence by the DIP Lender).
- 5.4 <u>Inventory</u>. The Debtors shall not, without the consent of the DIP Lender, (a) enter into any agreement to return any inventory to any of their creditors for application against any pre-petition indebtedness under any applicable provision of section 546 of the Bankruptcy Code, or (b) consent to any creditor taking any setoff against any of its pre-petition indebtedness based upon any such return pursuant to section 553(b)(1) of the Bankruptcy Code or otherwise.
- Order are in addition to and without prejudice to the rights of the DIP Lender to pursue any and all rights and remedies under the Bankruptcy Code, the DIP Loan Documents or any other applicable agreement or law, including, without limitation, rights to seek adequate protection and/or additional or different adequate protection, to seek relief from the automatic stay, to seek an injunction, to oppose any request for use of cash collateral or granting of any interest in the Collateral, as applicable, or priority in favor of any other party, to object to any sale of assets, and to object to applications for allowance and/or payment of compensation of Professional Persons or other parties seeking compensation or reimbursement from the Estates.

5.6 Binding Effect.

- (a) The provisions of this Final Order and the DIP Loan Documents, the Post-Petition Obligations, the DIP Loan Superpriority Claim, the Superpriority Claim and any and all rights, remedies, privileges and benefits in favor of the DIP Lender provided or acknowledged in this Final Order, and any actions taken pursuant thereto, shall be effective immediately upon entry of this Final Order notwithstanding Bankruptcy Rules 4001(a)(3), 6004(h) and 7062, shall continue in full force and effect, and shall survive entry of any such other order, including without limitation any order which may be entered confirming any plan of reorganization, converting one or more of the Chapter 11 Cases to any other chapter under the Bankruptcy Code, or dismissing one or more of the Chapter 11 Cases.
- (b) Any order dismissing one or more of the Chapter 11 Cases under section 1112 or otherwise shall be deemed to provide (in accordance with sections 105 and 349 of the Bankruptcy Code) that (a) the DIP Loan Superpriority Claim and the DIP Lender's liens on and security interests in the Collateral and all other claims, liens, adequate protections and other rights granted pursuant to the terms of this Final Order shall continue in full force and effect notwithstanding such dismissal until the Post-Petition Obligations are indefeasibly paid and satisfied in full, and (b) this Court shall retain jurisdiction, notwithstanding such dismissal, for the purposes of enforcing all such claims, liens, protections and rights.
- (c) To the extent this Court has, by entry of this Final Order, modified any of the provisions of the Interim Order or the DIP Loan Documents, such modifications shall not affect the rights or priorities of the DIP Lender or its interests in the Collateral authorized by or otherwise set forth in the Interim Order with respect to the Collateral or any portion of the Post-

Petition Obligations arising or incurred by the Loan Parties prior to such modification, and the Interim Order shall otherwise remain in full force and effect to such extent.

(d) This Final Order shall be binding upon the Debtors, all parties in interest in the Chapter 11 Cases and their respective successors and assigns, including any trustee or other fiduciary appointed in the Chapter 11 Cases or any subsequently converted bankruptcy case(s) of any Debtor. This Final Order shall also inure to the benefit of Debtors, the DIP Lender, and each of their respective successors and assigns.

5.7 Restrictions on Cash Collateral Use, Additional Financing, Plan Treatment.

under the DIP Credit Agreement and the other DIP Loan Documents are made in reliance on this Final Order and there shall not at any time be entered in the Chapter 11 Cases, or in any subsequently converted case under chapter 7 of the Bankruptcy Code, any order which (i) authorizes the use of cash collateral of Debtors in which the DIP Lender has an interest, or the sale, lease, or other disposition of property of any Debtor's Estate in which the DIP Lender has a lien or security interest, except as expressly permitted hereunder or in the DIP Loan Documents, or (ii) authorizes under section 364 of the Bankruptcy Code the obtaining of credit or the incurring of indebtedness secured by a lien or security interest which is equal or senior to a lien or security interest in property in which the DIP Lender holds a lien or security interest, or which is entitled to priority administrative claim status which is equal or superior to that granted to the DIP Lender herein; unless, in each instance (x) the DIP Lender shall have given its express prior written consent with respect thereto, no such consent being implied from any other action, inaction or acquiescence by the DIP Lender, or (y) such other order requires that all Post-Petition Obligations shall first be indefeasibly paid and satisfied in full in accordance with the terms of the DIP Credit

Agreement and the other DIP Loan Documents (other than unmatured indemnity obligations for which claims have not been asserted by the DIP Lender), including, without limitation, all debts and obligations of Debtors to the DIP Lender which arise or result from the obligations, loans, security interests and liens authorized herein, on terms and conditions acceptable to the DIP Lender. The security interests and liens granted to or for the benefit of the DIP Lender hereunder and the rights of the DIP Lender pursuant to this Final Order and the DIP Loan Documents with respect to the Post-Petition Obligations, Collateral, and claims against the Loan Parties are cumulative and shall not be altered, modified, extended, impaired, or affected by any plan of reorganization or liquidation of Debtors and shall continue after confirmation and consummation of any such plan unless (i) all Post-Petition Obligations indefeasibly paid in full prior to such time, or (ii) the DIP Lender expressly consents in writing authenticated by the DIP Lender to different treatment, and *provided* that the Roll-Up shall be subject to the Challenge Period.

- 5.8 No Owner/Operator Liability. The DIP Lender shall not, solely by reason of having made the Loans pursuant to the terms of the DIP Loan Documents and this Final Order, be deemed to be in control of the operations of the Debtors or to be acting as a "responsible person" or "owner or operator" with respect to the operation or management of the Debtors (as such terms, or any similar terms, are used in the United States Comprehensive Environmental Response, Compensation and Liability Act, 29 U.S.C. §§ 9601 et seq., as amended, or any similar federal or state statute).
- 5.9 <u>Marshalling</u>. In no event shall the DIP Lender or the Pre-Petition Lender be subject to the equitable doctrine of "marshalling" or any similar doctrine with respect to the Collateral or the Pre-Petition Collateral. The DIP Lender shall each be entitled to all of the rights and benefits of section 552(b) of the Bankruptcy Code, and the "equities of the case" exception

under section 552(b) of the Bankruptcy Code shall not apply to the Pre-Petition Lender with respect to proceeds, products, offspring or profits of any of the Collateral, as applicable.

Lender as of the Petition Date, including, without limitation, all funds deposited in, or credited to, an account of any Debtor with Pre-Petition Lender immediately prior to the filing of the Chapter 11 Cases (regardless of whether, as of the Petition Date, such funds had been collected or made available for withdrawal by any such Debtor), such funds (the "Deposited Funds") are subject to rights of setoff, which setoff rights shall be subject to the Challenge Period. By virtue of such setoff rights, the Deposited Funds are subject to a lien in favor of Pre-Petition Lender pursuant to sections 506(a) and 553 of the Bankruptcy Code.

5.11 Right to Credit Bid.

- (a) Subject to section 363(k) of the Bankruptcy Code, the DIP Lender shall have the right to "credit bid" the amount of its claims that are Post-Petition Obligations arising under the terms of the DIP Loan Documents, during any sale of all or substantially all of the Debtors' assets, including without limitation, sales occurring pursuant to section 363 of the Bankruptcy Code or included as part of any restructuring plan subject to confirmation under section 1129(b)(2)(A)(ii)-(iii) of the Bankruptcy Code.
- (b) Subject to section 363(k) of the Bankruptcy Code and the Challenge Period, the Pre-Petition Lender shall have the right to "credit bid" the amount of its and their claims that are Pre-Petition Obligations arising under the terms of the Pre-Petition Loan Documents, during any sale of all or substantially all of the Debtors' assets, including without limitation, sales occurring pursuant to section 363 of the Bankruptcy Code or included as part of any restructuring plan subject to confirmation under section 1129(b)(2)(A)(ii)-(iii) of the Bankruptcy Code.

Payment and Review of Lender Fees and Expenses. Each Loan Party shall 5.12 pay all fees, that may be reasonably required or necessary for the Loan Parties' performance of their Post-Petition Obligations under the DIP Credit Agreement, including, without limitation, the non-refundable payment to the (i) Pre-Petition Lender of the reasonable and documented fees and expenses set forth in the Pre-Petition Loan Documents, whether incurred pre-petition or postpetition, including without limitation, all reasonable and documented fees, costs, disbursements and expenses of Chapman and Cutler, LLP, as legal counsel, Womble Bond Dickinson (US) LLP, as legal counsel, Blake, Cassels & Graydon LLP, as Canadian legal counsel and FTI Consulting Canada Limited, as financial advisor, and (ii) the DIP Lender of the fees and expenses set forth in the DIP Loan Documents, whether incurred pre-petition or post-petition including, without limitation the reasonable fees, costs, disbursements and expenses of Chapman and Cutler, LLP, as legal counsel, Womble Bond Dickinson (US) LLP, as legal counsel, Blake, Cassels & Graydon LLP, as Canadian legal counsel and FTI Consulting Canada Limited, as financial advisor, in each case whether incurred before or after the Petition Date; provided, that Debtors shall pay all such reasonable fees and expenses within ten (10) business days of delivery of a statement or invoice for such fees and expenses (it being understood that no counsel or other professional be required to file any interim or final fee applications with the Court or otherwise seek the Court's approval of any such payments) to the Debtors, the U.S. Trustee and the Committee, unless, within such ten (10) business day period, the Debtors, the U.S. Trustee or the Committee serves a written objection upon the requesting party detailing the specific fees or expenses to which such party objects along with an explanation for the basis of such objection, in which case, the Debtors shall immediately pay only such amounts that are not the subject of any objection and only pay the balance of such statements or invoices at such time and in such amount as subsequently agreed to by the requesting party and any objecting party or as otherwise ordered by the Court to be paid.

- 5.13 <u>No Superior Rights of Reclamation</u>. Subject to the Challenge Period, in no event shall any alleged right of reclamation or return (whether asserted under section 546(c) of the Bankruptcy Code or otherwise) be deemed to have priority over the DIP Liens.
- 5.14 <u>Term; Termination</u>. Notwithstanding any provision of this Final Order to the contrary, the term of the financing arrangements among Debtors and the DIP Lender authorized by this Final Order may be terminated pursuant to the terms of the DIP Credit Agreement.
- 5.15 <u>Limited Effect</u>. In the event of a conflict between the terms and provisions of any of the DIP Loan Documents and this Final Order, the terms and provisions of this Final Order shall govern, interpreted as most consistent with the terms and provisions of the DIP Loan Documents.
- 5.16 <u>Retention of Jurisdiction</u>. The Court retains jurisdiction and power with respect to all matters arising from or related to the implementation or interpretation of this Final Order, the DIP Credit Agreement, and the other DIP Loan Documents.

5.17 <u>Aid and Assistance</u>. The Court hereby requests the aid and assistance of the Canadian Court administering the Canadian Proceedings to (i) grant a stay of proceedings in the Canadian Proceedings restricting and prohibiting any further actions or steps being taken in Canada against any of the Collateral of the Debtors which may be located in Canada, and (ii) create a charge in favor of the DIP Lender to secure any Post-Petition Obligations of the Debtors which may be located in Canada in furtherance of the liens and security interests of the DIP Lender granted under the DIP Loan Documents and this Final Order.

Dated: May 19th, 2023 Wilmington, Delaware CRAIG T. GOLDBLATT
UNITED STATES BANKRUPTCY JUDGE

Con Dellar

$\underline{\mathbf{EXHIBIT}\ \mathbf{1}}$

DIP Credit Agreement

Execution Version

SENIOR SECURED SUPERPRIORITY DIP FINANCING CREDIT AGREEMENT

BETWEEN:

STRUCTURLAM MASS TIMBER CORPORATION
- and STRUCTURLAM MASS TIMBER U.S., INC., as Borrowers

-and-

BANK OF MONTREAL, as Lender

TABLE OF CONTENTS

		Page
٨٣٣٢٢	LE 1 INTERPRETATION	1
1.1	Definitions	
1.2	Headings; Articles and Sections	20
1.3	Number; including	
1.4	Application of Accounting Principles; Accounting Changes	
1.5	References to Agreements and Enactments	
1.6	Per Annum Calculations; Currency	
1.7	Schedules	21
1.8	Time	
1.9	Payment for Value	
	LE 2 THE CREDIT FACILITY	
2.1	The Credit Facility	
2.1	Types of Availments	
2.2	Purpose	
2.3	Availability of the Credit Facility	
2.4	Minimum Drawdowns	
2.5	Drawdown Notices; Drawdowns	
2.7	Lender's Obligations with Respect to Cdn. \$ Loans and U.S. \$ Loans	
2.7	Irrevocability	23
2.9	Optional Repayment of the Credit Facility	23
2.10	Mandatory Repayment of Credit Facility	
2.10	Repayment of Borrowings In Excess of the Authorized Principal Amount	24
2.11	Cancellation of Credit Facility and Prepayment	24
2.12	Mandatory Prepayments	25
	CLE 3 CONDITIONS PRECEDENT TO DRAWDOWNS	
3.1	Conditions Precedent	
3.2	Conditions Precedent for all Drawdowns.	26
3.3	Waiver	
	CLE 4 EVIDENCE OF DRAWDOWNS	
	Accounts of Record	
4.1		
	CLE 5 PAYMENTS OF INTEREST AND FEES	
5.1	Interest on Cdn. \$ Loans	
5.2	Interest on U.S. \$ Loans	
5.3	Interest Act (Canada)	28
5.4	Nominal Rates; No Deemed Reinvestment	
5.5	Standby Fees	
5.6	Exit Fee	
5.7	Default Interest	
5.8	Waiver	29
5.9	Maximum Rate Permitted by Law	29
ARTIC	CLE 6 PLACE AND APPLICATION OF PAYMENTS	29
6.1	Place of Payment of Principal, Interest and Fees; Payments to Lender	
6.2	Funds	30
6.3	Application of Payments	
6.4	Payments Clear of Taxes	30
ARTIC	CLE 7 REPRESENTATIONS AND WARRANTIES	31
7.1	Representations and Warranties	

7.2	Deemed Repetition	36
7.3	Other Documents	37
7.4	Effective Time of Repetition	37
7.5	Nature of Representations and Warranties	37
ARTICI	E 8 GENERAL COVENANTS	37
8.1	Affirmative Covenants of the Obligors	37
8.2	Negative Covenants of the Obligors	42
8.3	Lender May Perform Covenants	44
ARTICI	E 9 SECURITY	
9.1	Security	45
9.2	Forms	45
9.3	No Merger	45
9.4	Permitted Encumbrances and Permitted Debt	46
9.5	Dealing with Security	
9.6	Effectiveness	46
9.7	Further Assurances – Security	46
9.8	Release and Discharge of Security	47
ARTICI	LE 10 EVENTS OF DEFAULT AND ACCELERATION	
10.1	Events of Default	47
10.2	Acceleration	50
10.3	Termination of Lender's Obligations	50
10.4	Remedies Cumulative and Waivers	51
10.5	Application of Payments Following Demand and Acceleration	51
10.6	Remedies	51
10.7	Set-Off	52
ARTICI	LE 11 CHANGE OF CIRCUMSTANCES	53
11.1	Change in Law	53
11.2	Illegality	54
ARTICI	LE 12 COSTS, EXPENSES AND INDEMNIFICATION	54
12.1	Costs and Expenses	54
12.2	General Indemnity	55
12.3	Environmental Indemnity	56
12.4	Judgment Currency	56
12.5	Waiver of Consequential Damages, Etc.	57
ARTICI	LE 13 INTERPRETATION; GENERAL	57
13.1	Survival of Undertakings	57
13.2	Failure to Act	57
13.3	Exchange and Confidentiality of Information	57
13.4	Notices	58
13.5	Governing Law; Jurisdiction	59
13.6	Benefit of this Agreement; Assignment; Participations	
13.7	Severability	
13.8	Whole Agreement; Amendments and Waivers	
13.9	Further Assurances	
13.10	Time of the Essence	
13.11	Agreement Governs	61
13.12	Counterparts	61
13 13	Resolving Discrepancies with Court Orders	61

THIS SENIOR SECURED SUPERPRIORITY DIP FINANCING CREDIT AGREEMENT is dated as of April ____, 2023

BETWEEN:

STRUCTURLAM MASS TIMBER CORPORATION, a corporation governed by the laws of the Province of British Columbia, having an office in Penticton, British Columbia, as Cdn. Borrower

AND

STRUCTURLAM MASS TIMBER U.S., INC., a corporation governed by the laws of the State of Delaware, having an office in Penticton, British Columbia, as U.S. Borrower

AND

BANK OF MONTREAL, as Lender

WHEREAS the Borrowers have requested that the Lender provide financing to fund certain of the Obligors' obligations during the pendency of the CCAA Proceedings and the U.S. Proceedings, in accordance with the terms and conditions set out herein; and

AND WHEREAS the Lender has agreed to establish a debtor-in-possession credit facility in order to fund certain obligations of the Obligors in the context of the Obligors' CCAA Proceedings and U.S. Proceedings on the terms and conditions set forth herein.

NOW THEREFORE, in consideration of the premises, the covenants herein contained and for other good and valuable consideration, the receipt and sufficiency of which is hereby acknowledged by each of the parties hereto, the parties agree as follows:

ARTICLE 1 INTERPRETATION

1.1 Definitions

"Acceleration Event" has the meaning set out in Section 10.2 of this Agreement

"Acceleration Notice" has the meaning set out in Section 10.2 of this Agreement

"Additional Compensation" has the meaning set out in Section 11.1(a) of this Agreement.

"Administration Charge" means an administration charge in an aggregate amount not to exceed Cdn. \$150,000 granted by the Canadian Court pursuant to the Initial Order over all present and future assets and property of the Obligors, real and personal, tangible or intangible, and whether now owned or which are hereafter acquired or otherwise become the property of an Obligor.

"Advance" means an advance of funds made by the Lender to a Borrower under this Agreement.

"Affiliate" means any Person which, directly or indirectly, controls, is controlled by or is under common control with another Person; and, for the purposes of this definition, "control" (including, with correlative meanings, the terms "controlled by" or "under common control with") means the power to direct or cause

the direction of the management and policies of any Person, whether through the ownership of shares or other economic interests, the holding of voting rights or contractual rights or otherwise.

- "Agreed Budget" means the weekly line item budget covering the period of at least 13 calendar weeks following the Effective Date annexed hereto as Schedule C, together with all amendments thereto approved by the Lender in writing in its sole and absolute discretion (for certainty, each Updated Budget contemplated to be delivered by the Borrowers hereby shall not constitute an amendment of the Agreed Budget unless approved by the Lender in writing in its sole and absolute discretion). The Agreed Budget shall be prepared on a cash-flow basis and shall include the payment of accrued professional fees to an escrow agent pursuant to procedures set forth more fully in the Interim U.S. DIP Order and Final U.S. DIP Order, as applicable, such professional fees shall be held by the agreed escrow agent until such time as they may be distributed to the relevant professionals in accordance with the orders of the U.S. Court. As used in this Agreement, and notwithstanding anything contained herein to the contrary, any requirements set forth herein for the Obligors to comply (including in respect of expenditures or use of proceeds) with the Agreed Budget shall be deemed to mean compliance with Agreed Budget subject to any permitted variance contemplated by this Agreement.
- "Agreement" means this senior secured superpriority DIP financing credit agreement and all Schedules annexed hereto, as the same may be amended, modified, supplemented or restated from time to time in accordance with the provisions thereof.
- "Alternative Restructuring Option" means any transaction involving the refinancing or recapitalization of an Obligor or any other restructuring of the Obligors' businesses and operations to be implemented pursuant to a Plan.
- "Anti-Money Laundering Laws and Regulations" means the *Proceeds of Crime (Money Laundering)* and Terrorist Financing Act (Canada), the USA PATRIOT Act, the U.S. Money Laundering Control Act of 1986, the U.S. Currency and Foreign Transactions Reporting Act of 1970 (the U.S. Bank Secrecy Act), and other applicable anti-money laundering, anti-terrorist financing, government sanction and "know-your-client" laws applicable in Canada or the United States and corresponding laws of jurisdictions in which any Obligor operates or in which the proceeds of the Loans will be used or from which repayments of the Secured Obligations will be derived.
- "APA" means documentation evidencing a sale of all or substantially all of the assets of the Obligors pursuant to Section 363 of the U.S. Bankruptcy Code or otherwise.
- "APA Closing" means the closing of the APA.
- "Applicable Laws" or "applicable law" means, in relation to any Person, transaction or event, (a) all applicable provisions of laws, statutes, rules and regulations from time to time in effect of any Governmental Authority and (b) all Governmental Authorizations to which the Person is a party or by which it or its property is bound or having application to the transaction or event.
- "Applicable Pricing Margin" means, (a) in respect of any Loan, 14% per annum, and (b) any standby fees payable under Section 5.5, 2.8% per annum, provided that (i) the above rates per annum are expressed on the basis of a year of 365 days; (ii) the standby fee will accrue on the daily undrawn portion of the Credit Facility and will be payable in arrears in accordance with Section 5.5; and (iii) upon the occurrence and during the continuance of any Default or Event of Default, the Applicable Pricing Margin specified in clause (a) of this definition will increase by 3% per annum.

- "Approved Fund" means any Fund that is administered or managed by (a) the Lender, (b) an Affiliate of the Lender or (c) a Person or an Affiliate of a Person that administers or manages the Lender.
- "Approved Sale" means the sale of all or substantially all of the assets of the Obligors pursuant to one or more sales approved by the Canadian Court and the U.S. Court and consented to by the Lender; provided, that no such consent of the Lender shall be required to the extent that the proceeds of an Approved Sale result in the repayment in full in cash of all Indebtedness owing under each of this Agreement and the Pre-Filing Secured Credit Agreement upon the closing of such Approved Sale.
- "Approved Securities" means obligations maturing within one year from their date of purchase or other acquisition by the Borrowers (or either of them) and which are (a) issued by the Government of Canada, by the government of any Province of Canada or, in each case, an instrumentality or agency thereof and guaranteed fully as to principal, premium, if any, and interest by the Government of Canada or the government of such Province or (b) term deposits, guaranteed investment certificates, certificates of deposit or bearer deposit notes, in each case, of any Canadian chartered bank or other Canadian financial institution which has a long term debt rating of at least A+ by S&P, Al by Moody's, or A (high) by DBRS.
- "Attributable Debt" means, in respect of any lease (whether characterized as an operating lease under GAAP or not) entered into by a Person as lessee, the present value (discounted at the rate of interest implicit in such transaction, determined in accordance with GAAP) of the lease payments of the lessee, including all rent and payments to be made by the lessee in connection with the return of the leased property, during the remaining term of the lease (including any period for which such lease has been extended or may, at the option of the lessor, be extended) but excluding for certainty, (a) amounts required to be paid on account of insurance, taxes, assessments, utility, operating and labour costs and similar charges and (b) amounts payable by the lessee in connection with the exercise of any end-of-term purchase option, early buyout option or any similar amounts payable at the election of the lessee.
- "Banking Day" means, (a) in respect of any U.S. \$ Loan, a day on which banks are open for business in Vancouver, British Columbia, Toronto, Ontario and New York, New York, and (b) for all other purposes, a day on which banks are open for business in Vancouver, British Columbia and Toronto, Ontario, but does not in any event include a Saturday or a Sunday.
- "Bankruptcy Code" means Title 11 of the United States Code entitled "Bankruptcy".
- "BIA" means the Bankruptcy and Insolvency Act (Canada).
- "Borrowers" means, collectively, the Cdn. Borrower and the U.S. Borrower, and "Borrower" means either of them.
- "Canadian Court" means the Supreme Court of British Columbia.
- "Canadian Dollars" and "Cdn. \$" mean the lawful money of Canada.
- "Capital Expenditures" means, for any period, all amounts that would be included as additions to property, plant and equipment and other capital expenditures on a combined consolidated statement of cash flows of the Borrowers, or on an unconsolidated statement of cash flows of the relevant Borrower, as applicable, during such period in accordance with GAAP (excluding capitalized interest but including the amount of assets leased under any Capital Lease).
- "Capital Lease" means any leasing or similar arrangement which, in accordance with GAAP in effect prior to the adoption of IFRS 16, is classified as a capital lease or finance lease.

"Carve-Out" means the "Carve-Out" as described and defined in the Interim U.S. DIP Order and Final U.S DIP Order, as applicable.

"CCAA" means the Companies' Creditors Arrangement Act (Canada).

"CCAA Proceedings" means the proceedings to be commenced in Canada by the Obligors before the Canadian Court pursuant to the CCAA.

"Cdn. Borrower" means Structurlam Mass Timber Corporation.

"Cdn. \$ Loan" means an Advance in Canadian Dollars made by the Lender to a Borrower.

"CFTC" means the U.S. Commodity Futures Trading Commission.

"Code" means the United States Internal Revenue Code of 1986, as amended from time to time.

"Collateral" has the meaning set forth in the definition of "Lender Charge" in this Agreement.

"Credit Facility" means the debtor-in-possession credit facility, including availability under the MasterCard Facilities (as defined in the Pre-Filing Secured Credit Agreement) up to the amounts of Cdn. \$25,000 and U.S. \$20,000, respectively, in the maximum principal amount of (1) Cdn. \$4,000,000 to be made available by the Lender to the Borrowers under this Agreement and the Interim U.S. DIP Order and (2) Cdn. \$3,500,000 to be made available by the Lender to the Borrowers under this Agreement and the Final U.S. DIP Order and, in each case, subject to the terms hereof.

"Credit Facility Indebtedness" means all Debt, liabilities and obligations of the Borrowers to the Lender with respect to the Credit Facility arising under this Agreement and any other Documents, including without limitation, all Loans and all interest, fees, costs, expenses and other amounts of any nature or kind payable by the Borrowers (or either of them).

"DBRS" means Dominion Bond Rating Service Limited and any successors thereto.

"Debt" means, with respect to any Person ("X"), all obligations, liabilities and indebtedness of X which would, in accordance with GAAP, be classified upon a consolidated balance sheet of X as liabilities of X and its Subsidiaries (or be classified upon an unconsolidated balance sheet of X as liabilities of X, if Debt is to be determined on an unconsolidated basis for X) and, whether or not so classified, shall include (without duplication):

- (a) indebtedness of X and its Subsidiaries (or of X on an unconsolidated basis) for borrowed money;
- (b) obligations of X and its Subsidiaries (or of X on an unconsolidated basis) arising pursuant or in relation to: (i) bankers' acceptances (including payment and reimbursement obligations in respect thereof), or (ii) letters of credit and letters of guarantee supporting obligations which would otherwise constitute Debt within the meaning of this definition or indemnities issued in connection therewith;
- (c) obligations of X and its Subsidiaries (or of X on an unconsolidated basis) with respect to drawings under all other letters of credit and letters of guarantee;
- (d) obligations of X and its Subsidiaries (or of X on an unconsolidated basis) under Guarantees, indemnities, assurances, legally binding comfort letters or other contingent obligations relating to

the indebtedness or other obligations of any other Person which would otherwise constitute Debt within the meaning of this definition and all Financial Assistance including endorsements of bills of exchange (other than for collection or deposit in the ordinary course of business);

- (e) (i) all indebtedness of X and its Subsidiaries (or of X on an unconsolidated basis) representing the deferred purchase price of any property to the extent that such indebtedness is or remains unpaid after the expiry of the customary time period for payment, provided however that such time period shall in no event exceed 90 days, and (ii) all obligations of X and its Subsidiaries (or of X on an unconsolidated basis) created or arising under any: (A) conditional sales agreement or other title retention agreement or (B) Capital Lease;
- (f) all Attributable Debt of X and its Subsidiaries (or of X on an unconsolidated basis) other than in respect of (i) leases of office space or (ii) operating leases, in each case entered into in the ordinary course of business (and, for certainty, no Sale-Leaseback shall be considered to be entered into in the ordinary course of business);
- (g) all other long-term obligations (including the current portion thereof) upon which interest charges are customarily paid prior to default by X and its Subsidiaries (or by X if on an unconsolidated basis); and
- (h) all indebtedness of other Persons secured by a Security Interest on any asset of X and its Subsidiaries (or of X if on an unconsolidated basis), whether or not such indebtedness is assumed thereby; provided that the amount of such indebtedness shall be the lesser of (i) the fair market value of such asset at such date of determination, and (ii) the amount of such indebtedness shall only be Debt to the extent recorded as a liability in accordance with GAAP,

but shall exclude each of the following, determined (as required) in accordance with GAAP:

- (i) accounts payable to trade creditors and accrued liabilities incurred in the ordinary course of business;
- (ii) current taxes payable and future taxes; and
- (iii) accrued interest not yet due and payable.

"Default" means any event or condition which, with the giving of notice, lapse of time or upon a declaration or determination being made (or any combination thereof), would constitute an Event of Default.

"Designated Persons" means a person or entity (a) listed in the annex to, or otherwise subject to the provisions of, any Executive Order; (b) named as a "Specially Designated National and Blocked Person" ("SDN") on the most current list published by OFAC at its official website or any replacement website or other replacement official publication of such list (the "SDN List") or is otherwise the subject of any Sanctions Laws and Regulations; (c) in which an entity or person on the SDN List has 50% or greater ownership interest or that is otherwise controlled by an SDN.

"Distribution" means:

- (a) the declaration, payment or setting aside for payment of any dividend or other distribution on or in respect of any Equity Interests in the capital of a Borrower or any other Obligor that is not wholly owned by a Borrower (including any return of capital);
- (b) the redemption, retraction, purchase, retirement or other acquisition, in whole or in part, of any limited partnership units or other Equity Interests in the capital of a Borrower or any other Obligor that is not wholly-owned by a Borrower or any securities, instruments or contractual rights capable of being converted into, exchanged or exercised for shares or other Equity Interests in the capital thereof, including, without limitation, options, warrants, conversion or exchange privileges and similar rights;
- (c) the making of any loan or advance or any other provision of credit or Financial Assistance by any Obligor to any Related Party, other than to another Obligor;
- (d) the payment of any principal, interest, fees or other amounts, or the redemption, exchange, purchase, retirement, defeasance, sinking fund or similar payment any loans, on or in respect of (i) subordinated debt or (ii) advances or other Debt, owing at any time by any Obligor to any Related Party, other than to another Obligor; or
- (e) (i) the payment of any amount, (ii) the sale, transfer, lease or other disposition of any property or assets, or (iii) any granting or creation of any rights or interests, at any time, by any Obligor to or in favour of any Related Party, other than to another Obligor, and whether any of the foregoing is made, paid or satisfied in or for cash, property or any combination thereof.

"Documents" means this Agreement, the Security, and all certificates, notices, instruments and other documents delivered or to be delivered to the Lender in connection herewith or therewith from time to time and all future renewals, extensions, or restatements of, or amendments, modifications or supplements to, all or any part of the foregoing.

"Drawdown" means an Advance of a Cdn. \$ Loan or U.S. \$ Loan.

"Drawdown Date" means the date on which a Drawdown is made by a Borrower pursuant to the provisions of this Agreement and which shall be a Banking Day.

"Drawdown Notice" means a notice substantially in the form annexed to this Agreement as Schedule A to be given to the Lender by a Borrower pursuant to this Agreement.

"Effective Date" has the meaning set forth in Section 3.1.

"Environmental Claims" means any and all administrative, regulatory or judicial actions, suits, demands, claims, liens, notices of noncompliance or violation, investigations, inspections, inquiries or proceedings relating in any way to any Environmental Laws or to any permit issued under any such Environmental Laws including, without limitation:

- (a) any claim by a Governmental Authority for enforcement, clean up, removal, response, remedial or other actions or damages pursuant to any Environmental Laws; and
- (b) any claim by a Person seeking damages, contribution, indemnification, cost recovery, compensation or injunctive or other relief resulting from or relating to Hazardous Materials,

including any Release thereof, or arising from alleged injury or threat of injury to human health or safety (arising from environmental matters) or the environment.

"Environmental Laws" means all Applicable Laws with respect to the environment or environmental or public health and safety matters contained in statutes, regulations, rules, ordinances, orders, judgments, approvals, notices, permits or policies, guidelines or directives having the force of law.

"Equity Interests" means, with respect to any Person, (a) shares of capital stock of (or other ownership or profit interests in) such Person, (b) warrants, options or other rights for the purchase or other acquisition from such Person of shares of capital stock of (or other ownership or profit interests in) such Person, (c) securities convertible into or exchangeable for shares of capital stock of (or other ownership or profit interests in) such Person or warrants, rights or options for the purchase or other acquisition from such Person of such shares (or such other interests), and (d) other ownership or profit interests in such Person (including, without limitation, partnership, member or trust interests therein), in each case, whether voting or non-voting, and whether or not such shares, warrants, options, rights or other interests are authorized or otherwise existing on any date of determination.

"Equivalent Amount" means, on any day, the equivalent amount in Canadian Dollars or United States Dollars, as the case may be, after giving effect to a conversion of a specified amount of United States Dollars to Canadian Dollars or of Canadian Dollars to United States Dollars, as the case may be, at the rate of exchange for Canadian interbank transactions established by the Bank of Canada for the Banking Day immediately prior to the day in question, or, if such rate is for any reason unavailable, at the spot rate quoted for wholesale transactions involving the applicable currency by the Lender at approximately noon (Toronto time) on that day in accordance with its normal practice.

"ERISA" means the Employee Retirement Income Security Act of 1974 of the United States of America, as amended from time to time, and any successor statute.

"ERISA Affiliate" means any trade or business (whether or not incorporated) that, together with any Obligor, is treated as a single employer under Section 414(b) or (c) of the Code or, solely for purposes of Section 302 of ERISA and Section 412 of the Code, is treated as a single employer under Section 414 of the Code; provided that, for certainty, an ERISA Affiliate shall only include those entities owned by Holdco, directly or indirectly.

"ERISA Event" means:

- (a) any "reportable event", as defined in Section 4043 of ERISA or the regulations issued thereunder as in effect on the Effective Date with respect to a Pension Plan (other than an event for which the 30-day notice period is waived under such regulations);
- (b) the failure to meet the minimum funding standards of Section 412 or 430 of the Code;
- (c) the filing pursuant to Section 412(c) of the Code or Section 302(c) of ERISA of an application for a waiver of the minimum funding standard with respect to any Pension Plan;
- (d) the incurrence by Holdco or any of its ERISA Affiliates of any liability under Title IV of ERISA with respect to the termination of any Pension Plan;
- (e) the receipt by Holdco or any ERISA Affiliate from the PBGC or a plan administrator of any notice relating to an intention to terminate any Pension Plan or Pension Plans or to appoint a trustee to administer any Pension Plan;

- (f) the incurrence by Holdco or any of its ERISA Affiliates of any liability with respect to the withdrawal or partial withdrawal from any Pension Plan or Multiemployer Plan;
- (g) the incurrence by Holdco or any of its ERISA Affiliates of any liability under Section 4062(e) of ERISA;
- (h) the receipt by Holdco or any ERISA Affiliate of any notice assessing Withdrawal Liability in connection with a complete or partial withdrawal from any Multiemployer Plan, or a determination that a Multiemployer Plan is, or is expected in the current Plan Year or the immediately succeeding Plan Year to be, insolvent or in reorganization, within the meaning of Title IV of ERISA; or
- (i) the conditions exist for an imposition of a lien under Section 303(k) of ERISA.

"Event of Default" has the meaning set out in Section 10.1.

"Executive Order" has the meaning assigned to such term in the definition of Sanctions Laws and Regulations.

"Exit Fee" has the meaning set forth in Section 5.6.

"FATCA" means Sections 1471 through 1474 of the Code, as of the date of this Agreement (or any amended or successor version that is substantively comparable and not materially more onerous to comply with), any current or future regulations or official interpretations thereof and any agreement entered into pursuant to Section 1471(b)(1) of the Code.

"Final U.S. DIP Order" means a final order of the U.S. Court presiding over U.S. Proceedings (i) approving liens in favor of the Lender on assets of the Obligors and super-priority claims against the Obligors and (ii) granting adequate protection to Pre-Filing Secured Creditor under the Pre-Filing Secured Credit Agreement, which order shall be satisfactory to the Lender and substantially consistent in form and substance (except with regard to finality) with the Interim U.S. DIP Order.

"Financial Assistance" means, with respect to any Person and without duplication, any loan, guarantee, indemnity, assurance, acceptance, extension of credit, loan purchase, share purchase, equity or capital contribution, investment or other form of direct or indirect financial assistance or support of any other Person or any obligation (contingent or otherwise) primarily for the purpose of enabling another Person to incur or pay any Debt or to comply with agreements relating thereto or otherwise to assure or protect creditors of the other Person against loss in respect of Debt of the other Person and includes any guarantee of or indemnity in respect of the Debt of the other Person and any absolute or contingent obligation to (directly or indirectly):

- (a) advance or supply funds for the payment or purchase of any Debt of any other Person;
- (b) purchase, sell or lease (as lessee or lessor) any property, assets, goods, services, materials or supplies primarily for the purpose of enabling any Person to make payment of Debt or to assure the holder thereof against loss;
- (c) guarantee, indemnify, hold harmless or otherwise become liable any creditor of any other Person from or against any losses, liabilities or damages in respect of Debt;
- (d) make a payment to another for goods, property or services regardless of the non-delivery or nonfurnishing thereof to that Person; or

(e) make an advance, loan or other extension of credit to or to make any subscription for equity, equity or capital contribution, or investment in or to maintain the capital, working capital, solvency or general financial condition of another Person.

The amount of any Financial Assistance is the amount of any loan or direct or indirect financial assistance or support, without duplication, given, or all Debt of the obligor to which the Financial Assistance relates, unless the Financial Assistance is limited to a determinable amount, in which case the amount of the Financial Assistance is the determinable amount.

"Financial Instrument" means any instrument or agreement which effects a rate swap transaction, basis swap, forward rate transaction, commodity swap, commodity option, equity or equity index swap, equity or equity index option, bond option, interest rate option, foreign exchange transaction, cap transaction, floor transaction, collar transaction, currency swap transaction, cross-currency rate swap transaction, currency option, forward sale, exchange traded futures contract or any other similar transaction (including any option with respect to any of these transactions or any combination of these transactions).

"Fund" means any Person (other than a natural person) that is (or will be) engaged in making, purchasing, holding or otherwise investing in commercial loans and similar extensions of credit in the ordinary course of its business.

"GAAP" means generally accepted accounting principles in Canada, which shall be deemed to be reference to the recommendations at the relevant time of the Chartered Professional Accountants of Canada (or any successor institute thereto) applicable on a consolidated basis (unless otherwise specifically provided or contemplated in this Agreement) as at the date on which any determination or calculation is made or required to be made in accordance with such principles, which, for purposes of this Agreement is determined to be Canadian accounting standards for private enterprises (ASPE), applied on a consistent basis.

"Governmental Authority" means any federal, provincial, state, regional, municipal or local government or any department, agency, board, tribunal, accreditation entity or authority thereof or other political subdivision thereof and any entity or Person exercising executive, legislative, judicial, regulatory or administrative functions of, or pertaining to, government or the operation thereof.

"Governmental Authorization" means an authorization, accreditation, order, permit, approval, grant, license, consent, right, franchise, privilege, certificate, judgment, writ, injunction, award, determination, direction, decree or demand or the like issued or granted by law or by rule or regulation of any Governmental Authority.

"Guarantee" means any guarantee, undertaking to assume, endorse, contingently agree to purchase or to provide funds for the payment of, or otherwise become liable in respect of, any obligation of any Person; provided that the amount of each Guarantee shall be deemed to be the amount of the obligation guaranteed thereby, unless the Guarantee is limited to a determinable amount in which case the amount of such Guarantee shall be deemed to be the lesser of such determinable amount or the amount of such obligation; and the term "Guaranteed" shall be construed accordingly.

"Guarantors" means, (a) with respect to all Secured Obligations of the Borrowers, (i) Holdco and (ii) all present and future Subsidiaries of the Borrowers (or of either of them), (b) with respect to all Secured Obligations of the Cdn. Borrower, the U.S. Borrower, and (c) with respect to all Secured Obligations of the U.S. Borrower, and "Guarantor" means any one of them.

"Hazardous Materials" means any substance or mixture of substances which, if released into the environment, would likely cause, immediately or at some future time, harm or degradation to the environment or to human health or safety and includes any substance defined as or determined to be a pollutant, contaminant, waste, hazardous waste, hazardous chemical, hazardous substance, toxic substance or dangerous good under any Environmental Law.

"Holdco" means SLP Holdings Ltd., the corporation which legally and beneficially owns 100% of the Equity Interests of the Cdn. Borrower, and which legally and beneficially owns 100% of the Equity Interests of the U.S. Borrower.

"Indemnified Parties" means the Lender, including a receiver, receiver manager or similar Person appointed under applicable law, and their respective shareholders, Affiliates, officers, directors, employees and agents, and "Indemnified Party" means any one of the foregoing.

"Indemnified Third Party" has the meaning set out in Section 12.3 of this Agreement.

"Information" has the meaning set out in Section 13.3(b) of this Agreement.

"Information Officer" means Alvarez & Marsal Canada Inc., in its capacity as court appointed information officer of the Obligors.

"Initial Order" means an initial recognition order and supplemental order of the Canadian Court substantially in the form annexed hereto as Schedule D and otherwise in form and substance satisfactory to the Lender that, among other things, (i) recognizes the Foreign Representative as a "foreign representative" of the Obligors pursuant to the provisions of the CCAA, (ii) recognizes the U.S. Proceedings as foreign main or foreign nonmain proceedings in respect of each of the Obligors pursuant to the provisions of the CCAA, (iii) recognizes and gives effect to certain orders granted in the U.S. Proceedings, including, among other things, the Interim U.S. DIP Order and the Cash Management Order, (iv) grants the Lender Charge (subject only to Permitted Priority Liens and the Carve-Out), and (v) appoints the Information Officer, as information officer (as such order or orders may be amended, modified, varied or restated with the consent of the Lender).

"Intellectual Property" means, collectively, patents, patents pending, copyrights, proprietary processes or programs, industrial designs, trademarks, trademark applications, trade names and other intellectual property of every nature and kind.

"Interest Payment Date" means, with respect to each Loan, the last Banking Day of each calendar month; provided that, the Maturity Date or, if applicable, any earlier date on which the Credit Facility is fully cancelled or permanently reduced in full, shall be an Interest Payment Date with respect to all Loans then outstanding under the Credit Facility.

"Interest Period" means, with respect to each Loan, the period commencing on the applicable Drawdown Date and terminating on the date selected by the relevant Borrower under this Agreement for the repayment of such Loan, <u>provided</u> that the last day of all Interest Periods for Loans outstanding under the Credit Facility shall expire on or prior to the Maturity Date.

"Interim U.S. DIP Order" means an interim order of the U.S. Court presiding over U.S. Proceedings authorizing the Obligors to enter into this agreement and, *inter alia*, (i) approving liens in favor of the Lender on assets of the Obligors and super-priority claims against the Obligors and (ii) granting adequate protection to Pre-Filing Secured Creditor under the Pre-Filing Secured Credit Agreement, which order shall be satisfactory to the Lender and substantially in the form annexed hereto as Schedule F.

"Interpretive Guidance" means the Interpretive Guidance and Policy Statement Regarding Compliance with Certain Swap Regulations, 78 Fed. Reg. 45292 (July 26, 2013), as amended or supplemented by the CFTC from time to time.

"Investment" means (a) any purchase or other acquisition of shares or other securities of any Person, (b) any form of Financial Assistance to or for the benefit of any Person, (c) any capital contribution to any other Person and (d) any purchase or other acquisition of any assets, property or undertaking (including Capital Expenditures) other than an acquisition in the ordinary course of business of the purchaser.

"Judgment Conversion Date" has the meaning set out in Section 12.4(a)(ii) of this Agreement.

"Judgment Currency" has the meaning set out in Section 12.4(a) of this Agreement.

"KEIP" means any key employee incentive plan acceptable to the Lender and approved by the U.S. Court in the U.S. Proceedings.

"KERP" means any key employee retention plan acceptable to the Lender and approved by the U.S. Court in the U.S. Proceedings.

"Lender" means Bank of Montreal.

"Lender Charge" means, collectively, the super-priority administrative claim, authorized by the U.S. Court pursuant to the Interim U.S. DIP Order or the Final U.S. DIP Order, as applicable, in accordance with § 364(c)(1) of the Bankruptcy Code, having priority in right of payment over any and all other obligations, liabilities and indebtedness of the Obligors (other than the Carve-Out), the liens authorized by the U.S. Court pursuant to the Interim U.S. DIP Order or the Final U.S. DIP Order, as applicable, in accordance with § 364(d) of the Bankruptcy Code, having priority in right of payment over any and all other obligations, liabilities and indebtedness of the Obligors (other than the Permitted Priority Liens and the Carve-Out) and the charge granted by the Canadian Court pursuant to the Initial Order over all present and future assets and property of the Obligors, real and personal, tangible or intangible, and whether now owned or which are hereafter acquired or otherwise become the property of an Obligor in favour of the Lender (collectively, the "Collateral").

"Lender's Account for Payments" means:

(a) for amounts payable in Canadian Dollars:

Bank of Montreal 595 Burrard Street Vancouver, British Columbia V7X 1L7

Transit No. and Bank Code: 00149 001 Beneficiary Account No.: 1999-815

(b) for amounts payable in United States Dollars:

Bank of Montreal 595 Burrard Street Vancouver, British Columbia V7X 1L7 - 12 -

Transit No. and Bank Code: 00109 001 Beneficiary Account No.: 4604-350

"Lender's Branch" means the Lender's branch located at 595 Burrard Street, Vancouver, B.C. V7X 1L7 or such other branch in Canada as the Lender may from time to time designate.

"Lender's Counsel" means the firms of Blake, Cassels & Graydon LLP, Chapman and Cutler LLP and Womble Bond Dickinson (US) LLP or such other firms of legal counsel as the Lender may from time to time designate.

"Lender's Counsel's Financial Advisor" means FTI Consulting Canada Limited, in its capacity as financial advisor to the Lender's Counsel.

"Loan" means a Cdn. \$ Loan or U.S. \$ Loan under this Agreement.

"Material Adverse Change" means a material adverse change in:

- (a) the business, financial condition, operations, assets, management or properties of the Obligors on a consolidated basis, taken as a whole;
- (b) the validity or enforceability of this Agreement or any other Document or the validity or enforceability of any of the Lender Charge or the ranking of any of the Security Interests granted thereby or the material rights or remedies intended or purported to be granted to the Lender under or pursuant to such Lender Charge; or
- (c) the ability of any Obligor to perform any material obligation under this Agreement, any other Document or any Restructuring Court Order, or the ability of any Obligor to carry out an Approved Sale or Alternative Restructuring Option,

but does not include any of the Obligors filing for relief under the CCAA or under Chapter 11 or Chapter 15 of the Bankruptcy Code or any facts or circumstances disclosed by the Obligors in the pleadings supporting any such filing.

"Material Adverse Effect" means a material adverse effect upon:

- (a) the business, financial condition, operations, assets, management or properties of the Obligors on a consolidated basis, taken as a whole;
- (b) the validity or enforceability of this Agreement or any other Document or the validity or enforceability of any of the Lender Charge or the ranking of any of the Security Interests granted thereby or the material rights or remedies intended or purported to be granted to the Lender under or pursuant to such Lender Charge; or
- (c) the ability of any Obligor to perform any material obligation under this Agreement, any other Document or any Restructuring Court Order, or the ability of any Obligor to carry out a an Approved Sale or Alternative Restructuring Option.

but does not include the effect of any of the Obligors filing for relief under the CCAA or under Chapter 11 or Chapter 15 of the Bankruptcy Code or the effect of any facts or circumstances disclosed by the Obligors in the pleadings supporting any such filing.

"Maturity Date" means, the earlier to occur of: (a) June 30, 2023; (b) the effective date of a Plan; (c) the APA Closing; and (d) the date of delivery of an Acceleration Notice or the occurrence of an Acceleration Event.

"Milestones" means, unless otherwise agreed to by the Lender, each of the following events:

- reasonable best efforts shall be made by the Borrowers to have the U.S. Court enter the Interim U.S. DIP Order on or prior to one (1) Banking Day after the Petition Date, and in any case on or prior to two (2) Banking Days after the Petition Date, the U.S. Court shall have entered the Interim U.S. DIP Order;
- (b) reasonable best efforts shall be made by the Borrowers to have the Canadian Court issue the Initial Order on or prior to two (2) Banking Days after the date the U.S. Court enters the Interim U.S. DIP Order, and in any case on or prior to three (3) Banking Days after the date the U.S. Court enters the Interim U.S. DIP Order, the Canadian Court shall have issued the Initial Order;
- on or prior to one (1) Banking Day after the Petition Date, the Obligors shall have filed with the U.S. Court a motion seeking entry of the Sale Procedures Order and the Sale Order;
- (d) on or prior to April 28, 2023, the Obligors shall have filed with the Canadian Court a motion seeking an order recognizing the Sale Procedures Order and Sale Order.
- (e) reasonable best efforts shall be made by the Borrowers to have the U.S. Court enter the Sale Procedures Order by April 28, 2023, and in any case on or prior to the first Banking Day that is ten (10) calendar days after the Petition Date, the U.S. Court shall have entered the Sale Procedures Order;
- reasonable best efforts shall be made by the Borrowers to have the Canadian Court issue an order recognizing the Sale Procedures Order by May 8, 2023, and in any case on or prior to May 12, 2023, the Canadian Court shall have issued an order recognizing the Sale Procedures Order;
- (g) subject only to availability of the U.S. Court, on or prior to twenty-one (21) days after the date the U.S. Court enters the Interim U.S. DIP Order, the U.S. Court shall have entered the Final U.S. DIP Order (if the U.S. Court does not have the requisite availability, this Milestone is to be discharged on the U.S. Court's next available date);
- (h) reasonable best efforts shall be made by the Borrowers to have the U.S. Court enter the Sale Order on or prior to May 24, 2023 and in any case on or prior to the first Banking Day that is 35 calendar days after the Petition Date the U.S. Court shall have entered the Sale Order;
- (i) reasonable best efforts shall be made by the Borrowers to have the Canadian Court issue an order recognizing the Sale Order on or prior to May 29, 2023 and in any case prior to May 31, 2023, the Canadian Court shall have issued an order recognizing the Sale Order; and
- (j) by the later of June 1, 2023, and the date that is 45 calendar days after the Petition Date, the final APA Closing shall have occurred and the Credit Facility Indebtedness shall have been paid in full in cash.

[&]quot;Month End" means the last day of each of calendar month.

[&]quot;Moody's" means Moody's Investors Service, Inc. and any successors thereto.

"Multiemployer Plan" means a multiemployer plan as defined in Section 4001(a)(3) of ERISA.

"Negative Cash Flow Variance" means, at any time, the cumulative negative variance in the Obligors' actual cash receipts and cash expenditures and outflows, in respect of any line item (other than in respect of (i) foreign exchange fluctuations and (ii) fees and expenses of professionals engaged by the Lender) to the Agreed Budget of 15% or greater. For greater certainty, the payment of professional fees into escrow on a monthly basis pursuant to procedures set forth more fully in the Interim U.S. DIP Order and Final U.S. DIP Order, as applicable, shall be included in any variance calculation as if those payments were made to the professionals directly.

"Obligor Guarantee" means, collectively, the unlimited guarantee and indemnity (including any addition agreements executed in connection therewith) governed by the laws of the Province of British Columbia, in form satisfactory to the Lender, executed by each Obligor in favour of the Lender in respect of the Secured Obligations of each other Obligor, and each other guarantee executed by an Obligor governed by the laws of any other jurisdiction, if requested by the Lender, acting reasonably.

"Obligors" means, collectively, the Borrowers and Guarantors, and "Obligor" means any one of them.

"OFAC" has the meaning set out in the definition of Sanctions Laws and Regulations.

"Officer's Certificate" means a certificate or notice signed by any one of the chief executive officer or chief financial officer of a Borrower or of the applicable Obligor; provided that Drawdown Notices and Repayment Notices shall be executed on behalf of the relevant Borrower by any one of the foregoing persons or such other persons as may from time to time be designated by written notice from the Borrowers to the Lender.

"Outside Date" means April 27, 2023.

"Outstanding Principal" means, at any time, the aggregate of (i) the principal amount of all outstanding Cdn. \$ Loans, and (ii) the Equivalent Amount in Canadian Dollars of the principal amount of all outstanding U.S. \$ Loans, in each case, advanced or issued and outstanding under the Credit Facility.

"Payment Protocol" has the meaning given to such term in Section 2.10.

"PBGC" means Pension Benefit Guaranty Corporation or any Governmental Authority succeeding to any of its principal functions under ERISA.

"Pension Plan" means any employee pension benefit plan (other than a Multiemployer Plan) sponsored or contributed to by Holdco or any ERISA Affiliate which is subject to the provisions of Title IV of ERISA or Section 412 of the Code or Section 302 of ERISA, and in respect of which Holdco or any ERISA Affiliate is (or, if such plan were terminated, would under Section 4069 of ERISA during the time period set out therein) would have liability for contributions under Section 412(c)(11) of the Code or as a "contributing sponsor" as defined in Section 4001(a)(13) of ERISA.

"Permitted Contest" means action taken by or on behalf the Obligors in good faith by appropriate proceedings diligently pursued to contest a Tax, claim or Security Interest, provided that:

(a) the Borrowers, on a combined consolidated basis, have established reasonable reserves therefor if and to the extent required by GAAP;

- (b) proceeding with such contest does not have, and would not reasonably be expected to have, a Material Adverse Effect; and
- (c) proceeding with such contest will not create a material risk of sale, forfeiture or loss of, or interference with the use or operation of, a material part of the property, assets and undertaking of any Obligor.

"Permitted Debt" means the following:

- (a) the Secured Obligations;
- (b) Debt owing to the Lender;
- (c) all Debt of the Obligors which has been issued, created, incurred, assumed or existed on or prior to the Effective Date and to the extent such Debt was Permitted Debt (under and as defined in the Pre-Filing Secured Credit Agreement);
- (d) Debt owing by an Obligor to any other Obligor in accordance with the Agreed Budget;
- (e) Indebtedness incurred in the ordinary course of business by an Obligor after the Effective Date with respect to trade payables;
- (f) all Debt secured by a Permitted Encumbrance; and
- (g) such other Debt of an Obligor which the Lender has consented to in writing.

"Permitted Dispositions" means, in respect of an Obligor, any sale, exchange, lease, transfer or other disposition of inventory in the ordinary course of business.

"Permitted Encumbrances" means, as at any particular time, any of the following encumbrances on the property or any part of the property of an Obligor:

- (a) the Lender Charge;
- (b) the other Permitted Priority Liens;
- valid and perfected Security Interests existing prior to the Effective Date and to the extent such Security Interests were Permitted Encumbrances (under and as defined in the Pre-Filing Secured Credit Agreement);
- (d) inchoate statutory Security Interests arising after the Effective Date in respect of any accounts payable arising after the Outside Date in the ordinary course of business, provided that all such amounts are paid by the applicable Obligor as and when due;
- (e) liens for taxes, assessments or governmental charges not at the time due or delinquent or, if due or delinquent, the validity of which is being contested at the time by a Permitted Contest;
- (f) deemed liens and trusts arising by operation of law in connection with workers' compensation, employment insurance and other social security legislation, in each case, which secure obligations not at the time due or delinquent or, if due or delinquent, the validity of which is being contested at the time by a Permitted Contest;

- (g) undetermined or inchoate liens and charges incidental to operations arising in the ordinary course of business which have not at such time been filed pursuant to law against any Obligor or which relate to obligations not due or delinquent or, if due or delinquent, the validity of which is being contested at the time by a Permitted Contest;
- (h) easements, rights of way, servitudes or other similar rights in land (including, without in any way limiting the generality of the foregoing, rights of way and servitudes for railways, sewers, drains, gas and oil and other pipelines, gas and water mains, electric light and power conduits, poles, wires and cables) granted to or reserved or taken by other Persons which individually or in the aggregate do not materially detract from the value of the land concerned or materially impair its use in the operation of the business of the Obligors;
- (i) security given by any Obligor to a public utility or any municipality or governmental or other public authority when required by such utility or municipality or other authority in connection with the operations of the Obligor, all in the ordinary course of its business which individually or in the aggregate do not materially detract from the value of the asset concerned or materially impair its use in the operation of the business of the Obligor;
- (j) pledges or deposits to secure performance of (i) bids, tenders, contracts (other than contracts for the payment of money) or (ii) leases of real property, in each case, to which an Obligor is a party;
- (k) Security Interests resulting from the deposit of cash or Approved Securities as security when any Obligor is required to do so by a Governmental Authority or by normal business practice in connection with contracts, licenses or tenders or similar matters in the ordinary course of business and for the purpose of carrying on the same, or to secure workers' compensation, surety or appeal bonds or to secure costs of litigation when required by Applicable Law;
- (l) minor defects of title which, individually and in the aggregate, do not materially affect the right of ownership of the Obligors in and to the property affected thereby or the right of any Obligor to utilize such property to conduct its business;

<u>provided</u> that nothing in this definition shall in and of itself cause the Secured Obligations under this Agreement or the other obligations secured by the Security to be subordinated in priority of payment to any such Permitted Encumbrance or cause any Security Interests in favour of the Lender to rank subordinate to any such Permitted Encumbrance.

"Permitted Financial Assistance" means any Financial Assistance provided by the Obligors on or prior to date hereof and which was Permitted Financial Assistance (under and as defined in the Pre-Filing Secured Credit Agreement) and any Financial Assistance provided to an Obligor in accordance with the Agreed Budget.

"Permitted Priority Liens" means:

- (a) the Administration Charge;
- (b) Permitted Encumbrances (under and as defined in the Pre-Filing Secured Credit Agreement) that were properly perfected and senior in priority to the Pre-Filing Security under applicable law as of the Petition Date;
- (c) statutory security interests that by operation of law cannot be subordinated or primed by the Lender Charge; and

(d) such other Security Interests as may be agreed to in writing by the Lender,

<u>provided</u> that, for greater certainty, except as expressly set forth herein, Security Interests arising from the construction, repair, maintenance and/or improvement of real or personal property, shall not be "Permitted Priority Liens".

"Person" includes an individual, a partnership, a joint venture, a trust, an unincorporated organization, a company, a corporation, an association, a government or any department or agency thereof including Canada Revenue Agency, and any other incorporated or unincorporated entity.

"Petition Date" means the date of commencement of the U.S. Proceedings by the filing of voluntary petitions for relief by the Obligors.

"Plan" means a plan of reorganization, liquidation, compromise or arrangement within the CCAA Proceedings or the U.S. Proceedings that either (i) is in form and substance satisfactory to the Lender or (ii) would result in the repayment in full in cash of all Indebtedness owing under each of this Agreement and the Pre-Filing Secured Credit Agreement on the effective date of such Plan.

"Pre-Filing Secured Credit Agreement" means the amended and restated credit agreement dated as of December 3, 2019 between Structurlam Mass Timber Corporation, as Canadian borrower, Structurlam Mass Timber U.S., Inc., as U.S. borrower, and Bank of Montreal, as lender (as amended prior to the date hereof).

"Pre-Filing Secured Creditor" means Bank of Montreal, as the lender under the Pre-Filing Secured Credit Agreement.

"Pre-Filing Security" means the Security Interests granted to the Pre-Filing Secured Creditor by the Obligors.

"Quarter End" means the last day of each of March, June, September and December in each fiscal year of the Borrowers.

"Receiver" means any receiver, trustee, examiner, interim receiver, manager, or receiver and manager of the Collateral or any part thereof or the business and undertaking of any Obligor, or any part thereof, whether appointed by the Lender under a Document or by a court pursuant to Applicable Law and any nominee of the Lender or any other person that is appointed by the Lender to exercise all or any of the powers, rights, benefits and discretion of the Lender under such Document.

"Related Party" means any Person which is any one or more of the following:

- (a) an Affiliate of a Borrower, or any Subsidiary thereof, other than an Obligor;
- (b) a shareholder or partner of any Obligor, which, together with all Affiliates of such Person, owns or controls, directly or indirectly, more than 10% of the shares, capital or other ownership interests (however designated) of such Obligor, or an Affiliate of any such shareholder or partner;
- (c) the Sponsor;
- (d) Holdco;
- (e) an officer or director of any of the foregoing; and

- (f) a Person which is not at arm's length from any Obligor.
- "Release" means any release, spill, emission, leak, pumping, injection, deposit, disposal, discharge, dispersal, leaching or migration into the environment including, without limitation, the movement of Hazardous Materials through ambient air, soil, surface water, ground water, wetlands, land or sub-surface strata.
- "Reorganization Transaction" has the meaning set forth in Section 8.2(d).
- "Repayment Notice" means a notice substantially in the form annexed to this Agreement as <u>Schedule B</u> to be given to the Lender by a Borrower pursuant to this Agreement.
- "Required Permits" means, in respect of any Obligor, all Governmental Authorizations which are necessary at any given time for such Obligor to own and operate its property, assets, rights and interests or to carry on its business and affairs.
- "Restructuring Court Orders" means, collectively, the court orders made in the CCAA Proceedings and the U.S. Proceedings applicable to an Obligor, and "Restructuring Court Order" means any one of such orders.
- "S&P" means the Standard & Poor's Rating Group (a division of The McGraw-Hill Companies, Inc.) and any successors thereto.
- "Sale-Leaseback" means an arrangement, transaction or series of arrangements or transactions under which title to any real property, tangible personal property or fixture is transferred by an Obligor to another Person which leases or otherwise grants the right to use such property to the Obligor (or its nominee) and, whether or not in connection therewith, the Obligor also acquires a right or is subject to an obligation to acquire such property or a material portion thereof, and regardless of the accounting treatment of such arrangement, transaction or series of arrangements or transactions.
- "Sale Order" means an order of the U.S. Court presiding over the U.S. Proceedings approving, pursuant to section 363 of the Bankruptcy Code, one or more sales of, collectively, all or substantially all of the assets of the Obligors to one or more purchasers on an APA Closing, that approves the payment in full in cash of all Indebtedness owing under each of this Agreement and the Pre-Filing Secured Credit Agreement or that is otherwise acceptable in form and substance to the Lender.
- "Sale Procedures Order" means an order or orders of the U.S. Court presiding over U.S. Proceedings each acceptable in form and substance to the Lender approving one or more APAs as a stalking horse bidder on terms and conditions acceptable to the Lender and approving bidding and sale procedures for one or more sales pursuant to Section 363 of the Bankruptcy Code on terms and conditions acceptable to the Lender.
- "Sanctions Laws and Regulations" means (a) any sanctions, prohibitions or requirements imposed by any executive order (an "Executive Order") or by any sanctions program administered by the U.S. government, including without limitation the U.S. Department of the Treasury Office of Foreign Assets Control ("OFAC") and (b) any sanctions measures imposed by the United Nations Security Council, European Union or the United Kingdom.
- "Security" has the meaning set forth in Section 9.1.
- "Security Interest" means mortgages, charges, pledges, hypothecs, assignments by way of security, conditional sales or other title retentions, security created under the Bank Act (Canada), liens,

encumbrances, security interests or other interests in property, howsoever created or arising, whether fixed or floating, perfected or not, which secure payment or performance of an obligation and, including, in any event:

- (a) deposits or transfers of cash, marketable securities or other financial assets under any agreement or arrangement whereby such cash, securities or assets may be withdrawn, returned or transferred only upon fulfilment of any condition as to the discharge of any other indebtedness or other obligation to any creditor;
- (b) (i) rights of set-off or (ii) any other right of or arrangement of any kind with any creditor, which in any case are made, created or entered into, as the case may be, for the purpose of or having the effect (directly or indirectly) of (A) securing Debt, (B) preferring some holders of Debt over other holders of Debt or (C) having the claims of any creditor be satisfied prior to the claims of other creditors with or from the proceeds of any properties, assets or revenues of any kind now owned or later acquired (other than, with respect to (C) only, rights of setoff granted or arising in the ordinary course of business);
- (c) the rights of lessors under Capital Leases, operating leases and any other lease financing; and
- (d) absolute assignments of accounts receivable.

"Seller Disclosure Schedule" has the meaning ascribed to that term in the Stalking Horse Bid.

"Sponsor" means KF Arc Holdings LP (or its Affiliates), or an assignee of KF Arc Holdings LP approved by the Lender in its sole discretion.

"Stalking Horse Bid" means the Asset Purchase Agreement between Mercer International Inc. and the Obligors dated April 21, 2023.

"Subsidiary" means, with respect to any Person ("X"):

- (a) any corporation of which at least a majority of the outstanding shares having by the terms thereof ordinary voting power to elect a majority of the board of directors of such corporation (irrespective of whether at the time shares of any other class or classes of such corporation might have voting power by reason of the happening of any contingency, unless the contingency has occurred and then only for as long as it continues) is at the time directly, indirectly or beneficially owned or controlled by X or one or more of its Subsidiaries, or X and one or more of its Subsidiaries;
- (b) any partnership of which, at the time, X, or one or more of its Subsidiaries, or X and one or more of its Subsidiaries: (i) directly, indirectly or beneficially own or control more than 50% of the income, capital, beneficial or ownership interests (however designated) thereof; and (ii) is a general partner, in the case of limited partnerships, or is a partner or has authority to bind the partnership, in all other cases; or
- (c) any other Person of which at least a majority of the income, capital, beneficial or ownership interests (however designated) are at the time directly, indirectly or beneficially owned or controlled by X, or one or more of its Subsidiaries, or X and one or more of its Subsidiaries,

<u>provided</u> that, unless otherwise expressly provided or the context otherwise requires references in this Agreement to "Subsidiary" or "Subsidiaries" shall be and shall be deemed to be references to Subsidiaries of the Cdn. Borrower or the other Obligors.

"Taxes" means all taxes, levies, imposts, stamp taxes, duties, fees, deductions, withholdings, charges, compulsory loans or restrictions or conditions resulting in a charge which are imposed, levied, collected, withheld or assessed by any country or political subdivision or taxing authority thereof now or at any time in the future, together with interest thereon and penalties, charges or other amounts with respect thereto, if any, and "Tax" and "Taxation" shall be construed accordingly.

"United States Dollars" and "U.S. \$" mean the lawful money of the United States of America.

"Updated Budget" means an update to the Agreed Budget prepared and delivered to the Lender on a weekly basis in accordance with Section 8.1(e)(ii)(B) for the period commencing at the end of the Friday of the previous week through and including 13 weeks thereafter, which shall reflect the Borrowers' good faith projections and be in form and detail consistent with the initial Agreed Budget and subject to the approval of the Lender (for certainty, the Updated Budget shall not constitute an amendment of the Agreed Budget unless consented to by the Lender in its sole and absolute discretion (exercised reasonably)). Each Updated Budget shall be prepared on a cash-flow basis and shall include the payment of accrued professional fees to an escrow agent pursuant to procedures set forth more fully in the Interim U.S. DIP Order and Final U.S. DIP Order, as applicable, to be held by the agreed escrow agent until such time as they may be distributed to the relevant professionals in accordance with the orders of the U.S. Court.

"USA PATRIOT Act" means the U.S. Uniting and Strengthening America by Providing Appropriate Tools Required to Intercept and Obstruct Terrorism Act of 2001 and the regulations and rules promulgated thereunder, as amended from time to time.

"U.S. \$ Loan" means an Advance in United States Dollars made by the Lender to a Borrower under the Credit Facility.

"U.S. Borrower" means Structurlam Mass Timber U.S., Inc., a Delaware corporation.

"U.S. Court" means the United States Bankruptcy Court for the District of Delaware.

"U.S. Guarantor" means Natural Outcomes, LLC, a Delaware limited liability corporation.

"U.S. Proceedings" means the proceedings to be commenced in the United States by the Borrowers and the other Obligors before the U.S. Court pursuant to chapter 11 of the Bankruptcy Code.

"Variance Testing Date" means the second Wednesday occurring after the Outside Date and each Wednesday thereafter.

"Weekly Cashflow Report" has the meaning set forth in Section 8.1(e)(ii)(A).

"Withdrawal Liability" means liability to a Multiemployer Plan as a result of a complete or partial withdrawal from such Multiemployer Plan, as such terms are defined in Part I of Subtitle E of Title IV of ERISA.

1.2 Headings; Articles and Sections

The division of this Agreement into Sections and Subsections and the insertion of headings are for convenience of reference only and shall not affect the construction or interpretation of this Agreement. The terms "this Agreement", "hereof", "hereunder" and similar expressions refer to this Agreement and not to any particular Article, Section or other portion hereof and include any agreement

supplemental hereto. Unless something in the subject matter or context is inconsistent therewith, references herein to Articles and Sections are to Articles and Sections of this Agreement.

1.3 Number; including

Words importing the singular number only shall include the plural and *vice versa*, words importing the masculine gender shall include the feminine and neuter genders and vice versa, and words and terms denoting inclusiveness (such as "include" or "includes" or "including"), whether or not so stated, are not limited by their context or by the words or phrases which precede or succeed them.

1.4 Application of Accounting Principles; Accounting Changes

Where the character or amount of any asset or liability or item of revenue or expense or amount of equity is required to be determined, or any consolidation or other accounting computation is required to be made for the purpose of this Agreement or any other Document, such determination or calculation shall, to the extent applicable and except as otherwise specified herein or as otherwise agreed in writing by the parties, be made in accordance with GAAP applied on a consistent basis. In the event that any "Accounting Change" (as defined below) shall occur and such change results in a change in the method of calculation of financial covenants, standards or terms in this Agreement, then the Borrowers and the Lender agree to enter into good faith negotiations in order to amend such provisions of this Agreement so as to reflect equitably such Accounting Changes with the desired result that the criteria for evaluating the Cdn. Borrower's consolidated financial condition shall be substantially the same after such Accounting Changes as if such Accounting Changes had not been made. Until such time as such an amendment shall have been executed and delivered by the Borrowers and the Lender, all financial covenants, standards and terms in this Agreement shall continue to be calculated or construed as if such Accounting Changes had not occurred. "Accounting Changes" refers to changes in accounting principles required by the promulgation of any rule, regulation, pronouncement or opinion by the Chartered Professional Accountants of Canada, and in all events including changes resulting from implementation of the International Financial Reporting Standards or accounting standards for private enterprises to the extent required by the Canadian Accounting Standards Board and to the extent applicable to the Borrowers.

1.5 References to Agreements and Enactments

Reference herein to any agreement, instrument, licence or other document shall be deemed to include reference to such agreement, instrument, licence or other document as the same may from time to time be amended, modified, supplemented or restated in accordance with the provisions of this Agreement if and to the extent such provisions are applicable; and reference herein to any enactment shall be deemed to include reference to such enactment as re-enacted, amended or extended from time to time and to any successor enactment.

1.6 Per Annum Calculations; Currency

Unless otherwise stated, wherever in this Agreement reference is made to a rate "per annum" or a similar expression is used, such rate is expressed on the basis of, and shall be calculated on the basis of, a year of 365 or 366 days, as applicable. Unless otherwise stated herein, all dollar amounts referenced in this Agreement are in Canadian currency.

1.7 Schedules

The following are the Schedules annexed hereto and incorporated by reference and deemed to be part hereof:

- 22 -

Schedules $7.1(e)$ to $7.1(x)$	=	Disclosure Schedules
Schedule A	-	Drawdown Notice
Schedule B	-	Repayment Notice
Schedule C	-	Agreed Budget
Schedule D	-	Initial Order
Schedule E	-	[Reserved]
Schedule F	-	Interim U.S. DIP Order
Schedule G	-	[Reserved]

1.8 Time

Unless otherwise provided herein, all references to a time in this Agreement shall mean local time in Vancouver, British Columbia.

1.9 Payment for Value

All payments required to be made hereunder shall be made for value on the required day in same day immediately available funds.

ARTICLE 2 THE CREDIT FACILITY

2.1 The Credit Facility

Subject to the terms and conditions hereof, the Lender shall make available to the Borrowers the Credit Facility. At any date, the Outstanding Principal under the Credit Facility shall not exceed the maximum principal amount thereof.

2.2 Types of Availments

- (a) Under the Credit Facility, a Borrower may, in Canadian Dollars, make Drawdowns of Cdn. \$ Loans and may, in United States Dollars, make Drawdowns of U.S. \$ Loans.
- (b) The Borrowers shall have the option, subject to the terms and conditions hereof, to determine which types of Loans shall be drawn down under the Credit Facility and in which combinations or proportions.

2.3 Purpose

The Credit Facility is being made available to the Borrowers solely for the purposes set forth in Section 8.1(s).

2.4 Availability of the Credit Facility

Subject to the terms and conditions hereof, the Borrowers may make Drawdowns under the Credit Facility after the Effective Date and prior to the Maturity Date for the purposes described in Section 2.3 above.

2.5 Minimum Drawdowns

Each Drawdown under the Credit Facility of the following types of Loans shall be in the following amounts indicated:

- (a) Cdn. \$ Loans in minimum principal amounts of \$50,000; and
- (b) U.S. \$ Loans in minimum principal amounts of U.S.\$50,000.

2.6 Drawdown Notices; Drawdowns

A Drawdown Notice shall be given to the Lender by 11:00 a.m. (Local Time) on Friday of each calendar week (or as otherwise required with the approval of the Information Officer, the Lender and the Lender's Counsel's Financial Advisor) substantially in the form of Schedule A annexed hereto. Subject to the satisfaction of the conditions precedent specified in this Agreement, the Lender shall make such amounts available to the Borrowers by depositing such amounts in the applicable Borrower's account with the Lender. Notwithstanding anything to the contrary herein, a Borrower may from time to time deliver a Drawdown Notice to the Lender whereby the proceeds of the Loan requested thereunder will be used for one or more of the purposes permitted pursuant to Section 8.1(s) but not contemplated by the Agreed Budget (or the most recent Updated Budget), and the Lender may, in its discretion, make the requested amount under such Drawdown Notice available to the Borrowers with the approval of the Information Officer.

2.7 Lender's Obligations with Respect to Cdn. \$ Loans and U.S. \$ Loans

The Lender shall, for same day value on the Drawdown Date specified by a Borrower in a Drawdown Notice with respect to a Cdn. \$ Loan or U.S. \$ Loan, make an advance to such Borrower of the full amount of the requested Loan in the relevant currency in accordance with any payment instructions set forth in the applicable Drawdown Notice.

2.8 Irrevocability

A Drawdown Notice or Repayment Notice given by a Borrower hereunder shall be irrevocable and, subject to any options the Lender may have hereunder in regard thereto and such Borrower's rights hereunder in regard thereto, shall oblige such Borrower to take the action contemplated on the date specified therein.

2.9 Optional Repayment of the Credit Facility

A Borrower may at any time and from time to time repay, without penalty, to the Lender the whole or any part of any Loan owing by it under the Credit Facility, together with accrued interest thereon to the date of such repayment provided that:

- (a) except as otherwise provided in Section 2.10, such Borrower shall give a Repayment Notice (executed in accordance with the definition of Officer's Certificate) to the Lender not later than 9:00 a.m. (Local Time) one Banking Day prior to the date of the proposed repayment of such Loan;
- (b) repayments pursuant to this Section may only be made on a Banking Day;
- (c) each such repayment shall be in a minimum amount of the lesser of: (i) the minimum amount required pursuant to Section 2.5 for Drawdowns of the type of Loan proposed to be repaid, and any repayment in excess of such amount shall be in integral multiples of \$100,000 or U.S.\$100,000, as the case may be, and (ii) the Outstanding Principal of all Loans outstanding under the relevant Credit Facility immediately prior to such repayment;

- (d) a Borrower may not repay a portion only of an outstanding Loan unless the unpaid portion is equal to or exceeds, in the relevant currency, the minimum amount required pursuant to Section 2.5 for Drawdowns of the type of Loan proposed to be repaid; and
- (e) unless the Lender otherwise agrees in writing, each repayment of the Credit Facility by the Borrower hereunder shall automatically result in a permanent reduction of the maximum principal amount of the Credit Facility by an amount equal to the principal amount repaid.

2.10 Mandatory Repayment of Credit Facility

Subject to Section 10.2 and Article 6, the Borrowers shall: (a) repay or pay, as the case may be, to the Lender all Loans and other Secured Obligations outstanding under the Credit Facility on or before the Maturity Date, and (b) subject to any funding requirements with respect to the Carve-Out set forth in the Interim U.S. DIP Order and the Final U.S. DIP Order, if and to the extent the aggregate amount of all receipts and deposits made to the Obligors' bank accounts during the period from and including the Effective Date through the Maturity Date exceeds the amount set out in the Agreed Budget (as such amount may be modified with the mutual consent of the Lender and the Borrowers), apply, when available, all such receipts and deposits in excess thereof in the following order (the "Payment Protocol") (i) against the prefiling indebtedness of the Obligors under the Pre-Filing Secured Credit Agreement up to the amount of C\$3,900,000, and (ii) as a mandatory prepayment of the Credit Facility (and, for certainty, any amounts so prepaid may not be reborrowed).

2.11 Repayment of Borrowings In Excess of the Authorized Principal Amount

If the Outstanding Principal under the Credit Facility (determined in Cdn. Dollars with all Loans denominated in U.S. Dollars under the Credit Facility being converted to the Equivalent Amount of Cdn. Dollars) is on any day in excess of the authorized principal amount of the Credit Facility, the Borrowers shall within five (5) Banking Days thereafter repay or otherwise reduce a portion of such Loans under the Credit Facility to the extent of the amount of such excess.

2.12 Cancellation of Credit Facility and Prepayment

Provided that the Information Officer is satisfied that there are sufficient cash reserves in the Obligors' bank accounts to satisfy amounts secured by the Permitted Priority Liens and to satisfy the Carve-Out, the Borrowers may, without penalty or premium, at any time during the term of this Agreement upon five (5) Banking Days prior written notice substantially in the form annexed hereto as Schedule B, cancel the Credit Facility or any portion thereof in minimum amounts of Cdn. \$500,000 and whole multiples of Cdn. \$100,000 thereafter if, on or prior to the last day of such notice period, the Borrowers have:

- (a) prepaid or otherwise reduced the applicable Loans outstanding under the Credit Facility in an amount equal to the amount by which the Loans outstanding under the Credit Facility would otherwise be in excess of the stated amount of the Credit Facility immediately after the reduction of the Credit Facility provided for in such notice; and
- (b) paid all accrued interest and other charges and fees in respect of the applicable Loans being repaid or reduced as aforesaid.

Any such notice of cancellation is irrevocable and the amount of the Credit Facility so cancelled and reduced may not be reinstated hereunder.

2.13 Mandatory Prepayments

- (a) On the Maturity Date, the Credit Facility and the Lender's commitment thereunder shall be permanently cancelled and the Borrowers shall repay all Loans and other Credit Facility Indebtedness outstanding under the Credit Facility to the Lender in full.
- (b) Unless otherwise consented to in writing by the Lender, and provided the Information Officer is satisfied that there are sufficient cash reserves in the Obligors' bank accounts to satisfy (x) amounts secured by the Permitted Priority Liens and to satisfy the Carve-Out, and (y) amounts anticipated on the date of the mandatory repayment under the Agreed Budget in respect of which Loans were made that have not yet been incurred or paid (including, for greater certainty, anticipated legal or advisory fees in the CCAA Proceedings or U.S. Proceedings):
 - (i) upon receipt by an Obligor of a refund or payment on account of taxes from any Governmental Authority, excluding refunds or payments on account of sales taxes;
 - (ii) to the extent directed by a final order of the U.S. Court in the U.S. Proceedings (which may include the Sale Order), upon receipt by an Obligor of the proceeds of any sale, assignment, transfer, conveyance, surrender, exchange, lease, sub-lease or other disposition of property other than Permitted Dispositions (the proceeds from which Permitted Distributions shall be applied as directed by the orders of the U.S. Court and the Canadian Court); and
 - (iii) upon receipt by an Obligor of the proceeds arising from the issuance of any equity or debt securities of such Obligor,

the Borrowers shall repay, or shall cause to be repaid, Loans in accordance with the Payment Protocol in an amount equal to the amount so received in cash and cash equivalents (less any amount determined by the Information Officer to be retained as an addition to cash reserves in the Obligor's bank accounts to satisfy amounts described in clauses (x) or (y) above and any other amounts reserved on account thereof in the Agreed Budget) within one Banking Day.

Each Borrower hereby authorizes and directs the Lender to debit such Borrower's or any other Obligor's accounts with the Lender to effect any of the foregoing to the extent that the Borrowers have not made any such mandatory prepayment in accordance with this Section 2.13(b). For greater certainty, amounts prepaid pursuant to this Section 2.13(b) shall permanently reduce the stated amount of the Credit Facility and may not be reborrowed.

ARTICLE 3 CONDITIONS PRECEDENT TO DRAWDOWNS

3.1 Conditions Precedent

This Agreement shall be effective on the date (the "Effective Date") upon which each of the following conditions are satisfied or waived in writing by the Lender:

(a) Initial Order. The Canadian Court shall have issued and entered the Initial Order by the Outside Date, satisfactory to the Lender and substantially in the form annexed hereto as <u>Schedule D</u>, and the Initial Order shall not have been stayed, vacated or otherwise caused to be ineffective or amended, restated or modified in a way that adversely impacts the rights and interests of the Lender, as determined by the Lender (acting reasonably), without the consent of the Lender.

- (b) U.S. Interim DIP Order. The U.S. Court shall have issued and entered the Interim U.S. DIP Order by the Outside Date, satisfactory to the Lender and substantially in the form annexed hereto as Schedule F, and the Interim U.S. DIP Order shall not have been stayed, vacated or otherwise caused to be ineffective or amended, restated or modified in a way that adversely impacts the rights and interests of the Lender, as determined by the Lender (acting reasonably), without the consent of the Lender.
- (c) [Reserved]
- (d) Representations and Warranties. The representations and warranties contained in Section 7.1 are true and correct as of such time, and the Lender shall have received a certificate from the Borrowers certifying the same.
- (e) **No Event of Default.** As of the Effective Date, there exists no Default or Event of Default, and the Lender shall have received a certificate from the Borrowers certifying the same.
- (f) No Material Adverse Change. No Material Adverse Change shall have occurred since the date of issuance of the Initial Order and the Lender shall have received a certificate of the Borrowers certifying the same.
- Receipt of Documentation. The Lender shall have received, in form and substance satisfactory to the Lender, (a) a duly executed original of this Agreement, (b) duly executed copies of the Security required pursuant to Section 9.1, (c) a certificate of status, certificate of good standing or similar document in respect of each Obligor issued under the laws of each jurisdiction where such Obligor is organized or formed; (d) an officer's certificate provided by or on behalf of each Obligor attaching thereto its constating documents and bylaws and other governing documents, any authorizing resolutions, and an incumbency certificate of the officer's executing the Documents; (e) a true and complete copy of the Initial Order, the Interim U.S. DIP Order, and the Agreed Budget; (f) an officer's certificate of the Borrowers certifying as to the matters set forth in Sections 3.1(d), (e) and (f); (g) a certificate or certificates of insurance in respect of the Obligors; (h) any applicable "know your client" or anti-money laundering information which the Lender may require; and (i) such other documents as are required under this Agreement or which the Lender may reasonably request.
- (h) Lender Charge. There are no Security Interests ranking in priority to the Lender Charge, other than Permitted Priority Liens.
- (i) Fees. The Borrowers have paid to the Lender a work fee in Canadian Dollars in respect of the Credit Facility calculated at a rate equal to 3% of the maximum authorized principal amount of the Credit Facility on the Effective Date.
- (j) Lenders Fees and Expenses. The Borrowers have paid to the Lenders all other fees and expenses of the Lender then due and payable (including all accrued and unpaid the fees and expenses of Lender's Counsel and Lender's Counsel's Financial Advisor).

3.2 Conditions Precedent for all Drawdowns

On or before each Drawdown under the Credit Facility the following conditions shall be satisfied:

- (a) the Lender shall have received a proper and timely Drawdown Notice from the relevant Borrower requesting the Drawdown;
- (b) the representations and warranties set forth in Section 7.1 shall be true and accurate in all material respects on and as of the date of the requested Drawdown, except to the extent that such representations and warranties refer specifically to an earlier date, in which case such representations and warranties shall be true and accurate in all material respects as of such earlier date; provided that any representation or warranty that is qualified as to "in all material respects", "materiality", "Material Adverse Effect" or similar language shall be true and accurate in all respects, as so qualified, on and as of the date of the requested Drawdown or as of such earlier date, as the case may be;
- (c) no Default or Event of Default shall have occurred and be continuing on and as of the date of the requested Drawdown nor shall the Drawdown result in the occurrence of a Default or Event of Default;
- (d) after giving effect to the proposed Drawdown, the Outstanding Principal of all Loans outstanding under the Credit Facility shall not exceed the maximum principal amount of the Credit Facility; and
- (e) all fees and expenses of the Lender then due and payable (including all accrued and unpaid fees and expenses of Lender's Counsel and Lender's Counsel's Financial Advisor) for which invoices have been provided to the Borrowers, shall have been paid or will be paid from the proceeds of the requested Drawdown.

3.3 Waiver

The conditions set forth in Sections 3.1 and 3.2 are inserted for the sole benefit of the Lender and may be waived by the Lender, in whole or in part (with or without terms or conditions).

ARTICLE 4 EVIDENCE OF DRAWDOWNS

4.1 Accounts of Record

The Lender shall open and maintain books of account at the Lender's Branch evidencing all Loans and all other amounts owing by the Borrowers to the Lender hereunder. The Lender shall enter in the foregoing accounts details of all amounts from time to time owing, paid or repaid by the Borrowers hereunder. The information entered in the foregoing accounts shall, absent manifest error, constitute *prima facie* evidence of the obligations of the Borrowers to the Lender hereunder with respect to all Loans and all other amounts owing by the Borrowers to the Lender hereunder. After a request by a Borrower, the Lender shall promptly advise the Borrowers of such entries made in the Lender's books of account.

ARTICLE 5 PAYMENTS OF INTEREST AND FEES

5.1 Interest on Cdn. \$ Loans

Each Borrower shall pay interest on each Cdn. \$ Loan owing by it during each Interest Period applicable thereto in Canadian Dollars at a rate per annum equal to the Applicable Pricing Margin to the Lender's Account for Payment. Such interest shall accrue daily and shall be payable in arrears on

- 28 -

each Interest Payment Date for such Loan for the period from and including the Drawdown Date or the preceding Interest Payment Date, as the case may be, for such Loan to and including the day preceding such Interest Payment Date and shall be calculated on the principal amount of the Cdn. \$ Loan outstanding during such period and on the basis of the actual number of days elapsed in a year of 365 or 366 days, as applicable.

5.2 Interest on U.S. \$ Loans

Each Borrower shall pay interest on each U.S. \$ Loan owing by it during each Interest Period applicable thereto in United States Dollars at a rate per annum equal to the Applicable Pricing Margin to the Lender's Account for Payment. Such interest shall accrue daily and be payable in arrears on each Interest Payment Date for such Loan for the period from and including the Drawdown Date or the preceding Interest Payment Date, as the case may be, for such Loan to and including the day preceding such Interest Payment Date and shall be calculated on the principal amount of the U.S. \$ Loan outstanding during such period and on the basis of the actual number of days elapsed in a year of 365 or 366 days, as applicable.

5.3 Interest Act (Canada)

- (a) Whenever a rate of interest or other rate per annum hereunder is calculated on the basis of a year (the "deemed year") which contains fewer days than the actual number of days in the calendar year of calculation, such rate of interest or other rate shall be expressed as a yearly rate for purposes of the *Interest Act* (Canada) by multiplying such rate of interest or other rate by the actual number of days in the calendar year of calculation and dividing it by the number of days in the deemed year.
- (b) Whenever a rate of interest or other rate per annum hereunder is expressed or calculated on the basis of a year of 360 days, such rate of interest or other rate shall be expressed as a rate per annum, calculated on the basis of a 365 day year, by multiplying such rate of interest or other rate by 365 and dividing it by 360.

5.4 Nominal Rates; No Deemed Reinvestment

The principle of deemed reinvestment of interest shall not apply to any interest calculation under this Agreement; all interest payments to be made hereunder shall be paid without allowance or deduction for deemed reinvestment or otherwise, before and after maturity, default and judgment. The rates of interest specified in this Agreement are intended to be nominal rates and not effective rates. Interest calculated hereunder shall be calculated using the nominal rate method and not the effective rate method of calculation.

5.5 Standby Fees

(a) The Borrowers shall pay to the Lender a standby fee in Canadian Dollars in respect of the Credit Facility calculated at a rate per annum equal to the Applicable Pricing Margin on the amount, if any, by which the amount of the Outstanding Principal under the Credit Facility for each day in the period of determination is less than the maximum amount for each such day of the Credit Facility (taking into account the actual amount of the Credit Facility that is actually authorized to be borrowed on such date). Fees determined in accordance with this Section 5.5 shall accrue daily from and after the Effective Date and be payable by the Borrowers monthly in arrears, on cancellation in full of the Credit Facility and on the Maturity Date.

(b) As of: (i) the last Banking Day of each calendar month, (ii) the date of the cancellation in full of the Credit Facility, and (iii) the Maturity Date, the Lender shall determine the standby fees under Section 5.5(a) for the period from and including the Effective Date or the date of the immediately preceding determination, as the case may be, to but excluding that date of determination and is hereby authorized to debit the current accounts of the Cdn. Borrower for payment of the standby fees so determined for the account of the Lender.

5.6 Exit Fee

The Borrowers shall pay to the Lender an exit fee in Canadian Dollars in respect of the Credit Facility calculated at a rate equal to 1% of the maximum authorized principal amount of the Credit Facility on the Effective Date (the "Exit Fee"), such Exit Fee to be payable in full on the Maturity Date.

5.7 Default Interest

Notwithstanding any other provision hereof, upon the occurrence of and during the continuation of any Event of Default, the Borrowers shall, at the option of the Lender by written notice to the Borrowers, pay interest on the Outstanding Principal of all outstanding Loans and any other Secured Obligations payable hereunder (including, without limitation, interest on interest), if and to the fullest extent permitted by applicable law, from the date of such Event of Default and such interest shall accrue daily, be calculated and compounded monthly and be payable on demand, after as well as before maturity, default and judgment, at a rate per annum that is equal to the Applicable Pricing Margin (for certainty, after giving effect to clause (iii) of the definition thereof).

5.8 Waiver

To the extent permitted by applicable law, the covenants of the Borrowers to pay interest at the rates provided herein shall not merge in any judgment relating to any obligation of the Borrowers (or either of them) to the Lender and any provision of the *Interest Act* (Canada) or *Court Order Interest Act* (British Columbia) which restricts any rate of interest set forth herein shall be inapplicable to this Agreement and is hereby waived by the Borrowers (and each of them).

5.9 Maximum Rate Permitted by Law

No interest or fee to be paid hereunder shall be paid at a rate exceeding the maximum rate permitted by applicable law. In the event that such interest or fee exceeds such maximum rate, such interest or fees shall be reduced or refunded, as the case may be, so as to be payable at the highest rate recoverable under applicable law.

ARTICLE 6 PLACE AND APPLICATION OF PAYMENTS

6.1 Place of Payment of Principal, Interest and Fees; Payments to Lender

All payments of principal, interest, fees and other amounts to be made by a Borrower to the Lender pursuant to this Agreement shall be made to the Lender for value on the day such amount is due, and if such day is not a Banking Day on the Banking Day next following, by deposit or transfer thereof to the Lender's Account for Payment or at such other place as the Borrowers and the Lender may from time to time agree.

6.2 Funds

Each amount advanced, disbursed or paid hereunder shall be advanced, disbursed or paid, as the case may be, in such form of funds as may from time to time be customarily used in Vancouver, British Columbia, and Toronto, Ontario in the settlement of banking transactions similar to the banking transactions required to give effect to the provisions of this Agreement on the day such advance, disbursement or payment is to be made (for certainty, each such amount advanced, disbursed or paid hereunder shall be advanced, disbursed or paid, as the case may be, in immediately available funds to the extent possible in the relevant jurisdiction).

6.3 Application of Payments

Except as otherwise agreed in writing by the Lender, if any Event of Default shall occur and be continuing, all payments made by the Borrowers to the Lender shall be applied in the following order:

- (a) to amounts due hereunder as fees;
- (b) to amounts due hereunder as costs and expenses;
- (c) to amounts due hereunder as default interest;
- (d) to amounts due hereunder as interest; and
- (e) to amounts due hereunder as principal.

6.4 Payments Clear of Taxes

Any and all payments by the Obligors to the Lender hereunder shall be made free and clear of, and (a) without deduction or withholding for or on account of, any and all present or future Taxes and all liabilities with respect thereto imposed, levied, collected, withheld or assessed by any Governmental Authority or under the laws of any international tax authority imposed on the Lender, or by or on behalf of the foregoing (and, for greater certainty, nothing in this Section 6.4(a) shall make a Borrower liable for any taxes imposed on or measured by the Lender's overall net income, capital taxes or franchise taxes imposed on it (in lieu of net income taxes) by the jurisdiction (or any political subdivision thereof) under the laws of which the Lender is organized, in which its applicable lending office is maintained or in which its principal office is located or as a result of a present or former connection between the Lender and the jurisdiction imposing such tax (other than any such connection arising solely from the Lender having executed, delivered or performed its obligations or received a payment under this Agreement)). In addition, each Obligor agrees to pay any present or future stamp, transfer, registration, excise, issues, documentary or other taxes, charges or similar levies which arise from any payment made under this Agreement or the Loans or in respect of the execution, delivery or registration or the compliance with this Agreement or the other Documents contemplated hereunder other than taxes imposed on or measured by the Lender's overall net income and franchise taxes imposed on it (in lieu of net income taxes) by the jurisdiction (or any political subdivision thereof) under the laws of which the Lender is organized, in which its applicable lending office is maintained or in which its principal office is located or as a result of a present or former connection between the Lender and the jurisdiction imposing such tax (other than any such connection arising solely from the Lender having executed, delivered or performed its obligations or received a payment under this Agreement). Each Obligor shall indemnify and hold harmless the Lender for the full amount of any Taxes or other amounts paid or payable by the

Lender and any liability (including penalties, interest, additions to tax and reasonable out-of-pocket expenses) resulting therefrom or with respect thereto which arise from any payment made under or pursuant to this Agreement or the Loans or in respect of the execution, delivery or registration of, or compliance with, this Agreement or the other Documents other than taxes imposed on or measured by the Lender's overall net income, capital taxes and franchise taxes imposed on it (in lieu of net income taxes) by the jurisdiction (or any political subdivision thereof) under the laws of which the Lender is organized, in which its applicable lending office is maintained or in which its principal office is located or as a result of a present or former connection between the Lender and the jurisdiction imposing such tax (other than any such connection arising solely from the Lender having executed, delivered or performed its obligations or received a payment under this Agreement); provided, however, that notwithstanding anything to the contrary in this Section 6.4(a), the Obligors do not indemnify or hold the Lender harmless for any amount of U.S. federal withholding Taxes imposed due to the Lender's failure to comply with FATCA.

- If any Obligor shall be required by law to deduct or withhold any amount for which it is liable (b) under Section 6.4(a) from any payment or other amount required to be paid to the Lender hereunder, or if any liability therefor shall be imposed or shall arise from or in respect of any sum payable hereunder, then the sum payable to the Lender hereunder shall be increased as may be necessary so that after making all required deductions, withholdings, and additional income tax payments attributable thereto (including deductions, withholdings or income tax payable for additional sums payable under this provision) the Lender receives an amount equal to the amount it would have received had no such deductions or withholdings been made or if such additional taxes had not been imposed; in addition, such Obligor shall pay the full amount deducted or withheld for such liabilities to the relevant taxation authority or other authority in accordance with applicable law, such payment to be made (if the liability is imposed on such Obligor) for its own account or (if the liability is imposed on the Lender) on behalf of and in the name of the Lender, as the case may be. If the liability is imposed on the Lender, such Obligor shall deliver to the Lender evidence satisfactory to the Lender, acting reasonably, of the payment to the relevant taxation authority or other authority of the full amount deducted or withheld.
- (c) The Lender shall provide any documentation reasonably requested by any Obligor sufficient for such Obligor to comply with its obligations, if any, under FATCA and to determine whether the Lender has complied with any applicable reporting requirements under FATCA.

ARTICLE 7 REPRESENTATIONS AND WARRANTIES

7.1 Representations and Warranties

Each of the Obligors represents and warrants as follows to the Lender and acknowledges and confirms that the Lender is relying upon such representations and warranties:

Existence and Good Standing. Each of the Obligors is a corporation validly existing and in good standing under the laws of its jurisdiction of incorporation, or is a partnership or trust validly existing under the laws of its jurisdiction of organization or formation; each Obligor is duly registered in all other jurisdictions where the nature of its property or character of its business requires registration, except for jurisdictions where the failure to be so registered or qualified would not have a Material Adverse Effect, and has all necessary power and authority to own its properties and carry on its business as presently carried on or as contemplated by the Documents.

- (b) <u>Authority</u>. Subject to issuance of the Initial Order and the Interim U.S. DIP Order, each Obligor has full power, legal right and authority to enter into the Documents and do all such acts and things as are required by such Documents to be done, observed or performed, in accordance with the terms thereof.
- (c) <u>Valid Authorization</u>. The execution, delivery and performance by each Obligor of each of the Documents to which it is a party:
 - (i) have been duly authorized by all necessary corporate, partnership, trust and other action, as applicable;
 - (ii) are within its corporate, partnership or trust power and capacity, as applicable;
 - (iii) do not violate any provision of law or of its articles of incorporation or by laws, partnership agreement or other constating or organizational documents, as applicable;
 - (iv) do not result in the breach of or constitute a default or require any consent under, or result in the creation of any Security Interest upon any of its property or assets pursuant to, any indenture or other agreement or instrument to which it is a party or by which it or its property may be bound or affected, and
 - (v) subject to the issuance of the Interim U.S. DIP Order and the Final U.S. DIP Order, as applicable, and the Initial Order, do not require any license, consent or approval of or advance notice to or advance filing with any governmental agency or regulatory authority except those which have already been made or obtained and which are in full force and effect.
- (d) <u>Execution and Delivery</u>. Each Document to which an Obligor is a party has been duly executed and delivered by such Obligor.
- (e) Validity of Agreement; Non-Conflict. Subject to the issuance of the Interim U.S. DIP Order and the Final U.S. DIP Order, as applicable, and the Initial Order, none of the authorization, execution or delivery of this Agreement or the other Documents or performance of any obligation pursuant hereto or thereto requires or will require, pursuant to applicable law now in effect, any approval or consent of any Governmental Authority having jurisdiction (except such as has already been obtained and are in full force and effect) nor is in conflict with or contravention of (i) any Obligor's articles, by-laws or other constating documents or any resolutions of directors or shareholders or (ii) the provisions of any material agreement, the contravention of which, in the case of clause (ii) above, has had or would reasonably be expected to have a Material Adverse Effect. The Documents, when executed and delivered, will constitute valid and legally binding obligations of each Obligor which is a party thereto enforceable against each such party in accordance with their respective terms, subject to applicable bankruptcy, insolvency and other laws of general application limiting the enforceability of creditors' rights and to the fact that equitable remedies are only available in the discretion of the court.
- (f) Ownership and Location of Property. Each Obligor has good and marketable title to its material property, subject to Permitted Encumbrances which, individually and in the aggregate, do not materially affect the rights of ownership of such Obligor to such property, the value thereof or its right or ability to utilize the same in the conduct of its business and affairs. Each location where tangible material property of any Obligor is kept is set forth in Schedule 7.1(e) annexed hereto.

- (g) <u>Debt</u>. None of the Obligors, has created, incurred, assumed, suffered to exist, or entered into any contract, instrument or undertaking pursuant to which any such Obligor is now or may hereafter become liable for any Debt other than Permitted Debt.
- (h) <u>Encumbrances</u>. None of the Obligors has created, incurred, assumed, suffered to exist, or entered into any contract, instrument or undertaking pursuant to which, any Person may have or be entitled to any Security Interest on or in respect of its property and assets or any part thereof except for Permitted Encumbrances.
- (i) <u>No Material Adverse Effect</u>. No event or circumstance has occurred or is continuing which has had or would reasonably be expected to have a Material Adverse Effect.
- (j) <u>No Omissions</u>. The Obligors have made available to the Lender all material information necessary to make any representations, warranties and statements contained in this Agreement not misleading in any material respect in light of the circumstances in which they are given.

(k) <u>Non-Default</u>.

- (i) No Default or Event of Default has occurred or is continuing or is, immediately after giving effect thereto, reasonably expected to occur following any Drawdown hereunder.
- (ii) No event or circumstance has occurred and is continuing which constitutes, or which with the giving of notice, the lapse of time, a relevant determination or any combination thereof would constitute, a contravention of or default under any material agreement which could reasonably be expected to have a Material Adverse Effect.
- (1) <u>Financial Condition</u>. The audited and unaudited combined consolidated and unconsolidated financial statements of the Borrowers delivered to the Lender pursuant hereto present fairly, in all material respects, the combined consolidated or unconsolidated financial condition of such Persons or Person as at the date thereof and the results of the combined consolidated or unconsolidated operations thereof for the fiscal year or fiscal quarter (as applicable) then ending, all in accordance with GAAP consistently applied.
- (m) <u>Information Provided</u>. All information, materials and documents, including all cash flow projections, economic models, capital and operating budgets and other information and data:
 - (i) prepared and provided to the Lender by the Obligors in respect of the transactions contemplated by this Agreement, or as required by the terms of this Agreement, were, in the case of financial projections, prepared in good faith based upon reasonable assumptions at the date of preparation, and, in all other cases, true, complete and correct in all material respects as of the respective dates thereof; and
 - (ii) to the extent prepared by Persons other than the Obligors and provided to the Lender by or on behalf of the Obligors in respect of the transactions contemplated by this Agreement, or as required by the terms of this Agreement, were, to the best of the knowledge of the Obligors, after due inquiry, in the case of financial projections, prepared in good faith based upon reasonable assumptions at the date of preparation, and, in all other cases, true, complete and correct in all material respects as of the respective dates thereof.
- (n) <u>Absence of Undisclosed Liabilities</u>. None of the Obligors has any material liabilities, losses or obligations of any nature (whether absolute, known or unknown, accrued, fixed, contingent,

liquidated, unliquidated, due or to become due, or otherwise), except for (i) liabilities included or reflected in the financial statements delivered to the Lender and adequately reserved against therein in accordance with GAAP consistently applied, (ii) liabilities or performance obligations arising subsequent to the date of the balance sheets delivered to the Lender in the ordinary course of business (and not as a result of a breach or default by any Obligor) out of or under agreements, contracts, leases, arrangements or commitments to which such Obligor is a party or (iii) otherwise constituting Permitted Debt.

- (o) Absence of Litigation. Except as set out in Schedule 3.6 of the Seller Disclosure Schedule and the general pendency of the CCAA Proceedings and the U.S. Proceedings, and actions arising in the ordinary course of the Business and that would not have, in the aggregate, a Material Adverse Effect, there are no actions, suits or proceedings pending or, to the knowledge of the Obligors, threatened against or affecting any of the Obligors or their respective property, undertakings and assets, at law, in equity or before any arbitrator or before or by any Governmental Authority having jurisdiction in the premises in respect of which there is a reasonable possibility of a determination adverse to such Person and which, if determined adversely, would have or would reasonably be expected to have a Material Adverse Effect.
- (p) Compliance with Applicable Laws, Court Orders and Material Agreements. Except as set out in Schedule 3.8(b) of the Seller Disclosure Schedule or Schedule 7.1(p) annexed hereto, and except as would not, individually or in the aggregate, reasonably be expected to have a Material Adverse Effect each of the Obligors and their respective property, businesses and operations are in compliance with all Applicable Laws (including, without limitation, all applicable Environmental Laws), all applicable directives, judgments, decrees, injunctions and orders rendered by any Governmental Authority or court of competent jurisdiction, its articles, by-laws, other constating documents, partnership agreements and other organizational documents, all material agreements, as applicable, and any employee benefit plans, except to the extent that any such non-compliance would not reasonably be expected to have a Material Adverse Effect or is otherwise stayed, including in accordance with section 362 of the Bankruptcy Code, or as contemplated by the Initial Order.

(q) Governmental Approvals; Required Permits in Effect.

- (i) subject to issuance of the Initial Order, no action, consent or approval of, registration or filing with or any other action by any Governmental Authority is or will be required in connection with the transactions contemplated by this Agreement or any of the Documents, in each case, except for (i) the filing of PPSA financing statements and similar security or collateral filings and registrations under applicable laws in other jurisdictions, and (ii) such actions, consents, approvals, registrations and filings as have been made or obtained and are in full force and effect; and
- (ii) all Required Permits are in full force and effect, except to the extent that the failure to have or maintain the same in full force and effect would not, when taken in the aggregate, have or reasonably be expected to have a Material Adverse Effect.
- (r) Remittances Up to Date. All of the material remittances required to be made by the Obligors to Governmental Authorities which are due and payable have been made, are currently up to date and there are no outstanding arrears, other than those which are being contested by Permitted Contest.
- (s) <u>Agreed Budget and Updated Budgets</u>. The Agreed Budget and each Updated Budget (when the same is delivered) are reasonable and have been prepared in good faith.

- (t) <u>Use of Borrowings</u>. The Borrowers' use of the proceeds of Loans obtained by it (and the request for Advances therefor) complies with the requirements of Section 2.3.
- (u) <u>Payroll Obligations</u>. No Obligor has defaulted in respect of its obligations for payroll and source deductions or is in arrears in respect of the payment of any such obligations.

(v) Environmental.

- (i) to the best of the knowledge and belief of the Obligors, after due inquiry, the Obligors and their properties, assets and undertakings taken as a whole comply in all material respects and the businesses, activities and operations of same and the use of such properties, assets and undertakings and the processes and undertakings performed thereon comply in all material respects with all Environmental Laws; further, no Obligor knows, or has reasonable grounds to know, of any facts which result in or constitute or are likely to give rise to material non-compliance with any Environmental Laws.
- (ii) no Obligor has received written notice and, except as previously disclosed to the Lender in writing, has no knowledge, after due inquiry, of any facts which could give rise to any notice of material non-compliance with any Environmental Laws and has not received any notice that any Obligor is a potentially responsible party for a federal, provincial, regional, municipal or local clean up or corrective action in connection with its properties, assets and undertakings.
- (w) Taxes. Except as disclosed in Schedule 3.16 of the Seller Disclosure Schedule, each Obligor has duly filed on a timely basis all tax returns required to be filed by it and have paid all material Taxes which are due and payable, and has paid all material assessments and reassessments, and all other material Taxes, governmental charges, governmental royalties, penalties, interest and fines claimed against it, other than those which (i) are being contested by it by Permitted Contest or (ii) those arising from a voluntary disclosure which have an aggregate liability of \$100,000 or less; it has made adequate provision for, and all required instalment payments have been made in respect of, Taxes payable for the current period for which returns are not yet required to be filed; there are no agreements, waivers or other arrangements providing for an extension of time with respect to the filing of any tax return by it or the payment of any Taxes; there are no actions or proceedings being taken by any taxation authority in any jurisdictions where any Obligor carries on business to enforce the payment of any Taxes by it other than those which are being contested by it by Permitted Contest.
- (x) <u>Structure</u>. Holdco is the sole shareholder of each of the Cdn. Borrower and the U.S. Borrower, and no Person (other than an Obligor) has any right or option to purchase any of the issued and outstanding Equity Interests of any Obligor.
- (y) <u>Subsidiaries</u>. The only Subsidiaries of each of Holdco, the Cdn. Borrower and the U.S. Borrower are as disclosed in <u>Schedule 7.1(y)</u> annexed hereto, as supplemented from time to time by the Borrowers within 10 Banking Days of the acquisition, creation or existence, or disposition, of each Subsidiary created, acquired or disposed of after the Effective Date. Each supplemented <u>Schedule 7.1(y)</u> sets forth a list of all of the Subsidiaries of each of Holdco, the Cdn. Borrower and the U.S. Borrower and identifies (A) the jurisdiction of incorporation, existence, organization or formation of each such Subsidiary and (B) each jurisdiction in which each such Person has assets or carries on a material part of its business. As at the Effective Date, the only Subsidiaries of Holdco are the Cdn. Borrower, the U.S. Borrower and the U.S. Guarantor.

- (z) <u>Insurance</u>. Each Obligor is insured with financially sound and reputable insurance companies that are not Affiliates of the Borrowers, in such minimum amounts, with such maximum deductibles and covering such risks as are consistent with the insurance carried by companies engaged in similar businesses and owning similar properties in localities where such Obligor operates.
- (aa) Intellectual Property. Each Obligor has or has the legal right to use all Intellectual Property necessary for the operation and conduct of its business, affairs, operations and processes and, to the best of the Obligors' knowledge and belief, no Person has asserted any claim or taken any step or proceedings to prohibit or limit the use of such Intellectual Property by the Obligors or any of them.
- (bb) Material Agreements. Except for the contracts listed in Schedule 3.8 of the Seller Disclosure Schedule, as supplemented from time to time by the Obligors within 10 Banking Days of executing a new material agreement after the Effective Date, the Obligors have no material agreements. Save and except for any material breach, default or event of default or similar event or condition (howsoever described) brought about by the Obligors commencing the CCAA Proceeding or the US Proceeding, each material agreement of the Obligors is in full force and effect and no material breach, default or event of default or similar event or condition (howsoever described) by any Obligor is subsisting thereunder, except as set forth in Schedule 7.1(bb) annexed hereto.
- (cc) <u>Transactions with Related Parties</u>. Except for the transactions referenced in <u>Schedule 7.1(cc)</u> annexed hereto, no Obligor is engaged in any transaction with any Related Party (other than another Obligor) on terms less favourable to such Obligor than those that would be obtainable by it in a comparable arm's-length transaction with a Person that is not a Related Party.
- (dd) <u>ERISA</u>. Except as disclosed in writing to the Lender, neither Holdco, nor any ERISA Affiliate maintains, sponsors, participates in or contributes to any Pension Plan or any Multiemployer Plan. No ERISA Event has occurred or is reasonably expected to occur that, when taken together with all other such ERISA Events for which liability is reasonably expected to occur, could reasonably be expected to result in a Material Adverse Effect, and each Pension Plan has complied in all material respects with the applicable provisions of ERISA and the Code. Neither Holdco, nor any Borrower sponsors a "welfare benefit plan", as defined in Section 4.1 of ERISA, that is subject to ERISA and which provides post-retirement medical benefits, other than liability for continuation coverage described in Part 6 of Subtitle B of Title I of ERISA.
- (ee) Sanctions and AML Laws and Regulations. No Obligor is (i) in violation of any Sanctions Laws and Regulations, (ii) in violation of any Anti-Money Laundering Laws and Regulations, (iii) a Designated Person, (iv) subject to or in violation of any sanctions, prohibitions or requirements imposed by any Executive Order or by any sanctions program administered by OFAC, (v) acting as agent for any person or government covered by any of the foregoing or (vi) subject to any action or investigation in respect of any of the foregoing.

7.2 Deemed Repetition

On the date of delivery by a Borrower of a Drawdown Notice to the Lender, and again on the date of any Drawdown made by a Borrower pursuant thereto:

(a) each of the representations and warranties contained in Section 7.1 shall be deemed to be repeated;

(b) the Obligors shall be deemed to have represented to the Lender that no event has occurred and remains outstanding which would constitute a Default or an Event of Default nor will any such event occur as a result of the aforementioned Drawdown.

7.3 Other Documents

All representations, warranties, certifications and statements of any Obligor contained in any other Document delivered pursuant hereto or thereto shall be deemed to constitute representations and warranties made by the Obligors to the Lender under Section 7.1 of this Agreement.

7.4 Effective Time of Repetition

All representations and warranties, when repeated or deemed to be repeated hereunder, shall be construed with reference to the facts and circumstances existing at the time of repetition, unless they are stated herein to be made as at the date hereof or as at another date.

7.5 Nature of Representations and Warranties

The representations and warranties set out in this Agreement or deemed to be made pursuant hereto shall survive the execution and delivery of this Agreement and the making of each Drawdown, notwithstanding any investigations or examinations which may be made by the Lender or Lender's Counsel. Such representations and warranties shall survive until this Agreement has been terminated, provided that the representations and warranties relating to environmental matters shall survive the termination of this Agreement.

ARTICLE 8 GENERAL COVENANTS

8.1 Affirmative Covenants of the Obligors

So long as any Secured Obligation is outstanding or the Credit Facility is available hereunder, each Obligor covenants and agrees with the Lender that, unless the Lender otherwise consents in writing:

- (a) Punctual Payment and Performance. Each Borrower shall duly and punctually pay the principal of all Loans, all interest thereon and all fees and other amounts required to be paid by such Borrower hereunder in the manner specified hereunder and each Obligor shall maintain, perform and observe all of its respective obligations under this Agreement and under any other Document to which it is a party.
- (b) <u>Books and Records</u>. It shall keep proper books of record and account in which complete and correct entries will be made of its transactions in accordance with GAAP.
- (c) <u>Maintenance and Operation</u>. It shall do or cause to be done all things necessary or required to have all its properties, assets and operations owned, operated and maintained, in all material respects, in accordance with diligent and prudent industry practice and Applicable Laws, and at all times cause the same to be owned, operated, maintained and used in compliance with all terms of any applicable insurance policy to the extent necessary to ensure that coverage under any such policy cannot be denied by the insurers thereunder.

- (d) Maintain Existence; Compliance with Legislation Generally; Required Permits. Except as otherwise expressly permitted by Section 8.2(d) or (n), each Obligor shall preserve and maintain its existence as a corporation, partnership or trust under the laws of its jurisdiction of incorporation, organization or formation. Each Obligor shall do or cause to be done all acts necessary or desirable to comply with all Applicable Laws, except where such failure to comply does not and would not reasonably be expected to have a Material Adverse Effect, and to preserve and keep in full force and effect all Required Permits and all other franchises, licences, rights, privileges, permits and Governmental Authorizations necessary to enable such Obligor to operate and conduct its business and own its properties and assets (including, without limitation, its material owned and leased real property) in all material respects, in accordance with Applicable Laws and prudent industry practice.
- (e) <u>Budgets, Financial Statements and Other Information</u>. The Obligors shall deliver to the Lender:
 - (i) Working Capital Reports when requested by the Lender, as soon as available and in any event within three (3) Banking Days of the Lender's request, a written report setting forth the Obligors' cash balances, accounts receivable and accounts payable, in each case, as at the end of the period covered by the Lender's request, such report and statement to include all supporting ledgers, analysis and other information for such period, each such report to be in a form and with such level of detail as shall be satisfactory to the Lender, acting reasonably;
 - (ii) Weekly Cashflow Reports and Projections commencing April 24, 2023 and on each Wednesday thereafter, by no later than 2:00 pm (Vancouver time) on each such Wednesday:
 - (A) a written report (the "Weekly Cashflow Report"), in form (including reasonable supporting information, to the extent requested by the Lender) satisfactory to the Lender in its sole discretion, summarizing the Obligors' actual cash receipts and cash expenditures and outflows, on a line by line basis, for (i) the period ending on Friday of the prior week, and (ii) the cumulative period commencing on April 24, 2023 and ending on Friday of the prior week, including details with respect to any variances, as compared to the Agreed Budget for the corresponding periods; and
 - (B) an Updated Budget together with a written summary of any changes to the Agreed Budget or the preceding Updated Budget, as applicable (for certainty, the Updated Budget shall not constitute an amendment of the Agreed Budget unless consented to by the Lender in its sole and absolute discretion (exercised reasonably)); and
 - (iii) Other such other information, reports, certificates, projections of income and cash flow or other matters affecting the business, affairs, financial condition, property or assets of the Obligors as the Lender may reasonably request.
- (f) Rights of Inspection. At any reasonable time and from time to time upon reasonable prior notice, each Obligor shall permit the Lender or any representative thereof (at the expense of the Borrowers) to (i) examine and make copies of and abstracts from the records and books of account of such Obligor, (ii) visit and inspect the premises and properties of such Obligor (in each case at the risk of the Borrowers, except for the gross negligence or willful misconduct of the inspecting party or the failure of any such inspecting party) and provided that the Lender (and its officers and employees) conducts such visit and inspection in accordance with Applicable Laws, prudent industry practice and the Borrowers' reasonable site safety requirements, and (iii) discuss the

affairs, operations, finances and accounts of the Obligors with any of the officers or directors of the Obligors.

- (g) Notice of Material Litigation. The Borrowers shall promptly give written notice to the Lender of any litigation, proceeding or dispute affecting any Obligor in respect of a demand or claim in respect of which there is a reasonable possibility of an adverse determination and which (i) involves potential liability on the part of any Obligor in excess of \$100,000, or (ii) if adversely determined would otherwise reasonably be expected to have a Material Adverse Effect, and shall, in each case, from time to time furnish to the Lender all reasonable information requested by the Lender concerning the status of any such litigation, proceeding or dispute.
- (h) Notice of Default or Event of Default. The Borrowers shall deliver to the Lender, as soon as reasonably practicable and in any event no later than 3 Banking Days after becoming aware of a Default or an Event of Default, an Officer's Certificate describing in detail such Default or such Event of Default and specifying the steps, if any, being taken to cure or remedy the same.
- (i) Notice of Management Letters; Material Adverse Effect. The Borrowers shall, as soon as reasonably practicable, promptly notify the Lender of (i) receipt of any accountant's or auditor's letters relating to the preparation of financial statements of any Obligor, or (ii) any event, circumstance or condition that has had or is reasonably likely to have a Material Adverse Effect.
- (j) <u>Notice of Change of Ownership</u>. The Borrowers shall promptly give written notice to the Lender of any change in ownership of the equity capital of any Obligor after the date hereof.
- (k) Notice of ERISA Events. The Borrowers shall, as soon as reasonably practicable, give written notice to the Lender of the occurrence of any ERISA Event that, alone or together with any other ERISA Events that have occurred, could reasonably be expected to result in liability of the Borrowers and Subsidiaries in an aggregate amount exceeding \$500,000 (or the Equivalent Amount thereof).
- (1) Payment of Taxes, Withholdings, etc. Each Obligor shall, from time to time, pay or cause to be paid all material Taxes, rents, rates, levies or assessments, ordinary or extraordinary, governmental fees or dues, and to make and remit all withholdings, lawfully levied, assessed or imposed upon such Obligor or any of its assets as and when the same become due and payable, except when and so long as the validity of any such material Taxes, rents, rates, levies, assessments, fees, dues or withholdings is being contested by such Obligor by a Permitted Contest.
- (m) Payment of Preferred Claims. Each Obligor shall, from time to time, pay when due or cause to be paid when due, in accordance with the Agreed Budget, all material amounts related to wages, workers' compensation obligations, government royalties or pension fund obligations and any other amount which may result in a lien, charge, Security Interest or similar encumbrance against the assets of such Obligor arising under statute or regulation, except when and so long as the validity of any such material amounts or other obligations is being contested by such Obligor by a Permitted Contest or when such material amounts or other obligations become due on termination of an employee and are of a kind that may become subject to a deemed Security Interest pursuant to section 81.3 or 81.4 of the of BIA.
- (n) <u>Environmental Covenants</u>. Without limiting the generality of Section 8.1(d) above, each Obligor shall, and shall use commercially reasonable efforts to cause any other party acting under its direction to, conduct its business and operations and maintain their properties and assets so as to comply at all times with all Environmental Laws in all material respects. If any Obligor shall:

- (i) receive or give any notice that a material violation of any Environmental Law has or may have been committed or is about to be committed by the same;
- (ii) receive any notice that a complaint, proceeding or order has been filed or is about to be filed against the same alleging a material violation of any Environmental Law; or
- (iii) receive any notice requiring any Obligor to take any action in connection with the Release of Hazardous Materials into the environment or alleging that any Obligor may be liable or responsible for costs associated with a response to or to clean up a Release of Hazardous Materials into the environment or any damages caused thereby, the Borrowers shall promptly provide the Lender with a copy of such notice and shall furnish to the Lender from time to time all reasonable information requested by the Lender relating to the same.
- (o) <u>Milestones</u>. The Borrower shall, and shall cause each other Obligor to, do all such further acts and things reasonably practicable to achieve all the Milestones, including complying with the reasonable best efforts undertakings set out in any Milestone.
- (p) <u>Compliance with Agreed Budget</u>. The Borrowers shall and shall cause each other Obligor to manage and operate their respective properties and assets and conduct their respective businesses and activities, in each case, in a manner which is consistent with the Agreed Budget.
- Qonference Calls. The Borrowers shall, and shall use commercially reasonable efforts, if requested by the Lender, to cause (i) its non-legal advisors to, participate on weekly conference calls with the Lender, and its non-legal advisors, to discuss the Agreed Budget and any Updated Budgets, the Borrowers' current and projected operational performance, and any related financial matters; and (ii) its sale advisor, Miller Buckfire & Co., LLC, to provide regular updates to the Lender and its advisors with respect to the sales process being undertaken by such advisors on behalf of the Obligors, including updates in a timely manner with respect to prospective bidders identified by the Borrowers or such advisors, communications with such prospective bidders and the identity of any bidders and details of bids received; further, the Borrowers shall provide and shall cause their financial advisors to provide the Lender and its advisors with draft copies of all materials prepared for distribution to potential bidders prior to such distribution, including any invitation or "teaser" letter, confidential information memorandum or management presentation.
- Notices, Filings and other Information. The Borrowers shall, on a timely basis, furnish to the (r) Lender: (i) as soon as practically possible upon the entering of the same, a true and complete copy of the Final U.S. DIP Order and the Initial Order; (ii) notice of all material developments with respect to the business and affairs of the Obligors, including (without limitation) the development of a Plan or an Alternative Restructuring Option; (iii) within a reasonable period of time prior to filing with the Canadian Court or U.S. Court, which, in any event, shall be: (A) in respect of any pleading related or connected to this Agreement, including in respect of any Plan or Approved Sale, at least 3 Banking Days (or such shorter time agreed by the Lender, which consent may be provided via email) prior to any such filing or proposed filing by, or on behalf of, an Obligor, (B) in the event of any other filing or proposed filing by, or on behalf of, an Obligor, as soon as reasonably practicable in advance of filing, copies of all material pleadings, material motions, material applications, material proposed orders or financing information and other material documents, in each case, which could reasonably be expected to affect the Lender or this Agreement, in each case, proposed to be filed by, or on behalf of any Obligor; (iv) regular (which, in any event shall be at least once weekly) update calls regarding the status of the CCAA Proceedings and U.S. Proceedings including, without limitation, reports on the progress of any Plan, Alternative Restructuring Option or Approved Sale and any information which may otherwise be confidential; and (v) copies of any

financial reporting provided to the Information Officer and any reports or commentary received from the Information Officer regarding the financial position of the Obligors. In addition to the foregoing and without in any way derogating therefrom, the Borrowers shall provide to the Lender copies of all such other information relating to the business, affairs, operations and financial condition of any Obligor as the Lender may reasonably request.

- Use of Credit Facility. Unless otherwise agreed to in writing by the Lender, the Borrowers shall (s) use the Credit Facility and the proceeds thereof solely: (i) (A) to finance operating expenses and restructuring costs in the CCAA Proceedings and U.S. Proceedings, (B) to pay (1) professional fees and (2) other items in accordance with the Agreed Budget and (C) for general corporate purposes of the Borrowers and the other Obligors, all in accordance with the Agreed Budget (including the funding of the Carve-Out); and (ii) to pay fees and expenses related to the Credit Facility, the CCAA Proceedings and the U.S. Proceedings in each case, as contemplated in the Initial Order or the Interim U.S. DIP Order or Final U.S. DIP Order, as applicable, provided that, no proceeds of any Loan will be used: (x) for any purpose which violates, or would be inconsistent with, Regulation T, Regulation U or Regulation X each as issued by the U.S. Federal Reserve, (y) by the Obligors to investigate, object to or challenge in any way any claims of the Lender against any of the Obligors in respect of the Credit Facility or of the Pre-Filing Secured Creditor under the Pre-Filing Secured Credit Agreement; or (z) to investigate, object to or challenge in any way the validity, perfection or enforceability of the Security Interests created pursuant to the Lender Charge or any Security Interests granted pursuant to the U.S. Proceedings.
- (t) <u>Compliance with CCAA Proceedings and U.S. Proceedings</u>. The Borrowers shall and shall cause each other Obligor to comply with the provisions of each Restructuring Court Order.
- (u) <u>Notice of Negative Cash Flow Variance</u>. Without limiting any other obligation of the Borrowers under the Documents, the Borrowers shall forthwith advise the Lender in writing if, as at the end of any period ending on Friday of the prior week, a Negative Cash Flow Variance exists.
- (v) Required Insurance. Each Obligor shall maintain, with financially sound and reputable insurers, insurance with respect to its properties and business and against such casualties and contingencies and in such types and such amounts as shall be required by the Lender (acting reasonably) and, in any event, as shall be in accordance with prudent business practices for Persons of the size and type of business and operations as such Obligor and in the case of the Cdn. Borrower or the U.S. Borrower, shall include business interruption insurance and third-party liability insurance. Each Obligor shall cause the Lender to be named loss payee or additional insured on each of its insurance policies required hereunder. In addition, the Borrowers shall provide to the Lender, promptly upon the Lender's request, (i) evidence of compliance with this Section 8.1(v), and (ii) a certified copy of each applicable policy of insurance.
- (w) <u>Intellectual Property</u>. Each Obligor shall own, or have the legal right to use, all Intellectual Property necessary for the operation and conduct of its business, affairs, operations and processes.
- (x) <u>Cash Management; Accounts</u>. To the extent authorized by the U.S. Court in the U.S. Proceedings, each of the Obligors shall cause all of its deposit accounts (other than the escrow account established in connection with the funding of the Carve-Out), securities accounts, and treasury accounts to be maintained with a branch of the Lender.

8.2 Negative Covenants of the Obligors

So long as any Secured Obligation is outstanding or any Credit Facility is available hereunder, each Obligor covenants and agrees with the Lender that, unless the Lender otherwise consents in writing:

- (a) Change of Business/Holdco to remain a Passive Holding Company.
 - (i) No Obligor shall change in any material respect the nature of its business or operations from the types of business and operations carried on by such Obligor on the date hereof.
 - (ii) Holdco shall not engage in any business or activities other than (A) its ownership of the Equity Interests of the Cdn. Borrower, (B) management of the Cdn. Borrower, (C) the issuance of Equity Interests and the investment of any proceeds therefrom in the Cdn. Borrower, and (D) entering into contracts and incurring other obligations in the ordinary course of business; provided that at no time shall Holdco own any property or assets other than the Equity Interests of the Cdn. Borrower.
- (b) <u>Negative Pledge</u>. No Obligor shall create, issue, incur, assume or permit to exist any Security Interests on any of its property, undertakings or assets other than Permitted Encumbrances.
- (c) <u>No Restrictions on Pledge</u>. No Obligor shall, directly or indirectly (except pursuant to the Documents):
 - (i) create or otherwise cause or suffer to exist or become effective any consensual restriction or encumbrance of any kind on the ability of such Obligor to make Distributions on any of its Equity Interests or to pay fees, including management fees, or make other payments to any other Obligor; or
 - (ii) enter into, assume or become subject to any agreement prohibiting or otherwise restricting the existence of any Security Interest upon any of its property, assets, or undertaking in favor of the Lender, whether now owned or hereafter acquired, except pursuant to agreements in connection with Debt of the type permitted under clause (g) of the definition of Permitted Debt, so long as such prohibition or restriction is limited to the assets financed with such Debt.
- (d) Restriction on Amalgamation, etc. No Obligor shall enter into or become party to any transaction (each a "Reorganization Transaction") or merger, amalgamation, consolidation, winding-up, plan of arrangement, reorganization or restructuring with any Person or enter into any transaction by way of transfer, liquidation, dissolution, sale, lease, disposition or otherwise whereby all or substantially all of its undertaking, property or assets would become the property of any other Person, or take any corporate, limited liability company, partnership or trust action in pursuance of any of the foregoing; provided that any Obligor may undertake a Reorganization Transaction if such Reorganization Transaction is undertaken pursuant to an (i) Approved Sale or (ii) an Alternative Restructuring Option that is either consented to by the Lender or that would result in the repayment in full in cash of all Indebtedness owing under each of this Agreement and the Pre-Filing Secured Credit Agreement upon the closing of such Reorganization Transaction.
- (e) <u>Restriction on Dispositions</u>. Except for Permitted Dispositions, no Obligor shall sell, transfer, exchange, lease, release or otherwise dispose of property or assets, whether in one or a series of transactions. Holdco shall not sell, transfer, exchange, release or otherwise dispose of its Equity

- Interests of the Cdn. Borrower. The Cdn. Borrower shall not sell, transfer, exchange, release or otherwise dispose of its Equity Interests of the U.S. Borrower.
- (f) <u>Limit on Sale-Leasebacks</u>. No Obligor shall enter into any Sale-Leaseback, synthetic lease or similar transaction involving any of its assets.
- (g) <u>Limit on Debt</u>. No Obligor shall have or incur any Debt other than Permitted Debt.
- (h) <u>Limit on Investment</u>. No Obligor shall make Investments other than:
 - (i) Investments consisting of Financial Assistance permitted under Section 8.2(j) below;
 - (ii) Investments in other Obligors contemplated by the Agreed Budget;
 - (iii) Investments contemplated by the Agreed Budget.
- (i) <u>Limits on Distributions and Payments on account of Subordinated Debt</u>. No Obligor shall make any Distribution, payment of management fees, or payment of principal or interest on account of subordinated debt other than (i) a Distribution that is permitted by a Restructuring Court Order, provided that any such Distribution does not cause a Default or Event of Default, (ii) Distributions that are provided for in the Agreed Budget, and (iii) Distributions to another Obligor in accordance with the Agreed Budget.
- (j) <u>Limit on Financial Assistance</u>. No Obligor shall provide any Financial Assistance to or in favour of any Person except for Permitted Financial Assistance.
- (k) <u>No Financial Instruments</u>. No Obligor shall enter into or transact any Financial Instruments after the Effective Date.
- (1) <u>Cash Management; Accounts</u>. No Obligor shall open or maintain any deposit account or treasury account or enter into any cash management agreement or arrangement with any financial institution other than the Lender, except to the extent required by a final order of the U.S. Court entered in the U.S. Proceedings.
- (m) Non-Arm's Length Transactions. No Obligor shall enter into any contract, agreement or transaction whatsoever, including for the sale, purchase, lease or other dealing in any property or the provision of any services, with any Related Party (other than another Obligor) except (i) contracts, agreements, or transactions upon fair and reasonable terms, which terms are not less favourable to such Obligor than it would obtain in an arm's length transaction and, if applicable, for consideration which equals the fair market value of such property or other than at a fair market rental as regards leased property.
- (n) No Amalgamation, Merger, etc. Other than the Stalking Horse Bid or another APA entered into in accordance with the Bidding Procedures (as defined in the Stalking Horse Bid), Obligor shall enter into, or agree to enter into, any transaction whereby all or substantially all of its undertaking, property and assets would become the property of any other Person whether by way of reconstruction, reorganization, recapitalization, consolidation, amalgamation, merger, transfer, sale or otherwise (and whether such transaction is consensual or hostile) except, in the case of any Obligor other than a Borrower, where the successor thereto or transferee thereof is a Borrower or a Guarantor.

- (o) <u>Fiscal Periods</u>. No Obligor shall cause any Quarter End to end on a date other than as set out in the definition of Quarter End without the prior written consent of the Lender.
- (p) Sanctions and AML Laws and Regulations.
 - (i) No Obligor shall, directly or indirectly, use the proceeds of the Loans, or lend, contribute or otherwise make available such proceeds to any Subsidiary, joint venture partner or other person or entity (A) to fund any activities or business of or with any Designated Person, or in any country or territory, that at the time of such funding is the subject of any sanctions under any Sanctions Laws and Regulations or (B) in any other manner that would result in a violation of any Sanctions Laws and Regulations by any party to this Agreement.
 - (ii) Each Obligor shall use commercially reasonable efforts to ensure that none of the funds or assets used to pay any amount due pursuant to this Agreement constitute funds derived from any unlawful activity, including without limitation (A) activities in violation of Anti-Money Laundering Laws and Regulations or (B) transactions with or relating to Designated Persons or countries which are the subject of sanctions under any Sanctions Laws and Regulations.
- (q) <u>Distribution of Funds Only to Obligors</u>. Except as set out in the Agreed Budget, no Obligor shall permit to occur, any payment or distribution of funds (whether from proceeds of any Loan under the Credit Facility or otherwise) from an Obligor to any Affiliate that is not an Obligor.
- (r) <u>Payments Consistent with Agreed Budget</u>. The Obligors shall not make any payment or distribution which is not consistent with the Agreed Budget.
- (s) Approved Sale Restriction. Without the prior written consent of the Lender, the Obligors shall not seek any authorization, approval or other order of the Canadian Court or the U.S. Court which will result in, or would reasonably be expected to result in, the authorization, approval or other order of such Canadian Court or U.S. Court of a sale of its assets, business and undertaking other than pursuant to an Approved Sale; *provided* that nothing in this Section 8.2(s) shall, or shall be deemed to, prohibit or restrict an Obligor's fulfillment of any fiduciary duty in accordance with Applicable Law, including pursuant to section 1107 of the Bankruptcy Code.
- (t) <u>Payments under Employment Benefit Plans</u>. The Obligors shall not make any payment or distribution in respect of post-employment benefit payments (excluding the KERP and KEIP).
- (u) <u>New Subsidiaries</u>. None of the Obligors shall acquire, create or cause to exist any Subsidiary which does not already exist on and as of the date hereof.

8.3 Lender May Perform Covenants

If the Obligors or any of them fail to perform any covenants on their part herein contained, subject to any consents or notice or cure periods required by Section 10.1, the Lender may give notice to the Borrowers of such failure and if such covenant remains unperformed, the Lender may, in its discretion but need not, perform any such covenant capable of being performed by the Lender and if the covenant requires the payment or expenditure of money, the Lender may make such payments or expenditure and all sums so expended shall be forthwith payable by the Borrowers to the Lender and shall bear interest at the applicable interest rate provided in Section 5.6. No such performance, payment or expenditure by the Lender shall be deemed to relieve the Obligors or any of them of any default hereunder or under the other Documents.

ARTICLE 9 SECURITY

9.1 Security

Subject to the Carve-Out, to secure the payment and performance of all amounts from time to time owing by the Obligors to the Lender (including without limitation, all Loans, interest thereon and all fees, legal and other costs, charges and expenses) and all obligations owing by the Obligors to the Lender, in each case, under or pursuant to the Documents and which may arise pursuant to the CCAA Proceedings and the U.S. Proceedings, including any amounts owing under the MasterCard Facilities, as defined in the Pre-Filing Secured Credit Agreement (collectively, the "Secured Obligations"), the Borrowers shall, or shall cause:

- (a) the Lender Charge to be granted to and in favour of the Lender;
- (b) each Guarantor to execute and deliver to the Lender an Obligor Guarantee; and
- (c) each Obligor to execute and deliver to the Lender such other security documents, agreements and instruments as the Lender may, acting reasonably, require,

(collectively, the "Security"). In addition to and without in any way derogating from the foregoing, the Lender's Account for Payments shall be subject to a priority Security Interest in favour of the Lender, subordinate only to the Permitted Priority Liens set forth in paragraphs (a) and (c) of the definition thereof and the Carve-Out.

9.2 Forms

The Security will be in such form or forms as will be required by the Lender, acting reasonably. Should the Lender determine at any time and from time to time that the form and nature of the then existing Security is deficient in any way or does not fully provide the Lender with the Guarantees and Security Interests and priority to which each is entitled hereunder, the Obligors will forthwith execute and deliver to the Lender, at the Borrowers' expense, such amendments to the Security or provide such new Guarantees or security as the Lender may reasonably request, in a form satisfactory to the Lender, acting reasonably.

9.3 No Merger

The taking of any Security as provided under this Agreement shall not operate by way of merger of any of the obligations of any Obligor under this Agreement or any other agreement evidencing Secured Obligations, or of any Security Interest, Guarantee, contract, promissory note, bill of exchange or security in any other form, whether or not similar to the foregoing, and no judgment recovered by the Lender shall operate by way of merger or in any way affect the obligations provided for in this Agreement or any other agreement evidencing Secured Obligations. For greater certainty, no judgment recovered by the Lender shall operate by way of merger or in any way affect the obligation of the Borrowers to pay interest at the rates, times and manner as provided in this Agreement or any other agreement evidencing Secured Obligations.

9.4 Permitted Encumbrances and Permitted Debt

None of:

- (a) the fact that any person is permitted to create or suffer to exist any Permitted Encumbrance or Permitted Debt;
- (b) the fact that any representation, warranty or covenant herein may make an exception for the existence of Permitted Encumbrances or Permitted Debt; or
- (c) the fact that the Security Interests created pursuant to the Security are stated to be subject to, or are not required to rank in priority to, Permitted Encumbrances;

shall in any manner, nor in any cause or proceeding, directly or indirectly, be taken to constitute a subordination of any Security Interest created pursuant to the Security to any Permitted Encumbrance or to any other Security Interest or other obligation whatsoever, or that the Secured Obligations are in any way subordinate or junior in right of payment to any Permitted Debt, it being the intention of the parties that all Security Interests created pursuant to the Security shall at all times, to the maximum extent permitted by applicable law and subject to Permitted Priority Liens and the Carve-Out, rank as first priority Security Interests in priority to Permitted Encumbrances and all other Security Interests or other obligations whatsoever and that the Secured Obligations will rank in right of payment at all times at least equally with such Permitted Debt.

9.5 Dealing with Security

The Lender may grant extensions of time or other indulgences, take and give up securities (including the Security or any part or parts thereof), accept compositions, grant releases and discharges and otherwise deal with the Obligors or any of them and other parties and with security (including without limitation, the Security and each part thereof) as the Lender may see fit, without prejudice to or in any way limiting the liability of the Obligors or any of them under this Agreement or the other Documents or under any of the Security or any other collateral security.

9.6 Effectiveness

The Security and the security created by any other Document constituted or required to be created shall be effective, and the undertakings as to the Security herein or in any other Document shall be continuing, whether any Loans are then outstanding or any amounts thereby secured or any part thereof shall be owing before or after, or at the same time as, the creation of such Security Interests or before or after or upon the date of execution of any amendments to this Agreement.

9.7 Further Assurances – Security

Each Obligor shall, forthwith and from time to time on the reasonable request of the Lender, grant to the Lender all such further rights and Security Interests necessary or of advantage to the Lender to permit it to operate the assets and business of the Obligors in a liquidation of assets or as a going concern following the occurrence of an Event of Default. In addition, each Obligor shall forthwith and from time to time on the reasonable request of the Lender execute and do or cause to be executed and done all assurances and things which in the opinion of the Lender may be necessary or of advantage to give the Lender the Security Interests and the priority intended hereunder to be created by the Security.

9.8 Release and Discharge of Security

No Obligor shall be discharged from the Security or any part thereof except by a written release and discharge signed by the Lender; provided that (1) to the extent such Security applies to a Permitted Disposition, the Lender shall provide a release or no interest letter in respect of the property and assets subject to such Permitted Disposition, promptly upon the request of the applicable Obligor, or (2) if all of the Secured Obligations have been unconditionally and indefeasibly repaid, paid, satisfied and discharged, as the case may be, in full and the Credit Facility and the Lender's commitment thereunder have been fully cancelled, then the Lender shall cause the Security and its interest therein to be released and discharged at the expense of the Borrowers.

ARTICLE 10 EVENTS OF DEFAULT AND ACCELERATION

10.1 Events of Default

The occurrence of any one or more of the following events (each such event being herein referred to as an "Event of Default") shall constitute a default under this Agreement:

- (a) <u>Principal Default</u>: if a Borrower fails to pay the principal of any Loan hereunder when due and payable;
- (b) Other Payment Default: if a Borrower fails to pay:
 - (i) any interest (including, if applicable, default interest) accrued on any Loan; or
 - (ii) any fees or other amounts not specifically referred to in paragraph (a) above or in this paragraph (b) payable by a Borrower hereunder;

in each case when due and payable, and such default is not remedied within two (2) Banking Days after written notice thereof is given by the Lender to the Borrowers specifying such default and requiring the relevant Borrower to remedy or cure the same;

- (c) <u>Certain Covenant Defaults</u>: if any Obligor fails to observe or perform any covenant in Section 8.2 or any negative covenant set forth in any Document;
- (d) Breach of Other Covenants: if any Obligor fails to observe or perform any covenant or obligation herein or in any other Document required on its part to be observed or performed (other than a covenant or condition whose breach or default in performance is specifically dealt with elsewhere in this Section) and the relevant Obligor shall fail to remedy such default within a period of three (3) Banking Days after the earlier of a Borrower or such Obligor having actual knowledge of such failure and delivery of notice by the Lender to the Borrowers specifying such default and requiring the relevant Obligor to remedy or cure the same;
- (e) <u>Incorrect Representations</u>: if any representation or warranty made by any Obligor herein or in any other Document shall prove to have been incorrect or misleading in any material respect on and as of the date made, or if any such representation or warranty is qualified as to "in all material respects", "materiality", "Material Adverse Effect" or similar language, such representation or warranty shall prove to have been incorrect or misleading in any respect on and as of the date made, and, in each case, the facts or circumstances which make such representation or warranty incorrect or misleading are not remedied and the representation or warranty in question remains incorrect or

- misleading more than five (5) Banking Days after a Borrower or any such Obligor becomes aware that such representation or warranty was incorrect or misleading in any respect;
- (f) <u>Disposition of Assets</u>: if any Obligor shall pass an effective resolution or initiate steps or proceedings (including applications to the Canadian Court and the U.S. Court) without the prior written consent of the Lender for the purpose of authorizing the disposition of all or a material amount of its property, assets and undertakings (except for a disposition in accordance with and as permitted by Section 8.2(e));
- (g) <u>Invalid Documents</u>: if any material provision of any Document continues to be invalid or unenforceable in whole or in a material part, or any of the Security Interests in and to any material Collateral constituted by the Security fails to attach thereto or to have the priority intended thereby;
- Adverse Proceedings: (i) if any proceeding, motion or application is commenced or filed by any (h) of the Obligors, or, if commenced by another party, supported or otherwise consented to by any Obligor, seeking the invalidation, subordination or other challenging of the terms of the Credit Facility, the Lender Charge, this Agreement, any other Document or the Initial Order; (ii) the filing of any Plan or application to the Canadian Court or the U.S. Court for approval of an Alternative Restructuring Option or sale of all or substantially all of the assets of the Obligors which does not either (A) have the prior written consent of the Lender or (B) result in the repayment in full in cash of all Indebtedness owing under each of this Agreement and the Pre-Filing Secured Credit Agreement upon the earlier of (x) the closing of such sale or Alternative Restructuring Transaction or upon the effective date of such Plan, and (y) the Maturity Date; or (iii) if any Obligor commences an action or takes any other proceeding to obtain any form of relief against the Lender, or the Pre-Filing Secured Creditor or any Affiliate thereof, including, without limitation, a proceeding to recover damages or to obtain payment of any amounts purported to be owing by the Lender, or the Pre-Filing Secured Creditor or any Affiliate thereof to any Obligor or any Affiliate thereof if the Lender, the Pre-Filing Secured Creditor or such Affiliate disputes any of the same;
- (i) Restructuring Court Orders: if: (i) any Restructuring Court Order contravenes this Agreement or any other Document so as to materially adversely impact the rights or interests of the Lender, as determined by the Lender, acting reasonably; or (ii) any Obligor breaches or otherwise violates in any way any Restructuring Court Order or (iii) the occurrence of any "Event of Default" as defined in the Interim U.S. DIP Order or the Final U.S. DIP Order;
- Court Order: the issuance of an order of the Canadian Court or the U.S. Court (including any (j) Restructuring Court Order) or any other court of competent jurisdiction: (i) dismissing the CCAA Proceedings or U.S. Proceedings, or lifting the stay in the CCAA Proceedings or U.S. Proceedings to permit (A) the enforcement of any Security Interest against an Obligor, or a material portion of their respective property, assets or undertaking, or (B) the appointment of a Receiver or the making of a bankruptcy order against an Obligor; (ii) granting any Security Interest which is senior to or pari passu with or senior to the Lender Charge, other than the Carve-Out; (iii) staying, reversing, vacating or otherwise modifying the Documents or any Restructuring Court Order in a manner materially adverse to the interests of the Lender, as determined by the Lender acting reasonably; (iv) materially adversely impacting the rights and interests of the Lender, as determined by the Lender acting reasonably, without the prior written consent of the Lender; (v) directing any Obligor to pay any post-employment benefits (excluding the KERP and KEIP); or (vi) other than the Carve-Out, granting, approving or allowing any administrative expense claim in the U.S. Proceedings, having any priority over, or being pari passu with, the superadministrative priority of the Credit Facility Indebtedness;

- (k) <u>Prohibited Filings</u>: the filing of any pleading by any Obligor seeking any of the matters set forth in Section 10.1(j) or failure of any Obligor to diligently oppose any Person that brings an application or motion for the relief set out in Section 10.1(j);
- (1) <u>Updated Budget Default</u>: if any information in the Updated Budget or Weekly Cashflow Report:
 (i) contemplates or forecasts an adverse change or changes from the then existing Agreed Budget and such change or changes constitute a Material Adverse Change; or (ii) forecasts that Advances under the Credit Facility will exceed the total authorized amount of the Credit Facility at any time (unless and until the Lender consents to increase the total authorized amount of the Credit Facility, which shall be in the Lender's sole and absolute discretion);
- (m) <u>Cash Flow Test</u>: if, after the date of this Agreement, as at the applicable calculation date, a Negative Cash Flow Variance exists, other than a variance solely due to: (i) changes in currency exchange rates, or (ii) the payment of professionals engaged by the Lender.
- (n) <u>Material Adverse Change</u>: if there occurs any Material Adverse Change;
- (o) <u>Excess Advances</u>: if at any time the Outstanding Principal exceeds the total authorized amount of the Credit Facility, provided that if such excess is a result of exchange rate fluctuations, if such excess is not repaid in accordance with Section 2.11;
- (p) <u>Plans and Alternative Restructuring Options</u>: if any Plan is filed with the U.S Court or the Canadian Court for approval or sanctioned or any Alternative Restructuring Option is filed with the U.S Court or the Canadian Court for approval or consummated by any of the Obligors that is not consistent with or contravenes any provision of this Agreement or any other Document;
- (q) <u>Prohibited Payments</u>: if except as set out in the Agreed Budget, or except or as otherwise agreed to in writing by the Lender, any Obligor is required by any Governmental Authority to make expenditures or pay damages, fines, claims, costs or expenses to remediate, in respect of any environmental liabilities, and such requirement is not stayed by a Restructuring Court Order;
- (r) <u>Permitted Priority Liens</u>: if any Obligor creates, permits to exist or seeks or supports a motion by another party to provide to any third party a Security Interest on the Collateral which is senior to or *pari passu* with the Lender Charge, other than the Permitted Priority Liens;
- (s) <u>Change in Name or Location</u>: if any Obligor changes its name, its trade name or its locations of business from those set forth in <u>Schedule 7.1(e)</u> annexed hereto without giving the Lender fifteen (15) days' prior notice thereof;
- (t) <u>Amend or Terminate Material Contracts</u>: if any Obligor disclaims, rejects, modifies, alters, amends, replaces, knowingly waives strict and timely performance of any compliance with (including, without limitation waive any default under) any material contract or terminates, cancels or suspends or assigns any material contract (except in accordance with its terms) or any material term, agreement, provision, item, obligation or covenant contained in any material contract, in each case, in any material respects without the consent of the Lender;
- (u) <u>Insurance Proceeds</u>: if any Obligor makes any application or use of any insurance proceeds (other than proceeds in respect of business interruption insurance) received by it in respect of any single claim or event which are not used to repair or replace any property which are the subject of such insurance claim until such application has been approved by the Lender in writing;

- (v) <u>Milestones</u>: the failure to achieve any one or more of the Milestones at the prescribed times including, for greater certainty, an Event of Default shall occur for the failure by the Obligors to use reasonable best efforts as prescribed by any particular Milestone;
- (w) <u>Bankruptcy</u>, <u>Receivership or Insolvency Proceedings</u>: if any Obligor becomes subject to any receivership proceedings or bankruptcy proceedings under the BIA or to any proceeding under chapter 7 or chapter 11 of the Bankruptcy Code or in any United States Bankruptcy Court other than the U.S. Court, or to any other bankruptcy, receivership or insolvency proceedings in any jurisdiction other than Canada, the United States or any State thereof, other than the U.S. Proceedings and CCAA Proceedings; or
- (x) <u>ERISA</u>: if (i) Holdco or any Borrower shall engage in any non-exempt "prohibited transaction" (as defined in Section 406 of ERISA or Section 4975 of the Code) involving any Pension Plan; or (ii) any ERISA Event shall occur and, taking together all such events, is reasonably expected to have a Material Adverse Effect.

10.2 Acceleration

If any Event of Default shall occur and for so long as it is continuing, subject to the Interim U.S. DIP Order and Final U.S. DIP Order, as applicable:

- (a) the entire principal amount of all Loans then outstanding from the Borrowers and all accrued and unpaid interest thereon; and
- (b) all other Secured Obligations outstanding hereunder,

shall, at the option of the Lender, become immediately due and payable (and, concurrently therewith, the Credit Facility and the Lender's commitment hereunder shall be fully and unconditionally cancelled) upon written notice (an "Acceleration Notice") to that effect from the Lender to the Borrowers, all without any other notice and without presentment, protest, demand, notice of dishonour or any other demand whatsoever (all of which are hereby expressly waived by the Borrowers); provided that, upon the occurrence of an Event of Default specified in Section 10.1(j)(i) (an "Acceleration Event") all Loans and other Secured Obligations outstanding hereunder shall automatically become due and payable (and, concurrently therewith, the Credit Facility and the Lender's commitment hereunder shall be fully and unconditionally cancelled) subject to any notice requirements set forth in the Interim U.S. DIP Order and Final U.S. DIP Order, as applicable. In such event and if the Borrowers do not immediately pay all such amounts upon receipt of any required notice, the Lender may, in its discretion, exercise any right or recourse and/or proceed by any action, suit, remedy or proceeding against the Borrowers authorized or permitted by law for the recovery of all the indebtedness and liabilities of the Borrowers to the Lender and proceed to exercise any and all rights hereunder and under the other Documents and no such remedy for the enforcement of the rights of the Lender shall be exclusive of or dependent on any other remedy but any one or more of such remedies may from time to time be exercised independently or in combination.

10.3 Termination of Lender's Obligations

The occurrence of a Default or Event of Default shall relieve the Lender of all obligations to provide any further Drawdowns to the Borrowers hereunder for so long as such Default or Event of Default is continuing; <u>provided</u> that the foregoing shall not prevent the Lender from disbursing money requested by the Borrowers and acceptable to the Lender.

10.4 Remedies Cumulative and Waivers

For greater certainty, it is expressly understood and agreed that the rights and remedies of the Lender hereunder or under any other Document are cumulative and are in addition to and not in substitution for any rights or remedies provided by law or by equity; and any single or partial exercise by the Lender of any right or remedy for a default or breach of any term, covenant, condition or agreement contained in this Agreement or other Document shall not be deemed to be a waiver of or to alter, affect or prejudice any other right or remedy or other rights or remedies to which the Lender may be lawfully entitled for such default or breach. Any waiver by the Lender of the strict observance, performance or compliance with any term, covenant, condition or other matter contained herein and any indulgence granted, either expressly or by course of conduct, by the Lender shall be effective only in the specific instance and for the purpose for which it was given and shall be deemed not to be a waiver of any rights and remedies of the Lender under this Agreement or any other Document as a result of any other default or breach hereunder or thereunder.

10.5 Application of Payments Following Demand and Acceleration

Subject to the Interim U.S. DIP Order and Final U.S. DIP Order, as applicable, except as otherwise agreed to by the Lender in its sole discretion, any sum received by the Lender at any time after delivery of an Acceleration Notice or the occurrence of an Acceleration Event which the Lender is obliged to apply in or towards satisfaction of sums due from the Borrowers hereunder shall be applied by the Lender in accordance with amounts owed to the Lender in respect of each category of amounts set forth below, each such application to be made in the following order with the balance remaining after application in respect of each category to be applied to the next succeeding category:

- (a) firstly, in or towards payment of any fees or expenses then due and payable to the Lender hereunder;
- (b) secondly, in or towards payment of amounts due and payable to the Lender as and by way of recoverable expenses hereunder;
- (c) thirdly, in or towards payment of amounts due and payable to the Lender by way of interest and overdue interest;
- (d) fourthly, in or towards repayment to the Lender of the Loans then outstanding hereunder; and
- (e) fifthly, in or towards payment of any other amounts not already referred to in this Section 10.5, which are then due and payable by the Borrowers hereunder.

10.6 Remedies

After an Event of Default:

- (a) Enforce Security: the Lender may take such actions and commence such proceedings as the Lender in its sole discretion may determine and may enforce or otherwise realize upon any Security, all without any obligation to marshal any Security Interests and without additional notice, presentation, demand or protest, all of which each of the Obligors hereby expressly waives (to the extent such rights may be waived under Applicable Law);
- (b) <u>General Remedies</u>: the rights and remedies of the Lender under the Documents are cumulative and are in addition to and not in substitution for any rights or remedies provided by law. The Lender

may, to the extent permitted and not prohibited by Applicable Law, bring suit at law, in equity or otherwise for any available relief or purpose including but not limited to:

- (i) <u>Specific Performance:</u> the specific performance of any covenant or agreement contained in the Documents;
- (ii) <u>Injunction:</u> enjoining a violation of any of the terms of the Documents;
- (iii) Assistance: aiding in the exercise of any power granted by the Documents or by law;
- (iv) <u>Judgment:</u> obtaining and recovering judgment for any and all amounts due in respect of the Loans or amounts otherwise due hereunder or under the Documents;
- (v) Receiver: seeking, by way of an application to the Canadian Court the appointment of a Receiver or similar official and such related proceedings in the U.S. Court as may be required;
- (vi) <u>BIA/Chapter 7:</u> seeking to adjudicate the Obligors bankrupt or convert the CCAA Proceedings and the U.S. Proceedings to proceedings under the BIA and/or proceedings under chapter 7 of the Bankruptcy Code;
- (vii) <u>Information Officer:</u> seeking to expand the powers of the Information Officer, pursuant to the terms of an order of the Canadian Court satisfactory to the Information Officer, to allow the Information Officer to realize on the Collateral and such ancillary and related relief before the U.S. Court as may be required; or
- (viii) Remedies at Law: exercising the powers and rights of a secured party under the Personal Property Security Act (British Columbia), the Uniform Commercial Code (United States) or any legislation of similar effect and all such other rights and remedies under the Documents, the Restructuring Court Orders and Applicable Law.

10.7 Set-Off

- (a) Subject to the Interim U.S. DIP Order and Final U.S. DIP Order, as applicable, in addition to any rights now or hereafter granted under applicable law and not by way of limitation of any such rights, upon the occurrence of an Event of Default which remains unremedied (whether or not the Loans have been accelerated hereunder), the Lender shall have the right (and is hereby authorized by each of the Obligors) at any time and from time to time to combine all or any of the Obligors' accounts with the Lender and to set-off and to appropriate and to apply any and all deposits (general or special, term or demand) including indebtedness evidenced by certificates of deposit whether matured or unmatured, and any other indebtedness at any time held by any of the Obligors or owing by the Lender to or for the credit or account of any of the Obligors against and towards the satisfaction of any Secured Obligations owing by any of the Obligors, and may do so notwithstanding that the balances of such accounts and the liabilities are expressed in different currencies, and the Lender is hereby authorized to effect any necessary currency conversions at the spot rate of exchange announced by the Bank of Canada for the Banking Day before the day of conversion.
- (b) The Lender shall provide advanced written notice to the Borrowers of any such set-off from the Obligors' accounts in accordance with the Interim U.S. DIP Order and the Final U.S. DIP Order, as applicable.

ARTICLE 11 CHANGE OF CIRCUMSTANCES

11.1 Change in Law

- (a) If the adoption of any applicable law, regulation, treaty or official directive (whether or not having the force of law) or any change therein or in the interpretation or application thereof by any court or by any Governmental Authority or any other entity charged with the interpretation or administration thereof or compliance by the Lender with any request or direction (whether or not having the force of law) of any such authority or entity in each case after the date hereof (including, without limitation, the *Dodd-Frank Wall Street Reform and Consumer Protection Act* and all requests, rules, regulations, guidelines and directives thereunder or issued in connection therewith and all requests, rules, regulations, guidelines and directives concerning capital adequacy promulgated by the Bank for International Settlements, the Basel Committee on Banking Supervision (or any successor or similar authority or any United States, Canadian or foreign regulatory authority), in each case regardless of the date enacted, adopted, or issued):
 - subjects the Lender to, or causes the withdrawal or termination of a previously granted exemption with respect to, any Taxes (other than Taxes on the Lender's overall income), or changes the basis of taxation of payments due to the Lender, or increases any existing Taxes (other than Taxes on the Lender's overall income) on payments of principal, interest or other amounts payable by a Borrower to the Lender under this Agreement;
 - (ii) imposes, modifies or deems applicable any reserve, liquidity, special deposit, regulatory or similar requirement against assets or liabilities held by, or deposits in or for the account of, or loans by the Lender, or any acquisition of funds for loans or commitments to fund loans or obligations in respect of undrawn, committed lines of credit;
 - (iii) imposes on the Lender or requires there to be maintained by the Lender any capital adequacy or additional capital requirements (including, without limitation, a requirement which affects the Lender's allocation of capital resources to its obligations) in respect of any Loan or obligation of the Lender hereunder, or any other condition with respect to this Agreement; or
 - (iv) directly or indirectly affects the cost to the Lender of making available, funding or maintaining any Loan (other than changes in Taxes on the Lender's overall income) or otherwise imposes on the Lender any other condition or requirement affecting this Agreement or any Loan or any obligation of the Lender hereunder;

and the result of (i), (ii), (iii) or (iv) above, in the sole determination of the Lender acting in good faith, is:

- (v) to increase the cost to the Lender of performing its obligations hereunder with respect to any Loan;
- (vi) to reduce any amount received or receivable by the Lender hereunder or its effective return hereunder or on its capital in respect of any Loan or the Credit Facility; or
- (vii) to cause the Lender to make any payment with respect to or to forego any return on or calculated by reference to, any amount received or receivable by the Lender hereunder with respect to any Loan or the Credit Facility;

the Lender shall determine that amount of money which shall compensate the Lender for such increase in cost, payments to be made or reduction in income or return or interest foregone (herein referred to as "Additional Compensation"). Upon the Lender having determined that it is entitled to Additional Compensation in accordance with the provisions of this Section, the Lender shall promptly so notify the Borrowers. The Lender shall provide the Borrowers with a photocopy of the relevant law, rule, guideline, regulation, treaty or official directive (or, if it is impracticable to provide a photocopy, a written summary of the same) and a certificate of a duly authorized officer of the Lender setting forth the Additional Compensation and the basis of calculation therefor, which shall be conclusive evidence of such Additional Compensation in the absence of manifest error. The Borrowers shall pay to the Lender within 10 Banking Days of the giving of such notice the Lender's Additional Compensation. The Lender shall be entitled to be paid such Additional Compensation from time to time to the extent that the provisions of this Section are then applicable notwithstanding that the Lender has previously been paid any Additional Compensation.

(b) The Lender agrees that it will not claim Additional Compensation from the Borrowers under Section 12.4(a) if it is not claiming similar compensation from its other customers in similar circumstances or in respect of any period greater than twelve months prior to the delivery of notice in respect thereof by the Lender, unless, in the latter case, the adoption, change or other event or circumstance giving rise to the claim for Additional Compensation is retroactive or is retroactive in effect.

11.2 Illegality

If the Lender determines, in good faith, that the adoption of any applicable law, regulation, treaty or official directive (whether or not having the force of law) or any change therein or in the interpretation or application thereof by any court or by any Governmental Authority or any other entity charged with the interpretation or administration thereof or compliance by the Lender with any request or direction (whether or not having the force of law) of any such authority or entity, now or hereafter makes it unlawful or impossible for the Lender to make, fund or maintain a Loan under the Credit Facility or to give effect to its obligations in respect of such a Loan, the Lender may, by written notice thereof to the Borrowers declare its obligations under this Agreement in respect of such Loan to be terminated whereupon the same shall forthwith terminate, and the relevant Borrower shall, within the time required by such law (or at the end of such longer period as the Lender at its discretion has agreed), prepay the principal of such Loan together with accrued interest, such Additional Compensation as may be applicable with respect to such Loan to the date of such payment and all costs, losses and expenses incurred by the Lender by reason of the liquidation or re deployment of deposits or other funds or for any other reason whatsoever resulting from the repayment of such Loan or any part thereof on other than the last day of the applicable Interest Period. If any such change shall only affect a portion of the Lender's obligations under this Agreement which is, in the opinion of the Lender, severable from the remainder of this Agreement so that the remainder of this Agreement may be continued in full force and effect without otherwise affecting any of the obligations of the Lender or the Borrowers hereunder, the Lender shall only declare its obligations under that portion so terminated.

ARTICLE 12 COSTS, EXPENSES AND INDEMNIFICATION

12.1 Costs and Expenses

The Borrowers shall pay promptly upon notice from the Lender all reasonable out-of-pocket costs and expenses of the Lender in connection with the Documents and the establishment of the Credit Facility, including in connection with (a) the preparation, printing, execution and delivery of this

Agreement and the other Documents and (b) the Lender's and any third party's due diligence investigation of the Obligors, in each case whether or not any Drawdown has been made hereunder, and also including, without limitation, the reasonable fees and out-of-pocket costs and expenses of Lender's Counsel with respect thereto and with respect to advising the Lender as to its rights and responsibilities under this Agreement and the other Documents. Except for ordinary expenses of the Lender relating to the day-to-day administration of this Agreement, the Borrowers further agree to pay within 10 days of demand by the Lender all reasonable out-of-pocket costs and expenses in connection with the preparation or review of waivers, consents and amendments pertaining to this Agreement, and in connection with the establishment of the validity and enforceability of this Agreement and the preservation or enforcement of rights of the Lender under this Agreement and other Documents, including, without limitation, all reasonable out-of-pocket costs and expenses sustained by the Lender as a result of any failure by the Obligors to perform or observe any of their respective obligations hereunder or in connection with any action, suit or proceeding (whether or not an Indemnified Party is a party or subject thereto), together with interest thereon from and after such 10th day if such payment is not made by such time.

12.2 General Indemnity

In addition to any liability of the Borrowers to the Lender under any other provision hereof, the Borrowers shall indemnify each Indemnified Party and hold each Indemnified Party harmless against any losses, claims, costs, damages or liabilities (including, without limitation, any expense or cost incurred in the liquidation and re deployment of funds acquired to fund or maintain any portion of a Loan and reasonable out-of-pocket expenses and reasonable legal fees on a solicitor and his own client basis) incurred by the same as a result of or in connection with the Credit Facility or the Documents (including any use of the proceeds of any Loan), including as a result of or in connection with:

- (a) any cost or expense incurred by reason of the liquidation or re-deployment in whole or in part of deposits or other funds required by the Lender to fund or maintain any Loan as a result of a Borrower's failure to complete a Drawdown or to make any payment, repayment or prepayment on the date required hereunder or specified by it in any notice given hereunder;
- (b) a Borrower's failure to pay any other amount, including without limitation any interest or fee, due hereunder on its due date after the expiration of any applicable grace or notice periods (subject, however, to the interest obligations of the Borrowers hereunder for overdue amounts);
- (c) a Borrower's failure to give any notice required to be given by it to the Lender hereunder;
- (d) the failure of a Borrower to make any other payment due hereunder;
- (e) any inaccuracy or incompleteness of the Obligors' representations and warranties contained in Article 7;
- (f) any failure of the Obligors to observe or fulfil their obligations under Article 8;
- (g) any failure of the Obligors to observe or fulfil any other obligation not specifically referred to above; or
- (h) the occurrence of any Default or Event of Default in respect of any Obligor,

provided that this Section shall not apply to any losses, claims, costs, damages or liabilities that arise by reason of the gross negligence or willful misconduct of the Indemnified Party claiming indemnity hereunder. The provisions of this Section shall survive repayment of the Secured Obligations.

12.3 Environmental Indemnity

The Borrowers shall indemnify and hold harmless the Indemnified Parties forthwith on demand by the Lender from and against any and all claims, suits, actions, debts, damages, costs, losses, liabilities, penalties, obligations, judgments, charges, expenses and disbursements (including without limitation, all reasonable legal fees and disbursements on a solicitor and his own client basis) of any nature whatsoever, suffered or incurred by the Indemnified Parties or any of them in connection with the Credit Facility, whether as beneficiaries under the Documents, as successors in interest of the Obligors, or voluntary transfer in lieu of foreclosure, or otherwise howsoever, with respect to any Environmental Claims relating to the property of the Obligors arising under any Environmental Laws as a result of the past, present or future operations of the Obligors (or any predecessor in interest thereto) relating to the property of the Obligors, or the past, present or future condition of any part of the property of the Obligors owned, operated or leased by the Obligors (or any such predecessor in interest), including any liabilities arising as a result of any indemnity covering Environmental Claims given to any Person by the Lender or a receiver, receiver manager or similar Person appointed hereunder or under applicable law (collectively, the "Indemnified Third Party"); but excluding any Environmental Claims or liabilities relating thereto to the extent that such Environmental Claims or liabilities arise by reason of the gross negligence or willful misconduct of the Indemnified Party or the Indemnified Third Party claiming indemnity hereunder. The provisions of this Section shall survive the repayment of the Secured Obligations.

12.4 Judgment Currency

- (a) If for the purpose of obtaining or enforcing judgment against the Obligors or any of them in any court in any jurisdiction, it becomes necessary to convert into any other currency (such other currency being hereinafter in this Section referred to as the "Judgment Currency") an amount due in Canadian Dollars under this Agreement, the conversion shall be made at the rate of exchange prevailing on the Banking Day immediately preceding:
 - (i) the date of actual payment of the amount due, in the case of any proceeding in the courts of any jurisdiction that will give effect to such conversion being made on such date; or
 - (ii) the date on which the judgment is given, in the case of any proceeding in the courts of any other jurisdiction (the date as of which such conversion is made pursuant to this Section being hereinafter in this Section referred to as the "Judgment Conversion Date").
- (b) If, in the case of any proceeding in the court of any jurisdiction referred to in Section 12.4(a)(ii), there is a change in the rate of exchange prevailing between the Judgment Conversion Date and the date of actual payment of the amount due, the relevant Obligors shall pay such additional amount (if any) as may be necessary to ensure that the amount paid in the Judgment Currency, when converted at the rate of exchange prevailing on the date of payment, will produce the amount of Canadian Dollars which could have been purchased with the amount of Judgment Currency stipulated in the judgment or judicial order at the rate of exchange prevailing on the Judgment Conversion Date.
- (c) Any amount due from the Obligors under the provisions of Section 12.4(b) shall be due as a separate debt and shall not be affected by judgment being obtained for any other amounts due under or in respect of this Agreement.
- (d) The term "rate of exchange" in this Section 12.4 means the rate of exchange for Canadian interbank transactions in Canadian Dollars in the Judgment Currency published by the Bank of Canada for

the day in question, or if such rate is not so published by the Bank of Canada, such term shall mean the rate of exchange determined by the Lender, acting reasonably.

12.5 Waiver of Consequential Damages, Etc.

To the fullest extent permitted by applicable law, the Obligors shall not assert, and each of the Obligors hereby waives, any claim against the Lender, on any theory of liability, for special, indirect, consequential or punitive damages (as opposed to direct or actual damages) arising out of, in connection with, or as a result of, this Agreement, any other Document or any agreement or instrument contemplated hereby, the transactions contemplated hereby or thereby, any Loan advanced hereunder, or the use of the proceeds thereof. The Lender shall not be liable for any damages arising from the use by unintended recipients of any information or other materials distributed by it through telecommunications, electronic or other information transmission systems in connection with this Agreement or the other Documents or the transactions contemplated hereby or thereby.

ARTICLE 13 INTERPRETATION; GENERAL

13.1 Survival of Undertakings

All covenants, undertakings, agreements, representations and warranties made pursuant to this Agreement survive the execution and delivery of this Agreement and continue in full force and effect until the full payment and satisfaction of all obligations of the Borrowers incurred pursuant to the Documents and the termination of this Agreement.

13.2 Failure to Act

No failure, omission or delay on the part of the Lender in exercising any right, power or privilege hereunder shall impair such right, power or privilege or operate as a waiver thereof nor shall any single or partial exercise of any right, power or privilege preclude any further exercise thereof or the exercise of any other right, power or privilege.

13.3 Exchange and Confidentiality of Information

- (a) The Obligors agree that the Lender may provide any assignee or participant or any *bona fide* prospective assignee or participant pursuant to Section 13.6 with any information concerning the financial condition of the Obligors provided such party agrees in writing with the Lender for the benefit of the Obligors to be bound by a like duty of confidentiality to that contained in this Section.
- (b) The Lender acknowledges the confidential nature of the financial, operational and other information and data provided and to be provided to them by the Obligors pursuant hereto (the "Information") and agrees to use all reasonable efforts to prevent the disclosure thereof; provided that:
 - (i) the Lender may disclose all or any part of the Information if, in their reasonable opinion, such disclosure is required in connection with any actual or threatened judicial, administrative or governmental proceedings including, without limitation, proceedings initiated under or in respect of this Agreement;

- (ii) the Lender shall incur no liability in respect of any Information required to be disclosed by any applicable law or regulation, or by applicable order, policy or directive having the force of law, to the extent of such requirement;
- (iii) the Lender may provide Lender's Counsel and their other agents and professional advisors with any Information; provided that such Persons shall be under a like duty of confidentiality to that contained in this Section;
- (iv) the Lender may provide its Affiliates (including BMO Financial Group business groups) and subsidiaries for the purpose of assisting the Lender in supporting the Obligors' strategic plans;
- (v) the Lender shall incur no liability in respect of any Information: (i) which is or becomes readily available to the public (other than by a breach hereof) or which has been made readily available to the public by the Obligors, (ii) which the Lender can show was, prior to receipt thereof from the Obligors, lawfully in the Lender's possession and not then subject to any obligation on its part to the Obligors to maintain confidentiality, or (iii) which the Lender received from a third party who was not, to the knowledge of the Lender, under a duty of confidentiality to the Obligors at the time the information was so received;
- (vi) the Lender may disclose the Information to other financial institutions and other Persons in connection with the assignment by the Lender of the Credit Facility and the Loans (or any portion thereof) or the granting by the Lender of a participation in the Credit Facility and the Loans where such financial institution or other Person agrees to be under a like duty of confidentiality to that contained in this Section; and
- (vii) the Lender may disclose all or any part of the Information so as to enable the Lender to initiate any lawsuit against the Obligors or any of them or to defend any lawsuit commenced by the Obligors or any of them the issues of which touch on the Information, but only to the extent such disclosure is necessary to the initiation or defense of such lawsuit.

13.4 Notices

Any demand, notice or communication to be made or given hereunder shall be in writing and may be made or given by personal delivery or by facsimile transmission or other electronic means of communication addressed to the respective parties as follows:

To the Obligors or any of them:

c/o SLP Holdings Ltd. 2176 Government Street Penticton, BC Canada V2A 8B5

Attention:

Shawn Turkington

Email:

sturkington@structurlam.com

- 59 -

To the Lender:

Bank of Montreal 595 Burrard Street, 6th Floor P.O. Box 49500 Vancouver, BC Canada V7X 1L7

Attention:

Shane Klein / Sandy Hayer

E-mail:

shane.klein@bmo.com / sandy.hayer@bmo.com

or to such other address or facsimile number as any party may from time to time notify the others in accordance with this Section. Any demand, notice or communication made or given by personal delivery or by facsimile transmission or other electronic means of communication during normal business hours at the place of receipt on a Banking Day shall be conclusively deemed to have been made or given at the time of actual delivery or transmittal, as the case may be, on such Banking Day. Any demand, notice or communication made or given by personal delivery or by facsimile transmission or other electronic means of communication after normal business hours at the place of receipt or otherwise than on a Banking Day shall be conclusively deemed to have been made or given at 9:00 a.m. (Local Time) on the first Banking Day following actual delivery or transmittal, as the case may be.

13.5 Governing Law; Jurisdiction

- (a) Governing Law. This Agreement shall be governed by and construed in accordance with the laws of the Province of British Columbia and the laws of Canada applicable therein, without prejudice to or limitation of any other rights or remedies available under the laws of any jurisdiction where property or assets of the Borrowers may be found.
- (b) <u>Submission.</u> The Borrowers and the Lender agree that the U.S. Court has jurisdiction to settle any disputes in connection with the Documents and accordingly the Borrowers and the Lender each submits to the non-exclusive jurisdiction of the U.S. Court.
- (c) Forum Convenience and Enforcement Abroad. Each Borrower:
 - (i) waives objection to the U.S. Court on grounds of inconvenient forum or otherwise as regards proceedings in connection with a Document; and
 - (ii) agrees that a judgment or order of the U.S. Court in connection with a Document is conclusive and binding on it (subject to any rights of appeal in respect thereof) and may be enforced against it in the courts of any other jurisdiction.
- (d) <u>Non-exclusivity.</u> Nothing in this Section 13.5 limits the right of the Lender to bring proceedings against the Borrowers in connection with any Document:
 - (i) in any other court of competent jurisdiction; or
 - (ii) concurrently in more than one jurisdiction.

13.6 Benefit of this Agreement; Assignment; Participations

- (a) This Agreement shall enure to the benefit of and be binding upon the Obligors, the Lender and their respective successors and permitted assigns.
- (b) The Lender may, without consent during the continuance of an Event of Default and at all other times with the prior written consent of the Borrowers, which consent shall not be unreasonably withheld or delayed, sell, assign, transfer or grant an interest in the Credit Facility, the Loans and its rights under this Agreement and the other Documents; provided that (a) any such sale, assignment, transfer or grant of interest shall be in a minimum principal amount of \$1,000,000 and (b) the purchaser, assignee, transferee, or grantee of such interest shall thereafter hold a minimum principal amount of \$1,000,000 of the Credit Facility. Notwithstanding the preceding sentence, the Lender may, at any time, without the prior written consent of the Borrowers, sell, assign, transfer or grant an interest in the Credit Facility, the Loans and its rights under this Agreement and the other Documents to an Affiliate of the Lender or an Approved Fund. Upon any such sale, assignment, transfer or grant, the Lender shall have no further obligation hereunder with respect to such interest. The Obligors shall not assign their rights or obligations hereunder without the prior written consent of the Lender.
- (c) The Lender may, without the consent of the Borrowers, grant one or more participations in the Loans to other Persons; provided that the granting of such a participation: (a) shall be at the Lender's own cost and (b) shall not affect the obligations of the Lender hereunder nor shall it increase the costs to the Borrowers hereunder or under any of the other Documents.

13.7 Severability

Any provision of this Agreement which is prohibited or unenforceable in any jurisdiction shall not invalidate the remaining provisions hereof and any such prohibition or unenforceability in any jurisdiction shall not invalidate or render unenforceable such provision in any other jurisdiction.

13.8 Whole Agreement; Amendments and Waivers

- (a) This Agreement and the other Documents constitute the whole and entire agreement between the parties hereto regarding the subject matter hereof and thereof and cancel and supersede any prior agreements, undertakings, declarations, commitments, representations, written or oral, in respect thereof.
- (b) Any provision of this Agreement may be amended only if the Borrowers and the Lender so agree in writing and may be waived only if the Lender agrees in writing. Any such waiver and any consent by the Lender under any provision of this Agreement must be in writing and may be given subject to any conditions thought fit by Lender. Any waiver or consent shall be effective only in the instance and for the purpose for which it is given.

13.9 Further Assurances

The Obligors and the Lender shall promptly cure any default by it in the execution and delivery of this Agreement, the other Documents or any of this Agreements provided for hereunder to which it is a party. Each Obligor, at the expense of the Borrowers, shall promptly execute and deliver to the Lender, upon request by the Lender (acting reasonably), all such other and further deeds, agreements, opinions, certificates, instruments, affidavits, registration materials and other documents reasonably necessary for such Obligor's compliance with, or accomplishment of the covenants and agreements of the Obligors

hereunder or more fully to state the obligations of the Obligors as set out herein or to make any registration, recording, file any notice or obtain any consent, all as may be reasonably necessary or appropriate in connection therewith.

13.10 Time of the Essence

Time shall be of the essence of this Agreement.

13.11 Agreement Governs

In the event of any conflict or inconsistency between the provisions of this Agreement and the provisions of the other Documents, the provisions of this Agreement, to the extent of the conflict or inconsistency, shall govern and prevail.

13.12 Counterparts

This Agreement may be executed in any number of counterparts, each of which shall be deemed to be an original and all of which taken together shall be deemed to constitute one and the same instrument, and it shall not be necessary in making proof of this Agreement to produce or account for more than one such counterpart. Delivery of an executed counterpart of a signature page of this Agreement by facsimile transmission or by sending a scanned copy by electronic mail shall be effective as delivery of a manually executed counterpart of this Agreement.

13.13 Resolving Discrepancies with Court Orders

In the event of any discrepancy between the terms of this Agreement and the terms of any order of the U.S. Court or the Canadian Court, such discrepancy shall be resolved in favour of the order of the U.S. Court.

[The remainder of this page has been intentionally left blank.]

IN WITNESS WHEREOF the parties hereto have caused this Agreement to be duly executed as of the date first written above.

BANK OF MONTREAL

By:

Name:

Title:

Sandy Hayer Director, SAMU National Accounts

By:

Name:

Shane Klein Managing Director Title:

DIP Financing Credit Agreement

	OCTORLA	AM MASS TIMBER CORPO	KА
By:	Name: Title:	Softwar Turn. No tea	
Зу:			
	Name:		
	Title:		
	201.0 010111111	rity to bind the corporation.	
STR		AM MASS TIMBER U.S., IN	с.
STR	UCTURL	AM MASS TIMBER U.S., IN	c.
			c.
STR	UCTURL Name:	AM MASS TIMBER U.S., IN	c.
STR	UCTURL Name:	AM MASS TIMBER U.S., IN	C.

I/We have authority to bind the corporation.

<u>GUA</u>	RANTORS	<u>:</u>	
SLP	HOLDING	S LTD.	
Ву:	R	2	hh
	Name: Title:	SHOWN	EMULLY TO
Bv.			

Name: Title:

I/We have authority to bind the corporation.

NATURAL OUTCOMES, LLC

Ву:		
,	Name:	
	Title:	
Ву:		
-	Name:	
	Title:	

I/We have authority to bind the corporation.

DocuSign Envelope ID: 55820990-8468-455E-930B-82058B6567AF-136-1 Filed 05/19/23 Page 69 of 71

DocuSign Envelope ID: FC7EF2B3-0124-4296-86A5-7076B2E5D206

GU	\mathbf{AR}	AN	TO	RS:

By: Shawn Turkington

Title: Interim CFO

By: Name:
Title: I/We have authority to bind the corporation.

NATURAL OUTCOMES, LLC

Ву:

Name: Matheur Waters Title: Plankager

By

Name:

Title:

I/We have authority to bind the corporation.

DIP Financing Credit Agreement

EXHIBIT 2

Final DIP Budget

Structurian et al Combined Chapter 11 Cash Flow Forceast - Prepared by Management For the period ending Angust 13, 2023 excent																	
Vesk grding	Actual Actual 28-Apr-23 05-May-23		Actual 12-May-23	Forecast 19-May-23	Forecast 26-May-23	Forecast 02-Jun-23	Forecast 09-Jun-23	Forecast 16-Jun-23	Forecast 23-Jun-23	Forecast 30-Jun-23	Forecast 07-Jul-23	Forecast 14-Jul-23	Forecast 21-Jul-23	Forecnst 28-Jul-23	Forecast 04-Aug-23	Forecast 11-Aug-23	Total
Receipts	•										,				•	6	1 164
Operating receipts	\$ 1,173		\$ 116	\$ 2,655	\$ 2,026				·		, 59			,		,	1,784
DIP Facility		2,000		2,000		3,500								•			7,500
Sales proceeds (net of BMO repayment, MB and 2nd KEIP)	١			-			3.674	,									18 058
Total Receipts	1,173	2,502	116	4,655	2,026	4,812	3,674				•					•	10,70
Disbursements		;	į	Ş	907	73						•		,			1.767
Materials	•	9	66	970	064	ŧ;		•	r	,			1	•	٠		2.625
Payroll, benefits and payroll taxes	323	511	255	551	2/0	9 (,	,	2,637
Manufacturing and other overhead		216	₹ 1	8//	99.	750			•		,				٠	,	613
Insurance	149	. !	314		051	, }			5						•		370
Rent and Dtilities	•	155			30	2			05		•				•		125
Sales taxes		2	8	. :		, ;			101	•	,						1 220
Bank interest and fees		155	6	446		019	٠			•		,	•				414
DIP Facility interest and fees		225	,			189	•										===
Sales advisor fees	,			101		. !		1	, ;			1986	•				3.778
Professional fees - debtor			,	1,178	•	1,053			292			455			•		87.9
Professional fees - Committee			,	135		405			3 :	,		. 5			•		1 322
Professional fees - other	•		•	386		351	. :		443		•	747			•		834
Key employee payments		•			•		834		•		. !	,					222
Post-closing administrative claims (placeholder)			•		. '	, :	•				1,373		•				27.
Contingency				25	25	25			, 100		1 12	502					17.420
Total Disbursements	472	1,302	1,295	4,234	1,831	4,189	\$24		1001		0/01						100
Net Cash Flow	701	1,200	(1,179)	422	194	623	2,840	•	(1,301)		(1,373)	(597)				r	nsctr
Cash Position Opening Cash Position	435	1,134	2,331	1.157	1,578	1,773	2,395	5,235	5,235	3,934	3,934	2,561	1,964	1,964	1,964	1,964	435
Foreign exchange gain (loss)	- 1	(2)	4	,	1			25.53	, c a	- 3 034	1 261	1 964	1 964	S 1.964	S 1.964	1.964	1.964
Closing Cash Position	\$ 1,134	5 2,331	5 1,157	5 1,578	5 1,73	6,095	- 11		9,734		1						
Interim Financing	٠		2.000	2.000	4.000	4.000	7,500				,	•	1	,	•		
Draws		2,000	•	2,000		3,500	. '		•	•	•	•	,	,			7,500
Planned repayment from sale proceeds		•		,	,		(7.500)										(1,000)
Ending Interim Financing	S	\$ 2,000	\$ 2,000	\$ 4,000	\$ 4,000	\$ 7,500		-		- -	. s	*		65	S		

This is Exhibit "___" referred to in the Affidavit of MICHÈLE HAY, sworn before me at Vancouver, BC, this 30th day of May, 2023.

A Commissioner for taking Affidavits in British Columbia

IN THE UNITED STATES BANKRUPTCY COURT FOR THE DISTRICT OF DELAWARE

In re:) Chapter 11	
STRUCTURLAM MASS TIMBER U.S., INC., et al., 1)) Case No. 23-10479 (CT	`G)
Debtors.) (Jointly Administered)	
	Re: Docket No. 15	

ORDER AUTHORIZING THE DEBTORS TO EMPLOY AND RETAIN KURTZMAN CARSON CONSULTANTS LLC AS CLAIMS AND NOTICING AGENT EFFECTIVE AS OF THE PETITION DATE

Upon the application (the "Application")² of the above-captioned debtors and debtors in possession (collectively, the "Debtors") for entry of an order (this "Order") for authority to employ and retain Kurtzman Carson Consultants LLC ("KCC") as claims and noticing agent (the "Claims and Noticing Agent") in the Debtors' Chapter 11 Cases effective as of the Petition Date, all as more fully set forth in the Application; and upon the First Day Declaration; and this Court having jurisdiction over this matter pursuant to 28 U.S.C. §§ 157 and 1334 and the Amended Standing Order of Reference from the United States District Court for the District of Delaware, dated February 29, 2012; and that this Court may enter a final order consistent with Article III of the United States Constitution; and this Court having found that venue of this proceeding and the Application in this district is proper pursuant to 28 U.S.C. §§ 1408 and 1409; and this Court having found that the relief requested in the Application is in the best interests of the Debtors' estates, their creditors, and other parties in interest; and this Court having found that the Debtors' notice of the Application and opportunity for a hearing on the Application were appropriate under the

The Debtors in these chapter 11 cases, along with the last four digits of each Debtor's federal tax identification number or Canadian business number, as applicable, include: Structurlam Mass Timber U.S., Inc. (6287); Natural Outcomes, LLC (n/a); Structurlam Mass Timber Corporation (5050); and SLP Holdings Ltd. (3114). The location of the Debtors' headquarters is: 2176 Government Street, Penticton, British Columbia, Canada V2A 8B5. The address of the registered agent for Structurlam Mass Timber U.S., Inc. is: 8 The Green, Suite A, Dover, Delaware 19901.

Capitalized terms used but not otherwise defined herein have the meanings ascribed to them in the Application.

circumstances and no other notice need be provided; and this Court having reviewed the Application and having heard the statements in support of the relief requested therein at a hearing before this Court (the "Hearing"); and this Court having determined that the legal and factual bases set forth in the Application and at the Hearing establish just cause for the relief granted herein; and upon all of the proceedings had before this Court; and after due deliberation and sufficient cause appearing therefor, it is HEREBY ORDERED THAT:

- 1. The Application is approved as set forth herein.
- 2. The Debtors are authorized, pursuant to 28 U.S.C. § 156(c) and Local Rule 2002-1(f), to retain KCC, and KCC is appointed as the Claims and Noticing Agent, effective as of the Petition Date, under the terms and conditions of the Services Agreement, attached hereto as **Exhibit 1**. Notwithstanding any terms of the Services Agreement, the Application is approved solely as set forth in this Order.
- 3. KCC, as the Claims and Noticing Agent, is directed to perform noticing services and to receive, maintain, record, and otherwise administer the proofs of claim filed in these cases (if any), and all related tasks, all as described in the Application (collectively, the "Claims and Noticing Services").
- 4. KCC shall serve as the custodian of court records and shall be designated as the authorized repository for all proofs of claim filed in these Chapter 11 Cases (if any) and is authorized and directed to maintain official claims registers for each of the Debtors, to provide public access to every proof of claim unless otherwise ordered by the Court and to provide the Clerk with a certified duplicate thereof upon the request of the Clerk.
- 5. KCC is authorized and directed to provide an electronic interface for filing proofs of claim and to obtain a post office box or address for the receipt of proofs of claim (if necessary).

- 6. KCC is authorized to take such other action to comply with all duties set forth in the Application.
- 7. KCC shall maintain records of all services performed, showing dates, categories of services, fees charged, and expenses incurred, and shall serve monthly invoices on the Debtors, the U.S. Trustee, counsel to the Debtors, counsel to any official committee monitoring the expenses of the Debtors, and any party in interest who specifically requests service of the monthly invoices.
- 8. The parties shall meet and confer in an attempt to resolve any dispute which may arise relating to the Services Agreement or monthly invoices, and the parties may seek resolution of such matter from this Court if resolution is not achieved.
- 9. Pursuant to section 503(b)(1)(A) of the Bankruptcy Code, the fees and expenses of KCC under this Order shall be an administrative expense of the Debtors' estates.
- 10. Without further order of this Court, the Debtors are authorized to compensate and reimburse KCC in accordance with the terms and conditions of the Services Agreement and consistent with the DIP order upon KCC's submission to the Debtors of invoices summarizing, in reasonable detail, the services rendered and the reasonable and necessary expenses incurred in connection therewith consistent with the DIP Order and without the necessity for KCC to file an application for compensation or reimbursement with this Court.
- 11. KCC seeks to first apply its retainer to all prepetition invoices and, thereafter, to have the retainer replenished to the original retainer amount and, thereafter, to hold the retainer under the Services Agreement during the Chapter 11 Cases as security for the payment of fees and expenses consistent with the DIP order incurred under the Services Agreement.

- 12. The Debtors shall indemnify KCC under the terms of the Services Agreement, as modified pursuant to this Order.
- 13. KCC shall not be entitled to indemnification, contribution, or reimbursement pursuant to the Services Agreement for services other than the Claims and Noticing Services, as provided under the Services Agreement, unless such services and the indemnification, contribution, or reimbursement therefore is approved by this Court.
- 14. Notwithstanding anything to the contrary in the Services Agreement, the Debtors shall have no obligation to indemnify KCC, or provide contribution or reimbursement to KCC, for any losses, claims, damages, judgments, liabilities, or expense that are either: (a) judicially determined (the determination having become final) to have arisen from KCC's gross negligence, willful misconduct, or fraud; (b) for a contractual dispute in which the Debtors allege the breach of KCC's contractual obligations, if this Court determines that indemnification, contribution, or reimbursement would not be permissible pursuant to *United Artists Theatre Co. v. Walton (In re United Artists Theatre Co.)*, 315 F.3d 217 (3d Cir. 2003); or (c) settled prior to a judicial determination under subsection (a) or (b), but determined by this Court, after notice and a hearing, to be a claim or expense for which KCC should not receive indemnity, contribution, or reimbursement under the terms of the Services Agreement as modified by this Order.
- 15. Before the earlier of: (a) the entry of an order confirming a chapter 11 plan in these Chapter 11 Cases (that order having become a final order no longer subject to appeal); and (b) the entry of an order closing these Chapter 11 Cases, should KCC believe that it is entitled to the payment of any amounts by the Debtors on account of the Debtors' indemnification, contribution, or reimbursement obligations under the Services Agreement (as modified by this Order), including, without limitation, the advancement of defense costs, KCC must file an application in

this Court, and the Debtors may not pay any such amounts to KCC before the entry of an order by this Court approving such application and the payment requested therein. This paragraph is intended only to specify the period of time under which this Court shall have jurisdiction over any request for fees and expenses by KCC for indemnification, contribution, or reimbursement, and not a provision limiting the duration of the Debtors' obligation to indemnify KCC. All parties in interest shall retain the right to object to any demand by KCC for indemnification, contribution, or reimbursement.

- 16. In the event KCC is unable to provide the Claims and Noticing Services, KCC will immediately notify the Clerk and the Debtors' counsel and, upon approval of this Court, cause to have all original proofs of claim and computer information turned over to another claims and noticing agent with the advice and consent of the Clerk and the Debtors' counsel.
- 17. The Debtors may submit a separate retention application, pursuant to section 327 of the Bankruptcy Code or any applicable law, for work that is to be performed by KCC but is not specifically authorized by this Order.
- 18. Notwithstanding any term in the Services Agreement to the contrary, during the Chapter 11 Cases, KCC's liability will not be limited to the amount paid or billed to the Debtors.
- 19. KCC shall not cease providing claims processing services during these Chapter 11 Cases for any reason, including nonpayment, without an order of this Court.
- 20. The Debtors and KCC are authorized to take all steps necessary or appropriate to carry out this Order.
- 21. In the event of any inconsistency between the Services Agreement, the Application, and this Order, the terms of this Order shall govern.

- 22. Notice of the Application as provided therein shall be deemed good and sufficient notice of such Application and the requirements of Bankruptcy Rule 6004(a) and the Local Rules are satisfied by such notice.
- 23. Notwithstanding any term in the Services Agreement to the contrary, this Court shall retain jurisdiction to hear and determine all matters arising from or related to the implementation, interpretation, or enforcement of this Order.

Dated: April 26th, 2023 Wilmington, Delaware CRAIG T. GOLDBLATT UNITED STATES BANKRUPTCY JUDGE

Cry Dollow

Exhibit 1

Services Agreement



This Agreement is entered into as of the 13 day of April 2023, between Sturcturlam Mass Timber Corporation (together with its affiliates and subsidiaries, the "Company"), and Kurtzman Carson Consultants LLC (together with its affiliates and subcontractors, "KCC"). In consideration of the premises set forth herein and other good and valuable consideration, the receipt and sufficiency of which are hereby acknowledged, the parties hereto agree as follows:

Terms and Conditions

I. SERVICES

- A. KCC agrees to provide the Company with consulting services regarding noticing, claims management and reconciliation, plan solicitation, balloting, disbursements and any other services agreed upon by the parties or otherwise required by applicable law, government regulations or court rules or orders.
- B. KCC further agrees to provide (i) computer software support and training in the use of the support software, (ii) KCC's standard reports as well as consulting and programming support for the Company requested reports, (iii) program modifications, (iv) data base modifications, and/or (v) other features and services in accordance with the fees outlined in a pricing schedule provided to the Company (the "KCC Fee Structure").
- C. Without limiting the generality of the foregoing, KCC may, upon request by the Company, (i) provide a communications plan including, but not limited to, preparation of communications materials, dissemination of information and a call center staffed by KCC and/or (ii) provide confidential on-line workspaces or virtual data rooms and publish documents to such workspaces or data rooms (which publication shall not be deemed to violate the confidentiality provisions of this Agreement).
- D. The price listed for each service in the KCC Fee Structure represents a bona fide proposal for such services, which may be accepted in whole or in part. Services will be provided when requested by the Company or required by applicable law, government regulations or court rules or orders. Services are mutually exclusive and are deemed delivered and accepted by the Company when provided by KCC.
- E. The Company acknowledges and agrees that KCC will often take direction from the Company's representatives, employees, agents and/or professionals (collectively, the "Company Parties") with respect to the services being provided under this Agreement. The parties agree that KCC may rely upon, and the Company agrees to be bound by, any requests, advice or information provided by the Company Parties to the same extent as if such requests, advice or information were provided by the Company. The Company agrees and understands that KCC shall not provide the Company or any other party with any legal advice.

¹ The term Company shall include, to the extent applicable, the Company, as debtor and debtor in possession in its chapter 11 case, together with any affiliated debtors and debtors in possession whose chapter 11 cases are jointly administered with the Company's chapter 11 case.

II. PRICES, CHARGES AND PAYMENT

- A. KCC agrees to charge and the Company agrees to pay KCC for its services at the rates and prices set by KCC that are in effect as of the date of this Agreement and in accordance with the KCC Fee Structure. KCC's prices are generally adjusted periodically to reflect changes in the business and economic environment and are inclusive of all charges. KCC reserves the right to reasonably increase its prices, charges and rates; provided, however, that if any such increase exceeds 15%, KCC will give thirty (30) days written notice to the Company.
- B. In addition to fees and charges for services, the Company agrees to pay KCC's reasonable transportation, lodging, and meal expenses incurred in connection with services provided under this Agreement.
- C. In addition to all fees for services and expenses hereunder, the Company shall pay to KCC (i) any fees and charges related to, arising out of, or as a result of any error or omission made by the Company or the Company Parties, as mutually determined by KCC and the Company, and (ii) all taxes that are applicable to this Agreement or that are measured by payments made under this Agreement and are required to be collected by KCC or paid by KCC to a taxing authority.
- D. Where the Company requires services that are unusual or beyond the normal business practices of KCC, or are otherwise not provided for in the KCC Fee Structure, the cost of such services shall be charged to the Company at a competitive rate.
- KCC agrees to submit its invoices to the Company monthly and the Company agrees that the amount invoiced is due and payable upon the Company's receipt of the invoice. KCC's invoices will contain reasonably detailed descriptions of charges for both hourly (fees) and nonhourly (expenses) case specific charges. Where total invoice amounts are expected to exceed \$10,000 in any single month and KCC reasonably believes it will not be paid, KCC may require advance payment from the Company due and payable upon demand and prior to the performance of services hereunder. If any amount is unpaid as of thirty (30) days from the receipt of the invoice, the Company further agrees to pay a late charge, calculated as one and one-half percent (1-1/2%) of the total amount unpaid every thirty (30) days. In the case of a dispute in the invoice amount, the Company shall give written notice to KCC within ten (10) days of receipt of the invoice by the Company. The undisputed portion of the invoice will remain due and payable immediately upon receipt of the invoice. Late charges shall not accrue on any amounts in dispute or any amounts unable to be paid due to Court order or applicable law. Unless otherwise agreed to in writing, the fees for print notice and media publication (including commissions) must be paid at least three (3) days in advance of those fees and expenses being incurred. Certain fees and charges may need to be adjusted due to availability related to the COVID-19 (novel coronavirus) global health issue.
- F. In the event that the Company files for protection pursuant to chapter 11 of the United States Bankruptcy Code (a "Chapter 11 Filing"), the parties intend that KCC shall be employed pursuant to 28 U.S.C. § 156(c) to the extent possible and otherwise in accordance with applicable Bankruptcy law and that all amounts due under this Agreement shall, to the extent possible, be paid as administrative expenses of the Company's chapter 11 estate. As soon as practicable following a Chapter 11 Filing (and otherwise in accordance with applicable law and rules and



orders of the Bankruptcy Court), the Company shall cause pleadings to be filed with the Bankruptcy Court seeking entry of an order or orders approving this Agreement (the "Retention Order"). The form and substance of the pleadings and the Retention Order shall be reasonably acceptable to KCC. If any Company chapter 11 case converts to a case under chapter 7 of the Bankruptcy Code, KCC will continue to be paid for its services in accordance with the terms of this Agreement. The parties recognize and agree that if there is a conflict between the terms of this Agreement and the terms of the Retention Order, the terms of the Retention Order shall govern during the chapter 11 or other proceeding.

G. To the extent permitted by applicable law, KCC shall receive a retainer in the amount of \$15,000 (the "Retainer") that may be held by KCC as security for the Company's payment obligations under the Agreement. The Retainer is due upon execution of this Agreement. In the event of a Chapter 11 Filing, KCC will first apply the Retainer to all pre-petition invoices, and thereafter, will have the Retainer replenished to the original amount. KCC shall be entitled to hold the Retainer until the termination of the Agreement. Following termination of the Agreement, KCC shall return to the Company any amount of the Retainer that remains following application of the Retainer to the payment of unpaid invoices.

III. RIGHTS OF OWNERSHIP

- A. The parties understand that the software programs and other materials furnished by KCC pursuant to this Agreement and/or developed during the course of this Agreement by KCC are the sole property of KCC. The term "program" shall include, without limitation, data processing programs, specifications, applications, routines, and documentation. The Company agrees not to copy or permit others to copy the source code from the support software or any other programs or materials furnished pursuant to this Agreement.
- B. The Company further agrees that any ideas, concepts, know-how or techniques relating to data processing or KCC's performance of its services developed or utilized during the term of this Agreement by KCC shall be the exclusive property of KCC. Fees and expenses paid by the Company do not vest in the Company any rights in such property, it being understood that such property is only being made available for the Company's use during and in connection with the services provided by KCC under this Agreement.

IV. NON-SOLICITATION

The Company agrees that neither it nor its subsidiaries or other affiliated companies shall directly or indirectly solicit for employment, employ or otherwise retain employees of KCC during the term of this Agreement and for a period of twelve (12) months after termination of this Agreement unless KCC provides prior written consent to such solicitation or retention.

V. CONFIDENTIALITY

Each of KCC and the Company, on behalf of themselves and their respective employees, agents, professionals and representatives, agrees to keep confidential all non-public records, systems, procedures, software and other information received from the other party in connection with the services provided under this Agreement; provided, however, that if either party reasonably believes that it is required to produce any such information by order of any governmental agency



or other regulatory body it may, upon not less than five (5) business days' written notice to the other party, release the required information.

VI. SUSPENSION OF SERVICE AND TERMINATION

- A. This Agreement shall remain in force until terminated or suspended by either party (i) upon thirty (30) days' written notice to the other party or (ii) immediately upon written notice for Cause (defined herein). As used herein, the term "Cause" means (i) gross negligence or willful misconduct of KCC that causes serious and material harm to the Company's reorganization under chapter 11 of the Bankruptcy Code, (ii) the failure of the Company to pay KCC invoices for more than sixty (60) days from the date of invoice, or (iii) the accrual of invoices or unpaid services in excess of the retainer held by KCC where KCC reasonably believes it will not be paid.
- B. In the event that this contract is terminated, regardless of the reason for such termination, KCC shall coordinate with the Company and, to the extent applicable, the clerk of the Bankruptcy Court, to maintain an orderly transfer of record keeping functions and KCC shall provide all necessary staff, services and assistance required for an orderly transfer. The Company agrees to pay for such services in accordance with KCC's then existing prices for such services. If such termination occurs following entry of the Retention Order, the Company shall immediately seek entry of an order (in form and substance reasonably acceptable to KCC) that discharges KCC from service and responsibility in the Company's bankruptcy case.
- C. Any data, programs, storage media or other materials furnished by the Company to KCC or received by KCC in connection with the services provided under the terms of this Agreement may be retained by KCC until the services provided are paid for, or until this Agreement is terminated with the services paid in full. The Company shall remain liable for all fees and expenses imposed under this Agreement as a result of data or physical media maintained or stored by KCC. KCC shall dispose of the data and media in the manner requested by the Company. The Company agrees to pay KCC for reasonable expenses incurred as a result of the disposition of data or media. If the Company has not utilized KCC's services under this Agreement for a period of at least ninety (90) days, KCC may dispose of the data or media, and be reimbursed by the Company for the expense of such disposition, after giving the Company thirty (30) days' notice. Notwithstanding any term herein to the contrary, following entry of the Retention Order, the disposition of any data or media by KCC shall be in accordance with any applicable instructions from the clerk of the Bankruptcy Court, local Bankruptcy Court rules and orders of the Bankruptcy Court.

VII. SYSTEM IMPROVEMENTS

KCC strives to provide continuous improvements in the quality of service to its clients. KCC, therefore, reserves the right to make changes in operating procedure, operating systems, programming languages, general purpose library programs, application programs, time period of accessibility, types of terminal and other equipment and the KCC data center serving the Company, so long as any such changes do not materially interfere with ongoing services provided to the Company in connection with the Company's chapter 11 case.



VIII. BANK ACCOUNTS

At the Company's request and subject to Court approval following any chapter 11 filing, KCC may be authorized to establish accounts with financial institutions in the name of and as agent for the Company. To the extent that certain financial products are provided to the Company pursuant to KCC's agreement with financial institutions, KCC may receive compensation from such financial institutions for the services KCC provides pursuant to such agreement.

IX. LIMITATIONS OF LIABILITY AND INDEMNIFICATION

- A. The Company shall indemnify and hold KCC, its affiliates, members, directors, officers, employees, consultants, subcontractors and agents (collectively, the "Indemnified Parties") harmless, to the fullest extent permitted by applicable law, from and against any and all losses, claims, damages, judgments, liabilities and expenses (including reasonable counsel fees and expenses) (collectively, "Losses") resulting from, arising out of or related to KCC's performance under this Agreement. Such indemnification shall exclude Losses resulting from KCC's gross negligence or willful misconduct. Without limiting the generality of the foregoing, Losses include any liabilities resulting from claims by any third-parties against any Indemnified Party. The Company shall notify KCC in writing promptly upon the assertion, threat or commencement of any claim, action, investigation or proceeding that the Company becomes aware of with respect to the services provided by KCC under this Agreement. The Company's indemnification obligations hereunder shall survive the termination of this Agreement.
- B. Except as provided herein, KCC's liability to the Company or any person making a claim through or under the Company for any Losses of any kind, even if KCC has been advised of the possibility of such Losses, whether direct or indirect and unless due to gross negligence or willful misconduct of KCC, shall be limited to the total amount billed or billable to the Company for the portion of the particular work which gave rise to the alleged Loss. In no event shall KCC be liable for any indirect, special or consequential damages such as loss of anticipated profits or other economic loss in connection with or arising out of the services provided for in this Agreement. In no event shall KCC's liability to the Company for any Losses, whether direct or indirect, arising out of this Agreement exceed the total amount billed to the Company and actually paid to KCC for the services contemplated under the Agreement; provided, however, that this limitation shall not apply to the Company during any chapter 11 case in which the Company is a debtor.
- C. The Company is responsible for the accuracy of the programs, data and information it or any Company Party submits for processing to KCC and for the output of such information. KCC does not verify information provided by the Company and, with respect to the preparation of schedules and statements, all decisions are at the sole discretion and direction of the Company. The Company reviews and approves all schedules and statements filed on behalf of, or by, the Company; KCC bears no responsibility for the accuracy or contents therein. The Company agrees to initiate and maintain backup files that would allow the Company to regenerate or duplicate all programs and data submitted by the Company to KCC.
- D. The Company agrees that except as expressly set forth herein, KCC makes no representations or warranties, express or implied, including, but not limited to, any implied or



express warranty of merchantability, fitness or adequacy for a particular purpose or use, quality, productiveness or capacity.

X. FORCE MAJEURE

KCC will not be liable for any delay or failure in performance when such delay or failure arises from circumstances beyond its reasonable control, including without limitation acts of God, acts of government in its sovereign or contractual capacity, acts of public enemy or terrorists, acts of civil or military authority, war, riots, civil strife, terrorism, blockades, sabotage, rationing, embargoes, epidemics, pandemics, outbreaks of infectious diseases or any other public health crises, earthquakes, fire, flood, other natural disaster, quarantine or any other employee restrictions, power shortages or failures, utility or communication failure or delays, labor disputes, strikes, or shortages, supply shortages, equipment failures, or software malfunctions.

XI. INDEPENDENT CONTRACTORS

The Company and KCC are and shall be independent contractors of each other and no agency, partnership, joint venture or employment relationship shall arise, directly or indirectly, as a result of this Agreement.

XII. NOTICES

All notices and requests in connection with this Agreement shall be given or made upon the respective parties in writing and shall be deemed as given as of the third day following the day it is deposited in the U.S. Mail, postage pre-paid or on the day it is given if sent by facsimile or electronic mail or on the day after the day it is sent if sent by overnight courier to the appropriate address set forth below:

Kurtzman Carson Consultants LLC Company
222 N. Pacific Coast Highway, 3rd Floor
El Segundo, CA 90245 City, ST Zip

Attn: Drake D. Foster Attn:
Tel: (310) 823-9000 Tel:
Fax: (310) 823-9133 Fax:

E-Mail: dfoster@kccllc.com

Or to such other address as the party to receive the notice or request so designates by written notice to the other.

XIII. APPLICABLE LAW

The validity, enforceability, and performance of this Agreement shall be governed by and construed in accordance with the laws of the State of California.

XIV. ENTIRE AGREEMENT/ MODIFICATIONS

Each party acknowledges that it has read this Agreement, understands it, and agrees to be bound by its terms and further agrees that it is the complete and exclusive statement of the agreement



between the parties, which supersedes and merges all prior proposals, understandings, other agreements, and communications oral and written between the parties relating to the subject matter of this Agreement. The Company represents that it has the authority to enter into this Agreement, and the Agreement is non-dischargeable under any applicable statute or law. If any provision of this Agreement shall be held to be invalid, illegal or unenforceable, the validity, legality and enforceability of the remaining provisions shall in no way be affected or impaired thereby. This Agreement may be modified only by a written instrument duly executed by an authorized representative of the Company and an officer of KCC.

XV. COUNTERPARTS; EFFECTIVENESS

This Agreement may be executed in two or more counterparts, each of which will be deemed an original but all of which together will constitute one and the same instrument. This Agreement will become effective when one or more counterparts have been signed by each of the parties and delivered to the other parties, which delivery may be made by exchange of copies of the signature page by facsimile or electronic mail.

XVI. ASSIGNMENT

This Agreement and the rights and duties hereunder shall not be assignable by the parties hereto except upon written consent of the other, with the exception that this Agreement can be assigned without written consent by KCC to a wholly-owned subsidiary or affiliate of KCC.

XVII. ATTORNEYS' FEES

In the event that any legal action, including an action for declaratory relief, is brought to enforce the performance or interpret the provisions of this Agreement, the parties agree to reimburse the prevailing party's reasonable attorneys' fees, court costs, and all other related expenses, which may be set by the court in the same action or in a separate action brought for that purpose, in addition to any other relief to which the prevailing party may be entitled.

[SIGNATURE PAGE FOLLOWS]



IN WITNESS WHEREOF, the parties hereto have executed this Agreement effective as of the first date mentioned above.

Kurtzman Carson Consultants LLC

— DocuSigned by:			
Evan Gershbein			
BY: Evan Gershbein	DATE: 15-Apr-2023	8:56:29	PM EDT
TITLE: EVP, Corporate Restructuring	Services		
		•	

Company -DocuSigned by: Shawn Turkington

______BY: Shawn Turkington DATE: 4/15/2023 TITLE: Interim Chief Financial Officer

Fee Structure



Consulting Services & Rates i
Position: Hourly Rate

Analyst

\$27 - \$45

The Analyst processes incoming mail, including proofs of claim, ballots, creditor correspondence and returned mail. Also assists with the generation of mailing services.

Technology/Programming Consultant 2

\$31.50 - \$85.50

The Technology/Programming Consultant assists with complex system requests, including unique claim/ballot reporting and custom website updates.

Consultant/Senior Consultant/Director

\$58.50 - \$175.50

The Consultant is the day-to-day contact for mailings, including the preparation and filing of affidavits of service (a critical due process component). He/she also responds to creditor and counsel inquiries, maintains the public access website, identifies actionable pleadings (i.e., claims objections, notices of transfer, withdrawals, etc.) and updates the official claims register. KCC's Consultants average over six years of experience.

The Senior Consultant manages the various data collection processes required by the chapter 11 process. This includes, among other things, compiling the creditor matrix and Schedules/SOFAs (and generating drafts of same for counsel and advisors), reviewing and processing claims, overseeing contract review, overseeing all mailings and generating custom claim and ballot reports. KCC's Senior Consultants average over seven years of experience.

The Director is the primary contact for the company, counsel and other professionals and oversees and supports the entirety of an engagement. KCC's Directors average over twelve years of experience and are generally former practitioners.

Securities/Solicitation Consultant

\$184.50

The Securities Director/Solicitation Consultant is the day-to-day contact and acts as advisor on transactions including balloting with treatment election, rights offers, exchange offers and complex plan distributions. This position handles service of related materials to banks, brokers and agents and manages tabulation and audit processes, preparing detailed reporting of results. In addition, the Solicitation Consultant provides support on all voting, tabulation, Schedule and SOFA services and other additional complex consulting tasks.

Securities Director/Solicitation Lead

\$193.50

The Solicitation Lead/Securities Director oversees all activities of the group and provides counsel with respect to solicitation and noticing events ensuring that processes employed are effective and practical for securities depositories, bank, brokers, nominees and their agents. In addition, the Solicitation Lead provides counsel on all voting, tabulation, Schedule and SOFA services and other additional complex consulting tasks.

Weekend, holidays and overtime

Waived

Fee Structure



Printing Services & Noticing Services	
Printing	\$0.10 per image (volume discounts apply)
Document folding and inserting	Waived
Envelopes	Varies by size
E-mail noticing	Waived 3
Fax noticing	\$0.05 per page
Public Securities Events	Varies by Event
Claim Acknowledgement Card	Waived
Insert creditor information into customized documents	Waived
Newspaper and legal notice publishing	Quote prior to publishing

Claims Administration & Management Expenses	
License fee and data storage	\$0.10 per record per month
Database and system access (unlimited users)	Waived
Custom client reports	Waived
Access to KCC CaseView (secure, password protected)	Waived

- Proprietary, secured, password protected portal for unlimited users
- Comprehensive case data, including extensive real time analytics on claim, solicitation and processing information
- Functionality to run or request customized reports summarizing case analytics

KCC eServices		
Case website set up & hosting	Waived	
Automated updates of case docket and claims register	Waived	
Online claims filing (ePOC)	Waived	

Document Management/Imaging	
Electronic imaging (scanning & bar coding)	\$0.10 per imaged page
Virtual Data Room	Quote prior to VDR set-up
CD-ROMS (mass document storage)	Varies upon requirements

Fee Structure



Call Center Support Services

Case-specific voice-mail box for creditors

Waived

Interactive Voice Response ("IVR")

Set-up and per minute fee waived

Monthly maintenance charge

Waived

Management of Call Center

Standard hourly rates

Disbursament

Check issuance

Quote prior to printing

W-9 mailing and maintenance of TIN database

See hourly rates and noticing charges

This is Exhibit "<u>K</u>" referred to in the Affidavit of **MICHÈLE HAY**, sworn before me at Vancouver, BC, this 30^{th} day of May 2023.

A Commissioner for taking Affidavits in British Columbia

IN THE UNITED STATES BANKRUPTCY COURT FOR THE DISTRICT OF DELAWARE

	1
In re:) Chapter 11
STRUCTURLAM MASS TIMBER U.S., INC., et al.,1) Case No. 23-10497 (CTG)
TO 11) (Jointly Administered)
Debtors.) Related Docket No. 93
) .

ORDER PURSUANT TO BANKRUPTCY RULE 3003(c)(3) AND LOCAL RULE 2002-1(e) ESTABLISHING BAR DATES AND RELATED PROCEDURES FOR FILING PROOFS OF CLAIM (INCLUDING FOR ADMINISTRATIVE EXPENSE CLAIMS ARISING UNDER SECTION 503(b)(9) OF THE BANKRUPTCY CODE) AND APPROVING THE FORM AND MANNER OF NOTICE THEREOF

Upon consideration of the motion (the **Motion**")² of the Debtors for entry of an order, pursuant to Bankruptcy Rule 3003(c)(3) and Local Rule 2002-1(e), establishing Bar Dates and related procedures by which creditors must file their Proofs of Claim, as more fully described in the Motion; and the Court having reviewed the Motion; and the Court having determined that there is good and sufficient cause for the relief set forth herein; and the Court having jurisdiction to consider the Motion and the relief requested therein in accordance with 28 U.S.C. §§ 157 and 1334 and the *Amended Standing Order of Reference* from the United States District Court for the District of Delaware dated as of February 29, 2012; and it appearing that proper and adequate notice of the

² Capitalized terms used but not otherwise defined herein shall have the same meanings ascribed to them in the Motion.



The Debtors in these chapter 11 cases, along with the last four digits of each Debtors' federal tax identification number or Canadian business number, as applicable, include: Structurlam Mass Timber U.S., Inc. (6287); Natural Outcomes, LLC (n/a); Structurlam Mass Timber Corporation (5050); and SLP Holdings Ltd. (3114). The location of the Debtors' headquarters is: 2176 Government Street, Penticton, British Columbia, Canada V2A 8B5. The address of the registered agent for Structurlam Mass Timber U.S., Inc. is: 8 The Green, Suite A, Dover, Delaware 19901.

Motion has been given and that no other or further notice is necessary; and after due deliberation thereon; and good and sufficient cause appearing therefor,

IT IS HEREBY ORDERED, ADJUDGED, AND DECREED THAT:

- 1. The Motion is GRANTED as set forth herein.
- 2. The Court hereby approves (i) the forms of the Bar Date Notice, the Proof of Claim Form, and the Publication Notice, substantially in the forms attached to this Order as **Exhibit 1**, **Exhibit 2** and **Exhibit 3**, respectively, and (ii) the manner of providing notice of the Bar Dates as described in this Order.
- 3. Pursuant to Bankruptcy Rule 3003(c)(2), any creditor (as defined in section 101(10) of the Bankruptcy Code) other than a governmental unit (as defined in section 101(27) of the Bankruptcy Code), who asserts a claim (as defined in section 101(5) of the Bankruptcy Code) against the Debtors that arose, or is deemed to have arisen, prior to April 21, 2023 (the "Petition Date") and does not fall under any of the categories set forth in paragraph 10 of this Order, must file a Proof of Claim on or prior to July 10, 2023 at 5:00 p.m. (prevailing Eastern Time) (the "General Bar Date") as provided in this Order.
- 4. Notwithstanding Paragraph 3 above, the deadline for governmental units (as defined in section 101(27) of the Bankruptcy Code) to file a Proof of Claim against the Debtors is **October 18, 2023** at 5:00 p.m. (*prevailing* Eastern Time) (the "Governmental Bar Date").
- 5. Any person or entity (including, without limitation, each individual, partnership, joint venture, corporation, estate, trust and governmental unit), that holds, or seeks to assert, a claim (as defined in section 101(5) of the Bankruptcy Code) against the Debtors that arose, or is deemed to have arisen, prior to the Petition Date including, without limitation, secured claims, unsecured priority claims (including, without limitation, claims entitled to priority under sections

507(a)(4) through 507(a)(10) and 503(b)(9) of the Bankruptcy Code) and unsecured non-priority claims (the holder of any such claim, the "Claimant"), must properly file a Proof of Claim on or before the applicable Bar Date in order to share in the Debtors' estates.

6. All Claimants must submit (by overnight mail, courier service, hand delivery, regular mail, in person or electronically through the online Proof of Claim Form available at https://www.kccllc.net/structurlam an original, written proof of claim that substantially conforms to the Proof of Claim Form so as to be *actually received* by KCC, the Debtors' claims and noticing agent, by no later than 5:00 p.m. (*prevailing* Eastern Time) on or before the applicable Bar Date at the following address:

Structurlam Mass Timber Claims Processing Center c/o KCC 222 N. Pacific Coast Highway, Suite 300 El Segundo, CA 90245

- 7. A Proof of Claim must satisfy all of the following requirements to be considered properly and timely filed in these Chapter 11 Cases:
 - a. be <u>actually received</u> by KCC as described in <u>paragraph 6</u> above by overnight mail, courier service, hand delivery, regular mail, in person or electronically through the Proof of Claim Form available on KCC's website on or before the applicable Bar Date;
 - b. be signed by the Claimant;
 - c. be written in the English language;
 - d. be denominated in lawful currency of the United States or Canada;
 - e. conform substantially to the Proof of Claim Form or Official Form 410;
 - f. specify the Debtor against which the Proof of Claim is filed as well as the bankruptcy case number corresponding to such Debtor;
 - g. set forth with specificity the legal and factual basis for the alleged claim; and

- h. include supporting documentation or an explanation as to why such documentation is not available.
- 8. Proofs of Claim sent to KCC by facsimile, telecopy, or electronic mail will <u>not</u> be accepted and will <u>not</u> be considered properly or timely filed for any purpose in these Chapter 11 Cases.
- 9. A Proof of Claim filed under the joint administration case number (23-10497 (CTG)), or otherwise without identifying a specific Debtor, will be deemed as filed only against debtor Structurlam Mass Timber U.S., Inc. To the extent more than one Debtor is listed on the Proof of Claim, such claim may be treated as if filed only against the first-listed Debtor.
- 10. Notwithstanding the above, holders of the following claims are <u>not</u> required to file a Proof of Claim on or before the applicable Bar Date solely with respect to such claim:
 - a. a claim against the Debtors for which a signed proof of claim has already been properly filed with the Clerk of the Bankruptcy Court for the District of Delaware or KCC in a form substantially similar to Official Bankruptcy Form No. 410;
 - b. a claim that is listed on the Debtors' schedules of assets and liabilities (collectively, the "Schedules") if and only if (i) such claim is not scheduled as "disputed," "contingent," or "unliquidated," (ii) the holder of such claim agrees with the amount, nature and priority of the claim as set forth in the Schedules <u>and</u> (iii) the holder of such claim agrees that the claim is an obligation of the specific Debtor that listed the claim in its Schedules;
 - c. an administrative expense claim allowable under sections 503(b) and 507(a)(2) of the Bankruptcy Code as an expense of administration (other than any claim allowable under section 503(b)(9) of the Bankruptcy Code);
 - d. an administrative expense claim for post-petition fees and expenses incurred by any professional allowable under sections 330, 331, and 503(b) of the Bankruptcy Code;
 - e. a claim that has been paid in full by the Debtors in accordance with the Bankruptcy Code or an order of this Court;
 - f. a claim that has been allowed by an order of this Court entered on or before the applicable Bar Date;

- g. a claim of any Debtor against another Debtor;
- h. any fees payable to the U.S. Trustee under 28 U.S.C. § 1930 or accrued interest thereon arising under 31 U.S.C. § 3717; and
- i. a claim for which specific deadlines have been fixed by an order of this Court entered on or before the applicable Bar Date.
- 11. Any Claimant exempted from filing a Proof of Claim pursuant to Paragraph 10 above must still properly and timely file a Proof of Claim for any other claim that does not fall within the exemptions provided by Paragraph 10 above.
- 12. Any person or entity holding an equity security (as defined in section 101(16) of the Bankruptcy Code and including, without limitation, common stock, preferred stock, warrants, or stock options) or other ownership interest in the Debtors (an "Interest Holder") is not required to file a proof of interest on or before the applicable Bar Date; provided, however, that the Debtors reserve the right to seek relief at a later date establishing a deadline for Interest Holders to file proofs of interest.
- 13. Any person or entity that holds a claim that arises from the rejection of an executory contract or unexpired lease must file a Proof of Claim based on such rejection by the later of (a) the General Bar Date or (b) 5:00 p.m. (prevailing Eastern Time) on the date that is twenty-one (21) days following service of an order approving rejection of any executory contract or unexpired lease of the Debtors.
- 14. If the Debtors amend their Schedules, then the deadline to submit a Proof of Claim for those creditors affected by any such amendment shall be the later of (i) the applicable Bar Date or (ii) 5:00 p.m. (*Prevailing* Eastern Time) on the date that is twenty-one (21) days from the date that the Debtors provide written notice to the affected creditor that the Schedules have been amended.

- 15. Not later than a date that is three (3) business days after entry of the Bar Date Order, the Debtors shall serve the Bar Date Notice, together with a copy of the Proof of Claim Form, by first class United States mail, postage prepaid (or equivalent service), to the following parties:
 - a. all known holders of potential claims and their counsel (if known), including all persons and entities listed in the Schedules at the addresses set forth therein (or any more recent address that is known to the Debtors), as potentially holding claims;
 - b. all parties that have requested notice of the proceedings in these Chapter 11 Cases pursuant to Bankruptcy Rule 2002 as of the date of the Bar Date Order;
 - c. all parties that have filed proofs of claim in these Chapter 11 Cases as of the date of the Bar Date Order;
 - d. all known parties to executory contracts and unexpired leases with the Debtors;
 - e. all known parties to litigation with the Debtors;
 - f. the District Director of the Internal Revenue Service for the District of Delaware;
 - g. all other known taxing authorities for the jurisdictions in which the Debtors maintain or conduct business;
 - h. the Securities and Exchange Commission;
 - i. the Environmental Protection Agency and similar state environmental agencies for states in which the Debtors conduct business;
 - j. the United States Attorney for the District of Delaware;
 - k. the Federal Communications Commission;
 - 1. the U.S. Trustee for the District of Delaware;
 - m. all other entities listed on the Debtors' matrix of creditors;
 - n. all persons who were employees of the Debtors as of the Petition Date, or within six months prior to the Petition Date; and
 - o. counsel (if known) to any of the foregoing.

- 16. In accordance with Bankruptcy Rule 2002(a)(7), service of the Bar Date Notice and Proof of Claim Form in the manner set forth in this Order is and shall be deemed to be good and sufficient notice of the Bar Date to known Claimants.
- 17. Pursuant to Bankruptcy Rule 2002(I), the Debtors shall cause the Publication Notice to be published once in one or more regional or local newspapers, trade journals, or similar publications, if any, as the Debtors deem appropriate, as soon as practicable after entry of this Order but no later than twenty-one (21) days before the General Bar Date. Such form and manner of publication notice is hereby approved and authorized and is and shall be deemed to be good and sufficient notice of the Bar Dates to unknown Claimants.
- 18. Properly filing an original, written Proof of Claim that substantially conforms to the Proof of Claim Form shall be deemed to satisfy the procedural requirements for the assertion of administrative priority claims under section 503(b)(9) of the Bankruptcy Code; *provided*, *however*, that all other administrative claims under section 503(b) of the Bankruptcy Code must be made by separate requests for payment in accordance with section 503(a) of the Bankruptcy Code and will not be deemed proper if made by Proof of Claim.
- 19. Claimants filing claims under section 503(b)(9) of the Bankruptcy Code shall attach to the Proof of Claim a supplemental statement setting forth with specificity: (i) the date of shipment of the goods the Claimant contends the Debtors received in the twenty (20) days before the Petition Date; (ii) the date, place, and method (including carrier name) of delivery of the goods the Claimant contends the Debtors received in the twenty (20) days before the Petition Date; (iii) the value of the goods the Claimant contends the Debtors received in the twenty (20) days before the Petition Date; and (iv) whether the Claimant timely made a demand to reclaim such goods

under section 546(c) of the Bankruptcy Code, including any documentation identifying such demand.

- 20. Pursuant to Bankruptcy Rule 3003(c)(2), absent further order of the Court to the contrary, any Claimant that is required to file a Proof of Claim in these Chapter 11 Cases pursuant to the Bankruptcy Code, the Bankruptcy Rules or this Order with respect to a particular claim against the Debtors, but that fails to do so properly by the applicable Bar Date, shall not be treated as a creditor in these Chapter 11 Cases with respect to such claim for purposes of voting and distribution; provided, however, that a holder of a claim shall be treated as a creditor for purposes of voting and distribution as to any undisputed, noncontingent and liquidated claims identified in the Schedules on behalf of such holder, in the amount set forth in the Schedules.
- 21. Nothing contained in this Order, the Publication Notice, or the Bar Date Notice is intended or shall be construed as a waiver of any of the Debtors' rights, including, without limitation, their rights to: (a) dispute, or assert offsets or defenses against, any filed claim or any claim listed or reflected in the Schedules as to the nature, amount, liability, or classification thereof; (b) subsequently designate any scheduled claim as disputed, contingent, or unliquidated; or (c) otherwise amend or supplement the Schedules. In addition, nothing contained in this Order, the Publication Notice or the Bar Date Notice is intended or shall be construed as an admission of the validity of any claim against the Debtors or an approval, assumption or rejection of any agreement, contract, or lease under section 365 of the Bankruptcy Code. All such rights and remedies are reserved.
- 22. The provisions of this Order apply to all claims of whatever character or nature against the Debtors or their assets, whether secured or unsecured, priority or non-priority, liquidated or unliquidated, fixed or contingent.

207

- 23. All Claimants who desire to rely on the Schedules with respect to filing a proof of claim in these Chapter 11 Cases shall have the sole responsibility for determining that their respective claim is accurately listed therein.
- 24. Notwithstanding the possible application of Bankruptcy Rules 6004(h), 7062 or 9014, the terms and conditions of this Order shall be immediately effective and enforceable upon its entry.
- 25. The Debtors and KCC are authorized to take all actions necessary or appropriate to effectuate the relief granted pursuant to this Order.
- 26. The Court shall retain jurisdiction with respect to any matters, claims, rights or disputes arising from or related to this Order.

Dated: May 25th, 2023 Wilmington, Delaware CRAIG T. GOLDBLATT
UNITED STATES BANKRUPTCY JUDGE

EXHIBIT 1

IN THE UNITED STATES BANKRUPTCY COURT FOR THE DISTRICT OF DELAWARE

)
In re:) Chapter 11
STRUCTURLAM MASS TIMBER U.S., INC., et al.,1) Case No. 23-10497 (CTG)
) (Jointly Administered)
Debtors.) Related Docket No.
)

NOTICE OF DEADLINE FOR THE FILING OF PROOFS OF CLAIM, INCLUDING FOR CLAIMS ASSERTED UNDER SECTION 503(b)(9) OF THE BANKRUPTCY CODE

(GENERAL BAR DATE IS JULY 10, 2023 AT 5:00 P.M. (PREVAILING EASTERN TIME))

TO: ALL HOLDERS OF POTENTIAL CLAIMS AGAINST THE DEBTORS (AS LISTED BELOW)

PLEASE TAKE NOTICE that on April 21, 2023 (the "Petition Date"), Structurlam Mass Timber U.S., Inc. and its affiliates, the debtors and debtors-in-possession (collectively, the "Debtors") filed voluntary petitions for relief under chapter 11 of the Bankruptcy Code (the "Chapter 11 Cases") with the United States Bankruptcy Court for the District of Delaware (the "Court").

PLEASE TAKE FURTHER NOTICE that on _______, 2023, the Court entered an order (the "Bar Date Order") establishing July 10, 2023 at 5:00 p.m. (prevailing Eastern Time) (the "General Bar Date") as the last date and time for each person or entity to file a proof of claim in the Chapter 11 Cases (the "Proof of Claim" or "Proofs of Claims," as applicable); provided that, solely with respect to a governmental unit, the last date and time for such governmental unit to file a Proof of Claim in the Chapter 11 Cases is October 18, 2023 at 5:00 p.m. (prevailing Eastern Time) (the "Governmental Bar Date," and together with the General Bar Date, the "Bar Dates").

For your convenience, enclosed with this Notice is a proof of claim form (the "**Proof of Claim Form**"). To the extent your claim is listed in the Debtors' schedules of assets and liabilities

The Debtors in these chapter 11 cases, along with the last four digits of each Debtors' federal tax identification number or Canadian business number, as applicable, include: Structurlam Mass Timber U.S., Inc. (6287); Natural Outcomes, LLC (n/a); Structurlam Mass Timber Corporation (5050); and SLP Holdings Ltd. (3114). The location of the Debtors' headquarters is: 2176 Government Street, Penticton, British Columbia, Canada V2A 8B5. The address of the registered agent for Structurlam Mass Timber U.S., Inc. is: 8 The Green, Suite A, Dover, Delaware 19901.

filed in these Chapter 11 Cases (collectively, the "Schedules"), such Proof of Claim Form identifies on its face the amount, nature and classification of your claim in the Schedules. A copy of the Debtors' Schedules can be viewed free of charge on the website of the Debtors' Claims Agent, KCC at https://www.kccllc.net/structurlam, under the Tab "Schedules & Statements". If you disagree with the amount, nature and classification of your claim as set forth on the enclosed Proof of Claim Form, you must file a Proof of Claim by the applicable Bar Date, in writing, at Structurlam Mass Timber Claims Processing Center, c/o KCC, 222 N. Pacific Coast Highway, Suite 300, El Segundo, CA 90245, or online at https://www.kccllc.net/structurlam. Additional Proof of Claim forms can also be found online at https://www.kccllc.net/structurlam under the Tab "Claim Form". Such Proofs of Claim should be prepared and filed in accordance with the procedures set forth herein.

PLEASE TAKE FURTHER NOTICE that the Bar Date Order and the procedures set forth therein and herein for the filing of Proofs of Claim apply to all claims (the holder of any such claim, a "Claimant") against the Debtors that arose, or are deemed to have arisen, prior to the Petition Date, regardless of their character or nature, whether secured or unsecured, priority or non-priority, liquidated or unliquidated, fixed or contingent, including, without limitation, claims entitled to administrative priority status under section 503(b)(9) of the Bankruptcy Code.²

As used in this Notice, the term "creditor" has the meaning given to it in section 101(10) of the Bankruptcy Code, and includes all persons, entities, estates, trusts, governmental units and the United States Trustee. In addition, the terms "persons," "entities," and "governmental units" are defined in sections 101(41), 101(15) and 101(27) of the Bankruptcy Code, respectively.

As used in this Notice, the term "claim" or "Claim" has the meaning given to it in section 101(5) of the Bankruptcy Code, and includes as to or against the Debtors: (a) any right to payment, whether or not such right is reduced to judgment, liquidated, unliquidated, fixed, contingent, matured, unmatured, disputed, undisputed, legal, equitable, secured, or unsecured; or (b) any right to an equitable remedy for breach of performance if such breach gives rise to a right to payment, whether or not such right to an equitable remedy is reduced to judgment, fixed, contingent, matured, unmatured, disputed, undisputed, secured, or unsecured.

YOU ARE RECEIVING THIS NOTICE BECAUSE YOU MAY HAVE OR YOU MAY ASSERT A CLAIM AGAINST THE DEBTORS IN THE ABOVE-CAPTIONED CHAPTER 11 CASES. THEREFORE, YOU SHOULD READ THIS NOTICE CAREFULLY AND DISCUSS IT WITH YOUR ATTORNEY. IF YOU DO NOT HAVE AN ATTORNEY, YOU MAY WISH TO CONSULT ONE.

General Information about the Debtors' Cases. The Debtors' cases are being jointly administered under case number 23-10497 (CTG).

<u>Individual Debtor Information</u>. The last four digits of each Debtor's federal tax identification number are set forth below. The Debtors' mailing address is 2176 Government Street, Penticton,

- 2 -

A claim arising under section 503(b)(9) of the Bankruptcy Code is a claim arising from the value of goods received by the Debtors within twenty (20) days before the Petition Date, provided that the goods were sold to the Debtors in the ordinary course of business.

British Columbia, Canada V2A 8B5.

<u>Debtor</u>	CASE No.	LAST 4 DIGITS <u>of EID #</u>
Structurlam Mass Timber U.S., Inc.	23-10497	6287
Natural Outcomes, LLC	23-10498	N/A
Structurlam Mass Timber Corporation	23-10499	5050
SLP Holdings Ltd.	23-10500	Business No: 781693114 (Canada)

A CLAIMANT SHOULD CONSULT AN ATTORNEY IF THE CLAIMANT HAS ANY QUESTIONS, INCLUDING WHETHER SUCH CLAIMANT SHOULD FILE A PROOF OF CLAIM.

1. PERSONS OR ENTITIES WHO MUST FILE A PROOF OF CLAIM.

Any person or entity that has or seeks to assert a claim against the Debtors which arose, or is deemed to have arisen, prior to the Petition Date, including, without limitation, a claim under section 503(b)(9) of the Bankruptcy Code, MUST FILE A PROOF OF CLAIM ON OR BEFORE THE APPLICABLE BAR DATE in order to potentially share in the Debtors' estates, except as set forth in (A) below.

Under the Bar Date Order, the filing of a Proof of Claim Form shall be deemed to satisfy the procedural requirements for the assertion of administrative priority claims under section 503(b)(9) of the Bankruptcy Code. All other administrative claims under section 503(b) of the Bankruptcy Code must be made by separate requests for payment in accordance with section 503(a) of the Bankruptcy Code and shall not be deemed proper if made by proof of claim. No deadline has yet been established for the filing of administrative claims other than claims under section 503(b)(9) of the Bankruptcy Code. Claims under section 503(b)(9) of the Bankruptcy Code must be filed by the Bar Date.

Acts or omissions of the Debtors that occurred or arose before the Petition Date may give rise to Claims against the Debtors that must be filed by the applicable Bar Date, notwithstanding that such Claims may not have matured, are contingent or have not become fixed or liquidated prior to or as of the Petition Date.

THE FACT THAT YOU HAVE RECEIVED THIS NOTICE DOES NOT MEAN THAT YOU HAVE A CLAIM OR THAT THE DEBTORS BELIEVE THAT YOU HAVE A CLAIM. A CLAIMANT SHOULD CONSULT AN ATTORNEY IF THE CLAIMANT HAS ANY QUESTIONS, INCLUDING WHETHER SUCH CLAIMANT SHOULD FILE A PROOF OF CLAIM.

A. Claims For Which No Proof of Claim is Required to be Filed.

Notwithstanding the above, holders of the following claims are **NOT** required to file a Proof of Claim on or before the applicable Bar Date solely with respect to such claim:

- a. a claim against the Debtors for which a signed proof of claim has already been properly filed with the Clerk of the Bankruptcy Court for the District of Delaware or Kurtzman Carson Consultants LLC ("KCC") in a form substantially similar to Official Bankruptcy Form No. 410;
- b. a claim that is listed on the Debtors' Schedules if and only if (i) such claim is not scheduled as "disputed," "contingent," or "unliquidated," (ii) the holder of such claim agrees with the amount, nature and priority of the claim as set forth in the Schedules, <u>and</u> (iii) the holder of such claim agrees that the claim is an obligation of the specific Debtor that listed the claim in its Schedules;
- c. an administrative expense claim allowable under sections 503(b) and 507(a)(2) of the Bankruptcy Code as an expense of administration (other than any claim allowable under section 503(b)(9) of the Bankruptcy Code);
- d. an administrative expense claim for post-petition fees and expenses incurred by any professional allowable under sections 330, 331, and 503(b) of the Bankruptcy Code;
- e. a claim that has been paid in full by the Debtors in accordance with the Bankruptcy Code or an order of this Court;
- f. a claim that has been allowed by an order of this Court entered on or before the applicable Bar Date;
- g. a claim of any Debtor against another Debtor;
- h. any fees payable to the U.S. Trustee under 28 U.S.C. § 1930 or accrued interest thereon arising under 31 U.S.C. § 3717; and
- i. a claim for which specific deadlines have been fixed by an order of this Court entered on or before the applicable Bar Date.

Please take notice that any Claimant exempted from filing a Proof of Claim pursuant to paragraph A above <u>must still properly and timely file</u> a Proof of Claim for any other claim that does not fall within the exemptions provided by paragraph A above. As set forth in clause (e) above, creditors are not required to file a proof of claim with respect to any amounts that have already been paid by the Debtors.

B. No Bar Date for Proof of Interest.

Any person or entity holding an equity security (as defined in section 101(16) of the Bankruptcy Code and including, without limitation, common stock, preferred stock, warrants, or stock options) or other ownership interest in the Debtors (an "Interest Holder") is not required to file a proof of interest on or before the applicable Bar Date; provided, however, that the Debtors have reserved the right to establish at a later time a bar date requiring Interest Holders to file proofs of interest. If such a bar date is established, Interest Holders will be notified in writing of the bar date for filing of proofs of interest at the appropriate time.

C. Claims Arising from Rejected Executory Contracts or Unexpired Leases.

Any person or entity that holds a claim that arises from the rejection of an executory contract or unexpired lease must file a Proof of Claim based on such rejection by the later of (a) the General Bar Date or (b) 5:00 p.m. (*Prevailing* Eastern Time) on the date that is twenty-one (21) days following service of an order approving rejection of any executory contract or unexpired lease of the Debtors.

D. Amendment to the Debtors' Schedules.

If the Debtors amend their Schedules, then the deadline to submit a Proof of Claim for those creditors affected by any such amendment shall be the later of (i) the applicable Bar Date or (ii) 5:00 p.m. (*Prevailing* Eastern Time) on the date that is twenty-one (21) days from the date that the Debtors provide written notice to the affected creditor that the Schedules have been amended.

2. WHEN AND WHERE TO FILE.

All Claimants must submit (by overnight mail, courier service, hand delivery, regular mail, or in person) an original, written Proof of Claim that substantially conforms to the Proof of Claim Form so as to be *actually received* by KCC, the Debtors' claims and notice agent, by no later than 5:00 p.m. (*Prevailing* Eastern Time) on or before the applicable Bar Date at the following address:

Structurlam Mass Timber Claims Processing Center c/o KCC

222 N. Pacific Coast Highway, Suite 300
El Segundo, CA 90245

Alternatively, Claimants may submit a Proof of Claim electronically by completing the Proof of Claim Form that can be accessed at KCC's website, https://www.kccllc.net/structurlam, under the Tab "Submit Electronic Proof of Claim (ePOC)".

Proofs of Claim will be deemed timely filed only if *actually received* by KCC on or before the applicable Bar Date. Proofs of Claim may <u>not</u> be delivered by facsimile, telecopy, or electronic mail transmission. Any facsimile, telecopy, or electronic mail submissions will not be accepted and will not be deemed filed until a proof of claim is submitted to KCC by overnight mail, courier service, hand delivery, regular mail, in person, or through KCC's website listed above.

Claimants wishing to receive acknowledgment that their Proofs of Claim were received by KCC must submit (i) a copy of the Proof of Claim and (ii) a self-addressed, stamped envelope (in addition to the original Proof of Claim sent to KCC).

3. CONTENTS OF A PROOF OF CLAIM.

As noted above, the Debtors are enclosing a Proof of Claim Form for use in these Chapter 11 Cases, or you may use another proof of claim form that substantially conforms to Official Bankruptcy Form No. 410. The Proof of Claim Form is available free of charge on KCC's website, https://www.kccllc.net/structurlam, under the Tab "Claim Form."

To be valid, your Proof of Claim MUST (i) be signed by the Claimant; (ii) be written in the English language; (iii) be denominated in lawful currency of the United States or Canada; (iv) conform substantially to the Proof of Claim Form or Official Form 410; (v) specify the Debtor against which the Proof of Claim is filed as well as the bankruptcy case number corresponding to such Debtor; (vi) set forth with specificity the legal and factual basis for the alleged claim; and (vii) include supporting documentation or an explanation as to why such documentation is not available.

Your Proof of Claim must identify the specific Debtor entity. A Proof of Claim filed under the joint administration case number (23-10497 (CTG)), or otherwise without identifying a specific Debtor, will be deemed as filed only against debtor Stucturlam Mass Timber U.S., Inc. To the extent more than one Debtor is listed on the Proof of Claim, such claim may be treated as if filed only against the first-listed Debtor.

If you are filing a Claim under section 503(b)(9) of the Bankruptcy Code, you <u>must</u> indicate in the Proof of Claim Form available on KCC's website at https://www.kccllc.net/structurlam the amount of the Claim that arises under section 503(b)(9) of the Bankruptcy Code. For each Claim under section 503(b)(9) of the Bankruptcy Code, you <u>must</u> attach to the Proof of Claim Form a supplemental statement setting forth with specificity: (i) the date of shipment of the goods you contend the Debtors received in the twenty (20) days before the Petition Date; (ii) the date, place, and method (including carrier name) of delivery of the goods you contend the Debtors received in the twenty (20) days before the Petition Date; (iii) the value of the goods you contend the Debtors received in the twenty (20) days before the Petition Date; and (iv) whether you timely made a demand to reclaim such goods under section 546(c) of the Bankruptcy Code, and, if so, include any documentation identifying such demand.

4. CONSEQUENCES OF FAILURE TO FILE PROOF OF CLAIM BY THE BAR DATE.

Absent order of the Court to the contrary, any Claimant that is required to file a Proof of Claim in these Chapter 11 Cases pursuant to the Bankruptcy Code, the Bankruptcy Rules, or the Bar Date Order with respect to a particular claim against the Debtors, but that fails to do so properly by the applicable Bar Date, shall not be treated as a creditor in these Chapter 11 Cases with respect to such claim for purposes of voting and distribution; *provided*, *however*, that a holder of a claim shall be treated as a creditor for purposes of voting and distribution as to any undisputed,

noncontingent and liquidated claims identified in the Schedules on behalf of such holder, in the amount set forth in the Schedules.

5. CONTINGENT CLAIMS.

Acts or omissions of or by the Debtors that occurred, or that are deemed to have occurred, prior to the Petition Date, including, without limitation, acts or omissions related to any indemnity agreement, guarantee, services provided to or rendered by the Debtors, or goods provided to or by the Debtors, may give rise to claims against the Debtors notwithstanding the fact that such claims (or any injuries on which they may be based) may be contingent or may not have matured or become fixed or liquidated prior to the Petition Date. Therefore, any person or entity that holds a claim or potential "claim," as that term is defined in section 101(5) of the Bankruptcy Code, against the Debtors, even if contingent, or unliquidated, MUST file a Proof of Claim on or before the applicable Bar Date.

6. THE DEBTORS' SCHEDULES.

You may be listed as the holder of a claim against the Debtors in the Schedules. The Schedules are available free of charge on KCC's website at https://www.kccllc.net/structurlam, under the Tab "Schedules & Statements". If you rely on the Schedules, it is your responsibility to determine that your claim is accurately listed in the Schedules. As described above, if (i) you agree with the nature, amount and priority of your claim as listed in the Schedules, and (ii) your claim is NOT described as "disputed," "contingent," or "unliquidated," and (iii) you agree that the claim is an obligation of the specific Debtor that listed the claim in its Schedules, then you are not required to file a Proof of Claim in these Chapter 11 Cases with respect to such claim. Otherwise, or if you decide to file a Proof of Claim, you must do so before the applicable Bar Date in accordance with the procedures set forth in this Notice and the Bar Date Order.

7. RESERVATION OF RIGHTS.

Nothing contained in this Notice or the Bar Date Order is intended or should be construed as a waiver of any of the Debtor's rights, including without limitation, their rights to: (a) dispute, or assert offsets or defenses against, any filed claim or any claim listed or reflected in the Schedules as to the nature, amount, liability, or classification thereof; (b) subsequently designate any scheduled claim as disputed, contingent, or unliquidated; or (c) otherwise amend or supplement the Schedules. Please see paragraph 1.D above regarding the time to file a Proof of Claim relating to any amendment to the Schedules. In addition, nothing contained herein is intended or should be construed as an admission of the validity of any claim against the Debtors or an approval, assumption, or rejection of any agreement, contract, or lease under section 365 of the Bankruptcy Code. All such rights and remedies are reserved.

8. ADDITIONAL INFORMATION.

The Schedules, the Proof of Claim Form, this Notice, and the Bar Date Order are available free of charge on KCC's website at https://www.kccllc.net/structurlam, under the Tab "Court Docket" or "Claim Form", as applicable. If you have questions concerning the filing or processing of Claims, you may contact the Debtors' claims agent, KCC, at (888) 647-1715. If you require

additional information regarding the filing of a Proof of Claim, you may contact counsel for the Debtors in writing at the addresses below.

such other and further relief as may be just and proper.

Dated: _____, 2023 Wilmington, Delaware CHIPMAN BROWN CICERO & COLE, LLP

/s/ DRAFT

William E. Chipman, Jr. (No. 3818)

Robert A. Weber (No. 4013)

Mark L. Desgrosseilliers (No. 4083)

Mark D. Olivere (No. 4291)

Hercules Plaza

1313 North Market Street, Suite 5400

Wilmington, Delaware 19801

Telephone:

(302) 295-0191

Email:

chipman@chipmanbrown.com

weber@chipmanbrown.com desgross@chipmanbrown.com olivere@chipmanbrown.com

Proposed Counsel to Debtors and Debtors in Possession

EXHIBIT 2

Filed on MM / DD / YYYY

Case 23-10497-CTG Doc 174-1 Filed 05/25/23 Page 11 of 19

Your claim can be filed electronically on KCC's website at https://epoc.kccllc.net/structurlam

		Unit	ted States Bankruptcy	y Court for the District	of Delaware			
_	Indicate Deb	tor against which you assert	t a claim by checking	the appropriate box b	elow. (Check only	one Debtor pe	r claim form.)	
	☐ Structuriam Mass Timber U.S ☐ Natural Outcomes, LLC (C		☐ Structurlam Ma:	ss Timber Corporation (C	ase No. 23-10499)	☐ SLP Holdi	ngs Ltd. (Case No	o, 23-10500)
	icial Form 410							
_	oof of Claim the instructions before							04/22
rtg ola er	s must leave out or reda ments that support the cla gages, and security agree hin in an attachment. I son who files a fraudulen mall the information abo	aim, such as promissory ements. Do not send ori at claim could be fined up	notes, purchase of ginal documents; o to \$500,000, impr	rders, invoices, iten ; they may be destr risoned for up to 5 y	nized statements oyed after scann	of running ac ing. If the doc	uments are not	available,
	art 1: Identify the Clair				····			
Pa		Mame of the current credit	or (the person or entit	y to be paid for this cla				
Pa	Identify the Clair Who is the current	Name of the current creditor to the creditor to the No	tor (the person or entit used with the debtor	y to be paid for this cla				
Pa 1.	Who is the current creditor? Has this claim been acquired from someone else? Where should notices and payments to the	Name of the current creditor to the creditor to the No	or (the person or entit used with the debtor om?	y to be paid for this cla			to the creditor	be sent? (if
2.	Who is the current creditor? Has this claim been acquired from someone else? Where should notices and	Name of the current creditor to the current creditor to the creditor to the current creditor creditor creditor creditor credit	or (the person or entit used with the debtor om?	y to be paid for this cla	Where shou		to the creditor	be sent? (if
Pa	Who is the current creditor? Has this claim been acquired from someone else? Where should notices and payments to the creditor be sent? Federal Rule of Bankruptcy Procedure	Name of the current creditor to the creditor t	or (the person or entit used with the debtor om?	y to be paid for this cla	Where shou different) Name		to the creditor	
2.	Who is the current creditor? Has this claim been acquired from someone else? Where should notices and payments to the creditor be sent? Federal Rule of	Name of the current creditor to Other names the creditor to No Yes. From who Where should notices	or (the person or entit used with the debtor om?	y to be paid for this cla	Where shou different) Name	ld payments	to the creditor	
Pa 1.	Who is the current creditor? Has this claim been acquired from someone else? Where should notices and payments to the creditor be sent? Federal Rule of Bankruptcy Procedure	Name of the current creditor to Other names the creditor to No	tor (the person or entity used with the debtor om?	y to be paid for this cla	Where shou different) Name	ld payments		be sent? (if
Pa	Who is the current creditor? Has this claim been acquired from someone else? Where should notices and payments to the creditor be sent? Federal Rule of Bankruptcy Procedure	Name of the current creditor to Other names the creditor to No Yes. From who Where should notices Name	tor (the person or entity used with the debtor om?	y to be paid for this cla	Where shou different) Name Number City	Id payments		
2.	Who is the current creditor? Has this claim been acquired from someone else? Where should notices and payments to the creditor be sent? Federal Rule of Bankruptcy Procedure	Name of the current creditor to Other names the creditor to No No Yes. From who Where should notices Name Number Street City Country	tor (the person or entity used with the debtor om?	y to be paid for this cla	Where shou different) Name Number City Country	Id payments		

Yes. Claim number on court claims registry (if known)

Yes. Who made the earlier filing?

filed?

5. Do you know if

this claim?

anyone else has filed a proof of claim for ☐ No

6.	Do you have any number you use to identify the debtor?	□ No					
		Yes. Last 4 digits of the debtor's account or any number you use to identify the debtor:					
	How much is the claim?						
		\$ Does this amount include interest or other charges?					
		Ŭ No					
		Yes. Attach statement itemizing interest, fees, expenses, or other charges required by Bankruptcy Rule 3001(c)(2)(A).					
3.	What is the basis of the claim?	ples: Goods sold, money loaned, lease, services performed, personal injury or wrongful death, or credit card.					
	Giaiiii:	Attach redacted copies of any documents supporting the claim required by Bankruptcy Rule 3001(c).					
	,	Limit disclosing information that is entitled to privacy, such as health care information.					
Э.	Is all or part of the claim	□ No					
	secured?	Yes. The claim is secured by a lien on property.					
		Nature of property:					
		Real estate: If the claim is secured by the debtor's principal residence, file a Mortgage Proof of Claim Attachment (Official Form 410-A) with this Proof of Claim.					
		Motor vehicle					
		Other. Describe:					
		· · · · · · · · · · · · · · · · · · ·					
		Basis for perfection: Attach redacted copies of documents, if any, that show evidence of perfection of a security interest (for example, a mortgage, lien, certificate of title, financing statement, or other document that shows the lien has been filed or recorded.)					
		Value of property: \$					
-		Amount of the claim that is secured: \$					
	•	Amount of the claim that is unsecured: \$(The sum of the secured and unsecured amount should match the amount in line 7.					
		Amount necessary to cure any default as of the date of the petition: \$					
	•	Annual Interest Rate (when case was filed)% Fixed					
		☐ Variable					
10	. Is this claim based on a	□ No					
	lease?	Yes. Amount necessary to cure any default as of the date of the petition.					
11	. Is this claim subject to a	□ No					
	right of setoff?	Yes. Identify the property:					

Case 23-10497-CTG Doc 174-1 Filed 05/25/23 Page 13 of 19

Is all or part of the claim entitled to priority under	□ No				
11 U.S.C. § 507(a)?	Yes. Check	call that apply:		•	Amount entitled to priority
A claim may be partly priority and partly	Domes 11 U.S	etic support obligations (in .C. § 507(a)(1)(A) or (a)(cluding alimony and c 1)(B).	hild support) under	\$
nonpriority. For example, in some categories, the law limits the amount	Up to \$	\$3,350* of deposits towar es for personal, family, or	rd purchase, lease, or household use. 11 U	rental of property or .S.C. § 507(a)(7).	\$
entitled to priority.	Wages days b	s, salaries, or commission efore the bankruptcy pet ever is earlier. 11 U.S.C.	ns (up to \$15,150*) ea ition is filed or the del	arned within 180	\$
		or penalties owed to gove		S.C. § 507(a)(8).	· \$
	_	butions to an employee b			\$
	· <u>=</u>	Specify subsection of 11			\$
	. —				un on or after the date of adjustment
3. Is all or part of the claim pursuant to 11 U.S.C. § 503(b)(9)?	days befor the ordina	ate the amount of your cl re the date of commence ry course of such Debtor	ment of the above ca	se, in which the good	ceived by the debtor within 20 s have been sold to the Debtor ing such claim.
	\$				<u>.</u>
Part 3: Sign Below					
he person completing	Check the approp	riate box:			
his proof of claim must sign and date it.	l am the cred	ditor.			
FRBP 9011(b).	l am the cred	ditor's attorney or authoriz	ed agent.		
f you file this claim electronically, FRBP 5005(a)(2) authorizes courts	I am the trus	itee, or the debtor, or thei	r authorized agent. Ba	nkruptcy Rule 3004.	
to establish local rules specifying what a signature	l am a guarantor, surety, endorser, or other codebtor. Bankruptcy Rule 3005. I understand that an authorized signature on this <i>Proof of Claim</i> serves as an acknowledgement that when calculating the amount of the claim, the creditor gave the debtor credit for any payments received toward the debt.				
is. A person who files a					
fraudulent claim could be fined up to \$500,000,	I have examined the information in this Proof of Claim and have reasonable belief that the information is true and correct				
imprisoned for up to 5 years, or both.	I declare under penalty of perjury that the foregoing is true and correct.				
years, or both. 18 U.S.C. §§ 152, 157, and 3571.	Executed on date	MM / DD / YYYY	_		
•			•	•	
	Signature				
	Print the name of	of the person who is cor	mpleting and signing	this claim:	
•					
	Name	First name	Middle name	La	st name
·	Title				
	1 IIIC				
	Company	Identify the corporate servi	cer as the company if the	authorized agent is a servi	cer.
	·	identity the corporate servi	oo, as the company it the c	and the second s	
	Address	Number Street			
N.		Number Street			
		City	Sta	te ZIP	Code Country
	Contact phone			Email	

Official Form 410

Instructions for Proof of Claim

United States Bankruptcy Court

12/15

These instructions and definitions generally explain the law. In certain circumstances, such as bankruptcy cases that debtors do not file voluntarily, exceptions to these general rules may apply. You should consider obtaining the advice of an attorney, especially if you are unfamiliar with the bankruptcy process and privacy regulations.

A person who files a fraudulent claim could be fined up to \$500,000, imprisoned for up to 5 years, or both. $18 \text{ U.S.C. } \S\S 152, 157 \text{ and } 3571$

How to fill out this form

- Fill in all of the information about the claim as of the date the case was filed.
- Fill in the caption at the top of the form
- If the claim has been acquired from someone else, then state the identity of the last party who owned the claim or was the holder of the claim and who transferred it to you before the initial claim was filed.
- Attach any supporting documents to this form.

 Attach redacted copies of any documents that show that the debt exists, a lien secures the debt, or both. (See the definition of *redaction* on the next page.)

Also attach redacted copies of any documents that show perfection of any security interest or any assignments or transfers of the debt. In addition to the documents, a summary may be added. Federal Rule of Bankruptcy Procedure (called "Bankruptcy Rule") 3001(c) and (d).

- Do not attach original documents because attachments may be destroyed after scanning.
- If the claim is based on delivery health care goods or services, do not disclose confidential health care information. Leave out or redact confidential information both in the claim and in the attached documents.

PLEASE SEND COMPLETED PROOF(S) OF CLAIM

Structurlam Mass Timber Claims Processing Center c/o KCC 222 N. Pacific Coast Hwy., Ste. 300 El Segundo, CA 90245

Alternatively, your claim can be filed electronically on KCC's website at https://epoc.kccllc.net/structurlam.

- A Proof of Claim form and any attached documents must show only the last 4 digits of any social security number, individual's tax identification number, or financial account number, and only the year of any person's date of birth. See Bankruptcy Rule 9037.
- For a minor child, fill in only the child's initials and the full name and address of the child's parent or guardian. For example, write A.B., a minor child (John Doe, parent, 123 Main St., City, State). See Bankruptcy Rule 9037.

Confirmation that the claim has been filed

To receive confirmation that the claim has been filed, either enclose a stamped self-addressed envelope and a copy of this form or you may view a list of filed claims in this case by visiting the Claims and Noticing and Agent's website at http://www.kccllc.net/structurlam

Understand the terms used in this form

Administrative expense: Generally, an expense that arises after a bankruptcy case is filed in connection with operating, liquidating, or distributing that bankruptcy estate.

11 U.S.C. § 503

Claim: A creditor's right to receive payment for a debt that the debtor owed on the date the debtor filed for bankruptcy. 11 U.S.C. §101 (5). A claim may be secured or unsecured.

Claim Pursuant to 11 U.S.C. §503(b)(9): A claim arising from the value of any goods received by the Debtor within 20 days before the date of commencement of the above case, in which the goods have been sold to the Debtor in the ordinary course of the Debtor's business. Attach documentation supporting such claim.

Creditor: A person, corporation, or other entity to whom a debtor owes a debt that was incurred on or before the date the debtor filed for bankruptcy. 11 U.S.C. §101 (10).

Debtor: A person, corporation, or other entity to who is in bankruptcy. Use the debtor's name and case number as shown in the bankruptcy notice you received. 11 U.S.C. §101 (13).

Evidence of perfection: Evidence of perfection of a security interest may include documents showing that a security interest has been filed or recorded, such as a mortgage, lien, certificate of title, or financing statement.

Information that is entitled to privacy: A *Proof of Claim* form and any attached documents must show only the last 4 digits of any social security number, an individual's tax identification number, or a financial account number, only the initials of a minor's name, and only the year of any person's date of birth. If a claim is based on delivering health care goods or services, limit the disclosure of the goods or services to avoid embarrassment or disclosure of confidential health care information. You may later be required to give more information if the trustee or someone else in interest objects to the claim.

Priority claim: A claim within a category of unsecured claims that is entitled to priority under 11 U.S.C. §507(a). These claims are paid from the available money or property in a bankruptcy case before other unsecured claims are paid. Common priority unsecured claims include alimony, child support, taxes, and certain unpaid wages.

Proof of claim: A form that shows the amount of debt the debtor owed to a creditor on the date of the bankruptcy filing. The form must be filed in the district where the case is pending.

Redaction of information: Masking, editing out, or deleting certain information to protect privacy. Filers must redact or leave out information entitled to **privacy** on the *Proof of Claim* form and any attached documents.

Do not file these instructions with your form.

Secured claim under 11 U.S.C. §506(a): A claim backed by a lien on particular property of the debtor. A claim is secured to the extent that a creditor has the right to be paid from the property before other creditors are paid. The amount of a secured claim usually cannot be more than the value of the particular property on which the creditor has a lien. Any amount owed to a creditor that is more than the value of the property normally may be an unsecured claim. But exceptions exist; for example, see 11 U.S.C. § 1322(b) and the final sentence of 1325(a).

Examples of liens on property include a mortgage on real estate a security interest in a car. A lien may be voluntarily granted by a debtor or may be obtained through a court proceeding. In states, a court judgment may be a lien.

Setoff: Occurs when a creditor pays itself with money belonging to the debtor that it is holding, or by canceling a debt it owes to the debtor.

Uniform claim identifier: An optional 24-character identifier that some creditors use to facilitate electronic payment.

Unsecured claim: A claim that does not meet the requirements of a secured claim. A claim may be unsecured in part to the extent that the amount of the claim is more than the value of the property on which a creditor has a lien.

Offers to purchase a claim

Certain entities purchase claims for an amount that is less than the face value of the claims. These entities may contact creditors offering to purchase their claims. Some written communications from these entities may easily be confused with official court documentation or communications from the debtor. These entities do not represent the bankruptcy court, the bankruptcy trustee, or the debtor. A creditor has no obligation to sell its claim. However, if a creditor decides to sell its claim, any transfer of that claim is subject to Bankruptcy Rule 3001(e), any provisions of the Bankruptcy Code (11 U.S.C. § 101 et seq.) that apply, and any orders of the bankruptcy court that apply.

EXHIBIT 3

IN THE UNITED STATES BANKRUPTCY COURT FOR THE DISTRICT OF DELAWARE

In re:) Chapter 11
STRUCTURLAM MASS TIMBER U.S., INC., et al., ¹) Case No. 23-10497 (CTG
)) (Jointly Administered)
Debtors.) Related Docket No.
)

NOTICE OF DEADLINE FOR THE FILING OF PROOFS OF CLAIM, INCLUDING FOR CLAIMS ASSERTED UNDER SECTION 503(b)(9) OF THE BANKRUPTCY CODE

(GENERAL BAR DATE IS JULY 10, 2023 AT 5:00 P.M. (PREVAILING EASTERN TIME)

PLEASE TAKE NOTICE OF THE FOLLOWING:

On April 21, 2023 (the "Petition Date"), Structurlam Mass Timber U.S., Inc. and its affiliates, the debtors and debtors-in-possession (collectively, the "Debtors"), filed voluntary petitions for relief under chapter 11 of the Bankruptcy Code with the United States Bankruptcy Court for the District of Delaware (the "Court"). On ______, 2023, the Court entered an order [Docket No. ____] (the "Bar Date Order") establishing certain deadlines for the filing of proofs of claim in the chapter 11 cases of the following Debtors:

<u>Debtor</u>	Case No.	LAST 4 DIGITS OF EID#
Structurlam Mass Timber U.S., Inc.	23-10497	6287
Natural Outcomes, LLC	23-10498	N/A
Structurlam Mass Timber Corporation	23-10499	5050
		Business No:
SLP Holdings Ltd.	23-10500	781693114 (Canada)

The Debtors in these chapter 11 cases, along with the last four digits of each Debtors' federal tax identification number or Canadian business number, as applicable, include: Structurlam Mass Timber U.S., Inc. (6287); Natural Outcomes, LLC (n/a); Structurlam Mass Timber Corporation (5050); and SLP Holdings Ltd. (3114). The location of the Debtors' headquarters is: 2176 Government Street, Penticton, British Columbia, Canada V2A 8B5. The address of the registered agent for Structurlam Mass Timber U.S., Inc. is: 8 The Green, Suite A, Dover, Delaware 19901.

Pursuant to the Bar Date Order, each person or entity (including, without limitation, each individual, partnership, joint venture, corporation, estate, and trust) that holds or seeks to assert a claim (as defined in section 101(5) of the Bankruptcy Code) against the Debtors that arose, or is deemed to have arisen, prior to the Petition Date (including, without limitation, claims entitled to administrative priority status under section 503(b)(9) of the Bankruptcy Code), MUST FILE A PROOF OF CLAIM on or before 5:00 p.m. (prevailing Eastern Time), on July 10, 2023 (the "General Bar Date"), by sending an original proof of claim form to Kurtzman Carson Consultants LLC ("KCC"), at the following address: Structurlam Mass Timber Claims Processing Center, c/o KCC. 222 N. Pacific Coast Highway, Suite 300, El Segundo, CA 90245, or by completing the online proof of claim form available at https://www.kccllc.net/structurlam, under Tab "Submit Electronic Proof of Claim (ePOC)" so that it is actually received on or before the General Bar Date; provided that, solely with respect to governmental units (as defined in section 101(27) of the Bankruptcy Code), the deadline for such governmental units to file a proof of claim against the Debtors is October 18, 2023 at 5:00 p.m. (prevailing Eastern Time) (the "Governmental Bar Date"). Proofs of claim must be sent by overnight mail, courier service, hand delivery, regular mail, or in person, or completed electronically through KCC's website. Proofs of claim sent by facsimile, telecopy, or electronic mail will not be accepted and will not be considered properly or timely filed for any purpose in these Chapter 11 Cases.

ABSENT FURTHER ORDER OF THE COURT, ANY PERSON OR ENTITY THAT IS REQUIRED TO FILE A PROOF OF CLAIM IN THESE CHAPTER 11 CASES WITH RESPECT TO A PARTICULAR CLAIM AGAINST THE DEBTORS, BUT THAT FAILS TO DO SO PROPERLY BY THE APPLICABLE BAR DATE, SHALL NOT BE TREATED AS A CREDITOR IN THESE CHAPTER 11 CASES WITH RESPECT TO SUCH CLAIM FOR PURPOSES OF VOTING AND DISTRIBUTION; PROVIDED, HOWEVER, THAT A HOLDER OF A CLAIM SHALL BE TREATED AS A CREDITOR FOR PURPOSES OF VOTING AND DISTRIBUTION AS TO ANY UNDISPUTED, NONCONTINGENT, AND LIQUIDATED CLAIMS IDENTIFIED IN THE SCHEDULES ON BEHALF OF SUCH HOLDER IN THE AMOUNT SET FORTH IN THE SCHEDULES.

The Schedules, the Proof of Claim Form, the Bar Date Notice, and the Bar Date Order may be obtained by contacting the Debtors' Claims Agent, in writing, at KCC, 222 N. Pacific Coast Highway, Suite 300, El Segundo, CA 90245 or online at https://www.kccllc.net/structurlam, under the Tab "Court Docket" or "Claim Form", as applicable. The Bar Date Order can also be viewed on the Court's website at www.deb.uscourts.gov. If you have questions concerning the filing or processing of claims, you may contact the Debtors' claims agent, KCC, at (888) 647-1715.

Dated: May ____, 2023 Wilmington, Delaware

CHIPMAN BROWN CICEDRO & COLE, LLP

William E. Chipman, Jr.

Robert A. Weber

Mark L. Desgrosseilliers

Mark D. Olivere

Hercules Plaza

1313 North Market Street, Suite 5400

Wilmington, DE 19801

Telephone: (302) 295-0191

Facsimile: (302) 295-0199

Email:

chipman@chipmanbrown.com

weber@chipmanbrown.com desgross@chipmanbrown.com olivere@chipmanbrown.com

ATTORNEYS FOR THE DEBTORS AND DEBTORS IN POSSESSION

This is Exhibit "____" referred to in the Affidavit of MICHÈLE HAY, sworn before me at Vancouver, BC, this 30th day of May, 2023

A Commissioner for taking Affidavits in British Columbia

IN THE UNITED STATES BANKRUPTCY COURT FOR THE DISTRICT OF DELAWARE

Debtors.) Re: Docket Nos. 16 and 87
) (Jointly Administered)
STRUCTURLAM MASS TIMBER U.S., INC., et al., Case No. 23-10497 (CTC
In re:) Chapter 11
1

ORDER (I) AUTHORIZING (A) SALE OF ASSETS FREE AND CLEAR OF ALL LIENS, CLAIMS, ENCUMBRANCES AND INTERESTS AND (B) THE DEBTORS' ASSUMPTION AND ASSIGNMENT OF CERTAIN EXECUTORY CONTRACTS AND UNEXPIRED LEASES; AND (II) GRANTING RELATED RELIEF

Upon the Debtors' Motion Seeking Entry of an Order (A)(I) Approving Bidding Procedures, (II) Approving Stalking Horse Protections and Debtors' Entry into Stalking Horse Purchase Agreement, (III) Scheduling the Bid Deadlines and the Auction, (IV) Scheduling Hearings and Objection Deadlines with Respect to the Sale, (V) Approving the Form and Manner of Notice Thereof, (VI) Approving Contract Assumption and Assignment Procedures, and (VII) Granting Related Relief and (B)(I) Approving the Sale of the Assets Free and Clear of All Liens, Claims, Interests, and Encumbrances, (II) Approving Assumption and Assignment of Executory Contracts and Unexpired Leases and (III) Granting Related Relief [Docket No. 16] (the "Motion")² of the above-captioned debtors and debtors in possession (collectively,

The Debtors in these Chapter 11 Cases, along with the last four digits of each Debtor's federal tax identification number or Canadian business number, as applicable, include: Structurlam Mass Timber U.S., Inc. (6287); Natural Outcomes, LLC (n/a); Structurlam Mass Timber Corporation (5050); and SLP Holdings Ltd. (3114); The location of the Debtors' headquarters is: 2176 Government Street, Penticton, British Columbia, Canada V2A 8B5. The address of the registered agent for Structurlam Mass Timber U.S., Inc. is: 8 The Green, Suite A, Dover, Delaware 19901.

² Capitalized terms used but not otherwise defined herein shall have the meanings ascribed to them in the Motion or the Purchase Agreement (as defined herein), as applicable; *provided* that in the event of any conflict with respect to the meaning of a capitalized term between the Motion and the Purchase Agreement, the meaning ascribed to such term in the Purchase Agreement shall control.

the "Debtors")³ for entry of an order (this "Order"), among other things, (a) authorizing the sale (the "Sale") of the Transferred Assets (as defined in the Purchase Agreement (as defined below)) to Mercer International Inc. (or any permitted assignee pursuant to the terms of the Purchase Agreement, the "Buyer"), pursuant to the Asset Purchase Agreement dated April 21, 2023 (as amended), attached hereto as Exhibit 1 and incorporated herein by reference as if set forth herein (together with all other documents contemplated thereby, as such agreement has been or may be amended, restated or supplemented, the "Purchase Agreement"), free and clear of all Liens, Claims, and Interests (each as defined below) with the exceptions of the Assumed Liabilities and Permitted Encumbrances (each as defined in the Purchase Agreement); (b) authorizing the assumption and assignment of certain executory contracts and unexpired leases to the Buyer; (c) authorizing the Debtors to pay the net proceeds from the Sale, up to the amount of the DIP Obligations and the Prepetition Obligations (each as defined below), to Bank of Montreal as lender under (i) the Debtors' debtor-in-possession financing facility (in such capacity, the "DIP Lender," such facility, the "DIP Facility," and the obligations under such facility, the "DIP Obligations") and (ii) the Debtors' prepetition secured lending facility (in such capacity, the "Prepetition Lender," such facility, the "Prepetition Facility," and the obligations under such facility, the "Prepetition Obligations"); and (d) granting related relief, all as more fully set forth in the Motion; and the Court having entered the Order (I) Approving Bidding Procedures, (II) Approving Stalking Horse Protections and Debtors' Entry into Stalking Horse Purchase Agreement, (III) Scheduling the Bid Deadlines and the Auction, (IV) Scheduling Hearings and Objection Deadlines with Respect to the Sale, (V) Approving the Form and Manner of Notice Thereof, (VI) Approving

All references to the "Debtors" shall include the debtors and their estates.

Contract Assumption and Assignment Procedures, and (VII) Granting Related Relief [Docket No. 87] (the "Bidding Procedures Order"); and the auction (the "Auction") for the Transferred Assets, if any, having been held on May 24, 2023; and the Buyer's bid having been declared the Winning Bid; and the Debtors having filed the Notice of Winning Bidder [Docket No. 178] identifying the Buyer as the Winning Bidder for the Transferred Assets in accordance with the Bidding Procedures Order; and the Court having found that the relief requested in the Motion is in the best interests of the Debtors and their estates, their creditors, and all other parties in interest; and the Court having jurisdiction to consider the matters raised in the Motion pursuant to 28 U.S.C. § 1334; and the Court having authority to hear the matters raised in the Motion pursuant to 28 U.S.C. § 157; and venue being proper before the Court pursuant to 28 U.S.C. §§ 1408 and 1409; and consideration of the Motions and the requested relief being a core proceeding that the Court can determine pursuant to 28 U.S.C. § 157(b)(2)(A), (M), (N) and (O); and due and proper notice of the Motion and the hearing to consider the Motion having been given, and it appearing that no other or further notice need be provided; and the Court having reviewed and considered the Motion and the Supplemental Declaration of Kevin Haggard in Support of the Debtors' Sale Motion [Docket No. 183] (the "Haggard Declaration"); and the Court having held a hearing to consider the Motion on May 30, 2023 (the "Sale Hearing") at which time all interested parties were offered an opportunity to be heard regarding the Motion, the Purchase Agreement and the transactions contemplated by the Purchase Agreement (the "Transactions"); and the Court having found that the relief requested in the Motion is in the best interests of the Debtors, their creditors, their estates, and all other parties in interest; and the Court having reviewed and considered the objections, if any, to the Motion (collectively, the "Objections"); and based upon and as demonstrated by the Haggard Declaration, the evidence proffered or adduced at the Sale Hearing, and the arguments of counsel made on the record at the Sale Hearing; and upon all of the proceedings had before the Court; and after due deliberation and sufficient cause appearing therefor, IT IS HEREBY FOUND AND DETERMINED THAT:

Findings of Fact and Conclusions of Law

A. The findings of fact and conclusions of law herein constitute the Court's findings of fact and conclusions of law for the purposes of Bankruptcy Rule 7052, made applicable to this proceeding pursuant to Bankruptcy Rule 9014. To the extent any findings of facts are conclusions of law, they are adopted as such. To the extent any conclusions of law are findings of fact, they are adopted as such.

Jurisdiction and Venue

B. This Court has jurisdiction over this matter pursuant to 28 U.S.C. §§ 157 and 1334. Without limiting the generality of the foregoing, this Court has exclusive *in rem* jurisdiction over the Transferred Assets pursuant to 28 U.S.C. § 1334(e), as such Transferred Assets are property of the Debtors' chapter 11 estates, and, as a result of such jurisdiction, this Court has all necessary power and authority to grant the relief contained herein. This matter is a core proceeding pursuant to 28 U.S.C. § 157(b)(2)(A) and (N). Venue in this district is proper under 28 U.S.C. §§ 1408 and 1409.

Statutory Predicates

C. The statutory and other legal bases for the relief requested in the Motion are sections 105(a), 363, and 365 of the Bankruptcy Code, as supplemented by Bankruptcy Rules 2002, 6004, 6006, 9007, 9008 and 9014. The consummation of the Transactions contemplated by the Purchase Agreement and this Order is legal, valid, and properly authorized under all applicable provisions of the Bankruptcy Code, the Bankruptcy Rules, and the Local

Rules, and the Debtors and the Buyer and their affiliates have complied with all of the applicable requirements of such sections and rules in respect of the Transactions.

Notice

- D. As evidenced by the affidavits and/or certificates of service filed with the Court, proper, timely, adequate, and sufficient notice of the Motion, the Bidding Procedures, the Auction, the Sale (and the Transactions contemplated in connection therewith), the assumption and assignment to the Buyer of the executory contracts and unexpired leases specified as of the date hereof pursuant to the Purchase Agreement or as may be subsequently designated to be an "Additional Assumed Contract" (as defined in the Purchase Agreement) in accordance with Section 1.6(b) of the Purchase Agreement (collectively, the "Assigned Contracts"), the Cure Costs (as defined below), the Sale Hearing, and all deadlines related thereto, has been provided, as relevant, in accordance with sections 102(1), 363, and 365 of the Bankruptcy Code and Bankruptcy Rules 2002, 6004, 6006, 9007, 9008, and 9014, and in compliance with the Bidding Procedures Order, to all interested persons and entities.
- E. The Debtors served notices substantially in the form included in the Notice of (I) Potential Assumption and Assignment of Contracts and (II) Cure Amounts and Adequate Assurance in Connection Therewith [Docket No. 92] (each a "Notice of Assumption and Assignment"), in accordance with the Bidding Procedures Order, identifying, among other things, the Cure Costs (as defined below). The Debtors served (or, with respect to any Additional Assumed Contracts (as defined in the Purchase Agreement), will serve) the Notice of Assumption and Assignment on each of the non-Debtor counterparties to the Assigned Contracts. The service of the Notice of Assumption and Assignment was sufficient under the circumstances and in full compliance with the Bidding Procedures Order, and no further notice need be provided in respect of the Debtors' assumption and assignment to the Buyer of the Assigned Contracts or the Cure

Costs. All non-Debtor counterparties to the Assigned Contracts have had an adequate opportunity to object to the assumption and assignment of the Assigned Contracts and the Cure Costs.

F. The notice described in the foregoing paragraphs is good, sufficient, and appropriate under the circumstances, and no other or further notice of the Motion, the Bidding Procedures, the Auction, the Sale (and the Transactions contemplated in connection therewith), the assumption and assignment to the Buyer of the Assigned Contracts, the Cure Costs, the Sale Hearing, and all deadlines related thereto is or shall be required.

Marketing and Sale Process

- G. The Sale of the Transferred Assets to the Buyer pursuant to the Bidding Procedures is duly authorized pursuant to sections 363(b)(1) and 363(f) of the Bankruptcy Code and Bankruptcy Rule 6004(f). As demonstrated by (i) evidence proffered or adduced at the Sale Hearing and (ii) the representations of counsel made on the record at the Sale Hearing, the Debtors and their professionals, agents, and other representatives have marketed the Transferred Assets and conducted all aspects of the sale process, including the solicitation of bids for an Auction, in good faith and in compliance with the Bidding Procedures and the Bidding Procedures Order. The marketing process undertaken by the Debtors and their professionals, agents, and other representatives with respect to the Transferred Assets has been adequate and appropriate and reasonably calculated to maximize value for the benefit of all stakeholders. The Bidding Procedures and the Auction were duly noticed, were substantively and procedurally fair to all parties, including all Potential Bidders and including with respect to all provisions governing credit bidding, and were conducted in a diligent, non-collusive, fair, and good-faith manner.
- H. The auction contemplated in the Bidding Procedures Order was conducted in good faith, without collusion, and in accordance with the Bidding Procedures Order on May 24, 2023.

 At the conclusion of the Auction, the Debtors (in consultation with the Consultation Parties)

selected the Buyer as the Winning Bidder for the Transferred Assets in accordance with the Bidding Procedures Order.

- I. The Debtors' determinations that the offer reflected in the Purchase Agreement constitutes the highest or otherwise best offer for the Transferred Assets and that the Sale Transaction and the Purchase Agreement maximize value for the benefit of the Debtors' estates are valid and sound exercises of the Debtors' business judgment (exercised in consultation with Bank of Montreal).
- J. The Winning Bid (as defined in the Bidding Procedures Order and which is memorialized in the Purchase Agreement) is in accordance and compliance with the Bidding Procedures Order and constitutes the highest or otherwise best offer for the Transferred Assets.
- K. The Debtors have demonstrated good, sufficient, and sound business purposes and justifications for consummation of the Sale and Transactions contemplated by the Purchase Agreement in accordance with the requirements of section 363(b) of the Bankruptcy Code.
- L. The Purchase Agreement provides fair and reasonable terms for the purchase and sale of the Transferred Assets, and reasonable opportunity has been given to any interested party to make a higher or otherwise better offer for the Transferred Assets.
- M. The Buyer has complied in all respects with the Bidding Procedures Order and all other applicable orders of the Court in negotiating and entering into the Purchase Agreement.
- N. After completion of the Auction and the Debtors' selection of the Buyer as the Winning Bidder, the Debtors and the Buyer negotiated and finalized in good faith and at arm's length the Purchase Agreement and all related documents necessary to consummate the Sale Transaction.

- O. Approval of the Purchase Agreement pursuant to sections 105(a) and 363 of the Bankruptcy Code is necessary to maximize the value of the Debtors' assets.
- P. Neither the Purchase Agreement nor the Transactions contemplated thereunder constitute a *sub rosa* chapter 11 plan. The Purchase Agreement does not specify the terms of, or any distributions under, any subsequent chapter 11 plan by the Debtors (other than provisions that are consistent with the sale of the Transferred Assets under the Purchase Agreement and the relief granted hereunder).
- Q. Approval of the Motion and the Purchase Agreement, and the prompt consummation of the Transactions contemplated thereby, will maximize the value of the Debtors' estates and are in the best interests of the Debtors, their chapter 11 estates, their creditors, and other parties in interest.
- R. The Debtors conducted a robust, sufficient marketing process, and (i) the Purchase Agreement and total consideration to be provided by the Buyer thereunder constitutes the highest or otherwise best offer received by the Debtors for the Transferred Assets, (ii) the bidding procedures utilized were designed to yield the highest or otherwise best bids for the Debtors' assets; (iii) the Purchase Agreement and the completion of the Transactions will present the best opportunity to realize the value of the Transferred Assets and avoid further decline and devaluation of the Transferred Assets; (iv) there is risk of deterioration of the value of the Transferred Assets if the Sale is not consummated promptly; and (v) the Purchase Agreement and the consummation of the Transactions will provide greater value to the Debtors' estates than would be provided by any other presently available alternative.
- S. The total consideration to be provided by the Buyer under the Purchase Agreement constitutes (i) fair value, (ii) fair, full, and adequate consideration, (iii) reasonably equivalent

value, and (iv) reasonable market value for the Transferred Assets for purposes of the Bankruptcy Code, the Uniform Fraudulent Transfer Act, the Uniform Fraudulent Conveyance Act, and any other applicable laws of the United States, any state, territory, or possession thereof or the District of Columbia.

T. The terms of the Purchase Agreement and the Transactions contemplated therein are fair and reasonable under the circumstances of the Debtors' businesses and these Chapter 11 Cases.

Corporate and Organizational Authority

U. The Transferred Assets constitute property of the Debtors' estates and title thereto is vested in the Debtors' estates within the meaning of section 541 of the Bankruptcy Code. The Debtors (i) have full corporate and other organizational power and authority to execute the Purchase Agreement and all other documents contemplated thereby, and the Sale to the Buyer has been duly and validly authorized by all necessary corporate and other organizational action, (ii) have all of the corporate and other organizational power and authority necessary to consummate the Sale and the Transactions contemplated by the Purchase Agreement, (iii) have taken all corporate and other organizational action necessary to authorize and approve the Purchase Agreement and the consummation by the Debtors of the Sale and the Transactions contemplated thereby, and (iv) require no consents or approvals, other than those expressly provided for in the Purchase Agreement, to consummate the Transactions.

Highest and Best Offer; Business Judgment

V. The Debtors have demonstrated a sufficient basis to enter into the Purchase Agreement, sell the Transferred Assets on the terms outlined therein, and assume and assign the Assigned Contracts to the Buyer under sections 363 and 365 of the Bankruptcy Code. All such actions are appropriate exercises of the Debtors' business judgment and in the best interests of the

Debtors, their creditors, their estates and other parties in interest. Approval of the Sale pursuant to the Purchase Agreement at this time is in the best interests of the Debtors, their creditors, their estates, and all other parties in interest.

W. The offer of the Buyer, upon the terms and conditions set forth in the Purchase Agreement, including the total consideration to be realized by the Debtors thereunder, (i) is the highest and best offer received by the Debtors after extensive marketing, including through the Bidding Procedures, (ii) is in the best interests of the Debtors, their creditors, their estates and other parties in interest and (iii) constitutes full and adequate consideration, is fair and reasonable and constitutes reasonably equivalent value, fair consideration, and fair value for the Transferred Assets under the Bankruptcy Code, the Uniform Fraudulent Transfer Act, and under the laws of the United States, any state, territory, possession, or the District of Columbia. Taking into consideration all relevant factors and circumstances, no other entity has offered to purchase the Transferred Assets for greater economic value to the Debtors or their estates.

X. There has been no showing that any of the Debtors or Buyer (i) has entered into the Purchase Agreement or proposes to consummate the Transactions for the purposes of hindering, delaying, or defrauding the Debtors' present or future creditors or (ii) is entering into the Purchase Agreement or proposing to consummate the Transactions fraudulently, for the purpose of statutory or common law fraudulent conveyance and fraudulent transfer claims, whether under the Bankruptcy Code or under the laws of the United States, any state, territory, possession thereof or the District of Columbia or any other applicable jurisdiction with laws substantially similar to the foregoing.

Y. The sale of the Transferred Assets outside a chapter 11 plan pursuant to the Purchase Agreement neither impermissibly restructures the rights of the Debtors' creditors nor impermissibly dictates the terms of a chapter 11 plan of the Debtors.

Opportunity to Object

Z. A reasonable opportunity to object or be heard with respect to the Motion, the Bidding Procedures, the Sale (and the Transactions contemplated in connection therewith), the assumption and assignment to the Buyer of the Assigned Contracts, the Cure Costs, the Sale Hearing, and all deadlines related thereto has been afforded to all interested persons and entities.

Good Faith Purchaser; Arm's Length Sale

- AA. The Purchase Agreement was negotiated, proposed, and entered into by the Debtors and the Buyer without collusion, in good faith, and from arm's length bargaining positions. Neither the Debtors, the Buyer, nor any parent or affiliate of the Buyer has engaged in any conduct that would cause or permit the Purchase Agreement or the Sale to be avoided under section 363(n) of the Bankruptcy Code.
- BB. The Buyer is a good-faith purchaser under section 363(m) of the Bankruptcy Code and, as such, is entitled to all of the protections afforded thereby.
- CC. Neither Buyer nor any of its affiliates, members, officers, directors, shareholders, or any of their respective successors or assigns is an "insider" or "affiliate" of any of the Debtors, as those terms are defined in sections 101(31) and 101(2) of the Bankruptcy Code, and the Buyer's professionals, agents, and other representatives have complied in all respects with the Bidding Procedures Order and all other applicable orders of this Court in negotiating and entering into the Purchase Agreement. The Purchase Agreement complies with the Bidding Procedures Order and all other applicable orders of this Court.

Free and Clear Transfer Required by Buyer

DD. The Buyer would not have entered into the Purchase Agreement and would not consummate the Sale, if each of the Sale and the assumption and assignment of the Assigned Contracts to the Buyer were not free and clear of all Liens, Claims, and Interests of any kind or nature whatsoever (with the exceptions of: (i) the Assumed Liabilities; (ii) any Permitted Encumbrances; or (iii) as more fully set forth in Paragraphs 9–11 of this Order), or if the Buyer would, or in the future could, be liable for any of the Excluded Liabilities except to the extent expressly provided in this Order. For the avoidance of doubt, subject to Paragraphs 9–11 of this Order, neither the Buyer nor any of its affiliates shall have any responsibility whatsoever with respect to the Excluded Liabilities, which shall remain the responsibility of the Debtors before, on, and after the Closing.

EE. As of the Closing, pursuant and subject to the terms of the Purchase Agreement and this Order, the transfer of the Transferred Assets and of the Assumed Liabilities and the Sale will effect a legal, valid, enforceable, and effective transfer of the Transferred Assets and will vest the Buyer with all of the Debtors' rights, title, and interests in the Transferred Assets free and clear of all Liens, Claims, and Interests of any kind or nature whatsoever (with the exceptions of the Assumed Liabilities and any Permitted Encumbrances).

Satisfaction of Section 363(f)

FF. The Debtors may sell the Transferred Assets free and clear of any and all Liens, Claims, and Interests of any kind or nature whatsoever (with the exception of any Permitted Encumbrances), including any rights or claims based on any putative successor or transferee liability, as set forth herein, because, in each case, one or more of the standards set forth in section 363(f)(1)–(5) of the Bankruptcy Code has been satisfied. All parties in interest, including, without limitation, any holders of Liens, Claims, and/or Interests, and any non-Debtor

counterparties to the Assigned Contracts, that did not object, or who withdrew their objection, to the Sale, the Motion, the assumption and assignment of the applicable Assigned Contract, or the associated Cure Cost are deemed to have consented to the relief granted herein pursuant to section 363(f)(2) of the Bankruptcy Code.

Assigned Contracts

GG. The Debtors have demonstrated (i) that it is an exercise of their sound business judgment to assume and assign the Assigned Contracts to the Buyer in each case in connection with the consummation of the Sale and (ii) that the assumption and assignment of the Assigned Contracts to the Buyer is in the best interests of the Debtors, their estates, and creditors, and other parties in interest. The Assigned Contracts being assigned to the Buyer are an integral part of the Transferred Assets being purchased by the Buyer and, accordingly, such assumption, assignment, and cure of any defaults under the Assigned Contracts are reasonable and enhance the value of the Debtors' estates. Any non-Debtor counterparty to an Assigned Contract that has not actually filed with the Court an objection to such assumption and assignment in accordance with the terms of the Motion is deemed to have consented to such assumption and assignment.

Cure Costs and Adequate Assurance

HH. The Debtors and the Buyer, as applicable, have, including by way of entering into the Purchase Agreement, and agreeing to the provisions therein relating to the Assigned Contracts, (i) cured, or provided adequate assurance of cure, of any default existing prior to the date hereof under any of the Assigned Contracts within the meaning of section 365(b)(1)(A) of the Bankruptcy Code and (ii) provided compensation or adequate assurance of compensation to any party for any actual pecuniary loss to such party resulting from a default prior to the date hereof under any of the Assigned Contracts within the meaning of section 365(b)(1)(B) of the Bankruptcy Code and the Buyer has, based upon the record of these proceedings, including the evidence

adduced at the Sale Hearing, provided adequate assurance of its future performance of and under the Assigned Contracts pursuant to sections 365(b)(1) and 365(f)(2) of the Bankruptcy Code. The Buyer's agreement under the Purchase Agreement to perform the obligations under the Assigned Contracts after the Closing shall constitute adequate assurance of future performance under the Assigned Contracts being assigned to the Buyer within the meaning of sections 365(b)(1)(C) and 365(f)(2)(B) of the Bankruptcy Code. The Cure Costs are hereby found to be the sole amounts necessary to cure any and all defaults under the Assigned Contracts under section 365(b) of the Bankruptcy Code.

Time Is of the Essence; Waiver of Stay

II. Time is of the essence in consummating the Sale. In order to maximize the value of the Transferred Assets, it is essential that the sale and assignment of the Transferred Assets occur within the time constraints set forth in the Purchase Agreement. Accordingly, there is cause to waive the stays contemplated by Bankruptcy Rules 6004 and 6006.

NOW THEREFORE, IT IS HEREBY ORDERED, ADJUDGED, AND DECREED THAT:

Motion is Granted

1. The relief requested by the Motion is granted as set forth herein.

Objections Overruled

- 2. All objections to the entry of this Order or to the relief granted herein, whether filed, stated on the record before this Court or otherwise, which have not been withdrawn, waived, or settled, and all reservations of rights included therein, are denied and overruled on the merits.
- 3. Notice of the Motion, the Bidding Procedures, the Sale (and the Transactions contemplated in connection therewith), the assumption and assignment to the Buyer of the Assigned Contracts, the Cure Costs, the Sale Hearing, and all deadlines related thereto was fair

and equitable under the circumstances and complied in all respects with section 102(1) of the Bankruptcy Code and Bankruptcy Rules 2002, 6004 and 6006.

Approval of the Purchase Agreement

- 4. The Purchase Agreement, as amended to reflect the increased Purchase Price as set forth on the record of the Auction, including all of the terms and conditions thereof, is hereby approved. Pursuant to sections 105, 363, and 365 of the Bankruptcy Code, the Debtors are authorized and directed to take any and all actions necessary to fulfill their obligations under, and comply with the terms of, the Purchase Agreement and to consummate the Sale pursuant to and in accordance with the terms and conditions of the Purchase Agreement and this Order, without further leave of the Court. The Debtors are further authorized and directed to pay, without further order of this Court, whether before, at, or after the Closing, any expenses or costs, if any, that are required to be paid to consummate the Transactions contemplated by the Purchase Agreement, as set forth in the Purchase Agreement and this Order, or otherwise perform their obligations under the Purchase Agreement and this Order.
- 5. The Debtors are authorized, in accordance with the Purchase Agreement, to execute and deliver, and empowered to perform under, consummate, and implement, the Purchase Agreement, together with all additional instruments, documents, and other agreements that may be reasonably necessary or desirable to implement the Purchase Agreement, and to take all further actions as may be reasonably requested by the Buyer for the purpose of assigning, transferring, granting, conveying, and conferring to the Buyer or reducing to possession, the Transferred Assets, or as may be reasonably necessary or appropriate to the performance of the obligations as contemplated by the Purchase Agreement.
- 6. Buyer and the Debtors have no obligation to proceed with the Closing unless and until all conditions precedent to their respective obligations to do so, as set forth in the Purchase

Agreement, have been met, satisfied, or waived in accordance with the terms of the Purchase Agreement.

Binding Effect of Order

7. This Order and the Purchase Agreement shall be binding upon all creditors of, and equity holders in, the Debtors and any and all other parties in interest, including, without limitation, any and all holders of Liens, Claims, and Interests (including holders of any rights or claims based on any putative successor or transferee liability) of any kind or nature whatsoever, all non-Debtor parties to the Assigned Contracts, the Buyer, the Debtors and their affiliates and subsidiaries, and any trustee or successor trustee appointed in the Debtors' Chapter 11 Cases or upon a conversion to cases under chapter 7 under the Bankruptcy Code.

Amendments to the Purchase Agreement

8. The Purchase Agreement and any related agreements, documents, or other instruments may be modified, amended, supplemented, or restated by the parties thereto in a writing signed by such parties and in accordance with the terms thereof, without further order of this Court, provided that any such modification, amendment, supplement, or restatement does not have a material adverse effect on the Debtors' estates or any other person or entity, including Bank of Montreal. The Purchase Agreement and this Order and the Debtors' and Buyer's obligations therein and herein shall not be altered, amended, rejected, discharged, or otherwise affected by any chapter 11 plan proposed or confirmed in these bankruptcy cases or any other relief sought from or granted by the Court without the prior written consent of the Buyer.

Transfer of the Transferred Assets Free and Clear

9. Subject to Paragraphs 9–11 hereof, the Buyer shall assume and be liable for only those liabilities expressly assumed pursuant to the Purchase Agreement. Except as expressly permitted or otherwise specifically provided for in the Purchase Agreement or this Order, pursuant

to sections 105(a), 363(b), 363(f), and 365(b) of the Bankruptcy Code, upon the Closing, the Transferred Assets shall be transferred to the Buyer free and clear of any and all Liens, Claims, and Interests of any kind or nature whatsoever, with the exceptions of the Assumed Liabilities and any Permitted Encumbrances, with such Liens, Claims and Interests to attach to the proceeds of such transfer to the same extent and with the same priority as such Liens, Claims and Interests had in the Transferred Assets. For purposes of this Order, "Liens," "Claims," and "Interests" shall mean, respectively:

- any and all Encumbrances, charges, liens (statutory or otherwise), claims, mortgages, leases, subleases, hypothecations, deeds of trust, pledges, security interests, options, hypothecations, rights of use or possession, rights of first offer or first refusal (or any other type of preferential arrangement), rights of consent, rights of offset and setoff (but only to the extent not actually taken pre-petition), successor liability, easements, servitudes, restrictive covenants, interests or rights under any operating agreement, encroachments, encumbrances, restrictions on transferability of any type, any dedication under any gathering, transportation, treating, purchasing or similar agreement that is not assumed by or assigned to the Buyer, any rights that purport to give any party a right or option to effect any forfeiture, modification, right of first offer or first refusal, or consents, or termination of the Debtors' or the Buyer's interest in the Transferred Assets, any similar rights, and third-party interests or any other restrictions or limitations of any kind with respect to the Transferred Assets (collectively, "Liens");
- any and all claims as defined in section 101(5) of the Bankruptcy Code and **b**. jurisprudence interpreting the Bankruptcy Code, including, without limitation, (i) any and all claims or causes of action based on or arising under any labor, employment or pension laws, labor or employment agreements, including any employee claims related to worker's compensation, occupational disease, or unemployment or temporary disability, including, without limitation, claims that might otherwise arise under or pursuant to (a) ERISA (as defined below), (b) the Fair Labor Standards Act, (c) Title VII of the Civil Rights Act of 1964, (d) the Federal Rehabilitation Act of 1973, (e) the National Labor Relations Act, (f) the Age Discrimination and Employment Act of 1967 and Age Discrimination in Employment Act, as amended, (g) the Americans with Disabilities Act of 1990, (h) the Consolidated Omnibus Budget Reconciliation Act of 1985, as amended, including, without limitation, the requirements of Part 6 of Subtitle B of Title I of ERISA and Section 4980B of the Internal Revenue Code and of any similar state law (collectively, "COBRA"), (i) state

discrimination laws, (j) state unemployment compensation laws or any other similar state laws, (k) any other state or federal benefits or claims relating to any employment with the Debtors or any of their predecessors, or (1) the WARN Act (29 U.S.C. §§2101 et seq.), (ii) any rights under any pension or multiemployer plan (as such term is defined in Section 3(37) or Section 4001(a)(3) of the Employee Retirement Income Security Act of 1974 (as amended, "ERISA"), health or welfare, compensation or other employee benefit plans, agreements, practices, and programs, including, without limitation, any pension plans of the Debtors or any multiemployer plan to which the Debtors have at any time contributed to or had any liability or potential liability, (iii) any and all claims or causes of action based upon or relating to any putative successor or transferee liability, (iv) any rights related to intercompany loans and receivables between the Debtors and any non-Debtor subsidiary or affiliate, (v) any Excluded Liabilities, (vi) any and all claims or causes of action based upon or relating to any unexpired and executory contract or unexpired lease to which a Debtor is a party that is not an Assigned Contract that will be assumed and assigned pursuant to this Order and the Purchase Agreement, (vii) any and all claims or causes of action based upon or relating to any bulk sales or similar law, (viii) any and all claims or causes of action based upon or relating to any tax statutes or ordinances, including, without limitation, the Internal Revenue Code of 1986, as amended, and any taxes arising under or out of, in connection with, or in any way relating to the operation of the Transferred Assets prior to the Closing, including, without limitation, any ad valorem taxes assessed by any applicable taxing authority (notwithstanding the foregoing, the Sale is not tax exempt under Section 1146(a) of the Bankruptcy Code, and nothing herein releases, waives, or enjoins enforcement of the Buyer's obligation to pay any transfer taxes that may be due and owing as a result of the Sale), and (ix) any and all other claims, causes of action, proceedings, warranties, guaranties, rights of recovery, setoff (but only to the extent not actually taken pre-petition), rights, remedies, obligations, liabilities, counterclaims, cross-claims, third party claims, demands, restrictions, responsibilities, or contribution, reimbursement, subrogation, or indemnification claims or liabilities based on or relating to any act or omission of any kind or nature whatsoever asserted against any of the Debtors or any of their respective affiliates, subsidiaries, directors, officers, agents, successors or assigns in connection with or relating to the Debtors, their operations, their business, their liabilities, the Debtors' marketing and bidding process with respect to the Transferred Assets, the Assigned Contracts, or the Transactions contemplated by the Purchase Agreement (collectively, "Claims"); and

c. any and all equity or other interests of any kind or nature whatsoever in or with respect to (x) any of the Debtors or their respective affiliates, subsidiaries, successors or assigns, (y) the Transferred Assets, or (z) the Assigned Contracts (collectively, "Interests");

whether in law or in equity, known or unknown, choate or inchoate, filed or unfiled, scheduled or

unscheduled, noticed or unnoticed, recorded or unrecorded, perfected or unperfected, allowed or disallowed, contingent or non-contingent, liquidated or unliquidated, matured or unmatured, material or non-material, disputed or undisputed, direct or indirect, and whether arising by agreement, understanding, law, equity or otherwise, and whether occurring or arising before, on or after the Petition Date, or occurring or arising prior to the Closing. On the Closing, the Buyer shall take title to and possession of the Transferred Assets, subject only to the Permitted Encumbrances and Assumed Liabilities, and Paragraphs 9–11 of this Order.

Vesting of Transferred Assets in the Buyer

- Agreement shall constitute a legal, valid, effective and final transfer of the Transferred Assets on the Closing, and shall vest the Buyer with all of the Debtors' rights, title and interests in the Transferred Assets free and clear of all Liens, Claims, and Interests of any kind or nature whatsoever (with the exceptions of the Assumed Liabilities, any Permitted Encumbrances, and subject to Paragraphs 9–11 of this Order), with such Liens, Claims and Interests to attach to the proceeds of such transfer to the same extent and with the same priority as such Liens, Claims and Interests had in the Transferred Assets.
- 11. In connection with the Closing and subject in all respects to any financing documents or organizational agreements, the Buyer is hereby authorized in connection with the consummation of the Sale to allocate the Transferred Assets, including the Assigned Contracts, to and among its direct and indirect parent and subsidiary corporations (collectively, "Designees"), in a manner as it in its sole discretion deems appropriate, and to assign, lease, sublease, license, sublicense, transfer, or otherwise dispose of any of the Transferred Assets, including the Assigned Contracts, to and among its Designees with all of the rights and protections accorded to the Buyer

under this Order and the Purchase Agreement with respect thereto, and the Debtors shall cooperate with and take all actions reasonably requested by the Buyer to effectuate any of the foregoing.

Assumption and Assignment of Assigned Contracts

- 12. Pursuant to sections 105(a) and 365 of the Bankruptcy Code, and subject to and conditioned upon the Closing, the Debtors' assumption and assignment to the Buyer of the Assigned Contracts is hereby approved, and the requirements of section 365(b)(1) of the Bankruptcy Code with respect thereto are hereby deemed satisfied.
- 13. The Debtors are hereby authorized, in accordance with the Purchase Agreement, and in accordance with sections 105(a) and 365 of the Bankruptcy Code, to (i) assume and assign to the Buyer the Assigned Contracts, effective upon and subject to the occurrence of the Closing, free and clear of all Liens, Claims, and Interests of any kind or nature whatsoever (with the exceptions of the Assumed Liabilities and Permitted Encumbrances), which Assigned Contracts, by operation of this Order, shall be deemed assumed and assigned to the Buyer effective as of the Closing, and (ii) execute and deliver to the Buyer such documents or other instruments as the Buyer may deem necessary to assign and transfer the Assigned Contracts to the Buyer.
 - 14. Subject to Paragraph 15 hereof:
 - a. The Debtors are authorized to and shall assume all of the Assigned Contracts in accordance with section 365 of the Bankruptcy Code.
 - b. The Debtors are authorized to and shall assign each Assigned Contract to the Buyer in accordance with sections 363 and 365 of the Bankruptcy Code, and any provisions in any Assigned Contract that prohibit or condition the assignment of such Assigned Contract to the Buyer on the consent of the counterparty thereto or allow the non-Debtor party to such Assigned Contract to terminate, recapture, impose any penalty, condition, renewal or extension, or modify any term or condition upon the assignment of such Assigned Contract, shall constitute unenforceable anti-assignment provisions which are expressly preempted under section 365 of the Bankruptcy Code.

- c. All requirements and conditions under sections 363 and 365 of the Bankruptcy Code for the assumption and assignment of the Assigned Contracts by the Debtors to the Buyer have been satisfied.
- d. To the extent provided in section 365(f)(1) of the Bankruptcy Code, upon the Closing, the Assigned Contracts shall be transferred and assigned to, and remain in full force and effect for the benefit of, the Buyer in accordance with their respective terms, notwithstanding any provision in any such Assigned Contract (including those of the type described in sections 365(b)(2), 365(e)(1) and 365(f) of the Bankruptcy Code) that prohibits, restricts, limits, or conditions such assignment or transfer.
- e. After the Debtors' transfer and assignment of the Assigned Contracts to the Buyer, in accordance with sections 363 and 365 of the Bankruptcy Code, the Buyer shall be fully and irrevocably vested in all right, title, and interest of each Assigned Contract free and clear of Liens, Claims, and Interests with the exceptions of the Assumed Liabilities and Permitted Encumbrances, with such Liens, Claims and Interests to attach to the proceeds of such transfer and assignment to the same extent and with the same priority as such Liens, Claims and Interests had in the Assigned Contracts.
- f. Any portion of any Assigned Contract which purports to permit a landlord thereunder to cancel the remaining term of such Assigned Contract if the lessee discontinues its use or operation of the leased premises is void and of no force and effect, and shall not be enforceable against the Buyer, or its assignees and sublessees; and the landlords under any such Assigned Contract shall not have the right to cancel or otherwise modify the Assigned Contract or increase the rent, assert any claim, or impose any penalty by reason of such discontinuation, the Debtors having suspended or ceased operations in, on, or about the leased premises prior to the transfer and assignment of such Assigned Contract to the Buyer, the assignment of such Assigned Contract to the Buyer, or the interruption of business activities at any of the leased premises.
- g. The failure of the Debtors to enforce, at any time prior to the Closing Date, one or more terms or conditions of any Assigned Contract shall not be a waiver of such terms or conditions, or of the Buyer's rights to enforce every term and condition of the Assigned Contracts.
- h. There shall be no rent accelerations, assignment fees, increases, or any other fees charged to the Buyer as a result of the assumption, assignment, and sale of the Assigned Contracts.
- 15. All defaults and all other obligations of the Debtors under the Assigned Contracts occurring, arising or accruing prior to the assignment thereof to the Buyer at Closing (without

giving effect to any acceleration clauses or any default provisions of the kind specified in section 365(b)(2) of the Bankruptcy Code) are deemed to have been cured or satisfied by the payment of the proposed amount necessary, if any, to cure all monetary defaults, if any, under each Assigned Contract in the amounts set forth in the Notice of Assumption and Assignment or any supplemental Notice of Assumption and Assignment (or any other cure cost reached by agreement after an objection to the proposed cure cost by a counterparty to an Assigned Contract), which was served in compliance with the Bidding Procedures Order (the "Cure Costs"), and which Cure Costs were satisfied, or shall be satisfied as soon as practicable, by the Buyer, as the case may be, as provided in the Purchase Agreement. For all Assigned Contracts for which a Notice of Assumption and Assignment was served, the Buyer is authorized and directed to pay the Cure Costs required to be paid by such parties in accordance with the Purchase Agreement upon the Closing. For any Assigned Contract that is designated by Buyer to be an Assigned Contract after the date of this Order in accordance with the Purchase Agreement, the Debtors shall file a Notice of Assumption and Assignment on the counterparties thereto that identifies the Cure Costs, and, to the extent no objection is received to such notice within seven (7) days of the date of mailing, such Assigned Contract shall be deemed assumed and assigned to Buyer as an Assigned Contract with the Cure Costs set forth in such notice and subject to all relief set forth herein that applies to Assigned Contracts effective as of the Closing. To the extent a timely objection to such notice is received and a is not resolved between the Debtors and such counterparty in accordance with the procedures governing the assumption and assignment of executory contracts and leases approved in the Bidding Procedures Order, such objection will be adjudicated by this Court.

- 16. Pursuant to section 365(k) of the Bankruptcy Code, the Debtors and their estates shall be relieved from any liability for any breach for any Assigned Contract that occurs after the effectiveness of such assumption and assignment to the Buyer.
- 17. Notwithstanding any provision in this Order, any and all agreements and obligations of Buyer shall be and are subject to Closing in accordance with the Purchase Agreement.

No Successorship or Transferee Liability

18. Neither the Buyer nor any of its affiliates is or shall be deemed, as a result of the consummation of the Transactions contemplated herein, to: (a) be legal successors to the Debtors or their estates by reason of any theory of law or equity, (b) have, *de facto* or otherwise, merged with or into the Debtors, or (c) be an alter ego or a mere continuation or substantial continuation or successor of the Debtors in any respect. Neither the Buyer nor any of its affiliates shall assume or in any way be responsible for any liability or obligation of any of the Debtors and/or their estates, except as otherwise expressly provided in the Purchase Agreement or this Order.

Modification of the Automatic Stay

19. The automatic stay provisions of section 362 of the Bankruptcy Code are lifted and modified to the extent necessary to implement the terms and conditions of the Purchase Agreement and the provisions of this Order.

Release of Liens by Creditors; Collection of Transferred Assets

Agreement, the holder of any valid Lien, Claim, or Interest in the Transferred Assets (other than a Permitted Encumbrance) shall, as of the Closing, without regard to whether such holder has executed or filed any applicable release, and such Liens, Claims and Interests shall be deemed to attach to the proceeds of such transfer to the same extent and with the same priority as such Liens,

Claims and Interests had in the Transferred Assets. Notwithstanding the foregoing and except as expressly provided in this Order or the Purchase Agreement, any such holder of such a Lien, Claim, or Interest is authorized and directed to execute and deliver any waivers, releases, or other related documentation reasonably requested by the Debtors to evidence the release of their Liens, Claims, or Interests in the Transferred Assets. Any person or entity that has filed any financing statements, mortgages, mechanic's liens, lis pendens, or any other documents or agreements evidencing a Lien on any of the Transferred Assets conveyed pursuant to the Purchase Agreement and this Order is directed to deliver to the Debtors prior to the Closing, in proper form for filing and executed by the appropriate parties, termination statements, instruments of satisfaction, or releases of all Liens which the person or entity has with respect to the Transferred Assets or otherwise.

- 21. In the event that such termination statements, instruments of satisfaction, or releases of all Liens (excluding Permitted Encumbrances) are not filed in accordance with the foregoing paragraph, the Debtors and the Buyer are hereby authorized to file, register, or otherwise record a certified copy of this Order, which, once filed, registered, or otherwise recorded, shall constitute conclusive evidence of the release of all Liens in the Transferred Assets of any kind or nature whatsoever (except any Permitted Encumbrances) except as expressly provided in this Order or the Purchase Agreement.
- 22. As of the Closing, the Buyer and its successors and assigns shall be designated and appointed as the Debtors' true and lawful attorney with full power of substitution in the Debtors' name and stead on behalf of and for the benefit of the Buyer, and its successors and assigns, for the following sole and limited purposes: to have the power to demand and receive any and all of the Transferred Assets and to give receipts and releases for and in respect of the Transferred Assets, or any part thereof, and from time to time to institute and prosecute against third parties for the

benefit of the Buyer, its successors and assigns, proceedings at law, in equity or otherwise, which the Buyer, and its successors and assigns, may deem proper for the collection or reduction to possession of any of the Transferred Assets.

Payment of Net Proceeds

Within three (3) business days of the Closing, the Debtors will deliver the net 23. proceeds from Sale, up to the amount of the DIP Obligations and the Prepetition Obligations, to the DIP Lender and the Prepetition Lender, respectively, for application to the DIP Obligations in accordance with the DIP Facility and the Prepetition Obligations in accordance with the Prepetition Facility (the "DIP and Prepetition Payoff"); provided that, notwithstanding the DIP and Prepetition Payoff, (a) the rights of the Committee and any other party in interest pursuant to section 4.1(b) of the Final Order (I) Authorizing Debtors and Debtors in Possession to (A) Obtain Postpetition Financing, (B) Use Cash Collateral, (C) Grant Liens and Super-Priority Claims, and (D) Grant Adequate Protection; (II) Modifying the Automatic Stay; and (C) Granting Related Relief [Docket No. 136] (the "DIP Order") shall be expressly reserved, (b) the rights of the DIP Lender and the Prepetition Lender are expressly reserved in connection with the same, including, without limitation, any rights (including but not limited to any rights of indemnification) arising under the DIP Order and DIP Credit Agreement (as defined in the DIP Order) in the event of a Challenge (as defined in the DIP Order) (the "Lender Rights") and (c) notwithstanding the DIP and Prepetition Payoff, the liens of the DIP Lender and the Prepetition Lender under the DIP Order and the Pre-Petition Loan Documents (as defined in the DIP Order) shall attach to the net proceeds from the Sale to secure the Lender Rights, and such liens shall be deemed perfected without any further notice, filing or action (subject, with respect to the rights of the Prepetition Lender under the Pre-Petition Loan Documents and the DIP Order, to the Challenge Period).

Effect of Recordation of Order

This Order, once filed, registered, or otherwise recorded, (a) shall be effective as a 24. conclusive determination that, upon the Closing, all Liens, Claims, and Interests of any kind or nature whatsoever (with the exceptions of the Assumed Liabilities and any Permitted Encumbrances and except as expressly provided in this Order) existing as to the Transferred Assets prior to the Closing have been unconditionally released, discharged, and terminated and that the conveyances described herein have been effected, and (b) shall be binding upon and shall govern the acts of all persons and entities including, without limitation, all filing agents, filing officers, title agents, title companies, recorders of mortgages, recorders of deeds, registrars of deeds, administrative agencies, governmental departments, secretaries of state, federal, state, local officials, notaries, protonotaries, and all other persons and entities who may be required by operation of law, the duties of their office, or contract, to accept, file, register or otherwise record or release any documents or instruments, or who may be required to report or insure any title or state of title in or to, the Transferred Assets. Each and every federal, state, and local governmental agency or department is hereby authorized to accept any and all documents and instruments necessary and appropriate to consummate the Transactions contemplated by the Purchase Agreement and this Order, including, without limitation, recordation of this Order.

Prohibition of Actions Against the Buyer

25. Except for the Assumed Liabilities and any Permitted Encumbrances or as expressly permitted or otherwise specifically provided for in the Purchase Agreement or this Order, the Buyer and its affiliates shall have no liability or responsibility for any liability or other obligation of the Debtors or any of their affiliates arising under or related to the Transferred Assets or otherwise. Without limiting the generality of the foregoing, and except as otherwise specifically provided in this Order or in the Purchase Agreement, the Buyer and its affiliates shall not be liable

for any claims against the Debtors or any of their predecessors or affiliates, and the Buyer and its affiliates shall have no successor or vicarious liabilities of any kind or character including, without limitation, any theory of antitrust, warranty, product liability, environmental, successor or transferee liability, labor law, ERISA, *de facto* merger, mere continuation, or substantial continuity, whether known or unknown as of the Closing, now existing or hereafter arising, whether fixed or contingent, with respect to the Debtors or any obligations of the Debtors, including, without limitation, liabilities on account of any taxes arising, accruing, or payable under, out of, in connection with, or in any way relating to the operation of the Debtors' business prior to the Closing or any claims under the WARN Act or any claims related to wages, benefits, severance, or vacation pay owed to employees or former employees of the Debtors.

26. EFFECTIVE UPON THE CLOSING, WITH THE SOLE EXCEPTIONS OF ANY ENFORCEMENT OF RIGHTS RELATED TO THE ASSUMED LIABILITIES OR ANY PERMITTED ENCUMBRANCES, AND EXCEPT AS EXPRESSLY PROVIDED BY THIS ORDER, ALL PERSONS AND ENTITIES HAVING LIENS, CLAIMS, ENCUMBRANCES, CAUSES OF ACTION OR INTERESTS OF ANY KIND OR NATURE WHATSOEVER AGAINST OR IN THE DEBTORS OR THE TRANSFERRED ASSETS OR ASSIGNED CONTRACTS SHALL BE, AND HEREBY ARE, FOREVER BARRED AND ENJOINED FROM (A) TAKING ANY ACTION THAT WOULD ADVERSELY AFFECT OR INTERFERE WITH THE ABILITY OF THE DEBTORS TO TRANSFER THE TRANSFERRED ASSETS TO THE BUYER IN ACCORDANCE WITH THE TERMS OF THIS ORDER AND THE PURCHASE AGREEMENT OR BUYER'S RIGHTS, USE, AND ENJOYMENT OF THE TRANSFERRED ASSETS AND ASSIGNED CONTRACTS IN ACCORDANCE WITH THE TERMS OF AND AS AUTHORIZED BY THIS ORDER TO THE EXTENT SUCH ACTION

RELATED TO A LIEN, CLAIM, ENCUMBRANCE, INTEREST, OR CAUSE OF ACTION ARISING ON OR PRIOR TO THE CLOSING DATE AND (B) ASSERTING, PROSECUTING, OR OTHERWISE PURSUING, WHETHER IN LAW OR IN EQUITY, IN ANY JUDICIAL, ADMINISTRATIVE, ARBITRAL, OR OTHER PROCEEDING, ANY LIENS, CLAIMS, ENCUMBRANCES, CAUSES OF ACTION, OR INTERESTS OF ANY KIND OR NATURE WHATSOEVER ARISING ON OR PRIOR TO THE CLOSING DATE AND RELATING TO (i) ANY LIEN, CLAIM, ENCUMBRANCE, CAUSE OF ACTION, OR INTEREST RELEASED PURSUANT TO THIS ORDER, (ii) THE PURCHASE AGREEMENT, (iii) THE SALE, (iv) THE TRANSACTIONS, (v) THE TRANSFERRED ASSETS, OR (vi) THE ASSIGNED CONTRACTS AGAINST THE BUYER AND ITS AFFILIATES, SUCCESSORS, DESIGNEES, ASSIGNS, OR PROPERTY, OR THE TRANSFERRED ASSETS OR ASSIGNED CONTRACTS CONVEYED UNDER THE AUTHORIZATION OF THIS ORDER IN ACCORDANCE WITH THE PURCHASE AGREEMENT.

No Interference

27. Following the Closing, subject to paragraphs 9–11 of this Order, no holder of a Lien, Claim, and/or Interest (with the exception of any holders of Permitted Encumbrances) in or against the Debtors or the Transferred Assets, or any other entity or person, shall interfere with the Buyer's title to or use and enjoyment of the Transferred Assets based on or related to such Lien, Claim, and/or Interest or any actions that the Debtors may take in their bankruptcy cases or any successor cases.

Retention of Jurisdiction

28. This Court retains jurisdiction prior to, on, and after Closing to, among other things, interpret, enforce, and implement the terms and provisions of the this Order and the Purchase Agreement, all amendments thereto, any waivers and consents thereunder, and each of the

agreements executed in connection therewith in all respects, including, without limitation, retaining jurisdiction to: (a) compel delivery of the Transferred Assets or performance of other obligations owed to the Buyer; (b) compel performance of obligations owed to the Debtors; (c) resolve any disputes arising under or related to the Purchase Agreement; (d) interpret, implement, and enforce the provisions of this Order; (e) determine whether any lien or Encumbrance constitutes a Permitted Encumbrance, (f) determine any claim, cause, cause of action, or controversy brought by the Buyer relating to the Purchase Agreement or the Transferred Assets or that constitute Transferred Assets, and (g) protect the Buyer and its affiliates against (i) any Liens, Claims, and Interests in or against the Debtors or the Transferred Assets of any kind or nature whatsoever and (ii) any creditors or other parties in interest regarding the turnover of the Transferred Assets that may be in their possession.

No Stay of Order

29. Notwithstanding Bankruptcy Rules 6004(h) and 6006(d), this Order shall be effective and enforceable immediately upon entry and its provisions shall be self-executing. In the absence of any person or entity obtaining a stay pending appeal, the Debtors and the Buyer are free to close the Sale under the Purchase Agreement at any time pursuant to the terms thereof.

Good Faith Purchaser

30. The Sale contemplated by the Purchase Agreement is undertaken by the Buyer in good faith, as that term is used in section 363(m) of the Bankruptcy Code, and Buyer has acted without collusion in undertaking and consummating the Sale and Transactions contemplated by the Purchase Agreement. Accordingly, the reversal or modification on appeal of the authorization provided herein to consummate the Sale shall not affect the validity of the Sale to the Buyer (including the assumption and assignment by the Debtors of any of the Assigned Contracts), unless such authorization is duly stayed pending such appeal. The Buyer is a buyer in good faith of the

Transferred Assets, and is entitled to all of the protections afforded by section 363(m) of the Bankruptcy Code.

31. There has been no showing that the Debtors or Buyer or its affiliates engaged in any action or inaction that would cause or permit the Transactions to be avoided or costs or damages to be imposed under section 363(n) of the Bankruptcy Code.

Inconsistencies with Prior Orders, Pleadings or Agreements

32. To the extent of any conflict between the Purchase Agreement and this Order, the terms of this Order shall govern. To the extent this Order is inconsistent or conflicts with any prior order or pleading in these Chapter 11 Cases, the terms of this Order shall govern to the extent required to permit consummation of the Sale.

Failure to Specify Provisions

33. The failure to specifically reference any particular provisions of the Purchase Agreement or other related documents in this Order shall not diminish or impair the effectiveness of such provisions, it being the intent of the Court that the Purchase Agreement and other related documents be authorized and approved in their entirety.

Resolution of Informal Objection by the Environmental Protection Agency

34. Nothing in this Order or the Asset Purchase Agreement releases, nullifies, precludes or enjoins the enforcement of any police or regulatory liability to a U.S. governmental unit that any entity would be subject to as the post-sale owner or operator of property after the date of entry of this Order. Nothing in this Order or the Asset Purchase Agreement authorizes the transfer or assignment of any U.S. governmental (a) license, (b) permit, (c) registration, (d) authorization, or (e) approval, or the discontinuation of any obligation thereunder, without compliance with all applicable legal requirements and approvals under police or regulatory law in the United States.

Nothing in this Order divests any tribunal of any jurisdiction it may have under police or regulatory law to interpret this Order or to adjudicate any defense asserted under this Order.

Dated: May 30th, 2023 Wilmington, Delaware CRAIG T. GOLDBLATT UNITED STATES BANKRUPTCY JUDGE

Cin Dulles

This is Exhibit "<u>M</u>" referred to in the Affidavit of **MICHÈLE HAY**, sworn before me at Vancouver, BC, this 30th day of May, 2023.

A Commissioner for taking Affidavits in British Columbia

IN THE UNITED STATES BANKRUPTCY COURT FOR THE DISTRICT OF DELAWARE

In re:) Chapter 11
STRUCTURLAM MASS TIMBER U.S., INC., et al., 1) Case No. 23-10497 (CTG)
) (Jointly Administered)
Debtors.) Related Docket No. 139
)

ORDER (I) AUTHORIZING THE DEBTORS TO IMPLEMENT KEY EMPLOYEE INCENTIVE AND RETENTION PLANS AND (II) GRANTING RELATED RELIEF

This matter coming before the Court upon the Debtors' Motion for Entry of an Order (i) Authorizing the Debtors to Implement Key Employee Incentive and Retention Plans and (ii) Granting Related Relief (the "Motion") filed by the above-captioned debtors (collectively, the "Debtors") for entry of an order (this "Order") (i) approving and authorizing the implementation of the KEIP and KERP; (ii) authorizing (but not directing) the Debtors to make payments under the KEIP and KERP to the Participants under the terms contained therein; (iii) granting administrative expense priority status to all payments to be made by the Debtors thereunder; and (iv) granting related relief, all as further described in the Motion; and upon consideration of the Declaration of Brian Cumberland in Support of Debtors' Motion for Entry of Order (I) Authorizing the Debtors to Implement Key Employee Incentive and Retention Plans and (II) Grating Related Relief (the "Cumberland Declaration"); and the Supplemental Declaration of Anthony Tillman in Support of Debtors' Motion for Entry of Order (I) Authorizing the Debtors to

The Debtors in these chapter 11 cases, along with the last four digits of each Debtors' federal tax identification number or Canadian business number, as applicable, include: Structurlam Mass Timber U.S., Inc. (6287); Natural Outcomes, LLC (n/a); Structurlam Mass Timber Corporation (5050); and SLP Holdings Ltd. (3114). The location of the Debtors' headquarters is: 2176 Government Street, Penticton, British Columbia, Canada V2A 8B5. The address of the registered agent for Structurlam Mass Timber U.S., Inc. is: 8 The Green, Suite A, Dover, Delaware 19901.

Implement Key Employee Incentive and Retention Plans and (II) Grating Related Relief (the "Tillman Declaration"), and the record of these Chapter 11 Cases; and this Court having found that (i) this Court has jurisdiction over the Debtors, their estates, and property of their estates, and to consider the Motion and the relief requested therein under 28 U.S.C. §§ 157 and 1334 and the Amended Standing Order of Reference from the United States District Court for the District of Delaware, dated February 29, 2012, (ii) this Court may enter a final order consistent with Article III of the United States Constitution, (iii) this is a core proceeding under 28 U.S.C. § 157(b)(2)(A), (iv) venue of this Motion in this District is proper under 28 U.S.C. §§ 1408 and 1409, and (v) no further or other notice of the Motion is required under the circumstances; and this Court having reviewed the Motion and having heard the statements in support of the relief requested in the Motion at a hearing before this Court; and having determined that the legal and factual bases set forth in the Motion and the Cumberland Declaration and the Tillman Declaration establish just cause for the relief granted in this Order; and this Court having found and determined that the relief sought in the Motion is in the best interests of the Debtors' estates, their creditors, and other parties in interest; and after due deliberation and sufficient cause appearing therefor,

IT IS HEREBY ORDERED THAT:

- 1. The Motion is GRANTED as set forth herein.
- 2. The KEIP as described in the Motion and modified herein is hereby approved, and the Debtors are authorized to implement the KEIP in their discretion.
- 3. The total amount of payments due to the KEIP Participants shall be capped at \$950,000.

- 4. The Debtors have released any avoidance action against Matthew Karmel for the \$175,000 bonus received prior to the Petition Date, as partial consideration for the agreed upon reduction in of the KEIP.
- 5. The KERP as described in the Motion is hereby approved, and the Debtors are authorized to implement the KERP in their discretion.
- 6. The Debtors are further authorized, but not directed, to take any and all actions they deem necessary, desirable, or appropriate to effect, implement, and/or consummate the KEIP and the KERP, including without limitation, making the Incentive Bonus and/or Retention Bonus payments that may become due thereunder, without further application or order of this Court.
- All amounts earned and payable under the KEIP and the KERP shall be afforded administrative expense priority under sections 503(b) and 507(a)(2) of the Bankruptcy Code for all purposes in these Chapter 11 Cases and in case of conversion or dismissal of these Chapter 11 Cases; provided, however, that notwithstanding the Court's approval of the KEIP and KERP, the KEIP Participants and/or KERP Participants shall not have a claim against the Debtors for such amounts unless and until the Debtors actually determine to award such payments to a particular KEIP Participant and/or KERP Participant.
- 8. The Debtors are authorized to take all actions necessary or appropriate to implement the relief granted in this Order.
- 9. This Order shall be binding upon any successors and assigns of the Debtors, including any trustee appointed in these Chapter 11 Cases or in any superseding proceeding under chapter 7 of the Bankruptcy Code.

- 10. Notwithstanding Bankruptcy Rule 6004(h) or any other Bankruptcy Rule(s) to the contrary, the terms and conditions of this Order shall be immediately effective and enforceable upon entry.
- 11. This Court shall retain jurisdiction with respect to all matters arising from or related to the implementation, interpretation, and enforcement of this Order.

Dated: May 30th, 2023 Wilmington, Delaware

CRAIG T. GOLDBLATT UNITED STATES BANKRUPTCY JUDGE

Cry Della

No. S233209 Vancouver Registry

IN THE SUPREME COURT OF BRITISH COLUMBIA

IN THE MATTER OF THE COMPANIES' CREDITORS ARRANGEMENT ACT, R.S.C. 1985, c. C - 36 AS AMENDED

AND

IN THE MATTER OF SLP HOLDINGS LTD, STRUCTURLAM MASS TIMBER CORPORATION, STRUCTURLAM MASS TIMBER U.S., INC., NATURAL OUTCOMES, LLC

APPLICATION OF SLP HOLDINGS LTD. UNDER SECTION 46 OF THE COMPANIES' CREDITORS ARRANGEMENT ACT, R.S.C. 1985, c. C-36, AS AMENDED

	PETITIONER
AFFIDAVIT	

GOWLING (WLG) CANADA LLP Suite 2300 550 Burrard Street Vancouver, BC V6C 2B5 Tel. 604.683-6498 Fax 604.683.3558

Attention: Jonathan B. Ross

File No. V56936

JBR/msh